Co-existence and niche separation of two subspecies of *Photorhabdus temperata* associated with *Heterorhabditis downesi* in a dune grassland



A thesis submitted to the National University of Ireland, Maynooth for the degree of Doctor of Philosophy

by

### Abigail M. D. Maher, B.Sc

Supervisor: Dr Christine T. Griffin

Head of Department: Prof. Paul Moynagh

October 2014

Declaration

This thesis has not been submitted in whole or in part to this or any other university for any other degree and is, except where otherwise stated, the original work of the author.

Signed: \_\_\_\_\_Abigail M.D. Maher\_\_\_\_\_

Date: \_\_\_\_<u>29-10-2014</u>\_\_\_\_\_

### Contents

Acknowledgement	i
Abbreviations	ii
Abstract	iv
Chapter 1 General Introduction	1
1.1 Symbiosis	1
1.2 Entomopathogenic nematode symbiosis	4
1.2.1 Life cycle	4
1.2.2 Benefits of Photorhabdus and Xenorhabdus for Heterorhabditis	and
Steinernema	7
1.2.3 Benefits of Heterorhabditis and Steinernema for Photorhabdus	and
Xenorhabdus	11
1.2.4 EPN Diversity	15
1.2.5 Association of Heterorhabditis with Photorhabdus	23
1.2.6 Specificity of the nematode-bacteria symbiosis	26
1.2.7 Colonisation	27
1.3 Ecology and use of EPN in biocontrol	28
1.4 Distribution	33
1.5 North Bull Island	34
1.6 Aims of the project	39

Chapter	2 Recovery of <i>Heterorhabditis downesi</i> from North Bu	ıll Island and
characte	erisation of their bacterial symbiont	41
2.1 Ir	ntroduction	
2.2 N	faterials and Methods	44
2.2	2.1 Detection of nematodes on North Bull Island	44
2.2	2.2 Nematode identification	
2.2	2.3 Symbiotic bacteria characterisation	61
2.2	2.4 Phylogenetic analysis of symbiotic bacteria	69
2.2	2.5 Statistical analysis	74
2.3 R	esults	
2.3	1 Entomopathogenic nematodes on North Bull Island	
2.3	3.2 Phylogenetic analysis of the nematode ITS1 region	
2.3	3.3 Phenotypic characterisation of symbiotic bacteria	
2.3	.4 Phylogenetic analysis of symbiotic bacteria	100
2.4 D	viscussion	104
Chapter	3 Cross compatibility of Heterorhabditis downesi –	Photorhabdus
tempera	<i>ita</i>	
3.1 Ir	ntroduction	118
3.2 N	faterials and Methods	122
3.2	2.1 Cross-compatibility of <i>Heterorhabditis downesi</i> –	Photorhabdus
ten	nperata strains	
3.2	2.2 Bacteria carried by emerging infective juveniles	

3.2.3 Tests of hybrid nematode/bacteria strains against G. mellonella
3.2.4 Survival of IJs in water
3.2.5 Bacterial quantification from individual infective juveniles
3.2.6 Statistical analysis
3.3 Results
3.3.1 Cross-compatibility of Heterorhabditis downesi – Photorhabdus temperata
strains
3.3.2 Bacteria carried by emerging infective juveniles 155
3.3.3 Tests of wild type and hybrid nematode/bacteria strains against $G$ .
mellonella
3.3.4 Survival of IJs in water
3.3.5 Bacterial quantification from individual infective juveniles 167
3.4 Discussion 171
Chapter 4 Infectivity and development of Heterorhabditis downesi associated
with two subspecies of <i>Photorhabdus</i> against a range of hosts
4.1 Introduction
4.2 Materials and Methods 188
4.2.1 Tests of wild type and hybrid combinations against G. mellonella
(infectivity and infective juvenile yields)
4.2.2 Tests of wild type and hybrid combinations against Hylobius abietis
(infectivity and infective juvenile yields)

4.2.3 Tests of PUR and YEL isolates against G. mellonella (infectivity and
infective juvenile yields)
4.2.4 Tests of PUR and YEL isolates against H. abietis (infectivity and infective
juvenile yields)
4.2.5 Tests of PUR and YEL isolates against <i>T. molitor</i> (infectivity) 192
4.2.6 Tests of PUR and YEL isolates against Kelp fly (Infectivity and infective
juvenile yields)
4.2.7 Statistical analysis 194
4.3 Results
4.3.1 Tests of wild type and hybrid combinations against <i>G. mellonella</i> 195
4.3.2 Virulence of wild type and hybrid combinations against <i>H. abietis</i> 198
4.3.3 Test of PUR and YEL isolates against <i>G. mellonella</i>
4.3.4 Test of PUR and YEL isolates against <i>H. abietis</i>
4.3.5 Host infectivity – <i>T. molitor</i>
4.3.6 Host infectivity – Kelp fly
4.4 Discussion
Chapter 5 Effect of host desiccation on IJ survival and emergence
5.1 Introduction
5.2 Materials and Methods 223
5.2.1 Desiccation in sand
5.2.2 Desiccation at 0% relative humidity
5.2.3 Statistical analysis

5.3 Results
5.3.1 l Nematode survival in cadavers stored at a range of moisture levels
(experiment 1)
5.3.2 Emergence of infective juveniles from cadavers stored in dry sand
(experiments 1, 3 & 4) 233
5.3.3 Emergence of infective juveniles from cadavers stored for 28 days in dry
sand (experiments 3 & 4)
5.3.4 Emergence of infective juveniles from cadavers stored for 28 days at 0%
relative humidity (experiment 2, 3 & 4)
5.3.5 Weight loss during desiccation
5.4 Discussion
Chapter 6 General Discussion
References
Appendix

## Acknowledgement

Firstly I would to thank my supervisor, Dr Christine Griffin. Without her guidance, support and endless patience throughout my time in NUI Maynooth from undergraduate to postgraduate this work would not have been possible. Her helpful advice and suggestion whenever I was feeling stuck got me going again. Thanks for everything. This thesis was funded by the Irish Research Council (IRC) and I thank the IRC and the Department of Education and Skills for their support.

Thanks are also due to all my friends in the Behavioural Ecology and Biocontrol laboratory, both past and present, from Aoife and Daragh who showed me the ropes, Chris, Adam, Ciaran and Chris with whom I shared the highs and lows of lab work, and Johan, Louise, Cathryn and Mohamed, who always offered words of encouragement as I was writing up. Special thanks to the summer students, JP Dunbar and Katie Burke, who helped with counting nematodes, and to the fourth year students, Donal Spring, Grace Daly and Sabine Kennedy who contributed to this work. Thanks also to Michelle Finnegan and all the technical staff and to Terry and all the admin staff in the Biology department who were always ready to help whenever I asked.

Going to college as a mature student is a daunting prospect but it was made easier by the support of my family and special thanks are due to them. To my parents, John and Geraldine Maguire, who fostered in all their children a love of nature and the outdoors and a thirst for knowledge. Thanks especially to my husband Paddy, for all his support and practical help from sampling sand dunes to midnight trips to the lab to count dead insects, to my daughter Vicky, my sons Hugh and Aonghus, and my brothers and sisters who encouraged me from the side-lines, particularly my sisters Careen and Finola, who led the way as mature students, and who have helped and encouraged me throughout my life.. Thank you all for believing I could do this, it kept me going when I felt like quitting. Like my brother Paul, I climbed this mountain because it was there.

I dedicate this work to the memory of my parents and my brother Paul Maguire, I will remember you in the wild spaces and the high places.

## Abbreviations

%	percent
<	less than
=	equals
>	greater than
$\chi^2$	Chi-square
$\overline{\chi}$	mean
χ.	median
ANOVA	Analysis of Variance
B. bassiana	Beauveria bassiana
B. subtilis	Bacillus subtilis
BO	Blowout
cm	centimetre
Е	Embryonic Dune
E. coli	Escherichia coli
EPN	entomopathogenic nematode
et al.	and others
FX	Fixed Dunes
g	gram
g/l	gram/litre
GLM	Generalised Linear Model
GP	Grassy Path
hr(s)	hour(s)
ID	internal diameter
IJ	infective invenile
ITS	internal transcribed spacer
1	litre
LB/Amn	Luria Broth Miller/Amnicillin
M	Marram Dunes
M. anisopliae	Metarhizium anisopliae
M. luteus	Micrococcus luteus
min	minute(s)
ml	millilitre
mm	millimetre
mM	millimolar
MW	Mann Whitney
NA	nutrient agar
1111	nutrient ugu
NBTA	blue triphenyl tetrazolium chloride agar
OD <sub>nnn</sub>	optical density
n.nn OD	outside diameter
°C	degrees centigrade
PBS	Phosphate buffered saline
PCR	polymerase chain reaction
	H. downesi nematode carrying purple Photorhabdus
PUR	<i>temperata</i> subsp. <i>temperata</i> colour variant

	purple Photorhabdus temperata subsp. temperata colour
pur	variant
rDNA	ribosomal DNA
rpm	rotations per minute
S.O.C. medium	Super Optimal Broth
sec	second(s)
SFX	Semi-Fixed Dunes
SP	Sandy Path
sp(p)	species
StDev.	Standard Deviation
subsp.	subspecies
xg	Relative centrifugal force
X. bovienii	Xenorhabdus bovienii
X-gal	5-bromo-4-chloro-3-indolyl-β-D-galactopyranoside
	H. downesi nematode carrying yellow Photorhabdus
YEL	temperata subsp. temperata colour variant
	yellow Photorhabdus temperata subsp. temperata colour
yel	variant
μΙ	microlitre

## Abstract

Entomopathogenic nematodes belonging to the families Heterorhabditidae and Steinernematidae form symbiotic relationships with entomopathogenic bacteria from the genera *Photorhabdus* and *Xenorhabdus*, respectively. Previous studies showed that *Heterorhabditis downesi* was associated with two different colour variants (*pur* and *yel*) of its *Photorhabdus* bacterial symbiont in sand dunes on North Bull Island in Dublin Bay, with evidence of the *pur* variant being more common to the front of the dunes. The main objectives of this study were to confirm the distribution of both colour variants of the nematode-bacterium complex on North Bull Island, identify them, investigate the specificity of the nematode-bacterium relationship and ascertain characteristics of the bacterial variants that might explain their distribution.

A section of the dune system on North Bull Island was surveyed in 2008 and resurveyed in 2012 (Chapter 2). In 2008 the colour variants were spatially segregated, with nematodes associated with the *pur* variant recovered from the front section of the dunes only and nematodes associated with the *yel* variant recovered from the rear section only. In 2012 nematodes associated with each colour variant were recovered from the front section of the dunes and only nematodes associated with the *yel* variant were recovered from the rear section of the dunes. The nematode-*pur* complex was more abundant than the nematode-*yel* complex in sections where both combinations co-occurred. Three laboratory lines of each nematode and its associated bacterial colour variant were established for further study. A phylogenetic analysis of the nematode ITS region showed that all the nematode isolates were monophyletic, with no difference between the isolates carrying the *pur* or *yel* bacterial symbionts. On the other hand, a phylogenetic analysis of a portion of the gyrB gene of the bacterial isolates showed that the two colour variants were distinct subspecies, Photorhabdus temperata subsp. temperata (yel) and Photorhabdus temperata subsp. cinerea (pur). This confirms Photorhabdus temperata subsp. temperata as the bacterial partner in the YEL H. downesi – Photorhabdus complex. In addition, this is the first report of Photorhabdus temperata subsp. cinerea in Ireland and also one of the few reports of the co-occurrence of two different symbiotic bacteria subspecies with a single nematode species (Chapter 2). In antibiosis tests against a range of organisms the *P. temperata* subsp. *cinerea* isolates had a greater inhibitory effect than the *P. temperata* subsp. *temperata* isolates. Only one pair of isolates, *yel1* and *pur2*, showed any antibiotic activity against the other subspecies and this pair were mutually inhibitory (Chapter 2). Phenotypic tests indicated that the subspecies had different growth requirements as the *pur* isolates grew better on MacConkey agar than on nutrient agar while there was no difference between the two media for the *yel* isolates. Biochemical characterisation was carried out using API 20E test strips. There was variability both within and between the colour variants, however a notable difference was that only the *yel* isolates were able to utilise trehalose, a major component of insect hemolymph (Chapter 2).

The cross-compatibility of the two different bacterial subspecies with the heterogeneous nematode isolates was tested by injecting *Galleria mellonella* larvae with 10,000 cells of either the *pur1* or *yel3* isolate or both and then infecting them with *H. downesi* nematodes carrying either *pur1* or *yel3* bacteria. The emerging nematodes carried the bacterial subspecies that had been injected into the host, irrespective of which bacterial subspecies the infecting nematode had carried and hybrid lines of nematode-bacterium complex were successfully maintained for the duration of this study (Chapter 3). In the case of cadavers infected with both

bacterial types all the emerging nematodes tended to carry a single subspecies. However, between 20 % and 30 % of the cadavers produced some nematodes carrying the *pur1* and some carrying the *yel3* isolate, (indicating that both bacterial subspecies were colonising the insect host and subsequently colonising the developing nematodes prior to emergence) and where this occurred a higher proportion of the nematodes carried the *yel3* isolate. Furthermore, there was an apparent shift in the colonisation of the nematodes within the host cadaver the proportion of nematodes carrying the *yel3* isolate was higher in later emerging nematodes (Chapter 3).

To test the hypothesis that the distribution of the colour variants was due to host specialisation (*H. downesi* complex differs from front to back of the dune system) the pathogenicity of the homogenous and heterogenous combinations against *Hylobius abietis* larvae was investigated. The LD<sub>50</sub> was significantly higher for the homogenous YEL/yel combination compared to the heterogeneous YEL/pur combination. There was no difference in reproduction rate of the nematodes carrying either subspecies in either *H. abietis* or *G. mellonella* and this was backed up by tests of all three isolates of each colour type. All six isolates from North Bull Island were tested against insects from three orders (Lepidoptera, Coleoptera (2) and Diptera), with no differences in mortality or reproduction rate. This is the first report of EPN being able to use the novel host kelp fly (*Coelopa* spp.). While there was no difference in mortality rates between the PUR and YEL strains in this host there did appear to be a difference in the rate of progression of the infection with nematodes carrying the *yel* bacteria emerging earlier than those carrying the *pur* bacteria (Chapter 4).

To test the hypothesis that the bacteria influenced nematode survival in desiccated hosts *G. mellonella* larvae were infected with the homogenous and heterogeneous combinations and stored in dry sand or at 0% relative humidity. There was no difference due to bacteria type in the number of cadavers with emergence from those stored in dry sand for 28 days. There was a highly significant difference due to bacteria type in the number of cadavers with emergence at 0% relative humidity for 28 days, with emergence for cadavers stored at 0% relative humidity for 28 days, with emergence from more cadaver infected with the *pur1* isolate. Overall, more nematodes emerged from cadavers infected with the *pur1* isolate following 28 days storage in dry sand or at 0% relative humidity (Chapter 5).

*Heterorhabditis downesi* is associated with two different subspecies of *Photorhabdus temperata* on North Bull Island: *Photorhabdus temperata* subsp. *temperata* and *Photorhabdus temperata* subsp. *cinerea* and the association of the nematode with a particular subspecies appears to vary with the soil conditions prevalent in the front and the rear sections of the dunes: Under normal conditions *P.t. temperata* appears to outcompete *P.t. cinerea* in forming a symbiotic association with the nematode when both subspecies are present in the same insect host, resulting in the *H. downesi-P.t. temperata* association predominating in the rear section of the dunes, where the increased organic content of the soil favours moisture retention. On the other hand the drier conditions that predominate in the sandier soil towards the front of the dune system favour *H. downesi-P.t. cinerea* association.

## Chapter 1 General Introduction

#### 1.1 Symbiosis

The term "symbiosis" was first used by Professor Heinrich Anton de Bary to describe mutualistic or parasitic associations in the plant kingdom (Paracer & Ahmadjian, 2000). The word literally means "living together" and is of Greek origin. The word symbiosis is used to describe a range of associations between organisms from different species and frequently involves organisms from different domains of life. By convention the term "host" is usually applied to the larger partner. Viral symbiotic associations are found in all three domains of life (Paracer & Ahmadjian, 2000), and microbial symbiosis is considered to have resulted in the evolution of eukaryotes through the incorporation of alpha-proteobacteria and cyanobacteria into other bacterial cells, giving rise to mitochondria in animal cells and chloroplasts in plant cells and protists respectively (Gray & Doolittle, 1982; Sapp, 2004). Symbiotic relationships can be described as mutualistic, where both partners benefit; commensal, where one benefits and the other is unaffected; or parasitic, where one benefits and the other is harmed (Paracer & Ahmadjian, 2000), although in practice the relationship can span a continuum covering all three, particularly in complex relationships involving multiple players. The associations can be intermittent or persistent, for example, the pollination of flowers by birds and insects or the dispersal of seeds by animals and birds is a seasonal event, some species of cleaner shrimp spend less than 2% of their time cleaning the fish they associate with (Jonasson, 1987), and the red-billed ox-pecker utilises 11 different hosts (Grobler, 1980), on the other hand, most endosymbionts (intracellular or extracellular) are persistent within their host, for example the head louse *Pediculus* humanus and its intracellular endosymbiont Candidatus Riesia pediculicola and the earthworm *Eisenia foetida* and its extracellular endosymbiont *Verminephrobacter* eiseniae (Bright & Bulgheresi, 2010). The relationship can be obligate, as in the case of anemone and damefish (Leung & Poulin, 2008) or facultative, for example the secondary symbionts of pea aphids (Oliver et al., 2010). The symbiotic relationship can allow one or both partners to exploit a niche which would otherwise be unsuitable, for example the photobiont (green algae or cyanobacteria) in a lichen is protected from desiccation and exposure to the sun's rays and thus can survive within the lichen thallus in exposed habitat such as tree trunks, rocks and tundra (Lewin, 1982) and associations with sulphur-oxidising bacteria have enabled a wide range of marine animals to live in inhospitable environments such as hydrothermal vents, cold seeps and coastal sediments (Dubilier et al., 2008). The association is not limited to two partners; many associations involve multiple interactions, such as plants with fungi (Parniske, 2008), mammals and the microbial communities that live in their gut (Ley et al., 2008) and insects and their intestinal microbial communities (Broderick et al., 2004). Symbiosis offers organisms a way of overcoming genetic constraints that limit them to a particular niche through the acquisition of new traits via the symbiotic partner (Moran, 2007; Janson et al., 2008).

Numerous examples of bacterial symbiosis with invertebrates are found, covering the continuum of relationships from commensal to mutualistic to parasitic. The bacteria frequently provide a metabolic service to the invertebrate host as in the case of sap-feeding insects such as the Cicadellinae, where the symbiont (*Baumannia*  spp.) provides nutrients that are not available to the insect from its sap diet (Moran, 2007). The nature of the relationship can change over time depending on the individual partnership as in the case of *Drosophila simulans* in California where its *Wolbachia* symbiont has increased the fecundity of the female hosts in approximately 20 years, indicating a switch from a parasitic to a mutualistic relationship (Weeks *et al.*, 2007). The insect gut microbiota also plays a role in protecting the insect host from infection; in larvae of the oriental tea fly tortrix, *Homona magnanima*, the normal gut microorganisms (*Streptococcus* spp. and *Staphylococcus* spp.) suppressed the growth of *Bacillus thuringiensis* in the insect but elimination of the normal gut microorganisms from the larvae allowed *B. thuringiensis* to multiply and kill the larvae (Takatsuka & Kunimi, 2000).

The specificity of the symbiotic relationship is dependent on many factors and can operate at the level of host specificity or symbiont specificity, as in the case of coralalgal symbiosis where some species of coral can host more than one type of *Symbiodinium* algae whereas others are only able to associate with a single type and likewise the algal partner can be associated with just a single host species or with a number of different genera (Baker, 2003). A similar pattern of generalists and specialist is seen in entomopathogenic nematodes (EPN) with some species of nematode being able to carry only its own symbiont while others can carry a different species and conversely the bacterial partner can be found associated with only a single nematode species or with several (Adams *et al.*, 2006; Lewis & Clarke, 2012). Lectins are a group of proteins that were initially discovered in plants and were shown to cause animal blood cells to clump *in vitro* (Sharon & Lis, 2004). Lectins can bind specific sugar residues in membrane glycoprotiens, as well as playing a role in cell–molecule and cell-cell recognition and specificity as well as having regulatory, defence and immune functions, reviewed in Sharon & Lis (1989, 2004), and although they were first identified in plants they are now known to be involved in many host-symbiont interactions from plants (Dazzo & Hubbell, 1975) to invertebrates (Nordbringhertz & Mattiasson, 1979; Bai et al., 2013) to mammals (Cash et al., 2006). Host-symbiont recognition can involve interactions between surface structures on either the host or the symbiont; for example, bacterial peptidoglycan and host cilia, which are induced to produce mucus as in the case of the bacterium Vibrio fischeri and the Hawaiian bobtail shrimp Euprymna scolopes, or pili on the nitrogen-fixing bacteria Azoarcus required for adherence to the lateral roots of Kallar grass Leptochloa fusca (Hurek & Reinhold-Hurek, 2003). Lipopolysaccharides are also involved in host-symbiont recognition and specificity and these can be either host-associated or symbiont-associated. Microbial symbionts mainly express lipopolysaccharides (LPS) and exopolysaccharides (EPS), while host surface sugars are usually in the form of glycosylated proteins (Bright & Bulgheresi, 2010). Quorum sensing also appears to play a role in host-symbiont communication as the production of EPS can be under the control of the quorum sensing system in some bacteria (Bright & Bulgheresi, 2010).

#### 1.2 Entomopathogenic nematode symbiosis

#### 1.2.1 Life cycle

The term entomopathogenic nematode is mainly applied to members of two families, Steinernematidae and Heterorhabditidae, which are mutualistically associated with *Gammaproteobacteria (Xenorhabdus* and *Photorhabdus*, respectively) (Forst *et al.*, 1997). Although both heterorhabditids and steinernematids have a similar life cycle (Figure 1.1) they are not closely related (Blaxter *et al.*, 1998). In both genera the only free living stage of the nematode is the non-feeding infective juvenile (IJ), which typically occurs in the soil. The bacterial partner is carried freely in the gut in *Heterorhabditis* spp. (Waterfield *et al.*, 2009), while in contrast the symbiotic bacteria is carried in a specialised vesicle called a receptacle in the case of *Steinernema* spp. (Snyder *et al.*, 2007). Both nematode-bacterium complexes form tri-trophic relationships, mutualistic between the nematode and the bacterium and parasitic within insects (Eleftherianos *et al.*, 2010). The IJ stage is analogous to the dauer stage in *C. elegans* – developmentally-arrested, non-feeding and long-lived (Viney *et al.*, 2005), but has a suite of specialised host-seeking and host-recognition behaviours. In both heterorhabditids and steinernematids the IJ enters an insect and "recovers" – resumes development and within the insect host the nematodes feed on a nutrient soup consisting of the symbiotic bacteria and degraded host tissue (Stock *et al.*, 2012).



**Figure 1.1:** Generalised life cycle of entomopathogenic nematodes (Shapiro-Ilan & Grewal, 2008)

Species of both genera are capable of killing a broad range of insects and, together with their bacterial symbionts, utilise the nutrients provided by the insect host for growth and reproduction (Poinar, 1993). Entomopathogenic nematodes invade their prey via the spiracles, the mouth or the anus or, in the case of *Heterorhabditis* spp., through the cuticle using a dorsal tooth, which *Steinernema* spp. lack (Kaya & Gaugler, 1993). There is also evidence that *Steinernema* spp. produce secretions such as proteases which aid their penetration of insect cuticle or gut wall (Peters & Ehlers, 1994; Wang & Gaugler, 1998). Once inside the insect IJ recovery occurs, the nematode releases the symbiotic bacteria, either by regurgitation in the case of heterorhabditids or via the anus in the case of steinernematids (Koppenhöfer & Gaugler, 2009), and resumes its development. In both heterorhabditids and steinernematids the bacteria rapidly colonise the insect and death due to septicaemia caused by bacterial toxins (Boemare *et al.*, 1996) occurs within 48 to 72 hrs, depending on the number of IJs invading, and the nematode and insect species.

In *Heterorhabditis* IJs develop into hermaphrodites which lay eggs and give rise to males, females and hermaphrodites in the next generation. In contrast, *Steinernema* IJs develop into adult males and females in the case of most species (Ciche *et al.*, 2006), although a hermaphrodite *Steinernema* species has been reported (Griffin *et al.*, 2001). In both *Steinernema* and *Heterorhabditis*, in a process known as *endotokia matricida*, in each generation some eggs hatch within the maternal body and consume it (Johnigk & Ehlers, 1999; Baliadi *et al.*, 2004). Nematodes of both genera go through one to three generations within the host cadaver, until low nutrient supply, low bacterial concentration and high nematode concentration trigger a switch to IJ production, where juvenile development arrests at the third stage. The developing IJs are colonised by a small number of the bacterial symbiont (Ciche &

Ensign, 2003; Martens *et al.*, 2003). Large numbers of IJs which have developed either from eggs laid by the mother or through *endotokia matricida* (Ciche *et al.*, 2006) emerge from the depleted cadaver in search of new hosts. The number of IJs produced depends on the size and species of the host insect and can range from tens of thousands to several hundred thousand per insect (Stuart *et al.*, 2006).

## **1.2.2 Benefits of** *Photorhabdus* and *Xenorhabdus* for *Heterorhabditis* and *Steinernema*

*Photorhabdus* and *Xenorhabdus* employ a range of mechanisms to kill the host and provide an accessible supply of nutrients for nematode development and reproduction. The bacteria face a number of challenges: firstly from the insect immune system, secondly from the endogenous microbial community associated with the insect and finally from exogenous bacterial and fungal organisms and scavenging insects. Colonisation of the insect is very rapid, for example *Photorhabdus luminescens* C9 can multiply to more than 1 x  $10^9$  cells per insect from an initial bacterial load of less than 500 cells per insect in 24 hours following nematode infection in the wax moth *Galleria mellonella* (Hu & Webster, 2000). The ability of the entomopathogenic bacteria to outgrow their competitors in the insect cadaver may be a factor in the lack of putrefaction of the insect by the nematode-bacterium complex.

A number of different types of compounds have been identified which have been shown to have a role in the complex interactions between bacteria, nematode and insect. For example culture broth from *P. luminescens* was shown to have haemolysin, phospholipase C, protease, lipase and nuclease activity (Bowen *et al.*, 1998), while *P. temperata* M1021 was shown to have catalase, protease and lipase

activity (Jang *et al.*, 2012). The genomes of *P. luminescens* TT01 (Duchaud *et al.*, 2003), *Xenorhabdus nematophila* ATCC19061 and *Xenorhabdus bovienii* SS-2004 (Chaston *et al.*, 2011), have been sequenced and this has led to the identification of genes predicted to be involved in the pathogenicity of these bacteria. Among the gene products in the *P. luminescens* TT01 sequence are large numbers of proteases, haemolysins, toxins, lipases and antibiotics and while similar genes have been identified in the *Xenorhabdus* sequences there are also differences in, for example, antibiotic and toxin production (Chaston *et al.*, 2011).

Inhibition of phospholipase A<sub>2</sub>, resulting in a reduced immune response in the insect, has been shown for *P. temperata* subsp. *temperata*, *P. luminescens* subsp. *luminescens*, *X. poinarii*, *X. beddingii*, and *X. nematophila* (Kim *et al.*, 2005). Both *Photorhabdus* and *Xenorhabdus* produce lipases which act in a variety of ways. In *X. nematophila* lipase XlpA contributes to nematode production (Richards & Goodrich-Blair, 2010), while secreted lipase from *P. temperata* strain K122 was shown to be toxic to *G. mellonella* (Clarke & Dowds, 1995). Proteases cause general degradation of the insect cadaver, releasing nutrients for the nematode development. They can also inhibit specific components of the insect immune system, such as lysozyme, cecropin and attacin. Proteases have been identified from *P. luminescens* (Abu Hatab *et al.*, 1998; Jarosz, 1998), and from *Xenorhabdus poinarii* (Abu Hatab *et al.*, 1998) and *X nematophila* (Caldas *et al.*, 2002; Park *et al.*, 2007).

Bacterial enzymes classed as haemolysins are toxic proteins that cause cell lysis and are detected on blood agar by the appearance of a transparent zone around the colony. A number of haemolysins, including  $\alpha$ -Xenorhabdolysin (C1) from X. *nematophila* strain F1, as well as haemolysins from Xenorhabdus japonica, X. *bovienii* and Xenorhabdus beddingii, all of which showed cytolytic activity against insect haemocytes, have been identified (Brillard *et al.*, 2001). Other haemolysins produced by the symbiotic bacteria include XhlA, identified from *X. nematophila* (Cowles & Goodrich-Blair, 2005) and PhIA haemolysin, identified from *P. luminescens* strain W14 (Brillard *et al.*, 2002). In a study of three strains of *P. luminescens*, NC-19, Hm and W-14, all isolated from *H. bacteriophora* nematodes, high molecular weight proteins with insecticidal properties were identified which were toxic orally or by injection to the larvae of tobacco hornworm, *Manduca sexta*, a model lepidopteran insect frequently used in EPN studies, (Bowen & Ensign, 1998). Four toxin complexes (Tc) have been identified from *P. luminescens* culture supernatant and two of the complexes, Tca and Tcd, showed oral toxicity to insects (ffrench-Constant & Bowen, 2000). Similar compounds are also present in *Xenorhabdus* spp. as Tc-like genes have been identified in *X. nematophila* strain Xu (Waterfield *et al.*, 2001), and toxin A24tox has been identified from *X. nematophila* strain A24 (Brown *et al.*, 2004).

A wide range of antibiotic activity has been reported from both *Photorhabdus* and *Xenorhabdus*, with members of both genera being able to inhibit a range of test organisms (Akhurst, 1982; Boemare & Akhurst, 1988). Common test organisms used by Akhurst (1982) and others (Simões & Rosa, 1996; Gualtieri *et al.*, 2009; Lanois *et al.*, 2010) include *Bacillus subtilis, Escherichia coli, Micrococcus luteus* and *Staphylococcus aureus*. These and other microorganisms form part of the microbial community in the insect gut in Lepidoptera such as *M. sexta* (van der Hoeven *et al.*, 2008) and the gypsy moth, *Lymantria dispar* (Broderick *et al.*, 2004), while Coleoptera, many of which are important plant pests, tend to have a more diverse microbial community as in for example the European cockchafer, *Melolontha melolontha*, (Egert *et al.*, 2005), and weevils, *Otiorhynchus* spp.,

(Hirsch et al., 2012). Antibiotic compounds that have been identified include for example 3,5-dihydroxy-4-isopropylstilbene (ST) and anthraquinones (AQ) isolated from P. luminescens (X. luminescens) strain HK (Richardson et al., 1988) and P. luminescens strain C9 (Li et al., 1995). Stilbenes occur in plants and Photorhabdus luminescens is the only known non-plant stilbene producer. Stilbenes have antibacterial, antifungal and insecticidal activity and may also function as signal molecules in nematode growth and development (Joyce et al., 2008). Anthraquinones have antibacterial properties and also deter ants and birds from scavenging on the host cadaver (Pankewitz & Hilker, 2008). Among the antibiotic compounds produced by *Xenorhabdus* are nematophin (Li et al., 1997), which has antibacterial and antifungal activity and two xenocoumacins (McInerney et al., 1991), both of which had antibacterial activity and one of which also had antifungal activity. Other Xenorhabdus antibiotic compounds include xenorhabdins, xenorxides, benzylideneacetone, iodinine, phenethylamides and indole derivatives, reviewed in Bode (2009).

Another important category of compounds produced in both genera are the bacteriocins, proteins with antibiotic action against closely related strains or species of the same bacteria. Both interspecific and intraspecific activity as well as activity against the other genus was observed in different species and strains of *Xenorhabdus* and *Photorhabdus* isolated from four different *Steinernema* species and three different species of *Heterorhabditis*, respectively, and members of each genus showed activity against the other (Akhurst, 1982). A number of bacteriocins have been identified including lumicins, isolated from four different *Photorhabdus* (Thaler *et al.*, 1995).

In addition to the insect-pathogenic compounds the bacteria also produce compounds which have nematicidal properties. Secondary metabolites, 3,5-dihydroxy-4-isopropylstilbene (ST) and indole, from *P. luminescens* strain MD, were found to be nematicidal to some species of nematodes including some EPN and plant-parasitic nematode *Meloidogyne incognita* (Hu *et al.*, 1999).

There is considerable variability in the contribution of the bacteria to insect death and nematode reproduction in both genera. Not all species of symbiotic bacteria are pathogenic to insects, for example *X. poinarii* was unable to kill *G. mellonella* larvae when whole cells were injected into the insect haemocoel (Akhurst, 1986) and neither whole cells nor liquid culture supernatant of *X. japonica* were pathogenic to *Spodoptera litura* when injected into the insect haemocoel (Yamanaka *et al.*, 1992). The mortality of *M. sexta* injected with Tc toxins from three *P. luminescens* strains varied considerably with the Tc toxins from one strain producing 30% mortality compared to 100% mortality for the other two strains (Bowen *et al.*, 1998), indicating the variability of these compounds among different strains of the same species of bacteria. Much of the investigation of bacterial products has focused on a small number of species and strains so it is unclear how widespread the mechanisms discussed here are among the members of each genus.

# **1.2.3 Benefits** of *Heterorhabditis* and *Steinernema* for *Photorhabdus* and *Xenorhabdus*

The nematode IJ provides a protective environment for the symbiont in the soil and transports it in to the insect, giving the bacteria access to the insect haemolymph (Forst & Clarke, 2002). The bacteria can be isolated from the nematode or an infected insect and cultured on suitable media in the laboratory. The bacteria are transported into the insect body cavity by the nematode. Once inside the insect the

symbiotic bacteria are released and start to multiply. Both *Heterorhabditis* and *Steinernema* have been shown to be able to evade the insect immune system through a number of mechanisms. Three species of *Heterorhabditis*, *H. bacteriophora*, *H. downesi* and *H. gerrardi*, were able to reduce the ability of haemocytes in *M. sexta* larvae to phagocytose their respective bacterial symbionts (Eleftherianos *et al.*, 2010). All three species nematodes with or without their symbiotic bacteria caused a reduction in phenoloxidase (PO) activity, nodulation and haemocyte aggregation, all key components of the insect immune response, indicating that the nematode had a role in this reduction. *Heterorhabditis marelatus* IJs were able to avoid encapsulation in Colorado potato beetle, *Leptinotarsa decemlineata*, (Armer *et al.*, 2004). Similarly, Brivio *et al.* (2002) found that *S. feltiae* strain UK caused a reduction in PO activity in *G. mellonella* by inhibiting haemolymph protease activity, thus interfering with the PO activation pathways, and showed that this was due to interactions with LPS-like molecules from the nematode cuticle and LPS-binding proteins in the insect haemolymph.

*Heterorhabditis* and *Steinernema* are also able to destroy antimicrobial compounds induced on infection in insects. In *G. mellonella, Heterorhabditis bacteriophora* secreted an extracellular proteinase that was shown to be active against cecropin-like antibacterial peptides from *G. mellonella* and cecropin B from *Hyalophora cecropia* (Lepidoptera) (Jarosz, 1998). *Steinernema carpocapsae* (*Neoaplectana carpocapsae*) strain MEX was able to destroy proteins of the immune system of *H. cecropia* which were shown to be active against the nematode's symbiont, *X. nematophila* (Gotz *et al.*, 1981). A transcript analysis of IJs of *H. bacteriophora* (strain Az29 and strain Az36), indentified a putative surface antigen with similarity to surface antigens in the parasitic nematodes *Trichinella spiralis, Necator americanus* and *Ancylostoma*  *caninum*, which were found to be involved in the ability of the parasite to evade the host immune response (Hao *et al.*, 2012). A similar analysis of *S. carpocapsae* strain Breton found significant numbers of genes predicted to encode immune suppression proteins such as proteases (Hao *et al.*, 2010). Differences occur in proteolytic abilities of different strains of the same nematode; two axenic strains of *S. carpocapsae* (Breton and Az27) produce different rates of mortality in *G. mellonella* and the more virulent strain (Breton) had a higher rate of protease activity, possibly related to toxicity (Simões *et al.*, 2000). Table 1.1 summarises some of the characteristics distinguishing *Heterorhabditis* and *Photorhabdus* from *Steinernema* and *Xenorhabdus* spp.

		1
Nematode	Heterorhabditis	Steinernema
	(19 species described by 2012)	(75 species described by 2012)
Infective	free living in soil, non-feeding,	free living in soil, non-feeding,
juveniles	has circular tooth,	no circular tooth, produce
	develop from eggs laid by	proteases, develop from eggs
	female or via endotokia	laid by female or via endotokia
	<i>matricida</i> , protects symbiont in	<i>matricida</i> , protects symbiont in
	soil, transports symbiont into	soil, transports symbiont into
	insect, reduces insect immune	insect, reduces insect immune
	system response to symbiont	system response to symbiont
	destroys insect antimicrobial	destroys insect antimicrohial
	compounds	compounds
First conception	Larmonhaditas	Malas and famalas
First generation	Hermaphrodites	Males and females
Symbiotic	Photorhabdus	Yanorhabdus
bacteria	(3 species and 12 subspecies	(21 species described by 2012)
bacterra	described by 2012)	(21 species described by 2012)
Symbiotic	Mutualistic with nematode host,	Mutualistic with nematode host,
relationship	nematode may be associated	bacteria may be associated with
	with more than one species or	more than one species of
	subspecies of bacteria.	nematode.
	Parasitic with insect hosts	Parasitic with insect hosts
Bacterial location	Free in the intestine, released by	In specialised vesicle in
	regurgitation	intestine released via the anus
	Biochemical activity*	
Bioluminescence	+	
catalase	+	
nhosnholinase	Phospholipase A activity	Phospholipase A activity
phospholipase	Phospholipase C activity	Thosphonpase A derivity
proteases	+	+
lipases	Lip-1	XlpA
nucleases	r +	+
haemolysin	PhIA haemolysin	Xenorhabdolysin
toxins	Tca, Tcd	A24tox
Pigments	Anthraquinones	_
Antibiotics	Hydroxystilbenes	Xenomacoumacins
	Indoles	Xenorhabdins
		Xenorxides
		Indoles
Bacteriocins	Lumicins	Xenorhabdicins
Insecticidal	Hydroxystilbene	Xenorhabdin 2
activity		
Nematicidal	Hydroxystilbene	Nematophin
activity	Indoles	Indoles
T not an exhausting	alist	

Table 1.1: Some of the characteristics that distinguish Heterorhabditis and Photorhabdus from Steinernema and Xenorhabdus species.

not an exhaustive list

+ activity reported for at least some strains

- no activity reported

#### **1.2.4 EPN Diversity**

There are many more species of Steinernema than Heterorhabditis and Steinernema tend to be more abundant in the environment (Hominick, 2002). New species of both genera of nematode and their associated symbionts are being described on an ongoing basis; in 2006 over 40 species of Steinernema and over 10 species of Heterorhabditis had been recognised (Adams et al., 2006). By 2012 this had increased to more than 60 species of Steinernema and more than 20 species of Heterorhabditis, (Campos Herrera et al., 2012; Lewis & Clarke, 2012) with 21 species of Xenorhabdus and three species and 12 subspecies of Photorhabdus (Tailliez et al., 2006; Tailliez et al., 2010). Although Heterorhabditis and Steinernema are both entomopathogenic nematodes they are not closely related (Blaxter et al., 1998; Blaxter, 2011; Chaston et al., 2011) (Figure 1.2). On the other hand, *Photorhabdus* and *Xenorhabdus* are sister groups within the family Enterobacteriaceae (Tailliez et al., 2010; Chaston et al., 2011) (Figure 1.2). The ancestors of *Heterorhabditis* and *Steinernema* each separately formed relationships with the gram-negative bacteria and through a process of convergent evolution gave rise to the mutualistic associations between Heterorhabditis and Photorhabdus and Steinernema and Xenorhabdus that we see today (Poinar, 1993).

The first entomopathogenic nematode to be described was *Steinernema kraussei* (*Aplectana kraussei*) by Steiner in 1923 (Poinar & Grewal, 2012). *Heterorhabditis* bacteriophora Poinar was first described in 1976 (Poinar, 1976). The first description of the symbiotic bacteria associated with either genus was of *X*. *nematophila* (*Achromobacter nematophilus*) from *S. Carpocapsae* in 1965 by Poinar and Thomas (Poinar & Grewal, 2012), and the first description of *Photorhabdus* was that of *P. luminescens* (*X. luminescens*) from *H. bacteriophora* (Poinar, 1976).

*Heterorhabditis downesi* Stock, Griffin & Burnell, 2002, was first isolated from grassland in the southeast of Ireland (Griffin *et al.*, 1991; 1994). It was originally called *Heterorhabditis* "Irish Group" but was later described as a new species on the basis of morphological characteristics, with K122 considered as the type isolate (Stock *et al.*, 2002).



**Figure 1.2:** A comparison of the relationships between *Heterorhabditis* and *Steinernema* and their symbiotic bacteria *Photorhabdus* and *Xenorhabdus* based on 16S RNA (bacteria) and 18S inter-ribosomal RNA (nematodes). From Chaston *et al.* (2011).

In 1993 X. luminescens was reclassified as P. luminescens on the basis of low DNA relatedness to other *Xenorhabdus* spp. as well as phenotypic characterisation (Boemare et al., 1993a). A major reclassification of the genus Photorhabdus occurred in 1999 (Fischer-Le Saux et al.), with the description of three species and four subspecies: P. luminescens subsp. luminescens; P. luminescens subsp. akhurstii; P. luminescens subsp. laumondii; P. temperata; P. temperata subsp. temperata; and *P. asymbiotica*. In that study bacteria isolated from *H. megidis* strains XINach (type strain) from Russia, H. megidis strains HL81 and HW79 from the Netherlands and an Italian Heterorhabditis species, strain ItH211 were identified as Photorhabdus temperata subsp. temperata. Nielsen & Lübeck (2002) used universally primed-PCR (UP-PCR) and DNA cross-hybridisation to characterise the symbionts of Heterorhabditis species and identified strain K122 (H. downesi K122, Ireland), strain HSH2 (H. megidis, Germany), strain HL81 (H. megidis, The Netherlands) and strain S172 (H. megidis, Denmark) as P. temperata subsp. temperata. Also Ehlers & Niemann (1998), using a primer specific to variation within the variable region of the 16s rRNA, found that the bacterial symbiont of *H. downesi* strains K122 and M145 from Ireland was the same as the bacterial symbiont of *H. megidis* strains HSH1, HSH2, HSH3, HN1 from Germany and HL81, HF85, FR86 and others from The Netherlands as well as *H. megidis* isolates from Finland, Switzerland and USA.

The gyrB gene has been suggested as a more useful tool than 16S rRNA to group bacterial isolates into subspecies and species because it has a higher frequency of nucleotide substitution (Yamamoto & Harayama, 1995; Dauga, 2002). In an analysis of *P. asymbiotica* Akhurst et al (2004) found that the use of 16S rRNA grouped subspecies into recognised groups but failed to place these within the correct species groups, on the other hand, use of gyrB sequences produced groups as previously described by Fischer-Le Saux (1999) who used DNA-DNA hybridisation and PCR-RFLP methods. The *gyrB* gene has also been used as part of a multigene approach (Peat *et al.*, 2010; Tailliez *et al.*, 2010). This multigene approach provides more robust phylogenetic trees than any of the trees obtained from analysis of a single sequence including *gyrB* and 16S rRNA, and a polyphasic approach using multiple genes for sequence analysis as well as phenotypic and biochemical characterisation is recommended for detailed species description (Adams *et al.*, 2006). However, for the identification at subspecies level the analysis of the *gyrB* gene is considered sufficient (Akhurst *et al.*, 2004; Maneesakorn *et al.*, 2011). The phylogenetic relationship of a number of *Photorhabdus* and *Xenorhabdus* species is shown in Figure 1.3 (Tailliez *et al.*, 2010).



**Figure 1.3:** The phylogenetic relationship of *Xenorhabdus* species and *Photorhabdus* species based on gyrB sequences. From Tailliez *et al.* (2010).

As with bacterial identification, molecular techniques are now routinely used to identify nematode species. Among the most widely used genetic markers are the ribosomal genes (Stock, 2009). These occur in high copy number and are comprised of a number of units that evolve at different rates and contain both variable and conserved regions. This means that different sections of the genes can be analysed depending on the objectives of a particular study. The internal transcribed spacer 1 (ITS1) unit has been used by many groups to identify nematodes at species and intraspecific level (Emelianoff *et al.*, 2008b; Malan *et al.*, 2008; Nguyen *et al.*, 2008; Maneesakorn *et al.*, 2011). The widespread use of the ITS1 region has resulted in a substantial database of sequences being available for phylogenetic analysis of new isolates. The phylogenetic relationship of a number of *Heterorhabditis* species is shown in Figure 1.4 (Nguyen *et al.*, 2008). This shows that *H. downesi* and *H. Megidis* are sister taxa.



**Figure 1.4:** Phylogenetic relationship of *Heterorhabditis* species based on ITS rDNA sequences. From Nguyen *et al.* (2008)
#### 1.2.5 Association of Heterorhabditis with Photorhabdus

The conventional view of the EPN nematode-bacteria symbiosis has been that each nematode species carries a single bacterial species (Adams et al., 2006). However, the relationships between Heterorhabditis spp. and Photorhabdus spp. are more complex (Table 1.2). Heterorhabditis downesi is associated with Photorhabdus temperata subspecies cinerea (P.t. cinerea) and Photorhabdus temperata subspecies temperata (P.t. temperata). Heterorhabditis bacteriophora is associated with three P. luminescens subspecies and two P. temperata subspecies, and H. megidis is associated with three P. temperata subspecies. These different Heterorhabditis-Photorhabdus associations tend to occur at geographically distant locations, for example H. megidis-P.t. temperata occurs in Europe and H. megidis-P.t. khanii occurs in the USA; and H. bacteriophora-P. luminescens subsp. kayii (P.l. kayii) occurs in Italy and H. bacteriophora-P. luminescens subsp. laumondii (P.l laumondii) occurs in Australia (Tailliez et al., 2010). Where more than one H. bacteriophora-Photorhabdus association has been recorded from the same region there has been evidence of genetic variation between the nematode strains. Molecular analysis of *H. bacteriophora* isolates from southern France showed that two strains of H. bacteriophora carrying different subspecies of P. luminescens had differences in their ITS region (Emelianoff et al., 2008b). In the relationship between Steinernema and Xenorhabdus there is a broadly one to one relationship with a single nematode species being associated with only one bacterial species, although a single bacterial species can be carried by more than one species of nematode. Table 1.3 lists currently described Steinernema-Xenorhabdus associations.

**Table 1.2:** List of described species of nematodes in the genera *Heterorhabditis* and their *Photorhabdus* bacterial symbionts.

Nematode	Bacteria
H. amazonensis	unknown
H. atacamensis	unknown
H. bacteriophora	P. luminescens subsp. caribbeanensis, P. luminescens subsp. kayaii, P. luminescens subsp. laumondii, P. luminescens subsp. luminescens, P. temperata subsp. khanii, P. temperata subsp. thracensis.
H. brevicauais	unknown
H. downesi	P. temperata subsp. cinerea, P. temperata subsp. temperata
H. floridensis	unknown
H. georgiana	P. luminescens subsp. akhurstii
H. gerradi	P. asymbiotica Kingscliff
H. heliothidis	P temperata subsp. khanii.
H. indica	P. luminescens subsp. akhurstii
H. marelatus	P. temperata subsp. tasmaniensis
H. megidis	<i>P. temperata</i> subsp. <i>cinerea</i> , <i>P. temperata</i> subsp. <i>khanii</i> , <i>P. temperata</i> subsp. <i>temperata</i> ,
H. mexicana	unknown
H. poinari	unknown
H. safricana	unknown
H. sonorensis	P. luminescens subsp. sonorensis
H. tayserea	unknown
H. zealandica	P. temperata subsp. tasmaniensis
<i>Heterorhabditis</i> sp.	<i>P. luminescens</i> subsp. <i>caribbeanensis</i> , <i>P. luminescens</i> subsp. <i>hainanensis</i> , <i>P. luminescens</i> subsp. <i>kayaii</i> , <i>P. temperata</i> subsp. <i>khanii</i> , <i>P. temperata</i> subsp. <i>thracensis</i>

Data in the table is drawn from Tailliez *et al* (2010), Campos Herrera *et al*. (2012), Lewis & Clarke (2012) and Orozco (2012)

Nematode	Bacteria	Nematode	Bacteria
		(continued)	(continued)
S. abasi	X. indica	S. kushidai	X. japonica
S. aciari	unknown	S. leizhouense	unknown
S. affine	X. bovienii	S. litorale	unknown
S. akhursti	unknown	S. loci	unknown
S. amazonense	unknown	S. longicaudum	X. beddingii
S. anatoliense	X. nematophilia	S. masoodi	unknown
S. apuliae	X. kozodoii	S. minutum	unknown
S. arenarium	X. hominickii	S. monticolum	X. hominickii
	X. kozodoii	S. neocurtillae	unknown
S. ashiuense	unknown	S. oregonense	X. bovienii
S. asiaticum	unknown	S. pakistanense	unknown
S. australe	unknown	S. puertoricense	X. romanii
S. backanense	unknown	S. pui	unknown
S. beddingi	unknown	S. puntauvense	X. bovienii
S. bicornutum	X. budapestensis	S. rarum	X. szentirmaii
S. boemarei	X. kozodoii	S. riobrave	X. cabanillasii
S. brazilense	unknown	S. ritteri	unknown
S. carpocapsae	X. nematophilia	S. robustispiculum	unknown
S. caudatum	unknown	S. sangi	X. vietnamensis
S. ceratophorum	X. budapestensis	S. sasonense	unknown
S. cholashanense	unknown	S. scapterisci	X. innexi
S. colombiense	unknown	S. scarabaei	X. koppenhoeferi
S. costaricense	unknown	S. seemae	unknown
S. cubanum	X. poinarii	S. serratum	X. ehlersii
S. cumgarense	unknown	S. siamkayai	X. stockiae
S. diaprepesi	X. doucetiae	S. sichuanense	X. bovienii
S. eapokense	unknown	S. silvaticum	unknown
S. everestense	unknown	S. tami	unknown
S. feltiae	X. bovienii	S. texanum	unknown
S. glaseri	X. poinarii	S. thanhi	unknown
S. guangdongense	unknown	S. thermophilum	X. indica
S. hebeiense	unknown	S. tielingense	unknown
S. hermaphroditum	X. griffiniae	S. unicornum	unknown
S. ichnusae	unknown	S. websteri	X. nematophilia
S. intermedium	X. bovienii	S. weiseri	X. bovienii
S. jollieti	X. bovienii	S. xechuanense	unknown
S. karii	X. hominickii	S. yirgalemense	unknown
S. khoisanae	unknown	Steinernema sp.	X. miraniensis
S. kraussei	X. bovienii	Steinernema sp.	X. mauleonii

**Table 1.3:** List of described species of nematodes in the genera *Steinernema* and their *Xenorhabdus* bacterial symbionts.

Data in the table is drawn from Tailliez et al. (2010), Campos Herrera et al. (2012), Lewis & Clarke (2012).

In symbiotic systems the holobiont (the host and its symbiotic bacteria) might reasonably be considered as the unit of selection in evolution (Feldhaar, 2011). Both partners benefit from acquisition of new traits from the other (Moran, 2007). Thus we might expect to find evidence of co-evolution between species of nematode and their associated bacteria. In a study of five *Heterorhabditis* species and their associated bacteria, a cophylogenetic analysis of 21 *Heterorhabditis-Photorhabdus* associations found cospeciation, duplication and host switching events, however, the analysis was limited to ITS and gyrB sequences only and did not include all the recognised *Heterorhabditis* species and *Photorhabdus* subspecies as the bacterial or nematode partner had not always been identified and there was no overall support for co-evolution (Maneesakorn *et al.*, 2011). A cophylogenetic analysis of 30 *Steinernema-Xenorhabdus* associations, using a multilocus approach, also found evidence of cospeciation, host-switching and sorting events but no overall support for co-evolution between *Steinernema and Xenorhabdus* (Lee & Stock, 2010).

## 1.2.6 Specificity of the nematode-bacteria symbiosis

The ability of the symbiont of one nematode species to support the growth of a different nematode species is limited. The specificity of the symbiotic relationship functions at three levels in EPN: IJ recovery, which prompts the IJs to exit the dauer stage, the supply of nutrients to support nematode growth and development, and the colonisation of the nematode by the symbiont (Grewal *et al.*, 1997).

Recovery of both *Heterorhabditis* and *Steinernema* is prompted by food signals from the symbiotic bacteria (Grewal *et al.*, 1997; Han & Ehlers, 1998). In the case of *P. luminescens* strain NP1/1, a small heat-stable molecule from *M. sexta* haemolymph was also shown to be involved in inducing IJ recovery (Ciche & Ensign, 2003). Nematodes of both genera are able to develop on non-natural symbiotic bacteria *in*  *vitro* (Gerritsen & Smits, 1993, 1997; Grewal *et al.*, 1997). It has been shown that *H. bacteriophora* was able to develop and retain the symbionts from two strains of *H. megidis* (Han & Ehlers, 2001). In some cases specific growth requirements have been uncovered. For example *H. downesi* K122 cannot grow on *P.t. temperata* K122 *exbD* mutants that are deficient in the ability to scavenge iron in the insect haemolymph whereas iron is not a requirement for the growth of *H. bacteriophora* on its symbiont, *P. luminescens* although it is required for virulence by the *H. bacteriophora-P. luminescens* complex (Watson *et al.*, 2010). *Steinernema* are less dependent on their natural symbiont for growth and development and will grow without a symbiont (Sicard *et al.*, 2003; Mitani *et al.*, 2004).

#### **1.2.7** Colonisation

A number of specific factors regulating symbiosis in the nematode-bacterium relationship have been elucidated as a result of the sequencing of the genome of *P. luminescens* strain TT01 (Duchaud *et al.*, 2003). The purine metabolic gene, *purL*, has been shown to play a role in the persistence of *P. temperata* in its host *H. bacteriophora* GP11, possibly through biofilm formation (An & Grewal, 2011b). In *P. luminescens* TT01 the *pbgPE1* gene, part of the *PbgPE* operon has been shown to have a role in the ability of the bacteria to colonise the nematode gut (Bennett & Clarke, 2005). *Photorhabdus luminescens* strain TT01 has been shown to switch from a P form, associated with pathogenicity in the insect host, to an M form, associated with colonising the nematode, through the expression of *mad* genes that regulate the expression of fimbriae on the cell surface, and *mad* deficient mutants were unable to colonise the *H. bacteriophora* host (Somvanshi *et al.*, 2012). Other *Photorhabdus* genes involved in IJ colonisation in the *H. bacteriophora–P. luminescens* strain TT01 nematode-bacterium complex have

been identified through screening colonisation incompetent TT01 mutants, and include genes in the *pbgPE* operon, *galE*, *galU*, *proQ*, *asmA*, and *hdfR*, and the *pbgPE*, *galE*, and *galU* genes are predicted to be involved in the production of LPS by the bacterium (Easom *et al.*, 2010). While many genes involved in the regulation of the colonisation events have been identified in *P. luminescens* strain TT01 it is unclear how universal these are among all *Photorhabdus* species and it is also likely that nutritional dependencies such as the reliance of *H. downesi* K122 on its symbiont for iron metabolism (Watson *et al.*, 2005) play a major role in the specificity of nematode-bacterium complexes.

Factors regulating symbiosis in the nematode-bacterium relationship have also been found through the sequencing of *X. nematophila* strain ATCC19061 and *X. bovienii* strain SS-2004 genomes (Chaston *et al.*, 2011). For example, two *X nematophila* genes, *nilB* and *nilC* encode membrane-localised proteins required for the colonisation of its host, *S. carpocapsae* (Heungens *et al.*, 2002), and expressing these genes in other *Xenorhabdus* spp. allowed them to colonise *S. carpocapsae* (Cowles & Goodrich-Blair, 2008). In *X. nematophila* strain AN6 Mrx fimbriae are involved in IJ colonisation (Snyder *et al.*, 2011).

## 1.3 Ecology and use of EPN in biocontrol

Entomopathogenic nematodes are known to kill a wide range of insects in the laboratory, although relatively little is known about their natural host range (Poinar, 1979; Peters, 1996). Their broad host range is due to the combined actions of the nematodes and their symbiotic bacteria in overcoming the insect immune mechanisms. Both *Steinernema* and *Heterorhabditis* species have been recovered from naturally infected insects (Peters, 1996; Lewis & Clarke, 2012). At least

thirteen species of *Steinernema* and four *Heterorhabditis* species have been recovered have from such natural infections in the field (Peters, 1996). Insect orders in which these natural infections occurred include Coleoptera, Diptera, Hymenoptera and Lepidoptera. In some cases infected insects were recovered during the course of general surveys for EPN, whilst in other cases the recovery occurred during the course of field trials against particular pest insects. Larvae of the wax moth *G. mellonella* are frequently used to recover EPN in the field as well as from soil samples in the laboratory (Bedding & Akhurst, 1975). Many studies have used insects from orders including Lepidoptera, Coleoptera and Diptera as bait to recover EPN from soil survey sites (Hominick, 2002).

Early work in the laboratory suggested that EPN had a broad host range, for example *S. carpocapsae* has been shown to infect more than 250 insect species in 13 orders (Poinar, 1979; Hodson *et al.*, 2011). However, the ability to kill an insect in the laboratory was not always matched in the field and indeed it was shown that differences occurred between strains as well as between species with no single species or strain being the most effective against all insects (Bedding *et al.*, 1983; Barbosa Negrisoli *et al.*, 2013), and it is probable that no single strain or species would out-perform all others across a range of traits (Shapiro-Ilan *et al.*, 2003). It is now recognised that both generalists and specialists occur. For example, *H. bacteriophora* and *S. carpocapsae* are considered generalists, with *H. bacteriophora* being naturally associated with these two orders as well as Hymenoptera and Diptera (Peters, 1996). On the other hand, *S. glaseri* has only been recovered from mole crickets (*Scapteriscus* spp). The ability of EPN to utilise a range of natural host species

would facilitate populations to persist in the soil. There is also evidence that many EPN will infect already dead insects (San-Blas & Gowen, 2008) as well as insects already infected by their own or other EPN species (Hay & Fenlon, 1995). Both of these strategies would help EPN to persist during natural fluctuations in host availability and utilise insects or life stages that they might not normally infect (Půža & Mráček, 2010). As discussed earlier (Section 1.2.2), the more diverse range of microorganisms in the intestinal community of Coleoptera compared to Lepidoptera may partly explain the specificity of action against certain pests (Egert *et al.*, 2003; Hirsch *et al.*, 2012).

EPN display a range of foraging behaviours, ranging from ambushing to cruising (Campbell & Kaya, 2002; Lewis, 2002). Ambush foragers nictate, where the IJ raises most of its body off the soil, this allows them to attach to passing insects (Campbell & Gaugler, 1993). Extreme ambush foragers spend several hours nictating, interspersed with short periods of crawling to relocate on the surface of the search area (Campbell & Gaugler, 1993). Cruise foragers on the other hand never nictate and instead actively search over a wider area for hosts in the soil (Campbell & Gaugler, 1993; Griffin et al., 2005). Some nematodes display an intermediate foraging strategy, with active searching behaviour interspersed with short periods of nictating (Lewis, 2002). Ambush foragers are more likely to find mobile hosts on the soil surface whereas cruise foragers are more likely to find sedentary hosts within the soil (Campbell & Gaugler, 1993) and those nematodes with an intermediate foraging strategy can locate hosts which themselves range for mobile to inactive. Host colonisation is the final event in a series of steps including dispersal, attraction to host habitat, host finding and host penetration and at each step the nematode responds to a range of chemical and physical cues (Lewis et al., 2006). These cues

include CO<sub>2</sub>, temperature and host faeces (Gaugler *et al.*, 1980; Byers & Poinar Jr, 1982; Grewal *et al.*, 1993). Hallem *et al.* (2011) found that CO<sub>2</sub> had a major role in attraction of *H. bacteriophora* to *G. mellonella*. Furthermore, plants damaged by below-ground herbivorous insects have been shown to recruit EPN: *H. megidis* responded to the roots of Strawberry plants (Boff *et al.*, 2001) and *Thuja* (van Tol *et al.*, 2001) infested with black vine weevil larvae *Otiorhynchus sulcatus*. Rasmann *et al.* (2005) found that European varieties of maize produce (E)- $\beta$ -carophyllene in response to root damage caused by western corn rootworm *Diabrotica virgifera virgifera* and *H. megidis* responded to this plant signal, resulting in infection of the rootworm by the nematode.

Effective use of EPN in biological control requires an understanding of the ecology of the target pest and the EPN species. Laboratory and field trials have shown that many EPN are effective against a wide range of insect pests, see for example chapters 7 -17 in Grewal *et al.* (2005) and Georgis *et al.* (2006). For example, *S. carpocapsae*, which performs well against a wide range of hosts in the laboratory including scarab beetles is ineffective against these pests in the field. Gaugler *et al.* (1997) suggest that the reasons for this is that the beetle larvae are sedentary and live below the surface of the soil while the nematode employs an ambusher foraging strategy, waiting near the soil surface for suitable hosts, the nematode is not strongly attracted to the scarab beetle as a host, the scarab immune response is effective against the nematode and if the nematode does succeed in infecting and killing the scarab beetle the rate of nematode reproduction is low. On the other hand, both *H. bacteriophora* and *S. glaseri* appear to be well adapted to scarab pests as they both employ a cruiser foraging strategy and respond to host cues from scarab beetle larvae, they can overcome the scarab immune system and both reproduce well in

scarab beetle larvae. Furthermore, it is also necessary to take biotic interactions with other soil dwelling organisms including endemic EPN, predators and scavengers and the impact on off-target species as well as abiotic factors such as soil type, soil moisture into account in assessing the suitability of a particular nematode species for use in biological control (Kung *et al.*, 1990, 1991; Kaya *et al.*, 2006).

A number of studies have shown that EPN do not tolerate direct desiccation well. Survival in air at reduced RH is usually measured in minutes (Menti et al., 1997; O'Leary et al., 2001) while survival in hygroscopic liquids is also reduced (Mukuka et al., 2010). Survival in soil of S. carpocapsae and S. glaseri was shown to decrease with decreasing RH with S. carpocapsae (All strain) and S. glaseri surviving for 2 days and 4 hrs respectively at 25% RH compared to 16 days for both nematodes at 97% RH (Kung et al., 1991). Soil type and soil moisture were also shown to have an effect on the survival and infectivity of EPN, with nematode persistence being different for four species, (S. scarabaei AMK001 strain, S. glaseri NC1 strain, H. zealandica X1 strain and H. bacteriophora GPS11 strain), tested in three different soil types at five different water potentials (Koppenhöfer & Fuzy, 2006, 2007). In these studies the authors found that *H. zealandica* persisted better in dry loamy sand (2.3% moisture) conditions than *H. bacteriophora* (72% compared to 40% survival after 56 days). Sandy soil does not retain water well due to its large particle size and Molyneux & Bedding (1984) found that the H. bacteriophora D1 and S glaseri KG strains did not infect insects in sand with a low moisture content (equivalent to -100kPa water potential) whereas the same strains did infect in loamy sand at the same moisture content (equivalent to -1000 kPa).

# 1.4 Distribution

Entomopathogenic nematodes are found in a wide range of ecosystems from the tropics to temperate, sub arctic and arctic regions and on all continents except Antarctica (Griffin et al., 1990; Hominick, 2002). There is evidence that some Heterorhabditis species are found mainly in sandy soils, in both coastal and inland sites (Griffin et al., 1991; 1999; 2000). However this association is not universal as H. bacteriophora occurs in loamy soils, for example in the Azores (Rosa et al., 2000). Three Heterorhabditis species, H. bacteriophora, H. downesi and H. megidis have been isolated from Europe. H. bacteriophora occurs from Germany (Glare et al., 1993), and France (Emelianoff et al., 2008b) in the north to Turkey (Susurluk et al., 2001; Hazir et al., 2004), Spain (Morton & García-del-Pino, 2009) and Portugal (Valadas et al., 2013) in the south. Heterorhabditis downesi appears to have a more northerly distribution occurring in Ireland (Griffin et al., 1991; 1994), the UK (Hominick et al., 1995), Denmark (Griffin et al., 1999), and Hungary (Griffin et al., 1999; Tóth, 2006), while *H. megidis* appears to have a wider distribution than either of the other two Heterorhabditis species, occurring in the UK (Hominick et al., 1995), across Northern Europe as far as Estonia (Hominick et al., 1995; Miduturi et al., 1996; Griffin et al., 1999; Sturhan, 1999) and as far south as Greece (Menti et al., 1997). Some Steinernema species are found in coastal locations (Emelianoff et al., 2008b) although for example in Spain and the Azores they tend to occur at higher altitudes than Heterorhabditis species and are frequently associated with woodlands (Rosa et al., 2000; Campos-Herrera et al., 2007)

To date the only *Heterorhabditis* species that has been detected in Ireland has been *H. downesi* and it occurs only in sandy coastal sites such as sand dunes and sandy grasslands (Griffin *et al.*, 1991; Griffin *et al.*, 1994; Rolston *et al.*, 2005; Rolston *et* 

*al.*, 2006). Insects infected with Irish *H. downesi* isolates turn yellow-orange, greenpurple or red or yellow and in at least one site two colour variants of *H. downesi* were detected (Rolston *et al.*, 2005). *Steinernema* are more common than *Heterorhabditis* in continental Europe, (reviewed in Hominick (2002)), and this is the case in Ireland also. Three species of EPN have been recorded for Ireland to date: *Heterorhabditis downesi, Steinernema affine* and *S. feltiae*, with a greater frequency of *Steinernema* than *Heterorhabditis* (Blackshaw, 1988; Griffin *et al.*, 1991; Griffin *et al.*, 1994; Rolston *et al.*, 2002; Dillon, 2003).

## 1.5 North Bull Island

North Bull Island lies in Dublin Bay, longitude -6.15, latitude 53.37 (National Grid Reference O 231371) (Figure 1.5). It is of conservation importance nationally and internationally as a site with a diversity of coastal habitats, lagoon, mudflats, saltmarsh, dune complex and beach, which support a wide range of flora and fauna. It is an important overwintering site for waders and wildfowl and contains rare plant and animal species. The site is covered by a number of regional, national, European and other international designations. It has been designated a Special Area Amenity Order site and a Statutory Wildfowl Sanctuary, it contains two Statutory Nature Reserves, it is a Natural Heritage Area, a Special Area of Conservation under the E.U. habitats Directive and a Special Protection Area under the E.U. Birds Directive. It is also a World Biosphere Reserve, a Ramsar Convention site, a Biogenetic Reserve (McCorry & Ryle, 2009).



Figure 1.5: North Bull Island lies in Dublin Bay, at the mouth of Dublin Harbour.

The island began as a sandbar, uncovered only at low tide. Construction of the South Wall and North Bull Wall at the mouth of Dublin Port during the  $18^{th}$  and  $19^{th}$  century caused increased sedimentation to the north of the channel and resulted in the formation of the island (Flood, 1977). North Bull Island is approximately 4.85 km long by 0.80 km wide, running parallel to the shore along a southwest to northeast axis. The dunes were reported to have a maximum height of 10.21 m above sea level and an average height of 6 - 7 m in 1977 (Harris, 1977). At the landward side of the island there are extensive salt flats. There are two access routes onto North Bull Island. The first is via a bridge which links the North Bull Wall to the shore. The second is via a causeway constructed in 1964. Construction of the causeway gave rise to the development of mudflats along its northern edge between the island and the shore.

The dune complex consists of 4 main habitat types: embryonic dunes, marram dunes, semi-fixed dunes and fixed dune, each with a characteristic community of plants (Moore, 1977). The first to be encountered as one moves from the sea towards the land are the embryonic dunes, dominated by pioneer species such as *Elytrigia juncea* 

and *Leymus arenarius*. The next habitat type is marram dunes where the dominant species is the dune builder Ammophila arenaria, which stabilises windblown accumulations of sand. *Elytrigia juncea* disappears as sand accumulates but L. arenarius continues in the marram dunes. Thirdly, semi-fixed dunes, with Agrostis spp. and *Festuca* spp. appear but A. arenaria remains as the dominant grass species. In this habitat type flowering plants appear, including *Hypochaeris radicata*, Senecio spp. and *Leontodon* spp. Finally fixed dunes develop, where A. arenaria no longer dominates and grasses include Poa spp. An abundant herbiage also develops, including Anthyllis vulneria, Galium verum, Viola tricolor as well as an extensive cover of lichens and mosses. Fixed dune habitat dominated by dune grassland species such as Lolium perenne, Bellis perennis and Trifolium repens is generally absent due to a number of factors. These include the construction of two golf clubs which occupy up to half the width of the island on the landward side. The habitat in these areas has been greatly modified by the application of fertiliser and regular mowing and is now characteristic of amenity grassland (Moore, 1977). In addition, dune grassland is typically composed of consolidated and flattened dunes which are frequently used for agricultural grazing, this leads to a species-rich close sward (Fossitt, 2000). Absence of grazing and intensive recreational use on North Bull Island has lead to rank grassland with bare areas caused by tramping in the zone immediately bordering the seaward perimeter of the golf courses.

The natural hosts of *H. downesi* and *S. feltiae* on North Bull Island is unknown, however, the dune system supports a very diverse invertebrate fauna, including protozoa, nematodes, annelids, molluscs and crustaceans, however, arthropods comprise over 90 % of the recorded invertebrate fauna for the island (Healy, 1975). A number of studies have been carried out on the arthropod fauna of the dune system

and the saltmarsh (Table 1.4). Grassland generalist insect species make up approximately 50% of the arthropod species recorded from the dunes and dune grassland, while the drier marram-covered dunes are dominated by dune specialists (Speight, 1997; O'Connor & Speight, 2002). In the region between the strand line and the embryonic dunes a number of the dune and grassland species forage at night. Kelp fly, *Coelopa frigida*, (Diptera) which is found in beds of rooting seaweed along the strand line around the coast of Ireland and the UK has also been recorded (Healy, 1975). Over 132 invertebrate species were recorded in a three year period ranging in depth from soil surface to 20 cm in the region between the strand line and the embryonic dunes (Healy, 1975). (See also Speight (1975) for an account of Syrphidae (Diptera) from Ireland, including species occurring on North Bull Island). Rolston (2005) noted the frequent occurrence of the black marram beetle Otiorhynchus atroapterus throughout the dune system. In the course of the present study adult *Psilothrix* spp. (Coleoptera) were observed in the front section of the dune system and larvae of the cinnabar moth Tyria jacobaeae and adult O. atroapterus were observed throughout the dune system (Maher, personal observation).

	Grassland generalist s	pecies		
Species	Order			
Platycheirus clypeatus	Diptera	Speight, 1997		
Philaenus spumarius	Hemiptera	Speight, 1997		
Neophilaenus lineatus	Hemiptera	Speight, 1997		
Myrmica ruginodis	Hymenoptera	Speight, 1997		
Bombus muscorum	Hymenoptera	Speight, 1997		
Philoscia muscorum	Isopoda	Speight, 1997		
Armadillidium vulgar	Isopoda	Speight, 1997		
Zygaena filipendulae	Lepidoptera	Speight, 1997		
	Dune specialist spec	cies		
Species	Order			
Dilta hibernica	Archaeognatha	Speight, 1997		
Demetrias atricapillus	Coleoptera	Speight, 1997		
Dromius melanocephalus	Coleoptera	Speight, 1997		
Broscus cephalotes	Coleoptera	Speight, 1997		
Phthiria pulicaria	Diptera	Speight, 1997		
Sphaerophoria rueppellii	Diptera	Speight, 1997		
Philonicus albiceps	Diptera	Speight, 1997		
Thereva nobilitata	Diptera	Speight, 1997		
Neoleria maritima	Diptera	O'Connor & Speight, 2002		
Psammochares plumbeus	Hymenoptera	Speight, 1997		
Armadillidium album	Isopoda	Speight, 1997		
Embryonic dune species				
Species	Order			
Laemostenus terricola	Coleoptera	Speight, 1997		
Silpha tristis	Coleoptera	Healy, 1975		
Thanatus rugosus	Coleoptera	Healy, 1975		
Coelopa frigida	Diptera	Healy, 1975		
Philoscia muscorum	Isopoda	Healy, 1975		
Armadillidium vulgare	Isopoda	Healy, 1975		

**Table 1.4:** Arthropod species associated with zones with the dune system onNorth Bull Island.

# 1.6 Aims of the project

Two colour variants of the *H. downesi - P. temperata* nematode - bacterium complex, designated *yel* and *pur*, have been detected in the sand dunes on North Bull Island (Rolston *et al.*, 2002). Surveys detected a pattern of spatial distribution of the two colour phenotypes in the dune system with *pur* tending to occur towards the front of the dunes (Rolston *et al.*, 2005; Rolston *et al.*, 2006). The main aim of this project was to test the hypothesis that differences in the bacterial symbiont associated with the *H. downesi* nematodes in the front and the rear sections of the dunes were driving the spatial distribution of the nematode – bacterium complexes in the dune system on North Bull Island and to identify the traits responsible.

## Specific objectives were to:

- Confirm the distribution of the two colour phenotypes in the dune system on North Bull Island. Identify the nematode and their bacterial symbionts by sequencing the ITS1 (nematode) and gyrB (bacteria) regions of the organisms. Investigate the biochemical and antibiosis profiles of the bacterial colour phenotypes and identify differences between the *yel* and *pur* phenotypes which might help to explain the spatial distribution of the nematode – bacterium complexes (Chapter 2).
- Test the specificity of the nematode bacterium relationship by investigating whether the nematode associated with each bacterial colour phenotype can carry the other phenotype. Investigate the cost to the nematode of carrying the bacterium by assessing the number of bacteria of each type carried and the influence of each type of bacteria on IJ survival (Chapter 3).

- Investigate the adaptation of the bacterial colour phenotypes to a range of hosts. Differences in the ability of the nematode bacterium combinations to kill and reproduce in a range of host could be due to differences in the ability of the bacterial phenotypes to overcome the insect host immune system and its intestinal microbial community and utilise the host tissue. As the host range of the dune system will vary from front to back differences in host adaptation between the two bacterial colour phenotypes may help to explain the trend for the nematodes carrying the *pur* bacteria to be more prevalent in the front section of the dune system (Chapter 4).
- Investigate the role of the bacterial partner in the adaptation of the nematode

   bacterium complex to the harsh abiotic conditions present in the sand dunes
   by testing the ability of different nematode bacterium combinations to
   survive and emerge from desiccated insect hosts. Differences in the ability of
   the nematode bacterium complex may also help to explain the trend for the
   nematodes carrying the *pur* bacteria to be more prevalent in the front section
   of the dune system where desiccating conditions are more prevalent (Chapter 5).

# Chapter 2 Recovery of *Heterorhabditis downesi* from North Bull Island and characterisation of their bacterial symbiont

# 2.1 Introduction

Heterorhabditis downesi are entomopathogenic nematodes which are mutualistically associated with bacteria of the genus Photorhabdus. By 2007 at least eleven species of Heterorhabditis had been described (Nguyen & Hunt, 2007), and additional species are being described on an ongoing basis (Malan et al., 2008; Nguyen et al., 2008; Lewis & Clarke, 2012). Three species of Photorhabdus, P. luminescens, P. temperata and P. asymbiotica, have been described, along with a number of subspecies of P. luminescens, and P. temperata (Tailliez & Boemare, 2009; Tailliez et al., 2010; Orozco, 2012). To date, only a single Heterorhabditis/Photorhabdus complex, Heterorhabditis downesi/Photorhabdus temperata subsp. temperata, has been recorded in Co. Wexford, Ireland. Griffin et al. (1991) recorded the occurrence of H. downesi at a single site in Ireland. This was a sandy coastal location. A subsequent survey in Ireland and Britain recorded H. downesi only at sandy coastal locations, including Portmarnock, Co. Dublin, approximately nine kilometres north of North Bull Island (Griffin et al., 1994). In mainland Europe H. downesi has been recorded in Denmark and Hungary (Griffin et al., 1999). In a study of entomopathogenic nematodes on North Bull Island two colour variants of Photorhabdus sp. were found to be associated with the nematode H. downesi, based on the colour of infected insect cadavers, (Rolston et al., 2005). These were

designated purple (*pur*) and yellow (*yel*). Studies of *H. downesi* in Ireland, Britain and mainland Europe reported the occurrence of four colour phenotypes, red, yellow-orange, green-purple, and grey (Griffin *et al.*, 1994; Tóth *et al.*, 2008). However, the work by the Rolston et al (2005) was the first to report the cooccurrence of two colour phenotypes within the same survey site. The two colour phenotypes appeared to be spatially segregated within the dune system with the purple phenotype tending to dominate in the front 50 m section.

North Bull Island is a low dune-covered island lying just off the coast in the northern half of Dublin Bay (see Section 1.5 for a full description). A number of gradients occur within the sand dune system, for example there is a progression from open sand to closed grassland as one moves from seaward to landward side of the dune system (Jeffrey, 1977). Associated with this gradient in soil cover there is a change from loosely compacted sand which loses moisture quickly to soil with a build-up of organic material both within the soil profile and overlaying the soil surface, which aids moisture retention. The combined effect of the increase in organic material and the leaching of minerals which occurs over time results in a decrease in pH and a decrease in salinity with distance into the dunes. Thus the dune system is comprised of a number of different microhabitats, with different plant and animal communities occurring across the dune system, both above and below the soil surface, with the number and range of species increasing with distance into the dune system.

In this chapter the continued presence of *H. downesi* in the dune system on North Bull Island will be confirmed by soil sampling. The soil sampling will seek to determine the spatial segregation of the colour phenotypes and investigate whether

42

the colour phenotypes have a spatial distribution or association with particular habitat type in the dune system.

A molecular approach will be used to confirm the identity of the *Heterorhabditis* sp nematodes on North Bull Island as *Heterorhabditis downesi* (see Section 1.2.4.). Nuclear ribosomal DNA (rDNA) has been widely used in the identification of nematodes (Joyce *et al.*, 1994; Hominick *et al.*, 1997; Powers *et al.*, 1997). DNA sequence analysis of the internal transcribed spacer 1 (ITS1) of the ribosomal genes has proved to be a useful molecular tool for the identification of *Heterorhabditis* at species level (Adams *et al.*, 1998; Maneesakorn *et al.*, 2011).

The symbiotic partners of nematodes of the genus *Heterorhabditis* are entomopathogenic bacteria of the genus *Photorhabdus* (see Section 1.2.5). A polyphasic approach will be employed to characterise the symbiont of *H. downesi*. Molecular analysis of the *gyrB* gene will be used to identify the symbiont (see Section 1.2.4.). This method has been used to investigate the phylogeny of the genus and has proved useful at the intrageneric level to identify closely related species and subspecies (Akhurst *et al.*, 2004; Tóth & Lakatos, 2008; Boemare & Tailliez, 2009; Maneesakorn *et al.*, 2011). Morphological, biochemical and physiological characteristics will be investigated to describe the two colour phenotypes found on North Bull Island. This polyphasic procedure may elucidate differences between the two phenotypes.

The change in habitat discussed above is also accompanied by a change in the community of microorganisms in the soil. Entomopathogenic nematodes also encounter the gut biota of the insects they invade. The entomopathogenic bacteria associated with *Heterorhabditis* and *Steinernema* are known to produce a range of

antimicrobial compounds which are effective in suppressing competition from other microorganisms and nematodes (Derzelle *et al.*, 2002; Ansari *et al.*, 2005; Eleftherianos *et al.*, 2007) (See Section 1.2.2). The antibiotic activity of the two phenotypes will be investigated using a range of bacterial and fungal targets. The hypothesis is that *yel* phenotype will be more effective in this regard as it would be expected to encounter a broader range of antagonistic organisms in the more organic soil in the rear half of the dune system and in the wider range of host insects present in the rear half of the dune system.

The objectives of this chapter are:

- To investigate the distribution of *H. downesi* in the dune system on North Bull Island and to recover fresh isolates of each colour phenotype for study.
- To identify the nematodes and their symbiotic bacteria using molecular techniques.
- iii) To look for differences between the *pur* and *yel* bacterial phenotypes, especially in antibiotic production.

# 2.2 Materials and Methods

### 2.2.1 Detection of nematodes on North Bull Island

## 2.2.1.1 Sampling Method

North Bull Island was sampled for entomopathogenic nematodes (EPN) during the period September 2008 to October 2008. Sampling was conducted in three transects. Each transect was 130 m long and ran from the beach across the dunes to the fescue-dominated grassland and consisted of fourteen sampling locations. Transect 1 began 100 m northeast of the more northerly beach access road. Transect 2 was located 500

m northeast of transect 1 and transect 3 was located 500 m southwest of transect 1 (Figure 2.1). The first sampling location in each transect was 2 m above the strand line at the base of the first embryonic dune. The sampling locations were at 10 m intervals running into the dune system along a bearing of 330°, perpendicular to the long axis of the island. Samples (3.7 cm<sup>3</sup>) were taken to a depth of approximately 10 cm with a bucket type soil corer (Figure 2.2). Thirty soil subsamples were taken at each sampling location, with samples being taken 10 m to either side of the transect along a line perpendicular to the transect. At each sampling location the subsamples were pooled in a plastic bag to make a single sample of approximately 500 g.

In 2012 transect 1 was resampled. On this occasion three indivdual samples were taken to a depth of approximately 10 cm with a bucket type soil corer, with one sample being taken along the transect and one sample being taken 10 m to either side of the transect along a line perpendicular to the transect. Samples were placed in individual 50 ml tubes (Sarstedt, Numbrecht, Germany).



**Figure 2.1**: North Bull Island. 1, 2 and 3 indicate the location of the transects sampled in this study. Transect 1 and transect 2 overlapped with the areas sampled in 2001 and 2002 in the Rolston (2005) study (see Appendix Figure A. 1)



**Figure 2.2:** Bucket type soil corer used for taking soil samples on North Bull Island. The corer was washed with 70% ethanol and allowed to air dry between sampling locations.

## 2.2.1.2 Soil baiting

In the laboratory the soil type was classified as sandy or sandy/organic by visual inspection. Each pooled soil sample (500 g approximately) was subdivided and placed into three separate 250 ml plastic food tubs with a diameter of 9 cm ("Econo" type 731, Huhtamaki, Espoo, Finland).

To recover the nematode-bacterium complex each tub was baited with five final instar larvae of the greater wax moth (*Galleria mellonella*). Wax moths were obtained from the Mealworm Company (Sheffield, UK), and were stored in wood shavings in batches of 250 at 15  $^{\circ}$ C for up to four weeks. The bait insects were placed on the soil and the tubs were lidded, inverted and stacked in random order in stacks of three in a plastic tray and covered with tinfoil to reduce evaporation. The baited tubs were incubated at 20  $^{\circ}$ C. In the case of soil samples taken in 2012 a single *G. mellonella* larva was place on the soil in each tube and the tubes were

inverted and incubated at 20  $^{\circ}$ C. The bait insects were removed after four days and stored individually in 5.5 cm Petri dishes (Figure 2.3) on moist tissue paper at 20  $^{\circ}$ C for a further four days. The baiting was repeated once.





**Figure 2.3:** Nematode-infected bait insects recovered from soil from North Bull Island. Colour phenotypes were designated PUR (A) and YEL (B)

## 2.2.1.3 Nematode recovery from bait insects

Nematode infected cadavers were attributed to *H. downesi* or *Steinernema* sp. on the basis of cadaver bioluminescence and colour. Bioluminescence was assessed by observing cadavers in a darkened room for 10 min. Colour phenotypes were designated PUR (Figure 2.3A) (cadaver achromatic darkening to purple) or YEL (Figure 2.3B) (cadaver straw coloured darkening to yellow). Fourteen days after baiting cadavers exhibiting evidence of a *H. downesi* infection (no putrefaction, body remaining soft, cadavers luminescent, cadaver colour PUR or YEL were placed on individual White traps (Kaya & Stock, 1997) and stored at 20 °C to collect emerging infective juvenile (IJ) nematodes. White traps for individual cadavers consisted of a lid of a 3.5 cm Petri dish covered with a 55 mm filter paper (FisherBrand FB59015, Fisher Scientific, Ireland) in a 9 cm Petri dish (Figure 2.4). The Petri dish was filled to approximately half its depth with tap water. Infected cadavers were placed on the

filter paper and emerging IJs were allowed to accumulate in the water for five days before harvesting.



Figure 2.4: White trap used for single infected cadavers

## 2.2.1.4 Nematode harvesting

White traps were harvested by pouring the IJ suspension into a 50 ml glass beaker. Each White trap was rinsed three times to collect all the IJs. The IJs were washed three times by sedimentation by filling the glass beaker with tap water, allowing the IJs to settle to the bottom of the beaker, pouring off the supernatant and refilling the beaker with tap water.

# 2.2.1.5 Surface sterilisation

IJ suspension was transferred to a 50 ml graduated cylinder and surface sterilised with 0.16 mM Hyamine 1622 solution (Sigma-Aldrich) (Enright & Griffin, 2004). The graduated cylinder was sealed with Parafilm (Pechiney Plastic Packaging, Menasha, USA) and inverted three times and then IJs were allowed to settle for 15 min. The supernatant was removed and the IJs were washed three times by resuspending in sterile tap water, inverting three times, allowing the IJs to settle, pouring off the supernatant and resuspending in sterile tap water again.

#### 2.2.1.6 Nematode counting and storage

Prior to storage nematode counts were carried out. The nematode suspension was made up to 50 ml with sterile tap water in a graduated cylinder which was sealed with parafilm and inverted three times. A 100 µl sample was removed and pipetted in a line of drops into a Petri dish. Live IJs were counted with the aid of a dissecting microscope by moving from one drop to the next. Three 100 µl samples were counted and the number of IJs in the total volume was calculated based on the mean of three counts. The nematode suspension was adjusted to approximately 1000 IJs/ml with sterile tap water and stored at 9 °C in 25 ml aliquots in 125 ml plastic food tubs with a diameter of 9 cm ("Econo" type 721, Huhtamaki, Espoo, Finland). Infective juveniles were stored for at least seven days at 9 °C before being used.

#### 2.2.1.7 Culturing of nematodes

Isolates were reared in final instar *G. mellonella* larva. The lid and base of a 9 cm Petri dish were each lined with a 90 mm filter paper. Ten final instar *G. mellonella* larvae were placed in the dish and 1 ml of IJ suspension (1000 IJs/ml) was pipetted evenly onto the lid and base, equivalent to a dose of 200 IJs per insect. The dishes were incubated at 20 °C. After four days any insects exhibiting signs of an EPN infection were placed on moist filter paper in a 9 cm Petri dish and stored at 20 °C. After a further ten days the cadavers were placed on a White trap consisting of a 5.5 cm Petri dish lid covered with a 70 mm filter paper (FisherBrand FB59017, Fisher Scientific, Ireland) in a 14.5 cm Petri dish. The Petri dish was filled to approximately half its depth with tap water. Infected cadavers were placed on the filter paper and the White trap was stored at 20 °C (Figure 2.5). Emerging IJs were allowed to accumulate in the water for five days before harvesting as described in Section 2.2.1.4. The nematode suspension collected from bulk infections with ten *G*. *mellonella* was made up to 250 ml in a graduated cylinder then counted and concentration adjusted to 1000 IJs/ml as described in Section 2.2.1.6 using normal tap water and stored for up to 1 month before use in experiments. Nematodes were cultured in this manner at least every three months.



Figure 2.5: White Trap for routine culturing of nematodes.

#### 2.2.1.8 Habitat survey

The flora at each sampling location was recorded. The soil temperature at a depth of 10 cm was recorded using a HI98509 Checktemp1<sup>®</sup> thermometer (Hanna Instruments, Bedfordshire, UK). Soil pH was measured using a pH meter (model 3510, Jenway, Dunmow, England). Samples were prepared by adding 20 ml deionised water to 10 g of soil in a 50 ml tube (Sarstedt, Numbrecht, Germany) and mixing thoroughly (Robertson *et al.*, 1999). The mixture was allowed to equilibrate for 30 min at room temperature before reading. Three replicate samples were prepared for each sampling location. The habitat along each transect was classified according to Fossitt (2000), modified according to Rodwell *et al* (2000). The categories used were: Embryonic Dunes, Marram Dunes, Semi-Fixed Dunes, Fixed Dunes, Sandy Path, Grassy Path and Blowout. All categories recorded at each

sampling location were used to produce a composite habitat type with thirteen designations.

#### 2.2.2 Nematode identification

*Heterorhabditis* sp. nematodes were identified on the basis of sequence analysis of the internal transcribed spacer 1 (ITS1) region of the ribosomal genes (rDNA).

## 2.2.2.1 DNA extraction from nematodes

DNA extraction was carried out on bulk samples of each of the nematode isolates listed in Section 2.2.1.3. DNA was extracted according to the Animal Tissue Protocol using a DNeasy<sup>®</sup> Blood and Tissue Mini Spin Column Kit (Qiagen, Crawley, UK). All buffers (ATl, AL, AW1, AW2, AE), proteinase K, spin columns and collection tubes were supplied with the kit. Details of reagent constituents are available in the DNeasy<sup>®</sup> Blood and Tissue Handbook. Sufficient Buffer AL was premixed with an equal volume of molecular grade ethanol (Sigma-Aldrich) (AL/EtOH) according to the manufacturer's instructions for the number of reactions being carried out. Buffers AW1 and AW2 were made up with molecular grade ethanol according to the manufacturer's instructions. A Sigma 1-15 microcentrifuge (Sigma Laboratory Centrifuges, Osterode, Germany) was used for all centrifugation steps.

One ml of surface sterilised nematode suspension (10,000 IJs /ml) was transferred to a 1.5 ml microcentrifuge tube (Fisher Scientific, UK) and collected by centrifugation at 17,968 x g for 5 min. The supernatant was discarded and the nematodes were placed at -70  $^{\circ}$ C for 15 min. The nematodes were semi-defrosted and ground with a micro pestle (Sigma-Aldrich, St. Louis, USA). The DNeasy protocol was then followed. Briefly, Buffer ATL (180 µl) and proteinase K (20 µl) were added, the sample was mixed by vortexing for 5- 10 sec then lysed at 56 °C for 1 hr with shaking at 100 rpm with 5 - 10 sec vortexing every 15 min. AL/EtOH (400 µl) was added and the sample was mixed by vortexing, transferred to a spin column in a 2 ml collection tube and centrifuged at 6000 x g for 1 min. The collection tube and flowthrough were discarded and the spin column was transferred to a fresh collection tube. Buffer AW1 (500  $\mu$ l) was added and the sample was centrifuged at 6000 x g for 1 min. The collection tube and flow-through were discarded and the spin column was transferred to a fresh collection tube. Buffer AW2 (500 µl) was added and the sample was centrifuged at 17,968 x g for 3 min. The collection tube and flowthrough were discarded and the spin column was transferred to a 1.5 ml microcentrifuge tube. Buffer AE (100 µl) was added and the sample was incubated at room temperature for 1 min then centrifuged at 6000 x g for 1 min. This step was repeated once. The spin column was discarded and the concentration of the DNA in the eluate was quantified using a NanoDrop® ND-1000 spectrophotometer (NanoDrop Technologies, Wilmington, USA). A portion of the eluate was adjusted to a concentration of 20 ng/µl. Both portions of DNA were stored at -20 °C until used. Three replicate extractions were carried out for each isolate.

## 2.2.2.2 PCR amplification of nematode DNA

Amplification of the nematode DNA extract was carried out using primers specific for the ITS1 region of the rDNA (Vrain *et al.*, 1992). All reactions were carried out using an Eppendorf Mastercycler Gradient machine (Eppendorf, Cambridge, UK). All reagents were kept on ice. All materials were from Sigma-Aldrich unless indicated otherwise. Sufficient master mix for three reactions per isolate was made up in a 1.5 ml microcentrifuge tube. Individual reactions were carried out in 0.5 ml thin-walled PCR tubes (Fisher Scientific, Ireland) using 45 µl of master mix and 5 µl of DNA extract (20 ng/ $\mu$ l). A negative control to which no DNA was added was also included. The contents of the tubes were mixed by vortexing and then briefly centrifuged. Table 2. gives details of the PCR mixture, along with suppliers for the ingredients, Table 2.2 lists the cycling conditions used and Table 2.3 gives details of primers used.

For each reaction 5 µl PCR product was mixed with 1 µl of 5 x tricolour loading buffer (Bioline, UK) and analysed on a 1 % agarose gel: agarose (Life Technologies) 1 g dissolved in 100 ml 1 x TAE buffer (40 mM Tris-acetate, 1mM EDTA) (see Appendix Table A 1 for Tris-acetate and EDTA buffer composition) stained with 1 x SYBR Safe DNA Gel stain (Life Technologies). Gels were run at 100V for 1 hr in 1 x TAE buffer. Gels were visualised on a Transilluminator TFX-35M (Vilber Lourmat, Marne-la-Vallée, France) and photographed using a Kodak Electrophoresis Documentation and Analysis System (EDAS) system (Eastman Kodak Company, Rochester, NY, USA)

**Table 2.1:** PCR mixture for amplification of ITS1 region of rDNA from nematode isolates from North Bull Island. Reactions were carried out in 50  $\mu$ l volumes. Reagents were supplied by Sigma-Aldrich unless otherwise indicated.

Reagent	Volume (µl)	Final concentration	Supplier
Buffer (10x)	5	1x	
MgCl2 (25mM)	4	2 mM	
dNTP Mix (10mM)	1	200 µM	Metabion
Forward primer (10mM)	5	1µM	Eurofins MWG Operon
Reverse primer (10mM)	5	1 µM	Eurofins MWG Operon
Taq Polymerase	0.25	0.025 U/µl	
DNA template (20 ng/µl)	5	2 ng/µl	
Molecular grade water	24.75		

Stage	Cycles	Temperature	Duration
Initial denaturation	1	94°C	2 min
Denaturation		94°C	30 sec
Annealing	40	45°C	1 min
Extension		72°C	90 sec
Final extension	1	72°C	5 min
Hold	1	4°C	indefinite

**Table 2.2:** PCR cycling conditions for amplification of ITS1 region of rDNA from nematode isolates from North Bull Island (Hominick *et al.*, 1997). Lid temperature was set to 95°C.

 Table 2.3: Primer sequences for nematode ITS1 rDNA (Vrain et al., 1992)

Primer	Sequence
18S (Forward)	5'-TTGATTACGTCCCTGCCCTTT-3'
28S (Reverse)	5'-TTTCACTCGCCGTTACTAAGG-3'

## 2.2.2.3 Cloning PCR product

Cloning reactions were set up using a TOPO TA Cloning<sup>®</sup> Kit (Life Technologies). One Shot<sup>®</sup> TOP10 chemically competent *Escherichia coli* (*E. coli*) cells, pCR<sup>®</sup>2.1-TOPO cloning vector, salt solution (1.2 M NaCl, 0.06 M MgCl<sub>2</sub>) and S.O.C. medium were included in the cloning kit. Details of cloning kit reagents are available in the TOPO TA Cloning<sup>®</sup> Kit User Manual. Ampicillin stock solution (50 mg/ml) was prepared by dissolving ampicillin in distilled water and filter sterilised using a 0.2 µm Filtropur S syringe filter (Sarstedt, Numbrecht, Germany). LB/ampicillin (LB/Amp) broth and agar were prepared using LB broth or LB agar (BD, Franklin Lakes, NJ, USA) made up to 1 L according to the manufacturer's instructions, autoclaved at 121 °C for 20 min and allowed to cool to 55 °C before adding ampicillin (50 µg/ml). LB/Amp agar was poured into 9 cm Petri dishes and allowed set. LB/Amp agar plates and LB/Amp broth were stored at 4 °C until required. LB/Amp/X-gal agar plates were prepared by spreading LB/Amp agar plates with X-gal (5-bromo-4-chloro-3-indolyl- $\beta$ -D-galactopyranoside) (Promega, Madison, WI, USA) (1.6 mg) and the plates were warmed at 37 °C for 30 min before use.

One ligation reaction was set up for each of the PCR products obtained in Section 2.2.2.2. Each reaction consisted of 1  $\mu$ l pCR<sup>®</sup>2.1-TOPO cloning vector, 1  $\mu$ l salt solution and 4  $\mu$ l fresh PCR product. The reagents were mixed gently by pipetting and then incubated at room temperature at 25 rpm for 20 min. The ligation products were then transformed into One Shot<sup>®</sup> TOP10 chemically competent *E. coli* cells according to the protocol for transforming One Shot Chemically Competent Cells (Life Technologies). Briefly, 2  $\mu$ l of the ligation reaction was added to a vial of freshly thawed cells. The reaction was mixed by inverting 3 times and incubated on ice for 20 min then heat-shocked for 30 sec at 42 °C and immediately returned to ice. S.O.C. medium (250  $\mu$ l) was added to each vial of transformed cells and vials were incubated horizontally at 37 °C with shaking (200 rpm) for 1 hr. Aliquots of 20  $\mu$ l, 35  $\mu$ l and 50  $\mu$ l of the transformed cells were spread on the LB/Amp/X-gal agar and incubated overnight at 37 °C for blue/white screening of clones.

## 2.2.2.4 Plasmid isolation

Plasmid DNA was purified using a QIAprep® Spin Miniprep Kit (Qiagen). All buffers (P1, P2, N3, PB, PE), LyseBlue, RNase A, spin columns and collection tubes were supplied with the kit. Details of reagent constituents are available in the QIAprep<sup>®</sup> Miniprep Handbook. RNase A (200  $\mu$ l) and LyseBlue (20  $\mu$ l) were added to buffer P1 (20 ml) according to the manufacturer's instructions.

For each isolate three single, well separated, white colonies were selected from the LB/Amp/X-gal overnight plates. Each colony was picked into a separate 50 ml centrifuge tube containing 5 ml of LB/Amp broth and incubated overnight at 37 °C with shaking (200 rpm). Overnight cultures were transferred to 15 ml centrifuge tubes and the cells were collected by centrifugation at 1462 x g for 10 min. The supernatant was discarded and the tubes were inverted and allowed to drain for 10 min. The cells were resuspended by pipetting in Buffer P1/RNase A/LyseBlue (250 µl) and transferred to a 1.5 ml microcentrifuge tube. Buffer P2 (250 µl) was added and the sample was mixed gently by inverting until a homogenous blue solution was obtained. Buffer N3 (350 µl) was added and the sample was mixed by inversion until a homogenous colourless suspension was obtained. The sample was centrifuged for 10 min at 17,968 x g and the supernatant was transferred to a spin column in a 2 ml collection tube. The sample was centrifuged for 1 min at 17,968 x g and the collection tube and the flow-through were discarded. The spin column was transferred to a clean collection tube. Buffer PE (0.75 ml) was added and the sample was centrifuged for 1 min at 17,968 x g. The flow-through was discarded and the sample was centrifuged for an additional 1 min at 17,968 x g. The collection tube and flow-through were discarded and the spin column was transferred to a clean 1.5 ml microcentrifuge tube. The DNA was eluted in molecular grade water (Sigma-Aldrich) by adding 50  $\mu$ l water to the spin column, incubating at room temperature for 1 min then centrifuging at 6000 x g for 1 min. The spin column was discarded and the concentration of the DNA in the eluate was quantified using a NanoDrop® ND-1000 spectrophotometer (NanoDrop Technologies, Wilmington, USA). Purified plasmids were stored at -20 °C until required.

#### 2.2.2.5 Plasmid analysis

Gel analysis of the plasmids was carried out to confirm presence of the ITS1 rDNA. Reactions were carried out in 50  $\mu$ l volumes using 1  $\mu$ l purified plasmid per reaction Sufficient PCR master mix was prepared for 18 samples using the recipe in Table 2. with the exception that the volume of water was increased to 28.75  $\mu$ l per reaction. Cycling conditions are given in Table 2.4. PCR products were analysed on a 1 % agarose gel as described in Section 2.2.2.2.

**Table 2.4:** PCR cycling conditions to confirm presence of nematode ITS1 region of rDNA in purified plasmids from clones. Lid temperature was set to 95°C.

Stage	Cycles	Temperature	Duration
Initial denaturation	1	94 °C	2 min
Denaturation		94 °C	30 sec
Annealing	2 - 35	45 °C	60 sec
Extension		72 °C	90 sec
Final extension	1	72 °C	5 min
Hold	1	4 °C	indefinite

## 2.2.2.6 Phylogenetic analysis of nematode ITS1 sequences

The plasmid DNA obtained in Section 2.2.2.4 was sequenced by a commercial sequencing company (LGC Genomics, Berlin, Germany). Sequencing was carried out using M13 forward and reverse primers. Vector sequences were identified using VecScreen (http://www.ncbi.nlm.nih.gov/VecScreen/VecScreen.html) and the sequences were manually edited to remove vector sequences. A consensus sequence of the clones for each individual isolate was obtained using CAP3 (http://pbil.univ-lyon1.fr/cap3.php) (Huang & Madan, 1999).
An initial multiple sequence alignment was carried out using the six isolates from North Bull Island in MEGA5 (Tamura et al., 2011) using ClustalW (Thompson et al., 1994; Larkin et al., 2007). Homologous nucleotide sequences in GenBank were identified the blastn algorithm in BLAST using (http://blast.ncbi.nlm.nih.gov/Blast.cgi) (Altschul et al., 1997). Table 2.5 contains a full list of the Heterorhabditis spp. sequences retrieved from GenBank (http://www.ncbi.nlm.nih.gov/Genbank/). Caenorhabditis elegans was included as an outgroup and a multiple sequence alignment with a total of 29 sequences was carried out in MEGA5 using ClustalW. The resulting alignment was trimmed to eliminate unaligned ends to leave 1012 positions shared by all sequences. Phylogenetic analysis was carried out in MEGA5, with the choice of model determined using the "Find Best DNA/Protein Models (ML)" option in the MEGA5 program. Evolutionary history was estimated using the maximum likelihood method (Kimura, 1980) based on the Kimura 2-parameter nucleotide substitution model with a Gamma distribution used to model non-uniform rates between sites. The parameters used for phylogenetic analysis are shown in Appendix Table A 2. A phylogenetic tree using bootstrap analysis with 1000 replicates was constructed.

Species /strain	GenBank accession no.	Bacterial symbiont ( <b>Photorhabdus</b> sp.)	Geographical origin	Reference
Heterorhabditis a t a c	amensis			
D099	HM230723	unknown	Chile	(Edgington <i>et al.,</i> 2011)
Heterorhabditis down	nesi			
H3107 <sup>T</sup>	EU921443	P. temperata subsp. cinerea	Hungary	(Tóth & Lakatos, 2008)
H3173	EU921444	P. temperata subsp. temperata	Hungary	(Tóth <i>et al.,</i> 2008)
K122	AY321482	P. temperata subsp. temperata	Ireland	(Nguyen <i>et al.,</i> 2004)
K122	EF043442	P. temperata subsp. temperata	Ireland	(Regeai & Burnell 2006, unpublished)
Heterorhabditis man	elatus			
OH10	AY321479	unknown	USA	(Nguyen <i>et al.,</i> 2004)
	EF043441	unknown	USA	(Regeai & Burnell 2006, unpublished)
Heterorhabditis meg	idis			
AGC	AY321480	unknown	USA	(Nguyen <i>et al.,</i> 2004)
Andong	AY293284	unknown	Korea	(Kang <i>et al.</i> 2003, unpublished)
HF85	EF043439	H. megidis	The Netherlands	(Regeal & Burnell 2006, unpublished)
H3014	EU921441	P. temperata subsp. ainerea	Hungary	(Tóth & Lakatos, 2008)
NIHHS-KHH05	AB698759	unknown	Korea	(Kim <i>et al.</i> 2012, unpublished)
N-UK76	HQ225905	P. temperata subsp. temperata	UK	(Maneesakorn <i>et al.,</i> 2011)
	KJ938577	unknown	Switzerland	(Campos-Herrera <i>et al.</i> 2014,
Heterorhabditis safric	ana			anpablisheay
SF52	EU860188	unknown	South Africa	(Malan & Moore, 2008, unpublished)
SF670	FJ473361	unknown	South Africa	(Malan 2008, unpublished)
SF724	FJ791249	unknown	South Africa	(Malan 2008, unpublished)
	EF488006	unknown	South Africa	(Malan <i>et al.,</i> 2008)
Heterorhabditis zeala	ndica			
Florida	AY321481	unknown	USA	(Nguyen <i>et al.,</i> 2004)
J37	EU727165	unknown	South Africa	(de Waal & Malan, 2008, unpublished)
NZH3	EF530041	unknown	New Zealand	Nguyen 2007, unpublished
X1	AY170330	unknown	New Zealand	Qui <i>et al.</i> 2006 unpublished

Table 2.5: Heterorhabditis	species and	strains u	ised in l	ITS1	analysis.
----------------------------	-------------	-----------	-----------	------	-----------

### 2.2.3 Symbiotic bacteria characterisation

### 2.2.3.1 Isolation of symbiotic bacteria

#### a) Mass isolation from nematodes

Prior to extracting the symbiotic bacteria the nematodes were treated with 1% NaOCl to desheath them, as described by Campbell & Gaugler (1991a; 1991b). Briefly, 6 ml of nematode suspension (1000 IJs /ml) was allowed to settle, 0.5 ml of the supernatant was removed and replaced by 0.5 ml 12% NaOCl (final concentration 1%), inverted three times and allowed to stand for 5 min. The nematode suspension was filtered (Supor<sup>®</sup> 200 membrane filter, Pall Life Sciences, USA) under vacuum and rinsed five times with 1 ml sterile Ringers solution per rinse. The nematodes were transferred to a sterile cold glass slide and crushed with a sterile glass rod. The crushed IJs were then streaked onto nutrient agar (NA) (Oxoid Ltd, Basingstoke, UK), nutrient agar supplemented with bromothymol blue (Sigma-Aldrich) (0.025 g/l) and triphenyltetrazolium chloride (Sigma-Aldrich) (0.04 g/l) (NBTA) or MacConkey agar (Oxoid Ltd) and incubated in the dark at 27 °C for 36 – 48 hr. (See Appendix Table A 3 for culture media composition).

# b) Isolation from individual infective juveniles

IJs were surface sterilised by adding 1 ml Hyamine 1622 solution (0.16 mM) (Sigma-Aldrich) to nine ml of nematode suspension (1000 IJs/ml), inverting 3 times and allowing to settle. The supernatant was discarded and the IJs were rinsed by adding 9 ml sterile phosphate buffered saline (PBS), inverting 3 times and allowing to settle. The rinse was repeated twice. Individual IJs were pipetted in 10  $\mu$ l aliquots with the aid of a dissecting microscope and added to 90  $\mu$ l PBS in a 1.5 ml microcentrifuge tube. Each IJ was crushed with a sterile micropestle for one minute. A 50  $\mu$ l aliquot of the product was plated onto Luria-Bertani (LB) agar (Difco)

supplemented with 0.1% sodium pyruvate (Sigma-Aldrich) (LB pyruvate agar) and incubated in the dark at 27 °C for 48 hr. The experiment was carried out once for each of the potential YEL isolates from the 2012 North Bull Island survey, with 10 IJs of each isolate.

### c) Isolation from haemolymph of infected insects

Infections to recover symbiotic bacteria from insect haemolymph were set up as described in Section 2.2.1.7, using 5.5 cm Petri dishes and 3 insects per isolate. Infected cadavers were checked for bioluminescence after 72 hours. Bacteria were recovered from all cadavers positive for *H. downesi* infection as follows: the cadaver was surface sterilised by washing in 95% ethanol, flaming and extinguishing by plunging into sterile distilled water. The cadaver was opened under aseptic conditions and a loopful of haemolymph was streaked onto NBTA or MacConkey agar plates and incubated in the dark at 27 °C for 36 - 48 hours (Kaya & Stock, 1997).

# 2.2.3.2 Routine culturing of symbiotic bacteria

Bacterial colonies were stored in the dark at 27 °C or at 15 °C and subcultured onto NA or MacConkey agar plates every 4 days or once a month, respectively. Liquid cultures were set up by inoculating single colonies from 36 - 48 hr plate cultures into 50 ml nutrient broth (NB) or MacConkey broth (MB) in a 250 ml flask and growing for 24 - 48 hr in the dark with shaking at 200 rpm at 27 °C. The optical density (OD) at 660 nm of liquid cultures was recorded as required using a Shimadzu UV mini 1240 Spectrophotometer (Fischer Scientific). Working and long-term stock cultures of 6 individual isolates of the bacteria were established by adding 0.5 ml of 24 hr liquid culture to 0.5 ml NB premixed with glycerol (30%), vortexing briefly and

storing at -20 °C and -80 °C respectively. Bacteria were recovered as required from frozen stocks by plating onto MacConkey agar and incubating in the dark at 27 °C.

## 2.2.3.3 Characterisation of symbiotic bacteria

## a) Morphological characterisation

Bacteria isolated either directly from nematodes or from the haemolymph of infected insects were designated *pur* and *yel* according to the colour phenotype of the infected cadaver, PUR and YEL respectively, (Figure 2.3) and were identified as *Photorhabdus* sp. on the basis of bioluminescence. Bioluminescence was assessed by observing colonies in a darkened room for 10 min. Bacteria were subcultured onto NA plates as described in Section 2.2.3.2 and grown in the dark at 27 °C for 36 – 48 hr until colonies visible to the naked eye were obtained. Stock strains of *Escherichia coli* (Gram-negative) and *Bacillus subtilis* (Gram-positive), obtained from the Preparation Laboratory, Biology Department, NUI Maynooth (NUIM), were used as reference strains for Gram staining. Gram staining, cell morphology and cell motility were compared under a Nikon OPTIPHOT-2 light microscope using a 100X oil immersion lens. Colony morphology was examined under a dissecting microscope.

### b) Microbial growth - direct cell count

One isolate of each colour phenotype was chosen to investigate growth rate. One liquid culture per isolate was set up as described in Section 2.2.3.2. An aliquot of cell suspension (1 ml) was drawn after 10 hours growth. A second 1 ml sample was drawn after a further 6 hours growth. At each time point a 1 in 10 serial dilution in <sup>1</sup>/<sub>4</sub> strength sterile Ringers solution was prepared until a suitable cell density for counting was obtained. To immobilise the cells for counting 0.5 ml of each dilution was added to 0.5 ml HCl (0.1 N) (Gerhardt, 1981) in a fresh tube. Samples were

placed on ice until counted. Cells were vortexed briefly before counting on a haemocytometer counting chamber under Nikon OPTIPHOT-2 light microscope using a 40X lens. A 10  $\mu$ l aliquot of the cell suspension was loaded into each chamber and cells were counted in the four corner squares. The average count for the two chambers was multiplied by the dilution factor and then by 10<sup>4</sup> to allow for the volume of the counting chamber. The dilution factor was calculated as 1/0.5 \* dilution to allow for the addition of HCl as described above. Growth rate (mean generation time) (g) was calculated using the following formula:

 $g = (\log_{10} N_t - \log_{10} N_0) / \log_{10}2$ where  $N_0 = \text{number of cells at time } t_0$  $N_t = \text{number of cells at time } t_1$  $t_0 = \text{initial time (hrs)}$  $t_1 = \text{final time (hrs)}$ 

From the growth rate the doubling time (hours/generation) (t/g) was calculated as

$$t / g$$
  
where  $t = t_1 - t_0$ 

This experiment was repeated once with cultures grown in NB and samples taken at 10 hours and 22 hours. The experiment was also run with cultures grown in MB and samples taken at 10 hours and 34 hours.

## c) Microbial growth – colony count

Samples for spread plating were drawn from the cultures in NB in Section 2.2.3.2b. Aliquots (100  $\mu$ l) of each of three dilutions were spread on triplicate NA plates (9 cm) at the initial time point (10 hrs). In addition, a final set of samples were drawn

from the cultures in experiment 2 at 64 hours. The final samples were spread on a selection of NA, NBTA and MacConkey agar plates. All plates were incubated at  $27^{0}$  C for 48 hours before counting. Plates which had fewer than 30 colonies or more than 350 colonies were not counted.

### d) Biochemical characterisation

Bacteria were subcultured onto NA plates as described in Section 2.2.3.2 and grown for 36 – 48 hours in the dark at 27 °C. Bacterial identification kits (API 20 E, API 50 CH) (bioMerieux, France) were used to analyse the biochemical and physiological properties of each isolate. Unless otherwise stated all reagents were supplied by bioMerieux. Test strip trays were prepared using distilled water according to the manufacturer's instructions. The API 20E strips (12 tests) were inoculated with a suspension of bacterial culture made by dispersing 1-4 single, well-isolated colonies in API 20E NaCl 0.85% medium (5 ml) to 0.5 McFarland. The test strips were incubated at 27° C and checked daily for 5 days. Tests for tryptophan deaminase, indole production, acetoin production and reduction of nitrates were carried out at the end of the incubation period. The API 50 CH strips (49 tests) were inoculated with a bacterial culture in API 50 CHB/E medium made by dispersing up to 20 single, well-isolated colonies in 1 ml sterile distilled water to 4 McFarland. This was then added to a vial of API 50 CHB/E medium and mixed thoroughly before use. The test strips were incubated at 27° C and checked daily for 5 days. The cytochrome oxidase activity was tested using tetramethyl-*p*-phenylenediamine dihydrochloride (Sigma-Aldrich). The test was carried out by soaking 1 cm squares of filter paper in a 1% aqueous solution of the reagent, allowing to dry, then transferring a colony from the NA plates to the paper using a sterile glass rod (Collins & Lyne, 1984).

### 2.2.3.4 Antibiosis

Bacterial isolates were screened for inhibition of bacteria and fungi using a range of organisms. Target organisms and culturing conditions are listed in Table 2.6. Inhibition was assessed by measuring the zone of inhibition at four points around the spot colony and taking the mean of the four measurements (bacterial test organisms) (Figure 2.6) or by measuring the zone of inhibition at 3 points along the bacterial streak (fungal test organisms) (Figure 2.7). Antibacterial tests were based on the method of Boemare & Lanois (1996). Briefly, 50 µl of 24 hr culture was spot inoculated onto NA in a 100 mm glass Petri dish (Fisher Scientific) and incubated in the dark at 27 °C for 48 hr. Immediately prior to the test the lid of the Petri dish was flooded with chloroform and the dish inverted over the lid for 30 min to kill the colony. The plates were left uncovered in a laminar flow hood for a further 30 min to allow the chloroform to evaporate. Soft agar (7 g agar / L dH2O) was inoculated with the target organism (1 ml / 50 ml agar) and poured onto the chloroform-killed spot colony. A negative control which had no spot colony was included in each treatment. Plates were examined under a Nikon OPTIPHOT-2 light microscope using a 40X lens after 20 hr (Escherichia coli, Bacillus subtilis), or 40 hr (Micrococcus luteus, Xenorhabdus bovienii, Photorhabdus isolates). For intraspecific tests pur/yel combinations were chosen by drawing lots and each combination was reversed so that all isolates were used as target organisms. Each test had three replicates and each test was run three times with different cultures in each experiment



Figure 2.6: Diagram showing zone of inhibition and four measuring points in antibacterial test

For antimycotic tests, the method of Ansari *et al.* (2005) was followed. Briefly, a loopful of *Photorhabdus* culture was streaked in a line 5 cm long and 3 cm from one edge of a 9 cm NA plate and incubated in the dark at 27 °C for 48 hr. One mycelial plug (1 cm square) was placed in line with the midpoint of the bacterial streak and centered 3cm from the bacterial streak and the wall of the Petri dish. A negative control which had no bacterial streak was included in each treatment. Plates were incubated in the dark at 25 °C and checked at 7 day intervals for a period of 28 days. Each test had three replicates and each test was run three times with different cultures in each experiment.



**Figure 2.7:** Diagram showing experimental setup with *P. temperata* streak, mycelial plug, fungal growth from the inoculating mycelial plug, zone of inhibition and three measuring points in antimycosis test.

Test organism	incubation temperature	incubation period
Escherichia coli <sup>1</sup> *	37° C	24 hr
Bacillus subtilis <sup>1</sup> *	37° C	24 hr
Micrococcus luteus <sup>1</sup> *	26° C	48 hr
Xenorhabdus bovienii <sup>2</sup> *	27° C	48 hr
Photorhabdus isolates <sup>3</sup> *	27° C	48 hr
Mucor sp. <sup>1</sup> †	25° C	7 days
Beauveria bassiana <sup>2</sup> ‡	25° C	14 days
Metarhizium anisopliae <sup>2</sup> ‡	25° C	14 days

Table 2.6: Target organisms and culturing conditions for used in antibiosis experiments

1. Obtained from the Preparation Laboratory, Biology Department, NUIM

2. Obtained from colleagues in the Behavioural Ecology Laboratory, NUIM

3. This study.

\* Cultures were grown on NA plates and 24 hr before the test a single was picked into 50 ml NB in a 250 ml flask and incubated with shaking at 200 rpm at the appropriate temperature for 24 hr.

<sup>†</sup> Cultures were grown on Malt Extract agar (Oxoid)

Cultures were grown on Sabouraud Dextrose Agar (SDA) (Oxoid)

### 2.2.4 Phylogenetic analysis of symbiotic bacteria

### 2.2.4.1 DNA extraction from symbiotic bacteria

Bacterial DNA was extracted using a Qiagen DNeasy Blood and Tissue 50 spin column kit. All reagents were supplied with the kit unless otherwise stated. Liquid cultures were set up as described in Section 2.2.3.2 in 5 ml of LB broth, Miller, (Luria-Bertani) (BD Difco, NJ, USA) and incubated overnight at  $27^{\circ}$  C with shaking at 200 rpm. Absorbance at 660 nm was checked and cell suspensions were adjusted to give approximately 2 x  $10^{9}$  cells by diluting in <sup>1</sup>/<sub>4</sub> strength sterile Ringers solution. (See Appendix Figure A. 2 for standard curves.) Cells were collected by spinning in a microcentrifuge at 5000 x *g* for 10 min, the supernatant was discarded and the extraction was then completed following using the DNeasy protocol described in Section 2.2.2.1 with the following variation: cells were lysed for 1.5 hr instead of 1 hr.

# 2.2.4.2 PCR amplification of symbiotic bacteria

Amplification of symbiotic bacteria DNA was carried out using primers specific for the *gyrB* gene (Tailliez & Boemare, 2009). Amplification reactions and gel analysis of PCR product were carried out as described in Section 2.2.2.2 with the following exceptions: cycling conditions (Table 2.7), PCR mixture (Table 2.8), forward and reverse primers (Table 2.9).

Table	2.7:	PCR	cycling	conditions	for	amplification	n of	gyrB	gene	from
symbic	otic ba	acteria	isolated	from H. do	wnes	<i>i</i> nematodes	from	North	Bull	Island
(Taillie	ez & E	Boema	re, 2009).	. Lid temper	rature	was set to 95	5°C.			

Stage	Cycles	Temperature	Duration
Initial denaturation	1	94 °C	5 min
Denaturation		94 °C	1 min
Annealing	30	60 °C	1 min
Extension		72 °C	2 min
Final extension	1	72 °C	7 min
Hold	1	4 °C	indefinite

**Table 2.8:** PCR mixture for amplification of gyrB gene from symbiotic bacteria isolated from *H. downesi* nematodes from North Bull Island (Tailliez & Boemare, 2009). Reactions were carried out in 50 µl volumes. Reagents were supplied by Sigma-Aldrich unless otherwise indicated.

Reagent	Volume (µl)	Final concentration	Supplier
Buffer (10x)	5	1x	
MgCl2 (25mM)	6	3 mM	
dNTP Mix (10mM)	1	200 µM	Metabion
Forward primer (10mM)	1	0.2 μΜ	Eurofins MWG Operon
Reverse primer (10mM)	1	0.2 μΜ	Eurofins MWG Operon
Taq Polymerase	0.25	0.025 U/µl	
DNA template (20 ng/µl)	5	2 ng/µl	
dH2O	30.75		

**Table 2.9:** Primer sequences for *gyrB* gene from symbiotic bacteria isolated from *H. downesi* nematodes from North Bull Island (Tailliez & Boemare, 2009).

Primer	Sequence
gyrBP1 (Forward)	5'-TACACGAAGAAGAAGGTGTTTCAG-3'
gyrBP2 (Reverse)	5'-TACTCATCCATTGCTTCATCATCT-3'

Cloning, plasmid isolation and plasmid analysis was carried out as described in Sections 2.2.2.3 - 2.2.2.5.

# 2.2.4.3 Phylogenetic analysis of gyrB gene sequences

The plasmid DNA obtained in Section 2.2.4.2 was sequenced by a commercial sequencing company (GATC Biotech, Konstanz, Germany). Sequencing was carried out using M13 forward and reverse primers plus four *gyrB* specific primers (Table 2.10) (Boemare & Tailliez, 2009) to obtain overlapping sequences. Sequences were edited as described in Section 2.2.2.6.

**Table 2.10:** Sequencing primers for gyrB gene (Boemare & Tailliez, 2009).

Primer	Sequence
gyrBSP1 (Forward)	5'-GATAACTCTTATAAAGTTTCCG-3'
gyrBSP2 (Reverse)	5'-CGGGTTGTATTCGTCACGGCC-3'
gyrBSP3 (Forward)	5'-CTCTACTTAGTGGAAGGGGA-3'
gyrBSP4 (Reverse)	5'-GCAGTAAATATTTTCCTGGA-3'

An initial multiple sequence alignment was carried out for just the six isolates from North Bull Island using ClustalW (Thompson *et al.*, 1994; Larkin *et al.*, 2007) in MEGA5 (Tamura *et al.*, 2011). Nucleotide sequences were translated and open reading frames in the protein sequence were identified using ORF Finder (http://www.ncbi.nlm.nih.gov/gorf/gorf.html). Homologous nucleotide sequences were assembled as described in Section 2.2.2.6. The resulting alignment was trimmed to eliminate unaligned ends to leave 847 positions shared by all the sequences. Table 2.11 contains a full list of the *Photorhabdus* spp. sequences retrieved from GenBank (http://www.ncbi.nlm.nih.gov/Genbank/). Phylogenetic analysis was carried out in MEGA5, with the choice of model determined using the "Find Best DNA/Protein Models (ML)" option in the MEGA5 program. Evolutionary history was estimated using the maximum likelihood method, based on the Kimura 2-parameter nucleotide substitution model (Kimura, 1980) with a Gamma distribution used to model non-uniform rates between sites. The parameters used for phylogenetic analysis are shown in Appendix Table A 4. A phylogenetic tree using bootstrap analysis with 1000 replicates was constructed, using *X. nematophila* as an outgroup.

Species	GenBank	Nematode host	Geographic	Reference
/strain	accession no.	( <b>Heterornabditis</b> sp.)	alongin	
P. temperata subsp. ar		11 day mari		(Táth 8 Lalatas 2009)
H3107	EUU053108	H. downesi		(TOUT & Lakatos, 2008)
H3240	EU0531/4	H. downesi	Hungary	(Toth & Lakatos, 2008)
H3014	EU155110	H. Megiais	Hungary	(TOTN & LAKATOS, 2008)
		11 hardenia harren NG1*		(4)
	AY278497	H. bacteriopriora NC1*	USA	(Akhurst <i>etal.,</i> 2004)
IVIEg	AY278512	H. megiais	USA	(Aknurst <i>et al.</i> , 2004)
Habana	AY278503	Heteromabalitis sp.	Cuba	(Akhurst <i>et al.,</i> 2004)
P. temperata subsp. ta	ismaniensis			
NZH3	AY278513	H. zealandica	NewZealand	(Akhurst <i>et al.</i> , 2004)
T327	EU930356	H. zealandica	Australia	(Tailliez <i>et al.</i> , 2010)
USCA01	EU930357	H. marelatus	USA	(Tailliez <i>et al.,</i> 2010)
P. temperata subsp. te	mperata		7	
HF85	AY278502	H. megidis	The Netherlands	(Akhurst <i>et al.,</i> 2004)
HL81	AY278504	H. megidis	Netherlands	(Akhurst <i>et al.,</i> 2004)
HW79	AY278507	H. megidis	ne Netherlands	(Akhurst <i>et al.,</i> 2004)
X1UT	AY278516	H. megidis	Lithuania	(Akhurst <i>et al.,</i> 2004)
X1Nach <sup>T</sup>	AY278517	H. megidis	Russia	(Akhurst <i>et al.,</i> 2004)
H3016	EU053166	H. megidis	Hungary	(Tóth <i>et al.,</i> 2008)
H3173	EU053169	H. downesi	Hungary	(Tóth <i>et al.,</i> 2008)
H3179	EU053170	H. downesi	Hungary	(Tóth <i>et al.,</i> 2008)
H3182	EU053171	H. downesi	Hungary	(Tóth <i>et al.,</i> 2008)
K122	EU930355	H. downesit	Ireland	(Tailliez <i>et al.,</i> 2010)
P. temperata subsp. th	nracensis			
H3210	EU053173	H. bacteriophora	Hungary	(Tóth <i>et al.,</i> 2008)
CIP108426 <sup>T</sup>	EU930351	H. bacteriophora	Turkey	(Tailliez <i>et al.,</i> 2010)
FR32	EU930352	entomopathogenic nematode	France	(Tailliez <i>et al.,</i> 2010)
P. temperata sp.				
H111	EF029048	H. megidis	Hungary	(Tóth <i>et al.,</i> 2008)
H267	EF029049	H. downesi	Hungary	(Tóth <i>et al.,</i> 2008)
H295	EF029050	H. megidis	Hungary	(Tóth <i>et al.,</i> 2008)
Photorhabdus sp.				
H3086	EU053167	H. downesi	Hungary	(Tóth <i>et al.,</i> 2008)

**Table 2.11:** *Photorhabdus* species and strains used in *gyr*B analysis.

\* The naming convention used by Akhurst et al (2004) is followed here.

## 2.2.5 Statistical analysis

Routine statistical analysis, including tests for normality, was carried out using Minitab statistical software, version 16.1.1 (Minitab Inc., 2010). Normality of data was tested using the Anderson-Darling method. Comparisons between two groups were carried out using a 2-Sample t-test or a paired t-test where appropriate. If the data were not normally distributed a Mann-Whitney U test was used. Analysis of variance was carried out using a one-way ANOVA or when groups had an uneven number of observations the general linear model with an unbalanced nested design was used to compare isolate and colour phenotype. Analysis of variance was followed by pair-wise multiple comparisons using Tukey's test. In experiments where nematode type and bacterial type were factors, results were analysed using a two-way ANOVA. Where no significant interactions occurred treatment means for each factor were compared using a one-way ANOVA and post-hoc Tukey's test for three or more treatments or 2-sample T-test for two treatments. When data could not be normalised treatment medians were compared using a Kruskal-Wallis test, followed by multiple pairwise comparisons using a Mann-Whitney U-test test with Bonferroni 95% confidence intervals, obtained by calculating  $(1 - \alpha / g)$  where  $\alpha$  is the level of significance and g is the number of pairwise comparisons carried out. In all tests a P value of < 0.05 was taken as significant unless indicated otherwise.

# 2.3 **Results**

### 2.3.1 Entomopathogenic nematodes on North Bull Island

## 2.3.1.1 Detection of entomopathogenic nematodes

Three transects from the seaward side to the landward side of the dunes on North Bull Island were sampled during September and October 2008. Two species of entomopathogenic nematode were recovered: *Heterorhabditis downesi* and *Steinernema feltiae*. A total of 35/42 (83.3%) of the soil samples were positive for nematodes when baited with *G. mellonella* (Table 2.12). Overall, *H. downesi* was detected in 13/42 (31%) of the soil samples and *S. feltiae* was detected in 34/42 (81%) of the soil samples (Table 2.12).

**Table 2.12:** Number of soil samples positive for entomopathogenic nematodes from North Bull Island. Two species were detected, *H. downesi* and *Steinernema feltiae*. Each transect consisted of 14 soil samples. Each sample was baited twice and each baiting consisted of 15 *G. mellonella*. (n = 90 bait insects per sampling location)

Transect	positive for EPN	H. downesi	S. feltiae
1	12	7	12
2	11	1	10
3	12	5	12
Total	35	13	34

A higher proportion of bait insects were killed by *S. feltiae* than *H. downesi*, 131/1260 (10.4%) and 22/1260 (1.7%) respectively (Table 2.13). Nematodes were more likely to be detected in the first baiting than in the second, 107/630 and 46/630 respectively (Table 2.14). This was true for both *H. downesi* (20/630 and 2/630 for baiting 1 & 2 respectively) and for *S. feltiae* (87/630 and 44/630 for baiting 1 & 2 respectively). Two sampling locations, 0 m and 10 m, were positive for in the second baiting only and only for *S. feltiae*.

	Number of d	lead bait insects	(/30)	Number of de	ead bait insects (/
Distance (m)*	transect 1	transect 2	transect 3	H. downesi	S. feltiae
0	1	0	0	0	1
10	0	0	1	0	1
20	5	3	11	3	16
30	2	1	0	1	2
40	11	3	2	4	12
50	3	1	1	1	4
60	5	3	2	2	8
70	6	1	4	2	9
80	4	6	6	4	12
90	17	2	3	2	20

**Table 2.13:** Number of dead bait insects recovered from soil samples from North Bull Island. Each sample was baited twice and each baiting consisted of 15 *G. mellonella.* (n = 30 bait insects per sampling location in each transect, giving a total n of 90 bait insects per distance (10 m) into the dunes)

\* The first sampling location in each transect (0 m) was 2 metres above the strand line.

total

	H. down	esi	S. feltiae	
Distance (m)*	bait 1	bait 2	bait 1	bait 2
0	0	0	0	1
10	0	0	0	1
20	3	0	9	7
30	1	0	2	0
40	4	0	6	6
50	1	0	4	0
60	2	0	5	3
70	1	1	8	1
80	3	1	6	6
90	2	0	12	8
100	0	0	13	2
110	0	0	5	1
120	3	0	9	6
130	0	0	8	2
total	20	2	87	44

**Table 2.14:** Detection of nematodes in first and second baiting. Each sample was baited twice and each baiting consisted of 15 *G. mellonella*. Data is pooled for three transects. (n = 30 bait insects per sampling location in each transect, giving a total n of 90 bait insects per distance (10 m) into the dunes.

\* The first sampling location in each transect (0 m) was 2 metres above the strand line.

# 2.3.1.2 Occurrence of *H. downesi* colour phenotypes - 2008

Two colour phenotypes of *H. downesi* were detected, designated PUR and YEL on the basis of cadaver colour (see Section 2.2.1.2, Figure 2.3) and pigmentation of the symbiotic bacteria on NA plates. The PUR phenotype was detected in all three transects whereas the YEL phenotype was detected in transect 1 only, and there was no overlap of the two colour phenotypes in this transect.

In transect 1 the PUR phenotype was detected in four soil samples, ranging in distance from 20 m to 60 m into the dune system and the YEL phenotype was

detected in three soil samples, ranging in distance from 80 m to 120 m into the dune system (Figure 2.8A). In transect 2 the PUR phenotype was detected in only one soil sample, at 50 m (Figure 2.8B), and in transect 3 the PUR phenotype was detected at 40 m, 60 m, 70 m, 80 m and 90 m (Figure 2.8C). The number of bait insects killed per sample ranged from 1 (1/30, 3%) to 3 (3/30, 10%) (Figures 2.8A-C).



**Figure 2.8** Percentage mortality of bait insects due to *H. downesi* PUR and YEL colour variants in A - transect 1, B - transect 2 and C - transect 3, North Bull Island. Each soil sample consisted of 30 pooled subsamples and was baited twice with 15 *G. mellonella* in each baiting. (n = 30). The first sampling location in each transect (0 m) was 2 metres above the strand line.

Bait insects from transect 1 were chosen to establish stock cultures of *H. downesi* as described in Section 2.2.1.3. Where multiple insects were infected in a particular soil sample one cadaver was chosen at random. Table 2.15 lists the isolates used to establish stock nematode cultures, along with the distance into the sand dunes of the soil sample from which they were recovered, and the symbiotic bacteria associated with each nematode isolate.

**Table 2.15:** *H. downesi* and *P. temperata* isolates from North Bull Island used in this study. The colour phenotype is indicated by the isolate name: PUR and *pur* for purple phenotype nematode and symbiotic bacteria respectively and YEL and *yel* for the yellow phenotype nematode and symbiotic bacteria respectively. Distance\* indicates the distance into the sand dunes of the soil sample from which the nematode was recovered. All isolates listed in the table were isolated from transect 1.

	isolates			
distance (m)*	nematode	bacteria		
20	PUR1	pur1		
30	PUR2	pur2		
60	PUR3	pur3		
80	YEL1	yel1		
90	YEL2	yel2		
120	YEL3	yel3		

\* The first sampling location in each transect (0 m) was 2 metres above the strand line.

### 2.3.1.3 Repeat sampling 2012

In 2012 transect 1 was re-sampled (Table 2.17). Both *H. downesi* and *S. feltiae* were detected. On this occasion both *H. downesi* colour phenotypes were found to be co-occurring in the front of the dunes. A total of 26/48 (54.2%) of the soil samples were positive for nematodes when baited with *G. mellonella*. Overall, *H. downesi* was detected in 9/48 (18.8%) and *S. feltiae* was detected in 17/48 (35.4%) of the soil samples, and more nematodes were detected in the first baiting than in the second

(27/48 and 2/48, respectively). The PUR phenotype was more abundant than the YEL in the front section of the dunes, with the PUR phenotype being detected in five of the individual soil samples, ranging in distance from 20 to 70 m into the dune system and the YEL phenotype being detected in four of the individual soil samples, ranging in distance from 20 to 130 m into the dune system. Overall there was no difference in the number of bait insects positive for either phenotype, as one soil sample was positive for the YEL phenotype in both baitings.

## 2.3.1.4 Distribution of *H. downesi* colour phenotypes

Overall, 13 sampling locations were positive for *H. downesi* in 2008 and 7 sampling locations were positive in 2012. Although there was some overlap in their occurrence the Pur phenotype tended to occur in the front section of the dunes while the Yel phenotype was found mainly in the rear section of the dunes and the distribution of the two phenotypes was significantly different when positive sampling locations (distance into dunes) of each colour phenotype were compared were compared (Mann Whitney U test, W = 101.5, P = 0.006). The first positive sample in 2012 was at 20 m above the strand line, in embryonic dunes, forward of the first dune ridge. In 2008 the first soil sample was taken at 2 m above the strandline. In 2012 the first soil sample was again taken at 2 m above the strandline, however, in 2012 the dunes had advanced such that starting from a point 2 m above the strandline two extra soil samples were taken on the seaward side of the dunes in order to reach the same end point in the rear of the dunes (Table 2.17). Cadavers were scored positive for *H. downesi* on the basis of bioluminescence. It was difficult to discriminate cadaver colour in some instances as the YEL cadavers were pale and early in an infection PUR cadavers tend to be achromatic. The YEL colour phenotype was confirmed by carrying out individual bacterial extractions from 10 IJs (Sections 2.2.3.1b above) for each isolate except for one of the positive cadavers at 20 m and the one at 90 m as there was no emergence from these two cadavers. These two cadavers were scored as YEL phenotypes on the basis of the bacterial extractions from the other potential YEL cadavers (data not shown) which yielded *yel* bacterial phenotype in all instances since all five cadavers had the same appearance.

**Table 2.16:** Number of dead bait insects recovered from soil samples from North Bull Island in 2012. Three individual soil samples were taken at each sampling location and each sample was baited twice with a single *G. mellonella* larva at each baiting (n = 6 bait insects per distance (10 m) into the dunes).

	Number of dead bait insects $(/3)$			
Distance (m)*	bait 1		bait	t 2
	H. downesi	S. feltiae	H. downesi	S. feltiae
0	0	0	0	0
10	0	0	0	0
20	2	0	0	0
30	2	0	0	0
40	1	1	1	0
50	0	2	0	0
60	0	0	0	0
70	1	0	0	0
80	0	1	0	0
90	0	1	1	0
100	0	2	0	0
110	1	1	0	0
120	0	3	0	0
130	1	1	0	0
140	0	3	0	0
150	0	2	0	0
Total	8	17	2	0

\* The first sampling location in each transect (0 m) was 2 metres above the strand line

**Table 2.17:** Detection of *H. downesi* PUR and YEL colour phenotype in sand dunes on North Bull Island in two surveys. Data shown is number of dead bait insects recovered from soil samples in transect 1. In  $2008^*$  each sample was baited twice and each baiting consisted of 15 *G. mellonella*. (n = 30 bait insects per distance). In 2012 each sample was baited twice and each baiting consisted of 1 *G. mellonella*. (n = 6 bait insects per distance).

		Transect 1			
dista	nce (m)	2008 2012		12	
2008	2012	PUR	YEL	PUR	YEL
	0			0	0
	10			0	0
$0$ $^{\dagger}$	20	0	0	2	0
10	30	0	0	2	0
20	40	3	0	0	2 ‡
30	50	1	0	0	0
40	60	3	0	0	0
50	70	0	0	1	0
60	80	1	0	0	0
70	90	0	0	0	1 ‡
80	100	0	1	0	0
90	110	0	1	0	1 ‡
100	120	0	0	0	0
110	130	0	0	0	1
120	140	0	3	0	0
130	150	0	0	0	0
Total		8	7	5	5

\* In 2008 30 subsamples per sampling location were pooled, mixed and then divided into three for baiting, with a total of 15 insects per distance per baiting. In 2012 three subsamples per sampling location were baited individually, with a total of 3 insects per distance per baiting.

<sup>†</sup> In both 2008 and 2012 the first soil sample was taken at 2 m above the strandline (= distance 0). In 2012 the dunes had advanced seaward such that two extra soil samples were taken to reach the same end point in the rear of the dune system.

‡ It was difficult to discriminate cadaver colour as the YEL cadavers were pale in some instances. The colour phenotype was confirmed by carrying out a bacterial extraction from 10 IJs for each isolate except for one of the positive cadavers at 20 m and the one at 90 m as there was no emergence from these two cadavers.

## 2.3.1.5 Habitat survey

The most common composite habitat recorded was Marram Dunes/Semi-Fixed Dunes/Fixed Dunes/Grassy Path, recorded at 10/42 sampling locations, followed by Marram Dunes/Semi-Fixed Dunes/Sandy Path, recorded at 8/ 42 sampling locations (Table 2.18). No *H. downesi* were detected in the categories that included Embryonic Dune habitat. Both colour phenotypes were detected in Marram and Semi-Fixed/Fixed Dune habitats that did or did not include paths. There was no clear difference between categories in the detection of the two colour phenotypes. The flora, soil temperature and pH records are given in Appendix Table A 5.

**Table 2.18**: Composite habitat types recorded at soil sampling locations on North Bull Island. The number of EPN infected bait insects is given. Detection of *H. downesi* PUR and YEL phenotypes is indicated by purple and yellow shading respectively. Absence of shading indicates detection of *S. feltiae* only. The first sampling location in eachtransect (0 m) was 2 metres above the strand line.

Distance (m	) Transect 1	Transect 2	Transect 3
0	Embry onic Dunes*	Embryonic Dunes	Embry onic Dunes
10	Embryonic Dunes, Marram Dunes, Sandy Path	Marram Dunes, Semi-Fixed Dunes, Sandy Path	Embryonic Dunes, Marram Dunes*
20	Marram Dunes, Semi-Fixed Dunes, Sandy Path*	Marram Dunes, Semi-Fixed Dunes, Fixed Dunes, Sandy Path*	Marram Dunes, Semi-Fixed Dunes, Sandy Path*
30	Marram Dunes, Semi-Fixed Dunes, Sandy Path*	Marram Dunes, Semi-Fixed Dunes, Fixed Dunes, Sandy Path*	Marram Dunes, Semi-Fixed Dunes, Sandy Path
40	Marram Dunes, Semi-Fixed Dunes, Sandy Path*	Marram Dunes, Semi-Fixed Dunes, Fixed Dunes*	Marram Dunes, Semi-Fixed Dunes, Fixed Dunes*
50	Marram Dunes, Semi-Fixed Dunes, Sandy Path*	Marram Dunes, Semi-Fixed Dunes, Fixed Dunes, Grassy Path	Marram Dunes, Semi-Fixed Dunes, Sandy Path*
60	Marram Dunes, Semi-Fixed Dunes, Fixed Dunes, Sandy Path, Blowout*	Marram Dunes, Semi-Fixed Dunes, Fixed Dunes, Grassy Path*	Marram Dunes, Semi-Fixed Dunes, Fixed Dunes, Grassy Path*
70	Marram Dunes, Semi-Fixed Dunes, Fixed Dunes, Sandy Path*	Marram Dunes, Semi-Fixed Dunes, Fixed Dunes, Grassy Path*	Marram Dunes, Semi-Fixed Dunes, Fixed Dunes, Grassy Path*
80	Marram Dunes, Semi-Fixed Dunes, Fixed Dunes, Sandy Path*	Marram Dunes, Semi-Fixed Dunes, Fixed Dunes, Grassy Path*	Marram Dunes, Semi-Fixed Dunes, Sandy Path, Grassy Path*
90	Marram Dunes, Semi-Fixed Dunes, Fixed Dunes*	Marram Dunes, Semi-Fixed Dunes, Fixed Dunes, Sandy Path, Grassy Path*	Marram Dunes, Semi-Fixed Dunes, Fixed Dunes, Sandy Path, Grassy Path*
100	Semi-Fixed Dunes, Fixed Dunes*	Marram Dunes, Semi-Fixed Dunes, Fixed Dunes, Grassy Path*	Marram Dunes, Semi-Fixed Dunes, Sandy Path, Grassy Path*
110	Semi-Fixed Dunes, Fixed Dunes, Grassy Path*	Marram Dunes, Semi-Fixed Dunes, Fixed Dunes, Grassy Path	Marram Dunes, Semi-Fixed Dunes, Grassy Path*
120	Semi-Fixed Dunes, Fixed Dunes, Grassy Path*	Marram Dunes, Semi-Fixed Dunes, Fixed Dunes, Sandy Path*	Marram Dunes, Semi-Fixed Dunes, Fixed Dunes, Grassy Path*
130	Semi-Fixed Dunes, Fixed Dunes*	Marram Dunes, Semi-Fixed Dunes, Fixed Dunes, Grassy Path*	Marram Dunes, Semi-Fixed Dunes, Grassy Path*

\* Sampling location positive for *S. feltiae*.

### 2.3.2 Phylogenetic analysis of the nematode ITS1 region

An initial alignment in ClustalW of the ITS1 sequences showed very little variation between the six isolates from North Bull Island with only 7 single nucleotide polymorphism in a total of 1031 shared positions (Appendix Figure A. 3). The BLAST search in the GenBank nucleotide database using the blastn algorithm showed that the ITS1 sequences from the PUR and YEL isolates were homologous with sequences within the ITS region from a range of Heterorhabditis species. Alignment of sequences in ClustalW is shown in Appendix Figure A. 4. There was no greater diversity within the North Bull Island isolates than there was between the two K122 isolates. All the North Bull Island nematode isolate shared a greater than 99% identity with each other and for one pair, PUR1 and YEL3 there was 100% identity over 705 bases in the final alignment. Similarly, there was a greater than 99% identity to two the sequences from H. downesi K122 produced by two different laboratories (see Section 2.2.2.6 Table 2.5). There was also a greater than 98% identity with two H. downesi isolates from Hungary, H3107 and H3173, the symbionts of which have been identified as P. temperata subsp. cinerea and P. temperata subsp. temperata, respectively.

The phylogenetic analysis showed that all six North Bull Island isolates belonged to the same cluster within the set of 29 sequences analysed (Figure 2.9). There appeared to be no separation of the PUR and YEL colour variants. The ITS1 sequences formed two main clusters. Cluster I contained all the North Bull Island isolates, the two K122 *H. downesi* isolates, the two Hungarian *H. downesi* isolates, H3107 and H3173, along with all *H. megidis* isolates, with 70% support for this cluster. Within cluster I the North Bull Island isolates, and the *H. downesi* isolates formed a separate clade from the *H. megidis* isolates with 76% support. Cluster II contained all the *H. zealandica* isolates. The other species *H. marelatus, H. atacamensis* and *H. safricana*, formed a separate group that branched more deeply. Neighbour-joining and Minimum Evolution methods of phylogenetic inference showed the same clustering with practically identical bootstrap values for 1000 replications (data not shown).



**Figure 2.9:** Maximum Likelihood phylogenetic tree based on a portion of the ITS1 sequence from six North Bull Island isolates and members of the *Heterorhabditis* species. *C. elegans* was used as an outgroup. Tree shown is the bootstrap consensus tree from 1000 replications.

### 2.3.3 Phenotypic characterisation of symbiotic bacteria

## a) Morphological characterisation

The symbiotic bacteria were isolated from each of the nematode isolates listed in Table 2.15. All isolates were identified as *Photorhabdus* sp. on the basis of bioluminescence. All isolates adsorbed dye from MacConkey agar and NBTA plates. All isolates formed large, convex colonies with irregular margins. All isolates were gram-negative, motile, rod shaped bacteria.

# b) Microbial growth - direct cell count

Direct cell counts for the isolates *pur1* and *yel3* were carried out in three separate experiments using liquid cultures set up as described in Section 2.2.3.3. Two of the experiments were carried out using cultures grown in nutrient broth and one used cultures grown in MacConkey broth. The elapsed time between sampling ranged from 6 hours to 24 hours. The doubling time was found to vary between experiments and growth media, ranging from 1.61 to 2.7 hrs/generation (*pur1*) and 1.62 to 2.59 hrs/generation (*yel3*) (Table 2.19). However, there was no significant difference in doubling time between the colour phenotypes (paired t-test, T = 1.61, P > 0.05). (see also Appendix Figure A. 5 for growth curves).

		Doubling tir	Doubling time (hrs/generation)	
Experiment	Growth medium	pur1	yel3	
1	Nutrient Broth	1.61	1.62	
2	Nutrient Broth	2.60	2.53	
3	MacConkey broth	2.55	2.59	

**Table 2.19:** Doubling times of *pur* and *yel* isolates. Cultures were sampled twice with at least 6 hours between time points. Doubling time was calculated from the average of two counts at each time point.

# c) Microbial growth – plate counts

In experiments 1 and 2 of the direct cell counts in Section 2.3.3b an aliquot of each cell culture was spread on replicate NA plates. In experiment 1 the colony counts showed both isolates grew on NA. Growth was similar for both isolates with a higher number of colony forming units (CFU) being recorded for the *yel* phenotype than for the *pur* (Figure 2.10). In experiment 2 at the second sampling point both cultures were spread on NA, NBTA and on MacConkey agar plates. It was noted that a greater number of colonies of the *pur* isolate grew on MacConkey agar than on NA or NBTA plates. This difference in number of colonies was not observed with the *yel* isolate.



**Figure 2.10:** Growth of *Photorhabdus pur* and *yel* isolates on nutrient agar (NA), nutrient agar with Bromothymol Blue (NBTA) and MacConkey agar (MA) in two trials. All samples were drawn from cultures in nutrient broth. Values are mean ( $\pm$  SE) colony forming units (CFU) per ml of culture medium Number of plates (n) in each trial is indicated above the bar.

This result was further investigated in a separate experiment by carrying out replicate plating on all three media. More colonies of the *pur* isolate grew on MacConkey agar than on either the nutrient agar or nutrient agar supplemented with Bromothymol Blue (Figure 2.11). This result was significant at the P < 0.05 level. (2-sample t-test, 4 df, T = 4.25, P < 0.05). The nutrient agar (NA) and nutrient agar supplemented with Bromothymol Blue (NBTA) plates were treated as one type as there was no difference between the NA and NBTA plates when these were treated as separate media and a one-way ANOVA was carried out on NA, NBTA and MA (results not shown). There was no difference in the number of *yel* colonies between nutrient agar and MacConkey agar.



**Figure 2.11:** Growth of *Photorhabdus pur* and *yel* isolates on nutrient agar (NA), nutrient agar with Bromothymol Blue (NBTA) and MacConkey agar (MA) Number of plates (n) in each trial is indicated above each bar. Within colour (*pur*, *yel*) same letters indicate no significant difference between plate type.

As the *pur* isolate appeared to be disadvantaged in nutrient agar or broth versus MacConkey agar or broth and growth rate in MacConkey broth was the same for both isolates (Table 2.19) this medium was used for subsequent experiments requiring an inoculum of liquid culture.

### d) Biochemical characterisation

In the API 20E tests all isolates were negative for oxidase activity,  $\beta$ -galactosidase activity on *o*-nitrophenyl- $\beta$ D-galactopyranoside, lysine decarboxylase, ornithine decarboxylase, hydrogen sulphide production, urease activity, acetoin production and nitrate reduction to nitrite (Table 2.20). All isolates were positive for arginine dihydrolase, citrate utilisation and gelatinase. Only the *yel* isolates were positive for

tryptophan deaminase. The *pur* isolates and a single *yel* isolate (*yel1*) were positive for indole production.

**Table 2.20:** Characteristics of *Photorhabdus* isolates from North Bull Island (API 20 E test strip). Results are indicated as follows: +, positive; -, negative; w, weakly positive. Unusual results are highlighted.

Reaction	pur1	pur2	pur3	yel1	yel2	yel3
β-galactosidase	-	-	-	-	-	-
Arginine dihydrolase	+	+	+	+	+	+
Lysine decarboxylase	-	-	-	-	-	-
Ornithine decarboxylase	-	-	-	-	-	-
Citrate utilisation	+	+	+	+	+	+
H <sub>2</sub> S production	-	-	-	-	-	-
Urease	-	-	-	-	-	-
Tryptophane deaminase	-	-	-	+	+	+
Indole production	W	W	W	+	-	-
Acetoin production	-	-	-	-	-	-
Gelatinase	+	+	+	+	+	+
Cytochrome-oxidase	-	-	-	-	-	-
NO <sub>2</sub> production	-	-	-	-	-	-
reduction to N <sub>2</sub> gas	-	-	-	-	-	-

In the API 50 CH tests acid was produced from 21 of the 49 carbohydrates in the test kit. Only substrates where at least one isolate released acid are shown (Table 2.21). All isolates produced acid from D-glucose, D-mannose, Esculin, D-fructose (weakly), D-ribose (weakly), inositol (weakly), N-acetylglucosamine, glycerol and D-maltose (weakly). Only *pur* isolates produced acid from salicin, D-sorbitol and methyl- $\alpha$ D-glucopyranoside and amygdalin. Only *yel* isolates produced acid from D-

trehalose, methyl-βD-xylopyranoside and L-lyxose. Some but not all isolates of each colour phenotype produced acid from D-turanose, L-fucose, gluconate, 2-ketogluconate and 3-ketogluconate.

**Table 2.21:** Carbohydrate fermentation by *Photorhabdus* isolates from North Bull Island (API 50 CH test strip). Results are indicated as follows: +, positive; -, negative; w, weakly positive. Substrates where all isolates were negative have been omitted. Substrates where only *pur* or *yel* were positive are outlined in purple or yellow outline respectively.

Acid produced from	pur1	pur2	pur3	yel1	yel2	yel3
D-Glucose	+	+	+	+	+	+
D-Mannose	+	+	+	+	+	+
Esculin	+	+	+	+	+	+
D-Fructose	W	W	W	W	W	W
D-Ribose	W	W	W	W	W	W
Inositol	W	W	W	W	W	W
D-Maltose	W	W	W	W	W	W
N-Acetylglucosamine	+	W	W	W	+	+
Glycerol	+	+	+	W	W	W
Salicin	W	W	W	-	-	-
D-Sorbitol	W	-	-	-	-	-
Methyl-aD-Glucopyranoside	+	-	-	-	-	-
Amygdalin	-	W	-	-	-	-
D-Trehalose	-	-	-	W	W	W
Methyl-βD-Xylopyranoside	-	-	-	W	-	-
D-Lyxose	-	-	-	-	-	w
D-Turanose	+	-	-	-	-	W
L-Fucose	W	W	-	W	+	+
Potassium Gluconate	W	-	W	W	W	W
Potassium 2-Ketogluconate	W	-	-	-	W	-
Potassium 3-Ketogluconate	W	W	-	-	W	W

# 2.3.3.1 Antibiosis

In antibacterial tests all *Photorhabdus* isolates inhibited the growth of Gram-negative (*E. coli*) and Gram-positive (*B. subtilis, M. luteus*) organisms (Figure 2.12). The mean size of the inhibition zone varied between target organisms. For both colour phenotypes the greatest inhibition was seen against *M. luteus* (Figure 2.12C),
followed by *E. coli* (Figure 2.12A). The least inhibition was seen against *B. subtilis* (Figure 2.12B).



**Figure 2.12:** Mean size of inhibition zone around individual *Photorhabdus* isolate spot colonies. Plates were examined after 20 hr (A - *E. coli*, B - *B. subtilis*, D - *X. bovienii*) or 40 hr (C - *M. luteus*). Results are mean of three trials with three *pur* and three *yel* isolates in each trial. Treatments which do not share the same letter are significantly different. (nested ANOVA, See Table 2.22).

For all four target organisms larger inhibition zones were recorded for the *pur* phenotype than for the *yel* phenotype. The difference between phenotypes was significant (P < 0.05), while that between individual isolates when phenotype was included as a factor was not significant (nested ANOVA, see Table 2.22). Figure 2.13 shows typical inhibition of target organisms.

**Table 2.22:** Inhibition zone for two *Photorhabdus* colour phenotypes (n = 3) against four target organisms. Inhibition zone was calculated from average inhibition in two (*X. bovienii*) or three trials. Data were analysed using nested ANOVA. Treatments which do not share the same letter are significantly different.

	Mean inhibition zone (mm)		Test statistic (df)	
	pur	yel	F	P-value
E. coli	11.3 a	3.4 b	94.336 <sub>1,4</sub>	0.001
B. subtilis	10.1 a	3.1 b	41.321 <sub>1,4</sub>	0.003
M. luteus	11.4a	6.9b	32.689 <sub>1,4</sub>	0.005
X. bovienii	5.3a	2.6b	$11.542_{1,4}$	0.027



Figure 2.13: Inhibition of *E. coli* by *purl* (A) and *yel3* (B).

In intraspecific tests inhibition was observed in only two combinations, *pur2/yel1* and *yel1/pur2* (Figure 2.14). The mean size of the inhibition zone was greater in the case of the *yel* isolate as target organism (Table 2.23).



**Figure 2.14:** There was very slight intraspecific inhibition between *pur* and *yel Photorhabdus* isolates from North Bull Island. Interactions are A: *pur2* inhibits *yel1*. B: *yel1* inhibits *pur2*. C: *yel1* does not inhibit *pur1*.

Spot colony	target	mean inhibition (mm)
purl	yel3	0.00
pur2	yell	0.50
pur3	yel2	0.00
yel1	pur2	0.20
yel2	pur3	0.00
yel3	pur1	0.00

**Table 2.23:** Intraspecific inhibition in *Photorhabdus* colour phenotypes (n = 2)

In antimycotic tests all target organisms were inhibited by all *Photorhabdus* isolates (Figure 2.15A-C and Figure 2.16A-C). The *pur* isolates produced larger zones of inhibition than the *yel*. For each colour phenotype the greatest inhibition was seen against *B. bassiana* (Figure 2.15A) followed by *M. anisopliae* (Figure 2.15B) and *Mucor* (Figure 2.15C). For all three target organisms larger inhibition zones were recorded for the *pur* phenotype than for the *yel* (Table 2.24). The difference between phenotypes was significant for all three target organisms (P < 0.05) while the difference between individual isolates was not significant (nested ANOVA, see Table 2.24).



A1 pur / Mucor



B1 pur / B. bassiana





B2 yel / B. bassiana



A3 Mucor control



B3 B. bassiana control



C1 pur / M anisopliae



C2 yel / M. anisopliae



C3 M. anisopliae control

**Figure 2.15:** Inhibition of A - *Mucor*, B – B. *bassiana*, C – M. *anisopliae* by *pur* (A1, B1, C1) and *yel* (A2, B2, C2, ) *Photorhabdus* isolates. Control plates (A3, B3, C3). All tests were run 3 times with 3 isolates of each colour phenotype per trial.



**Figure 2.16:** Inhibition of fungal growth by *Photorhabdus* isolates. Mean size of inhibition zone for individual *Photorhabdus* isolates. Data shown is for 7 days (A - *Mucor*) or 28 days (B - *B. bassiana*, C - *M. anisopliae*) growth. Results are mean of three trials with three *pur* and three *yel* isolates in each trial (2 *yel* in trial 1 *B. bassiana* only). Treatments which do not share the same letter are significantly different (nested ANOVA See Table 2.24).

**Table 2.24:** Inhibition zone for two *Photorhabdus* colour phenotypes (n = 3) against three fungal organisms. Inhibition zone was calculated from average inhibition in three trials. Data were analysed using nested ANOVA. Treatments which do not share the same letter are significantly different.

	Mean inhibition zone (mm)		Test statistic (df)	
	pur	yel	F	P-value
B. bassiana	18.4a	14.3b	13.18 <sub>1,4</sub>	0.020
M. anisopliae	14.9a	10.3b	$20.20_{1,4}$	0.011
Mucor sp.	10.1a	8.2b	$163.58_{1,4}$	< 0.001

#### 2.3.4 Phylogenetic analysis of symbiotic bacteria

Phylogenetic analysis of the *gyrB* gene sequences showed that the *pur* and *yel* bacterial isolates from North Bull Island belong to distinct subspecies of *Photorhabdus temperata*: *Photorhabdus temperata* subsp. *cinerea* and *Photorhabdus temperata* subsp. *temperata*, respectively. The initial alignment of the edited nucleotide sequence of the six isolates showed single nucleotide polymorphisms (SNPs) at a total of 78 sites (Appendix Figure A. 6). At each SNP site all three isolates of each colour variant shared the same nucleotide. Six of the SNP sites produced non-synonymous amino acids in the translated protein sequence identified by ORF Finder (Table 2.25).

amino acid position	pur isolates	yel isolates
39/460	Isoleucine (Ile)	Valine (Val)
83/460	Arginine (Arg)	Serine (Ser)
125/460	Glutamate (Glu)	Lysine (Lys)
155/460	Glycine (Gly)	Serine (Ser)
160/460	Alanine (Ala)	Glycine (Gly)
279/460	Serine (Ser)	Glycine (Gly)

**Table 2.25:** Position of non-synonymous amino acids in translated *gyrB* gene sequence from *Photorhabdus* isolates from North Bull Island (see also Appendix Figure A. 6)

The BLAST search in the GenBank nucleotide database using the blastn algorithm showed that the *pur* and *yel* isolates were homologous with *gyrB* gene sequences from a range of *Photorhabdus* species. Alignment of sequences in ClustalW is shown in Appendix Figure A. 7. The three purple isolates shared 100% identity over the 847 nucleotide sequence analysed with three *P. temperata* subsp. *cinerea* isolates and one other *P. temperata* sp. isolate and for one of these, H3014, there was 100% identity over the full 1402 base sequence. However, the *pur* isolates had an alignment score of only 93% against the three *yel* isolates and *P. temperata* subsp. *temperata* subsp. *temperata* isolate K122, the only Irish isolate for which a *gyrB* sequence was available in GenBank. Conversely, the three *yel* isolates had an alignment score of 100% against K122, and two other Hungarian isolates, *P. temperata* subsp. *temperata* isolate H3179, isolated from *H. downesi* and isolate H111, described as *P. temperata* sp. isolated from *H. megidis*.

The phylogenetic analysis showed that the *pur* and *yel* isolates belonged to separate clusters within the set of sequences analysed (Figure 2.17). The *gyrB* sequences

formed four main clusters with three P. temperata subsp. tasmaniensis forming a separate group that branched more deeply. Cluster I contained the three P. temperata subsp. khanii American isolates, Meg, C1 and Habana, isolated from H. megidis, H. bacteriophora and an unidentified Heterorhabditis species respectively. There was 90% support for this cluster. Cluster II contained the three *pur* isolates along with four Hungarian isolates: three P. temperata subsp. cinerea isolates, H3240, H3107, from H. downesi, and H3014, isolated from H. megidis, one Photorhabdus temperata isolate, H267, isolated from H. downesi and one Photorhabdus sp. isolate, H3086, isolated from H. downesi. There was 98% support for this cluster. Cluster III contained three P. temperata subsp. thracensis isolates, H3210, CIP108426 isolated from *H. bacteriophora* from Hungary and Turkey respectively and FR32, isolated from an unidentified nematode from France. There was 98% support for this cluster. Cluster IV showed more variation with the three yel isolates grouped together with two P. temperata subsp. temperata isolates, K122 isolated from *H. downesi* from Ireland and H3179 isolated from *H. downesi* from Hungary and one isolate described as *P. temperata* sp., H111, isolated from *H. megidis* from Hungary, 80% support. The isolate H3182, isolated from *H. downesi* from Hungary was distinct from this group with 80% support. These seven isolates were separated from the other eight isolates in Cluster IV with 97% support for the two clades. This last group contained seven P. temperata subsp. temperata isolates along with one isolate described as *P. temperata* sp., isolated from *H. megidis* and *H. downesi* from The Netherlands, Lithuania, Russia and Hungary. Neighbour-joining and Minimum Evolution methods of phylogenetic inference showed the same clustering with practically identical bootstrap values for 1000 replications (data not shown).



**Figure 2.17:** Maximum Likelihood phylogenetic tree, based on a portion of the *gyrB* gene from six North Bull Island isolates and members of the *Photorhabdus temperata* species. *X. nematophila* was used as an outgroup. Tree shown is the bootstrap consensus tree from 1000 replications.

# 2.4 **Discussion**

The detection of two species of entomopathogenic nematodes, Heterorhabditis downesi and Steinernema feltiae, in the sand dune complex on North Bull Island is consistent with a previous study (Rolston et al., 2005) (see Appendix Figure A. 1 for overlap of sampling areas in this study and the Rolston *et al.* (2005) study). These two species along with S. affine have previously been recorded in Ireland (Griffin et al., 1991). In the study reported here H. downesi was detected in 13/42 (31%) of soil samples and S. feltiae was detected in 34/42 (34%) of soil samples. This is much higher than the levels detected by Rolston et al. (2005) in surveys of the same site in 2001 and 2002 but comparable to those reported by Griffin et al. (1994; 1999) for sites in mainland Europe and Ireland and the UK respectively. In the current study in 2008 the detection of the two species, as percentage of bait insects killed, was 1.7% and 10.4% for H. downesi and S. feltiae respectively in 2008 (three transects) and 10.4% and 17.1% for *H. downesi* and *S. feltiae* respectively in 2012 (transect 1 only). The levels detected by Rolston et al. (2005) at the same site in two successive years, 2001 and 2002, as percentage of bait insects killed, was 0.48% and 1.88% respectively, while that of S. feltiae was 0.63% and 1.6% respectively. While the 2008 level of detection of *H. downesi*, as percentage of bait insects killed, was similar to that of the Rolston et al. (2005) study the level of detection of S. feltiae was higher. This was due in part to two high observations in transect 1 with detection levels of 36.7% and 56.7% as percentage of bait insects killed at the 40 m and 90 m sampling locations respectively. The level of detection for both nematode spp. was much higher in 2012.

The distribution of entomopathogenic nematodes is reported as patchy by several studies and can be expected to vary both spatially and temporally (Stuart & Gaugler,

1994; Lawrence *et al.*, 2006; Campos-Herrera *et al.*, 2007). This variation can be due to a number of factors such as host availability, presence of competitors and predators, as well as a range of abiotic factors such as soil type, moisture, temperature and land use (Mrácek *et al.*, 2005; Hoy *et al.*, 2008). Localised areas of high density of infective juveniles, such as that observed in transect 1 for *S. feltiae* (2008) may also be due to the death of infected insects and subsequent emergence and dispersal of infective juveniles (Timper *et al.*, 1988; Spiridonov *et al.*, 2007).

In this study and the Rolston *et al* (2005) work soil baiting with *G. mellonella* was used to detect entomopathogenic nematodes; however the Rolston *et al* (2005) study differed in that soil samples consisted of individual 10 x 10 x 150 mm cores and was conducted in July and August 2001 and 2002 whereas in 2008 this study used a bucket type corer (12 mm x 10 cm, Figure 2.2) and pooled soil samples and sampling was carried out in September and October 2008 and in 2012 a targeted sampling of transect 1 where both *H. downesi* colour phenotypes had previously been detected was carried out. Differences in detection levels of entomopathogenic nematodes between this study and the previous studies on this site may be due to differences in sampling and baiting method or to time of sampling. It is also the case that repeat sampling may fail to detect EPN in sites that were previously positive for them (Hominick & Briscoe, 1990; Rolston *et al.*, 2005).

In this study, bait insects infected by *H. downesi* showed two colour phenotypes, purple and yellow; designated PUR and YEL respectively. This is similar to Rolston *et al.* (2005). In 2008 the PUR phenotype was detected in all three transects sampled whereas the YEL phenotype was detected in transect 1 only, and the colour phenotypes were spatially segregated here. This is similar to Rolston *et al.* (2005) who, although they found no significant difference in the distribution of the colour

phenotypes into the dunes, found that the PUR was more dominant in the front 50 m of the dunes in the second year (2002) of that study. In a subsequent study in 2005 Rolston found that the purple phenotype was significantly more prevalent in terms of number of positive soil samples and in terms of number of bait insects invaded in the front section of the dune system (0 m - 40 m) than in the rear section (60 m -100 m) and that for the yellow phenotype significantly more bait insects were invaded in the rear section (Rolston et al., 2006). In 2012 the PUR and YEL phenotypes cooccurred in the front of the dunes and the PUR were more abundant here but not significantly so. None of the PUR phenotype was detected in the rear portion of the dunes in 2012. In the 2012 both H. downesi phenotypes had migrated forward with the advancing dunes and the PUR phenotype was detected 20 m forward of the first location that was positive for PUR in 2008 (Table 2.17). The YEL phenotype was detected 60 m forward of the first location that was positive for YEL in 2008. This forward movement of the nematodes with the advancing sand dunes may be due to the movement of infected insects prior to death or may be due to infected insects being dispersed by other predators such as birds and small mammals. It also appears to be the case the humans may play a role with higher occurrence of EPN reported where disturbance has occurred (Mracek & Webster, 1993; Rolston et al., 2005). These repeat surveys of the same site point to the variability of the population of entomopathogenic nematodes associated with a particular site and is consistent with other studies (Lawrence et al., 2006).

While it is difficult to directly correlate habitat with colour phenotype in this study, because soil samples were pooled at each sampling distance, a number of observations are possible. The PUR phenotype occurred where the ground cover was fragmented and *A. arenaria* was the dominant species. The ground in this front

section of the dunes was transected with many paths, leading to stretches of bare sand. In all sampling locations where the PUR phenotype was detected sandy paths comprised 1/5 to 1/3 of the cores taken. *Ammophila arenaria* was present throughout the dune system but was not the dominant species beyond 60 m into the dune system. The YEL phenotype was more likely to be associated with fixed dune type habitat. This habitat is characterised by broadleaved herbs such as *Anthyllis vulneraria*, *Galium verum*, *Ononis repens*, *Plantago lanceolata* and grasses such as *Festuca rubra* and *Agrostis* spp. (Fossitt, 2000; Rodwell *et al.*, 2000) and was the dominant habitat from 60 m into the dune system along transect 1. Ground cover was more or less complete in the rear 70 m of the dunes where a dense mat of living and dead plant material tended to accumulate, leaving few bare patches. Where this occurred the soil was well shaded and tended to be more moist. This may have been an important factor in the survival of the YEL phenotype. This is discussed in more detail in Chapter 5.

Little is known about the natural host range of entomopathogenic nematodes although they have been isolated from a range of insects in the field with *Heterorhabditis* spp. being recovered from Lepidoptera and Coleoptera (Peters, 1996; Adams & Nguyen, 2002). *Heterorhabditis downesi*-infected larvae of the garden chafer, *Phyllopertha horticola* have been found in modified dune grassland in Co. Clare by R. Enright (C. Griffin, pers. comm.). The harsher environment at the front of the dune system tends to favour sand-dune specialists whilst grassland species, some but not all of which are coastal species, tend to occupy the dune grassland of the fixed dunes (Speight, 1997). The black Marram weevil, *Otiorhynchus atroapterus* was observed throughout the dune system both in the Rolston *et al.* (2005) surveys and in the course of the present work larvae of the

cinnabar moth Tyria jacobaeae and adult O. atroapterus were observed throughout the dune system and adult *Psilothrix* spp. (Coleoptera) were observed in the front section of the dune system (Maher, personal observation). Many of the insects which are potential hosts for the entomopathogenic nematodes on North Bull Island may be nocturnal, live in the litter layer at the base of the grass tussocks or are viable hosts only in the larval stage. While this study did not include a systematic faunistic survey of North Bull Island adults of Lepidoptera, Diptera, Coleoptera and Hymenoptera were frequently observed and the larvae of these may be hosts for entomopathogenic nematodes. In a survey of the Czech Republic entomopathogenic nematodes occurred more frequently in habitats that had a high abundance of suitable hosts than those that had a low insect abundance (Mrácek et al., 2005). Emelianoff et al. (2008b) suggested that there was an association between nematode strain and habitat type in a survey of soils in Southern France. Factors such as host availability and soil type as well as physiological adaptations all play a role in nematode distribution (Adams et al., 2006). Differences in range of potential hosts between the front and the back of the dune system may be part of the explanation for the spatial segregation of the colour phenotypes.

An alignment of the *gyrB* sequence from the bacterial symbionts associated with the *H. downesi* nematode from North Bull Island showed that the *pur* and *yel* bacterial isolates could be separated on the basis of nucleotide sequence. Phylogenetic analysis of the six North Bull Island bacterial symbionts of *H. downesi* identified the three *pur* isolates as *P. temperata* subsp. *cinerea*, first described by Toth & Lakatos (2008), and identified the three *yel* isolates as *P. temperata*. Thus the *H. downesi* nematodes on North Bull Island are symbiotically associated with two different *Photorhabdus temperata* subspecies. This is the first finding of two

different subspecies of symbiotic bacteria associated with a single entomopathogenic nematode in Ireland. While the recovery of two different subspecies of bacterial symbiont from a single species of *Heterorhabditis* is unusual it is not unknown. It has been shown that a single *Photorhabdus* species can be associated with more than one Heterorhabditis species (Fischer-Le Saux et al., 1999; Akhurst et al., 2004; Tailliez et al., 2010). Conversely, a single Heterorhabditis species can be associated with more than one *Photorhabdus* species or subspecies (Hazir et al., 2004; Adams et al., 2006; Emelianoff et al., 2008b; Maneesakorn et al., 2011). The Emelianoff et al. (2008b) survey in Southern France found two strains of H. bacteriophora, at different sampling locations, one of which was associated with P. luminescens subsp. kayaii and was recovered from four different beach sites and one meadow while the other strain was associated with P. luminescens subsp. kayaii in a vineyard site and associated with P. luminescens subsp. laumondiii in a separate vineyard site and a non-agricultural field site. While the surveys by Emelianoff et al. (2008b) and Hazir et al. (2004) and others (Noujeim et al., 2011) reported the association of a single species of nematode with more than one species or subspecies of bacterial symbiont their findings showed that the different combinations occurred at separate sites. However, in Hungary Toth & Lakatos (2008), isolated P. temperata subsp. cinerea from both H. downesi and H. megidis. In a further study they reported the isolation of both P. temperata subsp. cinerea and P. temperata subsp. temperata from both *H. downesi* and *H. megidis* isolated from the two sites in central Hungary, an oak forest and a clearing 1 km distant from the forest site (Tóth & Lakatos, 2009). The Hungarian studies do not give any details of soil types associated with the occurrence of the nematode/bacterial combinations, or the spatial distribution of the bacterial subspecies within the sampling sites, however sampling areas were chosen on the basis of being the preferred habitat of the common cockchafer, *Melolontha melolontha*, a major horticultural pest in Hungary. The results from North Bull Island are a further confirmation of the association of a single *Heterorhabditis* species, *H. downesi*, with two different bacterial subspecies, *P. temperata* subsp. *temperata* and *P. temperata* subsp. *cinerea* in a single site.

The clustering obtained in the *gyrB* phylogenetic tree is broadly consistent with other studies (Akhurst *et al.*, 2004; Kuwata *et al.*, 2008; Tóth & Lakatos, 2008; Tóth *et al.*, 2008; An & Grewal, 2010a; Tailliez *et al.*, 2010; An & Grewal, 2011a). The grouping at subspecies level matches that obtained in these studies. The three *P. temperata* subsp. *tasmaniensis* isolates USCA01, NZH3 and T327 form a separate monophyletic group as do the American isolates C1, Meg and Habana, which are classified as *P. temperata* subsp. *khanii* by Tailliez *et al.* (2010) and as *P. temperata* subsp. *stackebrandtii* subsp. nov. by An & Grewal (2010b). Within the European species and subspecies the position of K122 and XINach in separate clades within the *P. temperata* subsp. *temperata* group and the position of the three Hungarian *P. temperata* subsp. *cinerea* isolates, H3107, H3240 and H3107 as a separate monophyletic group is well supported in the literature. Thus the identification of the *pur* and *yel* isolates as belonging to separate subspecies is based on a phylogeny that matches previous work.

To date only a single *Heterorhabditis* species, *H. downesi*, has been recorded from Ireland. The isolate K122, the type specimen for the species, has been sequenced by two other groups (Nguyen *et al.*, 2004; Regeai & Burnell, 2006). The analysis of the ITS1 region of the ribosomal genes identifies the nematode isolates in this study as *Heterorhabditis downesi* (Figure 2.9). The six North Bull Island isolates (PUR1, PUR2, PUR3, YEL1, YEL2 and YEL3) along with the two K122 isolate and two

Hungarian H. downesi isolates, H3017 and H3173, associated with P. temperata subsp. cinerea and P. temperata subsp. temperata respectively (Tóth & Lakatos, 2008; Tóth et al., 2008), were placed in a single clade, although the Hungarian isolates were placed on a separate branch of the clade. Based on an analysis of the ITS1 region of the *H. downesi* isolates recovered in this study there is no evidence of different nematode types being associated with the two different bacterial subspecies discussed above. This is in agreement with the findings of Toth et al. (2008), and Toth & Lakatos (2008; 2009), who first identified P. temperata subsp. cinerea and found both P. temperata subspecies associated with H. downesi at a single site. This is in contrast to Emelianoff et al. (2008b) where two H. bacteriophora strains which differed by four nucleotide sites in their ITS1 sequence were found to have a different bacterial subspecies associated with each nematode strain. In that study the two H. bacteriophora strains were isolated from sites with different habitats. Given the similarity of the nematode isolates in this study and their close association in a single site it might be expected that their symbiotic bacteria would be interchangeable. This is investigated in Chapter 3.

Biochemical characterisation of bacteria is useful in describing a species and can provide useful data for identifying new isolates (Akhurst *et al.*, 1996; Fischer-Le Saux *et al.*, 1999). In the case of the symbiotic bacteria associated with entomopathogenic nematodes they are also useful in differentiating between phase I and phase II variants (Boemare & Akhurst, 1988). Biochemical tests are widely used and are particularly useful when investigating new isolates. The use of this type of test adds to the profile of the strain and can be useful in identifying further isolates of a particular strain. The API 20 test strips were used to investigate the enzymatic profile of the North Bull Island isolates. The results showed that there was a great deal of similarity between all the isolates, but that certain tests were useful for distinguishing between the colour phenotypes. The most striking result was that all isolates were positive for arginine dihydrolase. This is in contrast to the type specimen for P. temperata subsp. temperata, XINach (Fischer-Le Saux et al., 1999). However, this character has been observed in other P. temperata subspecies, for example P. temperata subsp. stackebrandtii strain GPS11 (An & Grewal, 2010b). Another enzymatic character which may be used to distinguish P. temperata subspecies is tryptophan deaminase activity (Tóth & Lakatos, 2008). The three pur isolates were negative for this, while the three *yel* isolates were positive. In the Toth & Lakatos (2008) study the majority of P. temperata subsp. cinerea isolates were positive for tryptophan deaminase activity although the P. temperata subsp. cinerea type specimen was negative. The ability to carry out fermentative metabolism in the anaerobic conditions which persist in host cadavers is an important adaptation for organisms that exploit this niche. Entomopathogenic bacteria display an ability to ferment a number of carbohydrates under anaerobic conditions (Rosner et al., 1996). The use of test kits such as the API 50 strips allows a rapid analysis of the fermentative profile of bacteria. Only the *yel* isolates produced acid from trehalose, this is similar to the type strain for *P. temperata* subsp. temperata (Fischer-Le Saux et al., 1999), as well as other *Photorhabdus* species but is not universal in the genus (Akhurst et al., 1996). Trehalose is a major component of insect haemolymph (Wyatt & Kalf, 1957) and the ability to utilise this energy source could be an important ecological advantage, enabling a particular nematode-bacteria partnership to fully exploit the insect host. The production of acid from salicin and amygdalin was noted for only a few of the strains in the Fischer-Le Saux et al. (1999) study and none produced acid from D-sorbitol, methyl- $\alpha$ D-glucopyranoside. In contrast, in the case

of the North Bull Island symbionts at least one *pur* isolate produced acid from each of these substrates and acid was produced by all three pur isolates in the case of salicin. None of the North Bull Island *yel* isolates from produced acid from these four substrates. The characters which are exhibited by only one colour phenotype may prove useful in distinguishing bacterial isolates from *H. downesi* in the future. While acid production from a number of substrates was noted by only one or other of the bacterial colour phenotypes the results also showed within-type variation and between-type similarity. A number of studies have shown that biochemical characters can be variable between isolates of the same strain (Fischer-Le Saux et al., 1999; Kuwata et al., 2008; Tóth & Lakatos, 2008; Tóth et al., 2008) as well as between subspecies (Hazir et al., 2004; An & Grewal, 2011a) within the genus Photorhabdus, so this variation is not unexpected. In order for biochemical characterisation to be a useful tool for determining species and subspecies it should be used as part of a range of diagnostic methods and multiple isolates of each strain should be included. The results presented here are a useful step in building a profile of the bacterial symbiont associated with Heterorhabditis downesi in North Bull Island and the island of Ireland as a whole. The ability of the *yel* isolates but not the *pur* to ferment trehalose would appear to give the former a competitive advantage in utilising host resources and presents some interesting questions for future investigation.

The difference in the number of colonies of the *pur* isolate which grew on nutrient agar versus MacConkey agar which was noticed while determining the growth rate of the isolates in liquid culture was unexpected. There was no significant difference in growth rate of *pur* or *yel* isolates in the liquid cultures of these media (Table 2.19). One possible explanation for the differential growth on solid media was a greater

113

requirement for salt for optimal growth of the *pur* isolate, as a notable difference between media is the presence of bile salts (5 g/l) and NaCl (5 g/l) in MacConkey agar while nutrient agar contains 5 g/l NaCl only (Appendix Table A 3). Hazir et al. (2004) found that two Turkish species of Photorhabdus had heavier growth and a faster growth rate in the presence of 5 g/l NaCl in nutrient broth. Hodgson et al. (2003) found that three P. temperata strains grew more quickly in tryptone or peptone based media containing 5 g/l or 10 g/l NaCl. Some work was carried out by undergraduate students in the Behavioural Ecology laboratory under my cosupervision to test the hypothesis that the *pur* phenotype had a greater requirement for salt in the growth medium (Daly, 2010; Kennedy, 2011). Both bacterial colour variants were cultured in Luria broth with varying salt concentrations. This work was inconclusive in demonstrating a greater requirement for salt by the *pur* phenotype. Another major difference between nutrient agar and MacConkey agar is the make-up of nutrients available for bacterial growth and in particular the higher concentration of peptone in MacConkey agar (20 g/l) compared to nutrient agar (5 g/l) (Appendix Table A 3). MacConkey agar also contains lactose (10 g/l). Although nutrient agar contains yeast extract (2 g/l) and Lab-Lemco powder (1 g/l) the total concentration of complex nutrients is lower compared to MacConkey agar. Complex media such as peptone and yeast extract contain compatible solutes such as betaine and choline (Schmitz & Galinski, 1996). Schmitz & Galinski (1996) found that P. luminescens grown in nutrient broth supplemented with an additional 30 g/l NaCl accumulated compatible solutes, notably betaine and glutamine as well as choline, a precursor of betaine, presumably from the growth medium. Schmitz & Galinski (1996) also showed that when nutrient broth was supplemented with additional compatible solutes (1mM betaine or 1mM ecotine) the bacteria could tolerate 50 g/l NaCl and

114

accumulated compatible solutes including betaine, ecotine, glutamine and alanine. Crawford *et al.* (2010) found that *P. luminescens* and *X. nematophila* grown on agar containing 2 g/l tryptone and 5 g/l yeast extract could tolerate salt (NaCl) concentrations of 10 g/l when the agar was supplemented with L-proline (100mM), which is known to serve as an osmoregulator in *E. coli*. The salt concentration of insect haemolymph is high (Wyatt, 1961), and so the bacteria need to be able to tolerate conditions of high osmolarity. It may be the case that the *pur* phenotype requires nutrient conditions high in compatible solutes which can be used for osmoregulation to achieve optimum growth in the presence of salt. As the growth rate for both *pur* and *yel* isolates in MacConkey broth was the same and as nutrient agar appeared to be suboptimal for *pur* growth MacConkey broth was used for subsequent experiments requiring an inoculum of liquid culture.

The *yel* bacterial symbionts were expected to demonstrate higher antimicrobial effects than the *pur* isolates as the rear of the dune system where they mostly occur would be expected to harbour a greater range of antagonistic and/or competitor organisms. In fact the opposite result was observed. Antimicrobial compounds which are produced by entomopathogenic bacteria in the course of nematode infection of an insect are effective against both bacterial and fungal antagonists. These types of compounds are believed to be involved in out-competing other microorganisms for resources in the cadaver (Hu & Webster, 2000; Derzelle *et al.*, 2002). It has been shown that the midgut microbiota of certain Lepidoteran and Dipteran species can include between five and 22 different phyla, with *Enterococcus, Pseudomonas,* and *Serratia* occurring as common isolates (Walsh & Webster, 2003; Gouge & Snyder, 2006; van der Hoeven *et al.*, 2008; Erkosar *et al.*, 2013). The compounds produced by the North Bull Island *Photorhabdus* isolates have not been identified but it is

known that entomopathogenic bacteria produce a range of compounds with antifungal, antibacterial, insecticidal and nematicidal activity (Webster et al., 2002; Hinchliffe et al., 2010). The range of compounds produced by species of Xenorhabdus includes, for example, xenorhabdins, xenorxides, xenocoumacins and nematophins while species of *Photorhabdus* produce indole, and a range of stilbenes. In addition, members of both genera produce bacteriocins. Some of these compounds are believed to play a role in maintaining the specificity of the nematode-bacterium symbiosis as they are effective against other entomopathogenic microorganisms including closely related species and even different strains of the same species (Boemare et al., 1997; Hawlena et al., 2010b; Morales-Soto & Forst, 2011). This type of activity was evident in both the *pur* and *yel* phenotypes as both inhibited X. *bovienii*, the symbiont of *S. feltiae*, with the *pur* isolates having a greater inhibitory effect. There was limited inhibition in some pur/yel combinations. In contrast, Sharma et al (2002) found that all four strains inhibited each other in an in vitro study of the bacteriocin activity of P. luminescens subsp. akhurstii W14, P. luminescens subsp. laumondii TT01, P. temperata subsp. temperata K122 and P. asymbiotica ATCC43949. In the current study both phenotypes also inhibited two strains of entomopathogenic fungi, B. bassiana and M. anisopliae. This was similar to the findings of Ansari et al. (2005), who found that both P. luminescens and X. poinarii inhibited the growth of both of these fungi.

The enhanced antimicrobial properties of the *pur* isolates could also have a role in competition between *H. downesi* and *S. feltiae* and their respective bacterial symbionts *Photorhabdus* and *Xenorhabdus* for scarce resources as indole and stilbenes have nematicidal activity (Hu & Webster, 2000; Webster *et al.*, 2002). It is also the case that antimicrobial compounds produced *in vitro* and *in vivo* differ both

in terms of quantity and the range of compounds produced (Hu *et al.*, 1998). Studies by Crawford *et al.* (2010; 2012) showed that the production of some antimicrobial compounds was induced in response to molecular signals in insect haemolymph. Thus the *in vitro* investigation discussed here gives an indication of the range of antimicrobial activity of the *pur* and *yel* phenotypes but does not necessarily represent the *in vivo* activity. In chapter three the *in vivo* activity of a single strain of each bacterial subspecies towards the other host and bacterial subspecies will be investigated as part of a cross-compatibility study which will include co-infection of insects with both *P. temperata* subspecies.

# Chapter 3 Cross compatibility of *Heterorhabditis* downesi – Photorhabdus temperata

# 3.1 Introduction

Many multicellular organisms have over the course of their evolution formed close natural associations with microorganisms, to the extent that both partners now depend on the other for survival, with the larger organism providing a stable habitat with access to nutrients for the microbe while the microbe provides metabolic functions (Moran, 2006). (See Chapter 1 for a more general discussion of symbiosis). Many natural associations of entomopathogenic nematode-bacterium have been described, Lewis and Clarke (2012) for example, list 20 Steinernema-Xenorhabdus associations and seven Heterorhabditis-Photorhabdus associations. In some cases the nematode is associated with more than a single bacterial species and as has been seen in Chapter 2 this can occur at a single site, although this is unusual. The specificity of a symbiotic relationship is modulated by a number of recognition factors including tolerance or suppression of host defences, chemical or metabolic interactions and behavioural interactions. In the case of the entomopathogenic nematode-bacterium complex the specificity of the symbiotic relationship operates on a number of different levels: the food signals produced by the bacteria which influence IJ recovery in the insect (Strauch & Ehlers, 1998), the requirement by the nematode for the bacteria as a food source to complete its life cycle (Grewal et al., 1997) and the retention of the bacteria by the nematode IJ (Han & Ehlers, 1999). The level of specificity of the symbiotic relationship and indeed the requirement of the nematode for the bacteria appears to be variable.

In general, Steinernema spp. appear to be less dependent on their symbiont than Heterorhabditis spp., for example, Grewal et al (1997) found that S. scapterisci could develop on and retain Xenorhabdus spp. from S. carpocapsae and S. riobravis, whereas Han & Ehlers, (1998, 2000), found that while axenic H. bacteriophora H06 could reproduce on the symbiont of *H. megidis* HNA the next generation of infective juveniles did not retain the bacteria. The ability of a heterologous symbiont (nonnatural) to support the development of a particular nematode species and be transmitted to the next generation of infective juveniles appears to depend on the degree of relatedness of the homologous and heterologous bacteria (Sicard et al., 2004). At the time of the Han & Ehlers (1998) study all the symbionts of the different Heterorhabditis species were categorised as P. luminescens species. The authors suggested that the inability of the different symbionts to support the growth of each other's nematode supported the subdivision of the group into different species. Taxonomic studies showed that the genus Photorhabdus contained three species, each with a number of subspecies (Fischer-Le Saux et al., 1999; Akhurst et al., 2004; Tailliez et al., 2010). The two Heterorhabditis spp. investigated by Han & Ehlers (1998) were found to carry different species of symbiotic bacteria; the symbiont of H. megidis HNA is P. temperata (Fischer-Le Saux et al., 1999) while that of H. bacteriophora H06 is P. luminescens (Qiu et al., 2012). Evidence from coevolutionary studies of Heterorhabditis spp. - Photorhabdus spp. partnerships (Maneesakorn *et al.*, 2011) and *Steinernema* spp. - *Xenorhabdus* spp. partnerships (Lee & Stock, 2010) suggest that there is a tighter host-symbiont specificity in Heterorhabditis spp. - Photorhabdus spp. partnerships as there was evidence of a greater degree of cospeciation in the Maneesakorn study.

On the other hand, Gerritsen & Smits (1993) found that in some novel combinations of *Heterorhabditis – Photorhabdus* the IJs retained the heterologous bacteria. In the case of H. megidis strain HE, the IJs retained Xbac, the symbiont of H. bacteriophora strain Hbac. The H. bacteriophora IJs did not retain the HE symbiont although in a further study the same authors found that a very small percentage of H. bacteriophora IJ strains Hbac and Hmol retained the symbiont of H. megidis strain Pjun (Gerritsen et al., 1998). The same authors found that two strains of H. megidis, HE and HSH, could develop on and retain each other's symbionts (Gerritsen & Smits, 1997). Susurluk et al (2001) found that H. bacteriophora (TUR-H2), were able to grow on and retain the symbionts of both H. megidis HSH2 and H. *bacteriophora* HYB. The varying results in these studies demonstrate that the ability of both Heterorhabditis and Steinernema species and strains to form novel combinations is very variable and seems to be strain specific. In all of these cases (Gerritsen & Smits, 1993, 1997; Grewal et al., 1997; Gerritsen et al., 1998; Han & Ehlers, 1998, 2000; Susurluk et al., 2001) the recombination was carried out in vitro. Novel combinations have also been produced in vivo, for example (Chapuis et al., 2009) recombined aposymbiotic S. carpocapsae strain SK27 with a number of different X. nematophila strains and aposymbiotic S. feltiae strain VIN with a number of different X. bovienii strains from different geographical regions to that of the wild type combinations of both nematodes.

In nature it appears that each *Steinernema* species has a specific association with a single *Xenorhabdus* species although a single *Xenorhabdus* species can form an association with more than one nematode species (Adams *et al.*, 2006; Lee & Stock, 2010) (See Section 1.2.5 Table 1.2). In the case of *Heterorhabditis – Photorhabdus* associations the nematodes and the symbiotic bacterium will combine with multiple

partners rather than forming strict single species pairings (Adams *et al.*, 2006; Emelianoff *et al.*, 2008b; Tóth & Lakatos, 2008; Tóth & Lakatos, 2009), although the combinations appear to be strain-specific (Chaston & Goodrich-Blair, 2009). In a survey of forest sites in Hungary Toth & Lakatos (2009) found both *H. megidis* and *H. downesi* associated with both *P. temperata* subsp. *temperata* and *P. temperata* subsp. *cinerea*, indicating that both the nematode and the bacteria can form associations with more than one partner in the same area. Similarly, the *H. downesi* isolates from North Bull Island were found to be associated with two different subspecies of *P. temperata* (Chapter 2).

The symbiosis between the nematode and the bacteria has been shown to have a cost for both the nematode and the bacteria in terms of survival (Mitani et al., 2004; Emelianoff et al., 2007; An & Grewal, 2010a). However, this cost is not universal, as Emelianoff et al., (2007) found that while there was a difference in mortality between symbiotic and aposymbiotic IJs of S. carpocapsae strain SK27 there was no difference between symbiotic and aposymbiotic IJs of S. scapterisci strain SE61. In the Chapuis (2009) study the survival rate of the wild type S. carpocapsae-X. nematophila combination was lower than that of the wild type S. feltiae-X. bovienii combination. Furthermore, the wild type combination may not be the most pathogenic as a novel combination of *H. megidis* strain HE with the symbiont from H. megidis strain PE was found to be significantly more pathogenic against the dipteran Tipula oleracea than the wild type combination (Gerritsen et al., 1998). The specificity of the symbiosis between the North Bull Island nematode and the bacteria isolates will be investigated in this chapter using an *in vivo* approach. One bacterial subspecies will be injected into a host which will then be infected with IJs carrying the other subspecies. The ability of the nematode to recombine with and retain the heterologous bacteria will be investigated using IJs emerging from the cadavers. The pathogenicity of any novel combinations of nematode-bacterium obtained in the experiment will be investigated. The effect of the different subspecies of bacteria on the survival of the nematode will be investigated for the isolates from North Bull Island.

The objectives of this chapter are:

- i) To determine the ability of the PUR and YEL nematodes identified in Chapter 2 to carry each other's symbiont by means of a cross-compatibility experiment.
- ii) To look for differences in the infectivity of wild type and hybrid strains of the nematode-bacterium complex
- iii) To look for differences between the *pur* and *yel* bacterial subspecies in terms of infective juvenile survival and number of bacteria carried.

# 3.2 Materials and Methods

# 3.2.1 Cross-compatibility of *Heterorhabditis downesi – Photorhabdus temperata* strains

A no-choice experiment was carried out to investigate the preference of either the PUR or YEL nematode for its own colour bacteria variant within a host. The experiment consisted of four treatments (*pur* bacteria, *yel* bacteria, a mixture of both bacteria or no bacteria). Final instar larvae of *G. mellonella* were injected with the bacterial treatments and three days later the cadavers were infected with nematodes (Figure 3.). A subset of the injected insects was used to quantify the bacterial load in

the cadavers at 72 hours post injection. A subset of the injected and infected insects was used to estimate the invasion rate of the nematodes. The final subset of cadavers was used to quantify the progeny from each of the bacterial treatments. A single experiment was carried out separately for a single nematode/bacteria isolate of each colour phenotype (PUR1/*pur1* and YEL/*yel3*) (see Chapter 2, Table 2.15).



**Figure 3.1:** Experimental design for cross-compatibility experiment. Final instar *G. mellonella* larvae were injected with one of four treatments (*pur* bacteria, *yel* bacteria, a mixture of both bacteria or no bacteria) and subsequently infected with nematodes carrying either *pur* or *yel* bacteria. The bacterial phenotype carried by the emerging IJs was determined in subsequent experiments.

# **3.2.1.1 Preparation of bacterial cell suspension**

Liquid cultures of *P. temperata* isolates *pur1* and *yel3* were set up in MacConkey broth as described in Section 2.2.3.2 and grown for 48 hours. Cells were collected by centrifugation at 3,850 x g for 10 min. The supernatant was discarded and the cells were resuspended in  $\frac{1}{4}$  strength sterile Ringer's solution. The optical density was

measured at 660 nm ( $OD_{660}$ ) and the concentration was adjusted to 4 x 10<sup>7</sup> cells/ml, calculated from the standard curve (see Appendix Figure A. 2A and Figure A. 2B). For each experiment the bacteria were freshly cultured.

# 3.2.1.2 Injection of G. mellonella

Final instar *G. mellonella* larvae were injected with 2.5 µl bacterial cell suspension  $(4 \times 10^7 \text{ cells/ml}, \text{ equivalent to } 1 \times 10^5 \text{ cells})$ , prepared as outlined in Section 3.2.1.1. Larvae were injected behind the second proleg using a 5 µl microsyringe with fitted with a 70 mm plunger-in-needle (0.63 mm OD, 0.37 mm ID), (SGE Europe Ltd., Milton Keynes, UK). The syringe was rinsed five times with ethanol (70%) and five times with sterile distilled water between insects. Thirty insects per treatment received *P. temperata* isolates *pur1, yel3*, or *pur1/yel3* (1:1) bacteria. For each bacterial treatment ten insects were randomly allocated to one of three subsets for subsequent treatments:

- i) bacterial quantification
- ii) dissection to assess number of IJs invading
- iii) White traps to assess progeny

Twenty insects received a no bacteria treatment, consisting of an injection of 2.5  $\mu$ l sterile <sup>1</sup>/<sub>4</sub> strength Ringer's solution and were randomly assigned to subsets ii) and iii) above. Each insect was placed in a well of 24-well plates lined with two 1.5 cm filter papers (Whatman no. 1) (Corning Costar 3527, Fischer Scientific, Ireland). The plates were sealed with Parafilm and stored at 20 °C. Insects were checked for mortality after 3 days. All insects exhibiting evidence of death due to *P. temperata* infection (see Section 2.2.1.3) were used in the subsequent stages of the experiment. The colour of each insect cadaver was recorded as purple (P), yellow (Y) or

purple/yellow (P/Y), indicating mainly purple, mainly yellow or neither colour predominating. For each bacterial treatment the ten cadavers allocated for dissection were set aside. The remaining cadavers were used in the infection stage of the experiment.

#### **3.2.1.3 Infection of injected cadavers**

Three days after injection of the respective bacteria (Section 3.2.1.2), 50  $\mu$ l of 2000 IJ/ml nematode suspension (= 100 IJs/insect) was added to each of 20 cadavers per bacterial treatment in the 24-well plates. Nematode suspension (50  $\mu$ l of 2000 IJ/ml) was also added to all insects in the no bacteria treatment, which were still alive at this stage. The 24-well plates were resealed with parafilm which was punctured along each side for ventilation and stored at 20 °C. All insects were removed from the 24-well plates 24 hours after application of the nematodes, rinsed 3 times in sterile distilled water to remove any IJs on the exterior, placed individually on moist tissue paper in 3.5 cm Petri dishes and stored at 20 °C. The colour of each cadaver in the *pur1/yel3* (1:1) treatment was recorded as purple (P) when the cadaver was achromatic/purple, yellow (Y) when the cadaver had a distinctive yellow colour or purple/yellow (P/Y) when the cadaver colour was indeterminate (see also Chapter 2, Section 2.2.1.3 and Figure 2.3A-B).

# 3.2.1.4 Bacterial quantification from injected G. mellonella

Three days after injection of the respective bacteria the insects previously allocated to the bacterial quantification subset (Section 3.2.1.2i) were surfaced sterilised by rinsing in 70% ethanol, igniting and extinguishing by plunging into sterile <sup>1</sup>/<sub>4</sub> strength Ringer's solution and placed in a 3.5 cm Petri dish. Each cadaver was partially macerated with forceps and then transferred to a sterile 15 ml plastic vial (Sarstedt,

Numbrecht, Germany). The Petri dish was rinsed with 2 x 1 ml aliquots of <sup>1</sup>/<sub>4</sub> strength Ringer's solution which was also transferred to the plastic vial. All work was carried out beside a flame on the bench. The body contents were allowed to soften at room temperature for 4 hr with gentle agitation on a Grant Bio PS-3d platform rotator at 30 rpm. The contents were then vortexed for 2 min and allowed to stand at room temperature for 20 min. A 20 µl aliquot of the homogenate was added to 0.98 ml sterile <sup>1</sup>/<sub>4</sub> strength Ringer's solution in a 1.5 ml Eppendorf tube (Fisher Scientific, Ireland) and a 10-fold serial dilution was made. Three replicate 100 µl aliquots of each of 3 dilutions were spread on NA, NBTA and MA plates and incubated at 27 °C for 48 - 72 hours to assess the bacterial load in the insects at day 3 post injection.

### 3.2.1.5 Invasion rate

At day 4 following nematode infection the insects previously allocated to the dissection subset (Section 3.2.1.2ii) were examined to assess the number of nematodes invading in each treatment. Dissections were carried out in sterile <sup>1</sup>/<sub>4</sub> with the aid of a dissecting microscope. The insect cadaver was placed in a 9 cm Petri dish to which approximately 25 ml of <sup>1</sup>/<sub>4</sub> strength Ringer's solution was added. The cadaver was oriented with the ventral side uppermost and held steady with a coarse forceps. The body was opened along the line of the spiracles from the anus to the head using a fine forceps and the body contents were removed and dispersed in the <sup>1</sup>/<sub>4</sub> strength Ringer's solution and the number of nematodes per cadaver was recorded.

# 3.2.1.6 Infective juvenile emergence and harvest

Fourteen days after nematode infection (Section 3.2.1.3) the remaining ten cadavers for each treatment were placed on individual White traps (Section 3.2.1.2iii) (Figure 2.4). The individual White traps were harvested by carefully pouring the IJ

suspension into a 50 ml tube. The White trap was rinsed three times with tap water to transfer any remaining IJs to the 50 ml tube. Each White trap was harvested at day 4 following first emergence of IJs. A further harvest was carried out 24 hours later. This was repeated at day 11 and 12 following first emergence and at day 18 and 19 following first emergence. A final harvest was carried out on day 26 following first emergence. All IJs were stored at 9 °C prior to counting or use in subsequent experiments. Nematode counts were carried out as follows:

- a) Sample count (1 ml): The volume of nematode suspension from an individual White trap was noted and the 50 ml tube was inverted three times. A 1 ml sample was drawn and pipetted in a line of drops into a Petri dish. Live IJs in each drop were counted with the aid of a dissecting microscope. Three 1 ml samples were counted and the number of IJs in the total volume was calculated based on the mean of the three counts.
- b) Total count: The entire volume in the 50 ml tube was transferred to a 9 cm Petri dish marked with a grid and the total number of IJs in the sample was counted.
- c) Sample count (100 µl): The volume in the 50 ml tube was adjusted to 50 ml with tap water and three 100 µl samples were counted and the total number of IJs calculated as in a). If the initial suspension was too concentrated it was diluted 10-fold by drawing a 1 ml sample, transferring to a 10 ml graduated cylinder which was topped up to 10 ml with tap water, sealed and inverted three times prior to counting as outlined above.
- d) Semi quantitative estimation: A series of nematode suspensions was made up in 50 ml tubes to the concentrations listed in Table 3.. The total number of IJs present in a pooled harvest was estimated by eye by comparing the density of

the nematode suspension against the prepared concentrations and scored accordingly.

**Table 3.1:** Scoring used for semi-quantitative estimation of nematode harvest from White traps.

score	Nematode concentration (IJs/ml)
1	100
2	250
3	500
4	1000
5	2500
6	5000

Methods a) or b) were used where the IJ concentration was low and method c) was used where the IJ concentration was high. Method d) was used where individual counts were not carried out.

# 3.2.2 Bacteria carried by emerging infective juveniles

The bacterial symbiont carried by the emerging IJs harvested as described in Section 3.2.1.6 was assessed using *in vitro* and *in vivo* assays.

#### 3.2.2.1 In vitro assay

The IJs were stored for 1 week at 9 °C. A 600 $\mu$ l aliquot of IJ suspension was desheathed as described in Section 2.2.3.1a. The IJs were collected by centrifugation at 1000 x *g* for 5 min and the supernatant was removed. The IJs were rinsed three times with <sup>1</sup>/<sub>4</sub> strength sterile Ringer's solution, collecting by centrifugation after each rinse. The IJs were crushed with a sterile micro pestle for 1 min, the volume was adjusted to 600 $\mu$ l with <sup>1</sup>/<sub>4</sub> strength sterile Ringer's solution and two 1 in 10 serial

dilutions with <sup>1</sup>/<sub>4</sub> strength sterile Ringer's solution were made. A 100µl aliquot of each dilution was plated in triplicate on MacConkey agar plates and incubated in the dark at 27 °C for 48 hr.

The assay was carried out with IJs emerging from cadavers in all the bacterial treatments infected with PUR1 IJs. Several problems occurred with the method, for example it was noted that some IJs were lost at each step in the collection and rinsing process, also some IJs remained uncrushed after the final step in the method. Both of these issues gave rise to very low colony counts. As one of the objectives in carrying out the assay was to determine the bacterial partner of the IJs an *in vivo* assay was used instead. A separate experiment to assess the level of bacterial colonisation of individual IJs was carried out using the method outlined in Section 2.2.3.1b). Briefly, IJs were surface sterilised using Hyamine 1622 solution (0.16 mM), rinsed three times with sterile PBS and single IJs were crushed in 100 µl sterile PBS and 50 µl of the homogenate was plated onto Luria-Bertani (LB) agar (Difco) supplemented with 0.1% sodium pyruvate (Sigma-Aldrich) (LB pyruvate agar) and incubated in the dark at 27 °C for 48 hr.

#### 3.2.2.2 In vivo assay

A series of single nematode infections based on the one-on-one bioassay described by Grewal (2005) were set up as follows: with the aid of a dissecting microscope one IJ, picked up by pipette in a 10  $\mu$ l volume from a dilute suspension, was transferred to a well of 24-well plates lined with two 1.5 cm filter papers (Whatman no. 1) (Corning Costar 3527, Fischer Scientific, Ireland). The pipette was rinsed with 2 x 10  $\mu$ l tap water which was also transferred to the well to ensure the IJ did not remain in the pipette tip. A single final instar *G. mellonella* larva was added to each well, the lid was replaced and the plate was sealed with parafilm which was punctured along each side for ventilation. Plates were randomly stacked in blocks consisting of one plate of each of the wild type and hybrid combinations of nematode/bacterial treatment and one negative control plate consisting of a 24-well plate with two filter papers, 30 µl of tap water and one G. mellonella larva per well. One empty 24-well plate with filter papers and 30 µl tap water only was placed on the top and bottom of each stack as a buffer. Each stack was wrapped in tinfoil to ensure constant light conditions and stored at 20 °C. Plates were checked daily for eight days and insect mortality was recorded. Tap water (30 µl) was added to the wells as required to avoid the filter paper drying out. Presence of P. temperata infection was confirmed as by bioluminescence by observing the plate in a darkened room for ten min (see also Section 2.2.1.3). The colour of the insect cadaver was recorded. Twenty four infections were set up with progeny from each of four separate cadavers for each of the wild type and hybrid combinations of nematode/bacterial treatment (see Table 3.2) and for four cadavers in the *purl/yel3* 1:1 bacterial mixture treatment (= 96) insects per nematode/bacterial treatment). The experiment carried out once with IJs harvested at day 4 and repeated with IJs harvested at day 19 (Section 3.2.1.6). In addition, the experiment was also repeated with progeny harvested at day 4 and at day 19 from each of the remaining cadavers in the purl/yel3 1:1 bacterial mixture treatment in each run of the cross-compatibility experiment (= 144 insects using progeny from PUR IJs and 120 insects using progeny from YEL IJs).

# 3.2.3 Tests of hybrid nematode/bacteria strains against G. mellonella

Throughout the following text capitals are used to describe the nematode colour phenotypes PUR and YEL for purple and yellow respectively, and italics are used to describe the bacteria colour phenotypes *pur* and *yel* for purple and yellow respectively. The nematode/bacteria designations of the wild type and hybrid strains
along with details of the North Bull Island isolates from which they originated are given in Table 3.2.

**Table 3.2:** Designations of wild type and hybrid combinations of nematode/bacteria cultures established from North Bull Island isolates (see also Table 2-14).

Nematode/bacteria combination		Parent			
		nematode	bacteria		
Wild type	PUR/pur (Pp)	PUR1	pur1		
	YEL/yel (Yy)	YEL3	yel3		
Hybrid	PUR/yel (Py)	PUR1	yel3		
	YEL/pur (Yp)	YEL3	pur1		

Four separate infections of each type (Pp, Py, Yy, Yp) were set up to assess the infectivity of progeny from each of the wild type and hybrid nematode/bacteria combinations produced in Section 3.2.1.6 (see also Table 3.2). The lid and base of a 9 cm Petri dish were each lined with a 90 mm filter paper. Ten final instar *G. mellonella* larvae were placed in each dish (= 40 insects per hybrid) and 1 ml of IJ suspension (1000 IJs/ml) was evenly pipetted onto the lid and base, equivalent to a dose of 100 IJs per insect. The dishes were incubated at 20 °C. Insect mortality was recorded every four hours between the periods from 40 hours to 48 hours following infection. Mortality due to P. *temperata* infection was assessed previously described (Section 3.2.2.2). The experiment was carried out once with IJs harvested at day 4 (Section 3.2.1.6).

#### 3.2.3.1 Cultivation of wild type and hybrid nematode/bacteria strains

Laboratory lines of wild type and hybrid nematode/bacteria combinations were established from the IJs which developed from the infections set up in Section 3.2.3. All ten cadavers from each infection were placed on White traps (see Section 2.2.1.7), harvested as described in Section 2.2.1.4 and counted as described in Section 2.2.1.6. Equal volumes of nematode suspension (1000 IJs/ml) from the four White traps for each of the wild type or hybrid nematode/bacteria combinations were pooled to give a single nematode suspension for each wild type or hybrid nematode/bacteria combination.

### 3.2.4 Survival of IJs in water

The survival of IJs of the *H. downesi* nematode isolates from North Bull Island was investigated. A 200 µl aliquot of nematode suspension (1000 IJ/ml) was added to 8 ml of tap water in a 5 cm Petri dish. The experiment was set up in randomised block design with 50 blocks. A block consisted of one dish of each of the six nematode isolates listed in Table 2.14 randomly arranged in a stack with one dish of tap water only on top and bottom of each stack as a buffer. Nine blocks were randomly arranged in each of 5 plastic food boxes. Each box was covered with a close fitting lid, placed in a plastic tray, covered with tinfoil to give constant light conditions and stored at 20 °C. The remaining five blocks were assessed immediately. A different block was removed from each food box every two weeks for 14 weeks and the number of surviving IJs in each dish was counted with the aid of a dissecting microscope. The block was replaced in the box so that there nine blocks in each box at all times.

#### **3.2.5** Bacterial quantification from individual infective juveniles

IJs were surface sterilised by adding Hyamine 1622 solution (0.16 mM) (Sigma-Aldrich) to nine ml of nematode suspension (1000 IJs/ml), inverting 3 times and allowing to settle. The supernatant was discarded and the IJs were rinsed by adding 9 ml sterile phosphate buffered saline (PBS), inverting 3 times and allowing to settle. The rinse was repeated twice. Individual IJs were pipetted in 10 µl aliquots with the aid of a dissecting microscope and added to 90 µl PBS in a 1.5 ml microcentrifuge tube. Each IJ was crushed by grinding with a sterile micro pestle for 1 min. A 50 µl aliquot of the product was plated onto Luria-Bertani (LB) agar (Difco) supplemented with 0.1% sodium pyruvate (Sigma-Aldrich) (LB pyruvate agar) and incubated in the dark at 27 °C for 48 hr. The experiment was carried out twice for each of the six isolates from North Bull Island and once for each of the hybrid nematode/bacteria combinations listed in Table 3.2, with 20 IJs of each isolate or hybrid in each experiment.

## 3.2.6 Statistical analysis

Routine statistical analysis, including tests for normality, was carried out using Minitab statistical software, version 16.1.1 (Minitab Inc., 2010). Normality of data was tested using the Anderson-darling method. Analysis of variance was carried out using a one-way ANOVA or when groups had an uneven number of observations the general linear model with an unbalanced nested design was used to compare isolate and colour phenotype. Analysis of variance was followed by pair-wise multiple comparisons using Tukey's test. In experiments where nematode type and bacterial type were factors results were analysed using a two-way ANOVA. Where no significant interactions occurred treatment means for each factor were compared using a one-way ANOVA and post-hoc Tukey's test for three or more treatments or 2-sample T-test for two treatments. When data could not be normalised treatment medians were compared using a Kruskal-Wallis test, followed by multiple pairwise comparisons using a Mann-Whitney U-test test with Bonferroni 95% confidence intervals (CI), obtained by calculating  $(1 - \alpha / g)$  where  $\alpha$  is the level of significance and g is the number of pairwise comparisons carried out. Spearman's Rank Correlation was carried out online (http://vassarstats.net/corr\_rank.html). In all tests a P value of < 0.05 was taken as significant unless indicated otherwise.

# 3.3 **Results**

# 3.3.1 Cross-compatibility of Heterorhabditis downesi – Photorhabdus temperata strains

In the cross-compatibility experiments both the PUR (experiment 1) and YEL (experiment 2) nematode colour phenotypes invaded and reproduced in *G. mellonella* cadavers that had been injected with any of the three *P. temperata* treatments and the no bacteria treatment described in Section 3.2.1.2. Invasion is described in Section 3.3.1.2, reproduction is described in Sections 3.3.1.3 and 3.3.1.4. At the outset, the colonisation of the *G. mellonella* haemocoel by the *P. temperata* isolates following injection is described.

#### 3.3.1.1 Bacterial quantification from injected G. mellonella

In each experiment a number of the *G. mellonella* cadavers in the injection only subset (Section 3.2.1.2i) were homogenised 72 hours post injection and an aliquot of the homogenate was spread on agar plates. Initial trials using NBTA and MacConkey agar plates showed that it was not possible to accurately quantify the individual colour phenotypes of the *P. temperata* colonies in the *purl/yel3* 1:1 treatment on the selective agar plates as the colonies absorbed dye from the NBTA

and MacConkey agar, which masked the pigmentation of the *pur* and *yel* isolates. Replicate plating from the selective plates onto NA plates was also unsuccessful for enumerating colour phenotypes. Aliquots of the homogenate were also spread directly on NA plates. In the case of the NA plates, contamination from the endogenous flora present in the *G. mellonella* cadavers prevented accurate enumeration of the separate colour phenotypes as young *pur* colonies are achromatic and can be mistaken for the *yel* phenotype. Colonies need to grow for 48 - 96 hr before the distinctive purple pigmentation of the *P. temperata pur* phenotype is apparent. This extra incubation period allowed the endogenous flora to overgrow the *P. temperata* colonies in some cases. A number of observations were possible however.

In the first experiment, set up for the PUR1 nematodes, colonies of both colour phenotypes were isolated from insects which received an injection of the *pur1/yel3* 1:1 bacterial mixture (Figure 3.2A). More of the *pur phenotype* than the *yel* phenotype appeared to be present in cadavers from the *pur1/yel3* 1:1 treatment, based on colony counts (data not shown). In the individual treatments more *P*. *temperata* colonies grew on plates from the *insects* that received the *pur1* bacteria counts. There did not appear to be a difference in luminescence between the *pur and yel* colonies (plates were checked in the dark as outlined in Section 2.2.3.3a). Overall, more colonies grew on plates from the insects that received the *pur1/yel3* 1:1 bacterial mixture than either of the individual bacterial types. The data were analysed using a one-way ANOVA. There was no difference between the bacterial treatments (P > 0.05).

In the second experiment, set up for the YEL3 nematodes, the same trend in the relative number of colonies of the *pur* and *yel* phenotypes was observed (Figure 3.1B), with more *pur* colonies than *yel* in the *pur1/yel3* 1:1 treatment, overall more colonies in the *pur1/yel3* 1:1 treatment than the individual treatments and more colonies in the *pur1* individual treatment than the *yel3* individual treatment. The data were not normally distributed but were normalised using a log to the base 10 transformation. The log to the base 10 of the data were analysed using a one-way ANOVA. There was a significant difference in the number of colonies between the *pur1/yel3* 1:1 treatment and the individual *yel3* treatment (one-way ANOVA,  $F_{2,14} = 5.04$ , P = 0.022).



**Figure 3.2:** Mean number ( $\pm$  SE) of colony forming units of *P. temperata* colour phenotypes *pur* and *yel* in *G. mellonella* cadavers 72 hours post injection. Each insect had been injected with 1 x 10<sup>5</sup> cells *P. temperata* isolates *pur1, yel3* or *pur1/yel3* 1:1 mixture. Data shown is mean number of colony forming units on NA plates. In experiment 1 (A) n = 6 for *pur1/yel3* 1:1 treatment, n = 4 for *pur1* and *yel3* treatments. In experiment 2 (B) n = 9 for *pur1/yel3* 1:1 treatment, n = 3 for *pur1* treatment and n = 5 for *yel3* treatment. Within each graph treatments which do not share the same letter were significantly different (One-way ANOVA, P < 0.05). Data from experiment 2 were log to the base 10 transformed for analysis.

As a similar trend in the number of colony forming units was observed in each experiment, the log to the base 10 data for both experiments were combined and analysed with bacterial treatment and experiment as fixed factors, using a general linear model as n varied between treatments and experiments. This showed that both bacterial treatment and experiment had an effect on the number of colony forming units observed, with more colonies in the *pur1/yel3* 1:1 treatment compared to the *yel3* treatment and this difference was highly significant (GLM,  $F_{2,27} = 6.02$ , P = 0.007) (Table 3.3).

**Table 3.3:** Mean number ( $\pm$  SE) of colony forming units (cfu) of *P. temperata* colour phenotypes *pur* and *yel* in *G. mellonella* cadavers 72 hours post injection. Each insect had been injected with 1 x 10<sup>5</sup> cells *P. temperata* isolates *pur1, yel3* or *pur1/yel3* 1:1 mixture. Data shown is mean number of colony forming units on NA plates. n = 7 for *pur1* treatment, n = 9 for *yel3* treatment, and n = 15 for *pur1/yel3* 1:1 treatment for two experiments. Results of GLM analysis are also given. Data were log to the base 10 transformed for analysis. Treatments which do not share the same letter were significantly different.

Bacterial treatment	mean number cfu per cadaver			
pur1	2.27E+07 <sup>ab</sup> (7.03E+06)			
yel3	1.25E+07 <sup>b</sup> (3.89E+06)			
pur1/yel3	4.66E+07 <sup>a</sup> (1.09E+07)			
Factor	Test statistic	P value		
Bacterial treatment	$F_{2,27} = 6.02$ 0.007			
Experiment	$F_{1,27} = 7.09$ 0.013			

Because there was an effect of experiment when the data for both experiments were combined by bacterial treatment the log to the base 10 transformed data for all three bacterial treatments were pooled for each experiment and analysed using a one-way ANOVA. The overall level of colonisation of the insects by the *P. temperata* isolates appeared be to significantly greater in the first experiment than in the second (one-way ANOVA,  $F_{1,29} = 4.48$ , P = 0.043) (Figure 3.3).



**Figure 3.3:** Mean number ( $\pm$  SE) of colony forming units of *P. temperata* isolates per insect in *G. mellonella* cadavers 72 hours post injection in two experiments. Each insect had been injected with one of three treatments of 1 x 10<sup>5</sup> cells *P. temperata*. Data shown is pooled for all three bacterial treatments in each experiment with n = 14 in experiment 1 and n = 17 in experiment 2. Treatments which do not share the same letter were significantly different (Oneway ANOVA, P < 0.05). Data were log to the base 10 transformed for analysis.

#### 3.3.1.2 Invasion rate

In the first experiment, using PUR1 IJs, on average more nematodes invaded insects which had previously been injected with one of the three *P. temperata* treatments compared to the insects in the no bacteria treatment (Figure 3.4A). The PUR1 nematodes did not seem to have a preference for their own bacterial type or the other bacterial type as the mean number of nematodes invading in the *pur1/yel3* 1:1 bacterial treatment was approximately 38 compared to means of approximately 25 and 22 nematodes in the *pur1* and *yel3* treatments respectively and the mean number of nematodes invading in the number of nematodes invading in the mean number of nematodes respectively and the mean number of nematodes invading in the no bacteria treatment was approximately 11. The

difference between treatments was highly significant (one-way ANOVA,  $F_{3,36} = 6.8$ , P = 0.001) for the number of nematodes invading insects in the no bacteria treatment compared to the number of nematodes invading insects in the *pur1/yel3* 1:1 treatment but not the other two treatments.

In the second experiment, using YEL3 IJs, on average more nematodes invaded insects in the no bacteria treatment compared to insects in the three *P. temperata* treatments (Figure 3.4B). Similarly to the PUR1 nematodes, the YEL3 nematodes did not seem to have a preference for their own bacterial type or the other bacterial type as the mean number of nematodes invading in the *pur1/yel3* 1:1 bacterial treatment was approximately 15 compared to means of approximately 10 and 5 nematodes invading the no bacteria treatment was approximately 15 means of approximately 17. There was a highly significant difference in mean number of nematodes invading (one-way ANOVA,  $F_{3,36} = 9.24$ , P < 0.001) comparing the *yel3* treatment to the *pur1/yel3* 1:1 treatment and the no bacteria treatment but not to the *pur1* individual treatment.



**Figure 3.4:** Mean number ( $\pm$  SE) of adult nematodes per insect in *G. mellonella* at day 4 following infection with 100 IJs per insect of (A) PUR1 and (B) YEL3 isolates from North Bull Island with n = 10 for all treatments. Insects had previously been injected with 1 x 10<sup>5</sup> cells of *P. temperata* isolates *pur1, yel3*, or a *pur1/yel3* 1:1 mixture. Insects in the no bacteria treatment were injected with an equal volume of <sup>1</sup>/<sub>4</sub> sterile Ringer's solution only. For each nematode type, PUR1 or YEL3, bacterial treatments which do not share the same letter are significantly different. (One-way ANOVA, P < 0.05)

Both nematode types appeared to prefer the bacterial mixture rather than their own or the other bacterial type. As the trend in invasion rate in the three *P. temperata* treatments only, (pur1/yel3 > pur1 > yel3), was similar for both nematode types, the

data were analysed excluding the no bacteria treatment to see the effect of bacterial treatment on nematode invasion. For the YEL3 nematodes the difference in the mean number of nematodes invading was still found to be highly significant comparing the *yel3* treatment to the *pur1/yel3* 1:1 treatment but not the *pur1* treatment (one-way ANOVA,  $F_{2,27} = 10.51$ , P < 0.001). For the PUR1 nematodes there was a difference at the 10% level with fewer nematodes invading in the *yel3* treatment compared to the *pur1/yel3* 1:1 treatment but not to the *pur1* individual treatment (on-way ANOVA,  $F_{2,27} = 2.92$ , P = 0.071).

The data from both experiments were combined and the number of nematodes invading were analysed using a two-way ANOVA with bacterial treatment and nematode type as factors. This showed that both bacterial treatment and nematode type had a highly significant effect on the number of nematodes invading the cadavers (Table 3.4).

**Table 3.4:** Results of two-way ANOVA comparing the effect of both bacterial treatment and nematode type on the number of IJs invading *G. mellonella* in three bacterial treatments.

Factor	Test statistic	Р
Bacterial treatment	$F_{5,54} = 6.47$	0.003
Nematode type	$F_{1,54} = 40.76$	< 0.001
Interaction term	$F_{2,54} = 0.67$	0.516

Since there was no interaction the combined data were analysed by bacterial type and by nematode type. When the numbers of nematodes invading the injected insects from both experiments were pooled by bacterial treatment more nematodes invaded insects in the *pur1/yel3* 1:1 treatment than either of the individual bacterial treatments and the difference compared to the *yel3* treatment was significant (one-way ANOVA,  $F_{2,57} = 5.16$ , P = 0.009) (Figure 3.5). The data were log to the base 10 transformed for analysis. The mean number of nematodes invading the insects in the *pur1/yel3* 1:1 treatment was almost twice that in the *yel3* treatment and more nematodes also invaded insects in the *pur1* treatment compared to the *yel3* treatment, but not significantly so. This preference for the *pur1/yel3* 1:1 treatment when comparing the pooled nematode data was in line with the results for the individual nematode types.



**Figure 3.5:** Mean number ( $\pm$  SE) of adult nematodes per insect in *G. mellonella* at day 4 following infection with 100 IJs per insect. Insects had previously been injected with 1 x 10<sup>5</sup> cells of *P. temperata* isolates *pur1, yel3*, or a *pur1/yel3* 1:1 mixture. Data for both nematode types are pooled by bacterial treatment and are a mean of two experiments with n = 20 for all treatments. Treatments which do not share the same letter were significantly different (one-way ANOVA, P < 0.05).

When the numbers of nematodes invading the injected insects from both experiments were pooled by nematode type more PUR1 nematodes than YEL3 nematodes invaded and the difference was highly significant (one-way ANOVA,  $F_{1,58} = 34.62$ , P < 0.001) (Figure 3.6).



**Figure 3.6:** Mean number ( $\pm$  SE) of adult nematodes per insect in *G. mellonella* at day 4 following infection with 100 IJs per insect. Insects had previously been injected with 1 x 10<sup>5</sup> cells of *P. temperata* isolates *pur1, yel3*, or a *pur1/yel3* 1:1 mixture. Nematode used were PUR1 (experiment 1) and YEL3 (experiment 2). Data for all three bacterial treatments are pooled by nematode type and are a mean of two experiments with n = 30 in each experiment. Treatments which do not share the same letter were significantly different (one-way ANOVA, P < 0.001).

As a similar trend was observed for number of colony forming units from injected insects (Figure 3.3) and number of nematodes invading injected insects (Figure 3.6) in both experiments, (experiment 1 > experiment 2 for each parameter), the relationship between the level of bacterial colonisation of the insects and nematode

invasion rate was assessed using a Spearman's correlation (Figure 3.7). The test showed that there was a positive correlation between the mean number of colony forming units of bacteria in an insect and the invasion rate of the nematodes ( $\rho = 0.6406$ , n = 60, P < 0.001).





**Figure 3.7:** Correlation between invasion rate of nematodes in *G. mellonella* injected with one of three *P. temperata* treatments and subsequently infected with 100 IJs per insect and mean colony forming units in insects injected with one of three *P. temperata* treatments in two experiments. Nematode used were PUR1 (experiment 1) and YEL3 (experiment 2) with n = 10 in each experiment and each bacterial treatment.

## **3.3.1.3** Time of infective juvenile first emergence

Both of the *H. downesi* isolates PUR1 and YEL3 successfully reproduced and emerged from insects in all treatments, with first emergence of IJs occurring in some insects in all four treatments at 16 days post infection with nematodes (Appendix Table A 6). However, there was some variation in time to first emergence for all insects in each treatment with first emergence occurring at 31 days for one insect in

the *pur1/yel3* 1:1 treatment. The data were not normally distributed and could not be transformed to fit the normal distribution therefore a series of non parametric tests were used to analyse the data. The median time to first emergence was compared for each nematode type separately using a Kruskal-Wallis test. This showed that in each case there was a highly significant difference in the time to first emergence of infective juveniles between bacterial treatments (P = 0.006 and P = 0.004 for PUR1 and YEL3 IJs respectively). To determine which treatments differed a series of multiple pairwise comparisons using a two-tailed Mann-Whitney U-test with a Bonferroni 95% CI were carried out (Appendix Table A 7A and Table A 7B). In the case of the PUR1 nematodes the median time to first emergence was longer in the no bacteria treatment insects (19 days) than any of the P. temperata treatments (16 to 18 days) and this difference was significant compared to the yel3 treatment (P =0.0041) (Table 3.5). The opposite trend was observed for the YEL3 nematodes where the shortest median times to first emergence occurred in the no bacteria treatment insects and the *yel3* treatment (16.5 and 17 days respectively) compared to the other two bacterial treatments (18 and 19.5 days for the purl/yel3 1:1 and purl treatments, respectively) and the difference was highly significant for the no bacteria treatment compared to the *pur1* treatment (P = 0.0003).

**Table 3.5:** Time to first emergence for PUR1 and YEL3 IJs from *G. mellonella* infected with 100 IJs. Insects had previously been injected with 1 x  $10^5$  cells of *P. temperata* isolates *pur1*, *yel3* or a *pur1/yel3* 1:1 mixture. Insects in the no bacteria treatment were injected with an equal volume of <sup>1</sup>/<sub>4</sub> sterile Ringer's solution only.

Bacterial treatment	Median time to first	Median time to first emergence (days)				
	Ner	natode type				
	PUR1	YEL3				
pur1	18.5 ab*	19.5 a				
yel3	16.0 b	17.0 ab				
<i>pur1/yel3</i> 1:1	17.5 ab	18.0 ab				
no bacteria	19.0 a	16.5 b				

\* Treatments which do not share the same letter were significantly different (Mann-Whitney U-test, 95% CI Bonferroni-adjusted significance level of  $\alpha = 0.008$ ).

However, when comparing the time to first emergence in the *P. temperata* treatments only the trend for both nematode types was *pur1* > *pur1/yel3* 1:1 > *yel3*. Since the median time to first emergence for both the PUR1 IJs and the YEL3 IJs showed a similar trend the no bacteria treatment was omitted and the data were pooled and analysed by bacterial treatment and by nematode type. When time to first emergence for all nematodes was pooled by bacterial treatment the median time ranged from 17 days in the *yel3* treatment to 18 days in the *pur1/yel3* 1:1 treatment to 19 days in the *pur1* treatment (Figure 3.8A) suggesting that the nematodes were more likely to emerge sooner from insects infected with *yel3* bacteria. A Kruskal-Wallis test showed that there was a highly significant difference between the three bacterial treatments (H = 10.16, df = 2, P = 0.006). Subsequent pairwise comparisons using a Mann-Whitney U-test with Bonferroni 95% CI showed that the difference in the median time to first emergence was significant for the *pur1* compared to the *yel3* treatment (P = 0.0031) (Appendix Table A 8). When the time to first emergence was pooled by nematode type the median time for the PUR1 IJs was 17 days while the median time for the YEL3 IJs was 18 days (Figure 3.8B) however this difference was not significant (Mann-Whitney U-test, P > 0.05).



Days to first emergence of IJs

**Figure 3.8:** Median time to first emergence of IJs from insects following infection with 100 IJs per insect of PUR1 or YEL3 isolates from North Bull Island. Insects had previously been injected with 1 x 10<sup>5</sup> cells of *P. temperata* isolates *pur1, yel3*, or a *pur1/yel3* 1:1 mixture. Data shown is (A) both nematode types pooled by bacterial treatment with n = 20 for each treatment and (B) all three bacterial treatments pooled by nematode type with n = 30 for each nematode type. (A) Treatments which do not share the same letter are significantly different (Two-tailed Mann-Whitney U-test, P < 0.05, with 95% CI Bonferroni-adjusted significance level of  $\alpha = 0.017$ ).

#### **3.3.1.4** Numbers of IJs emerging

For each of the nematode and bacterial treatments the White traps were harvested at a number of time points. Numbers of IJs emerging are dealt with as follows:

- i) Estimates of total harvests from all four treatments at week 1, week 2 and week 3.
- ii) Day 4 and Day 19 counts of IJs from *pur1* and *yel3* treatments only.
- iii) Counts of pooled total emergence for 26 days from *pur1* and *yel3* treatments only.
- i) Estimates of total harvests from all four treatments at week 1, week 2 and week3.

Emergence was estimated as described in Section 3.2.1.6d) for each cadaver each week for three weeks in all four treatments (Table 3.6). In both the PUR and YEL nematodes the emergence in week 1 was highest in the no bacteria treatment and the emergence in this treatment declined in subsequent weeks. For both nematode types more IJs emerged from insects in the *pur1* treatment in week 1 compared to the *yel3* treatment or the *pur1/yel3* 1:1 treatment. Unlike the no bacteria treatment, emergence for all IJs except the YEL3 nematode/*pur1* bacteria combination was highest in week 2. By week 3 IJ emergence had declined in all three bacterial treatments and at this time point more IJs emerged from both the *yel3* treatment and the *pur1/yel3* 1:1 compared to the *pur1* treatment for both nematode types.

**Table 3.6:** Estimated IJ emergence over three weeks from *G. mellonella* infected with 100 nematodes per insect. Insects had previously been injected with 1 x  $10^5$  cells of *P. temperata* isolates *pur1, yel3*, or a *pur1/yel3* 1:1 mixture. Insects in the no bacteria treatment were injected with an equal volume of <sup>1</sup>/<sub>4</sub> sterile Ringer's solution only. N = 10 for each treatment for each nematode type except YEL3/*pur1/yel3* where n = 9.

Nematode type	Bacterial treatment	Week 1	Week 2	Week 3	total (± SE)
	pur1	486467	896250	142027	1524744 <sub>31603</sub>
DI ID 1	yel3	226341	1492150	294753	2013244 <sub>19823</sub>
PUKI	<i>pur1/yel3</i> 1:1	119638	705140	246802	$1071580_{21356}$
	no bacteria	780804	752500	198540	1731845 <sub>27920</sub>
YEL3	pur1	599945	261325	219528	1080798 <sub>34133</sub>
	yel3	229115	446800	309875	985790 <sub>19556</sub>
	<i>pur1/yel3</i> 1:1	522750	1118450	327627	1968827 <sub>72544</sub>
	no bacteria	791050	584350	252100	$1627500_{16908}$

There appeared to be a trend for greater numbers of IJs from both the PUR1 and the YEL3 nematode infections to emerge earlier in the *pur1* treatment compared to the *yel3* or the *pur1/yel3* treatments, therefore the estimated emergence in week 1 as a proportion of the total estimated emergence for three weeks in all four treatments was analysed to investigate the trend for higher numbers of IJs to emerge early from cadavers in the *pur1* treatment. As n varied between treatments the data were analysed using a general linear model with nematode type and bacterial treatment as factors. The analysis showed that bacterial treatment had a highly significant effect on the proportion of IJs emerging early while nematode type did not (Table 3.7). All proportion data were arcsine square root transformed for analysis

**Table 3.7:** Results of analysis of variance using a general linear model comparing the effect of both bacterial treatment and nematode type on estimated numbers of IJs emerging early. All proportion data were arcsine square root transformed for analysis

Factor	Test statistic	Р
Bacterial treatment	F <sub>3,74</sub> =5.25	0.002
Nematode type	$F_{1,74} = 0.50$	0.481

When the data were pooled and analysed by bacterial treatment the overall mean estimated proportion of IJs emerging early (week 1) in the *pur1* and no bacteria treatments was more than double that of the *pur1/yel3* 1:1 mixture and individual *yel3* treatments and this difference was highly significant (one-way ANOVA,  $F_{3,75} = 5.31$ , P = 0.002) (Table 3.8). All proportion data were arcsine square root transformed for analysis.

**Table 3.8:** Estimated proportion of IJs emerging early from *G. mellonella* infected with 100 IJs per insect. Insects had previously been injected with 1 x 105 cells of *P. temperata* isolates *pur1*, *yel3* or *pur1/yel3* 1:1 mixture. Insects in the no bacteria treatment were injected with an equal volume of <sup>1</sup>/<sub>4</sub> sterile Ringer's solution only. Data for both nematode types are pooled by bacterial treatment and are number of IJs emerging in week 1 as a percentage of total emergence over three weeks with n = 20 for all treatments except *pur1/yel3* (n = 19). All proportion data were arcsine square root transformed for analysis. Treatments which do not share the same letter are significantly different (one-way ANOVA,  $F_{3,75} = 5.31$ , P = 0.002).

Bacterial treatment	Mean estimated proportion of IJs emerging early (% week 1/total) (± SE)
pur1	42.0(7.32) a
yel3	18.7(3.33) b
pur1/yel3	19.7(4.40) b
no bacteria	48.6(7.35) a

ii) Day 4 and Day 19 counts of IJs from *pur1* and *yel3* treatments only.

Individual counts were carried out for harvests from each of four White traps for the wild type and hybrid combinations listed in Table 3.2 for early (day 4) and late (day 19) harvests (Figure 3.9**Error! Reference source not found.**A and Figure 3.9B). The day 4 harvest contained IJs emerging over the first four days for each White trap while the day 19 harvest contained IJs emerging over a 24 hour period from day 18 to day 19. While there was a large variation in the emergence there appeared to be a trend for greater numbers of IJs from both the PUR1 and the YEL3 nematode infections to emerge earlier in the *pur1* bacterial treatment (Table 3.9).



**Figure 3.9:** Mean ( $\pm$  S.E.) emergence per insect of (A) PUR1 and (B) YEL3 IJs from insects following infection with 100 nematodes per insect. Insects had previously been injected with 1 x 10<sup>5</sup> cells of *P. temperata* isolates *pur1* or *yel3*. Data shown are for harvests at day 4 and day 19, n = 4 for two bacterial treatments at each time point.

**Table 3.9:** Mean ( $\pm$  SE) percentage of IJs emerging early from insects following infection with 100 nematodes per insect. Insects had previously been injected with 1 x 10<sup>5</sup> cells of *P. temperata* isolates *pur1* or *yel3*. Data shown are for harvests at day 4 as a percentage of the total harvest at day 4 and day 19, n = 4 for each bacterial treatments.

	Bacterial treatment				
Nematode type	pur1	yel3			
PUR1	92.5(3.08)	72.5(16.38)			
YEL3	81.7(17.19)	61.4(18.34)			
Overall	87.1(8.34)	66.9(11.57)			

The data were combined and the number of PUR1 and YEL3 IJs emerged by day 4 as a proportion of the total harvest for each IJ in each bacterial treatment at both time points was compared using a two-way ANOVA with bacterial treatment and nematode type as factors (Appendix Table A 9). The analysis showed neither bacterial type nor nematode type had a significant effect on the number of IJs emerging early and there was no interaction between bacterial treatment and nematode type (P > 0.05 in all cases).

iii) Counts of pooled total emergence for 26 days from *pur1* and *yel3* treatments only.

The data from experiment 1 and experiment 2 were combined and the total emergence for 26 days for each White trap was pooled and counted (Table 3.10). A Two-way ANOVA comparing number of IJs emerging with bacterial treatment and nematode type as factors showed neither bacterial type nor nematode type had a significant effect on the number of IJs over the 26 day period and there was no

interaction between bacterial treatment and nematode type (P > 0.05 in all cases) (Appendix Table A 10).

**Table 3.10:** Mean number of IJs ( $\pm$  SE) emerging per insect from *G. mellonella* over 26 days following infection with 100 IJs per insect. Nematodes used were PUR1 (experiment 1) or YEL3 (experiment 2) isolates from North Bull Island. Insects had previously been injected with 1 x 10<sup>5</sup> cells of *P. temperata* isolates *pur1*or *yel3*. Data is from two experiments with n = 5 for all treatments.

	Bacterial treatment			
Nematode type	pur1	yel3		
PUR1	103095(21356)	104016(14456)		
YEL3	85062(12769)	65863 <sub>(22152)</sub>		
Overall	94078(14813)	84940(11095)		

#### **3.3.2 Bacteria carried by emerging infective juveniles**

A series of continuous exposure single nematode infections were set up as described in Section 3.2.2.2 and insect mortality due to entomopathogenic nematode infection was recorded for a period of eight days for IJs harvested at each time point (day 4 and day 19). Mean mortality ranged from approximately 19 to 44 % (Table 3.11). The overall mean mortality for both nematode types and both harvest time points ranged from 31% for progeny from the *pur1* treatment to 32.6% for progeny from the *yel3* treatment to 36.5% for progeny from the *pur1/yel3* 1:1 treatment (Figure 3.10). The data for both harvest time points were combined for analysis. As n varied between treatments the data were analysed using a general linear model with nematode type, bacterial treatment and harvest time point as factors. The analysis showed there was no effect of any of these factors on the percentage mortality of the

insects and no interaction between any factors (P > 0.05 in all cases).

**Table 3.11:** Mean insect mortality ( $\pm$  SE) in single nematode infections using IJs from insects that had been injected with 1 x 10<sup>5</sup> cells of *P. temperata* isolates *pur1, yel3* or *pur1/yel3* 1:1 mixture and infected three days later with 100 IJs per insect. Nematodes used were PUR1 or YEL3 isolates from North Bull Island. Data shown is pooled by bacterial treatment for IJs from PUR1 and YEL3 infection types for each harvest time point, with n = 4 for each nematode type in the individual bacterial treatments at each time point and n = 10 and 9 for the PUR1 and YEL3 nematodes respectively in the pur1/yel3 1:1 bacterial treatment at each harvest time point.

		Bacterial treatment				
Nematode type	Harvest time point	pur1	yel3	pur1/yel3		
PUR1	Day 4	39.58 <sub>(4.96)</sub>	32.29(6.88)	34.58(4.12)		
	Day 19	18.75 <sub>(7.12)</sub>	29.17 <sub>(4.50)</sub>	31.67(1.55)		
YEL3	Day 4	30.21(5.48)	33.33 <sub>(2.41)</sub>	37.04(5.49)		
	Day 19	30.56(3.67)	35.40(4.72)	43.52(4.72)		



**Figure 3.10:** Mean *G. mellonella* mortality ( $\pm$  SE) in single nematode infections using IJs from insects that had been injected with 1 x 105 cells of *P. temperata* isolates *pur1, yel3* or *pur1/yel3* 1:1 mixture and infected three days later with 100 IJs per insect. Nematodes used were PUR1 or YEL3 isolates from North Bull Island. Data shown is pooled by bacterial treatment for PUR1 and YEL3 IJs from two harvest time points, n for each bacterial treatment is displayed above the bars.

In this experiment the identity of the bacteria predominating in the IJ is ascertained based on the colour of the insect cadaver. In each of the single bacterial treatments the IJs always carried the bacterial phenotype which had been injected into the insect three days prior to nematode infection, irrespective of the colour phenotype of the nematode used in the subsequent infection. In the case of the *pur1/yel3* 1:1 bacterial treatment the IJs from any given source insect all tended to carry a single bacterial colour phenotype (Figure 3.11 and Table 3.12), although both bacterial colour phenotypes were detected in infections using IJs from 3/10 cadavers in the case of PUR1 IJs and from 2/9 cadavers in the case YEL3 IJs.



**Figure 3.11:** Cadaver colour in single nematode infections using *G. mellonella*. The IJs used for the infection were harvested on day 4 following emergence from cadavers which had been infected with  $1 \times 10^5$  cells of *P. temperata* isolates *pur1/yel3* 1:1 mixture and infected three days later with 100 PUR1 IJs per insect. Images shown are for IJs from two separate cadavers: (C3) all infections resulted in yellow cadavers and (C4) all infections resulted in purple cadavers. Cells marked with a cross (+) indicate insects killed by nematode infection on day 4 of the experiment.

**Table 3.12:** Colour phenotype of *G. mellonella* in single nematode infections using emerging IJs from insects in the cross-compatibility experiment that had been injected with 1 x  $10^5$  cells of *P. temperata* isolates *pur1/yel3* 1:1 bacterial mixture and infected three days later with 100 IJs per insect with PUR1 or YEL3 isolates from North Bull Island. N = 24 insects at each time point per harvest from 10 (PUR IJs) or 9 (YEL3) cadavers.

	Cadaver colour in single nematode infection $(/24 \text{ for each infection at each time point})$								
		Harvest				<i>jointy</i>			
		da	iy 4	da	dav 19		otal		
Nematode type	Cadaver	р	у	р	у	р	у	source cadaver colour <sup>†</sup>	Cadaver colour in single nematode infections <sup>‡</sup>
PUR1 IJs	1	4	0	0	8	4	8	$\mathbf{P}^1$	$p \rightarrow y^4$
	2	5	0	7	0	12	0	Р	р
	3	0	7	0	8	0	15	$P/Y^2$	у
	4	0	8	0	7	0	15	P/Y	у
	5	12	0	0	6	12	6	Р	$p \rightarrow y$
	6	11	0	9	0	20	0	Р	p
	7	2	6	0	6	2	12	Р	$p + y \rightarrow y^5$
	8	0	14	0	7	0	21	P/Y	у
	9	0	7	0	9	0	16	Р	у
	10	0	7	0	9	0	16	P/Y	У
YEL3 IJs	1	0	4	0	9	0	13	P/Y	у
	2	2	4	0	9	2	13	Y <sup>3</sup>	$p + y \rightarrow y$
	3	0	6	0	12	0	18	Y	У
	4	11	0	13	0	24	0	Р	p
	5	0	15	0	10	0	25	Р	У
	6	0	8	0	8	0	16	P/Y	У
	7	0	15	0	17	0	32	Y	У
	8	7	0	0	11	7	11	Р	$p \rightarrow y$
	9	8	0	5	0	13	0	Р	р
Overall		62	101	34	136	96	237		

<sup>†</sup> The colour of the cadavers was assessed at day 4 of the cross-compatibility experiment:

1. P = achromatic cadaver colouration typically seen early in infection with *H. downesi* PUR isolates.

2. Cadaver colour was recorded as P/Y when the colouration was intermediate between typical PUR or YEL infection phenotypes.

3. Y (yellow) = yellow cadaver colouration typically seen early in infection with *H. downesi* YEL isolates.

 $\ddagger$  The cadaver colour in the one-on-one bioassay was recorded as p or y for PUR or YEL nematode infection respectively.

- 4. An arrow in the final column indicates the colour of the cadavers was different in the infection with IJs from the second harvest.
- 5. A plus (+) symbol indicates cadavers of both colour phenotypes.

Overall, ten source cadavers produced yellow IJs only (based on the cadaver colour resulting from the single nematode infections), four produced purple IJs only and five produced IJs of both colours, as noted already (Table 3.12). The proportion of cadavers with the *y* colour phenotype was higher for each nematode type from each harvest time point. The proportion of each cadaver colour phenotype, (*p*, *y*), was tested at each harvest time point for each nematode type (Table 3.13). The difference between the proportion of *y* and *p* colour phenotypes was significant at the 10% level for PUR1 IJs at the day 4 harvest time point. The difference between the proportion of *y* and *p* colour phenotypes was highly significant for both nematode types at all other harvest time points ( $\chi^2$ , P < 0.01).

**Table 3.13:** Number of *G. mellonella* cadavers of each colour phenotype in single nematodes infections using emerging IJs from insects in the cross-compatibility experiment that had been injected with  $1 \times 10^5$  cells of *P. temperata* isolates *pur1/yel3* 1:1 bacterial mixture and infected three days later with 100 IJs per insect with PUR1 or YEL3 isolates from North Bull Island. n = 240 (PUR IJs) or 216 (YEL3) cadavers at each time point.

	Cadaver colour in single nematode infections						
-	Harvest						
-	day 4 day 19						
Nematode type	р	У	р	у			
PUR1 IJs	34	49	16	60			
Test result	$\chi^2 = 3.278$ , df = 1, P = 0.070		$\chi^2 = 30.266, dt$	f = 1, P = 0.000			
YEL IJs	28	52	18	76			
	$\chi^2 = 8.836$ , df	= 1, P = 0.003	$\chi^2 = 45.740, dt$	f = 1, P = 0.000			

The data for IJs from the *pur1/yel3* 1:1 treatment were pooled and the colour of source cadaver (Section 3.2.1.3) was compared to the colour carried by the IJs (Table 3.14). All of the nine source cadavers classified as yellow (Y) or purple/yellow (P/Y)

produced IJs carrying *yel3* bacteria, whereas only eight of the sixteen source cadavers classified as purple (P) or P/Y produced IJs carrying *pur1* bacteria.

**Table 3.14:** Comparison of source cadaver colour and cadaver colour resulting from a single nematode infection using emerging IJs from the cross-compatibility experiment. Data shown is for IJs emerging from *pur1/yel3* 1:1 bacterial treatment in the cross-compatibility experiment and is pooled for PUR1 and YEL3 nematodes in that experiment. (Insect used were *G. mellonella* larvae)

Source cadaver colour	cadaver colour in single nematode infection		
	р	p + y	У
Р	4	4	2
P/Y	0	0	6
Y	0	1	2

A number of two-way comparisons were carried out to test whether cadaver colour tends to predict the IJ colour (Table 3.15, see also Appendix Table A 11). The analysis showed that the cadaver colour tended to predict IJ colour as where the source cadaver appeared yellow (Y) or purple/yellow (P/Y) it is likely that all of the progeny from a cadaver will be carrying the *yel3* (*y*) bacteria, although in one case a yellow source cadaver produced some progeny with *pur1* (*p*) bacteria. Conversely, when there is no sign of yellow in the source cadaver it is likely that at least some of the progeny will be carrying the *pur1* bacteria, although in two cases for purple source cadavers no *pur1* progeny were detected using single nematode infections. These results suggest that the *yel3* bacteria are better at colonising the nematodes within the co-infected cadavers despite the apparent dominance of the *pur1* bacteria in colonising the insect (Section 3.3.1.1 above).

**Table 3.15:** Results of two-way comparisons of source cadaver colour and single nematode infection cadaver colour (Fisher's exact test). Data shown is for IJs emerging from *pur1/yel3* 1:1 bacterial treatment in the cross-compatibility experiment and is pooled for PUR1 and YEL3 nematodes in that experiment. (Insect used were *G. mellonella* larvae in all cases).

source cadaver colour	cadaver colour in single nematode infection	P-value
yellow or not	purple or not	0.005
yellow or not	yellow or not	0.087
purple or not	purple or not	1
purple or not	yellow or not	1

# 3.3.3 Tests of wild type and hybrid nematode/bacteria strains against G. mellonella

For each of the wild type and hybrid combinations listed in Table 3.2 four infections using *G. mellonella* were set up as described in Section 3.2.3. The mortality of insects in the various treatments was compared at a number of time points. The data at 40 hours and 48 hours was not normal and could not be normalised. The mean or median percentage of insects dead varied between the individual combinations, ranging from 0% at 40 hours to 100% at 48 hours (Table 3.16).

**Table 3.16:** Median<sup>†</sup> or mean<sup>‡</sup> ( $\pm$  SE) percentage of *G. mellonella* dead at three time points for wild type and hybrid combinations of nematode and bacteria. All insects were infected with 100 IJs per insect using IJs emerging from cadavers in the *pur1* and *yel3* bacterial treatments in the cross-compatibility experiment. There were four replications of each combination with ten insects per replication. Results of Kruskal-Wallis test or one-way ANOVA comparison between treatment medians or means are shown for proportion dead at each time point.

		Time point (hrs	5)
Nematode/ bacteria combination	$40^{\dagger}$	44 <sup>‡</sup>	$48^{\dagger}$
Рр	30	75 (11.9)	100
Yp	15	60 (1.08)	85
Yy	0	47.5 (11.1)	75
Ру	0	75 (6.45)	95
P value	0.121	0.226	0.051
Test statistic (3 df)	H = 5.81	$F_{3,12} = 1.67$	H = 7.77

In all four treatments all insects had died within 64 hours of infection and the PUR nematodes tended to produce higher mortality rate, irrespective of bacterial colour. As the data at 44 hours were normally distributed the percentage of insects dead at this time point were analysed using a two-way ANOVA with nematode type and bacteria type as factors. The analysis showed there was no effect of bacteria type and no evidence of an interaction between bacteria type and nematode type. Nematode type was significant at the 10% level but not at the 5% level (P = 0.061) (Appendix Table A 12). There was no significant difference in the median number of insects dead at 44 hours at the 5% level, although there was a significant difference between treatments at the 10% level for the number of insects dead by 48 hours. However, pairwise post-hoc comparisons (Two-tailed Mann-Whitney U-test, with Bonferroni 95% CI) showed no difference between combinations (Appendix Table A 13).

The data were pooled by nematode type (Table 3.17) and by bacterial type (Table 3.18) and median or mean percentage mortality compared using either a Mann-Whitney U-test or a one-way ANOVA. When the data were pooled by nematode type a higher percentage of insects were dead at 40 hours in the YEL3 treatment compared to the PUR1 treatment but the difference was not significant. However, at 44 hours and 48 hours a significantly higher percentage of insects were dead in the PUR1 treatment compared to the YEL 3 treatment.

**Table 3.17:** Median<sup>†</sup> or mean<sup>‡</sup> ( $\pm$  SE) percentage of *G. mellonella* dead at 3 time points for wild type and hybrid infections. Data is pooled by nematode type. Results of Mann-Whitney U-test or one-way ANOVA comparison between treatment medians or means are shown for number dead at each time point. N = 8 at each time point.

	Time point (hrs)		
Nematode type	$40^{\dagger}$	44 <sup>‡</sup>	$48^{\dagger}$
PUR1	5	75.00 (6.27)	100
YEL3	10	53.75 (7.54)	80
P value	0.740	0.049	0.0116
Test statistic $(df = 1)$	W = 64.5	T = 2.17	W = 91.5

When the data were pooled by bacterial type a greater percent of insects were dead at 40 hours in the *pur1* treatment compared to the *yel3* treatment and this difference was highly significant. These results suggest that both nematode type and bacteria play a role in insect mortality and that the *pur1* bacteria kill the insects quicker than the *yel3* bacteria do.

**Table 3.18:** Median<sup>†</sup> or mean<sup>‡</sup> ( $\pm$  SE) percentage of *G. mellonella* dead at 3 time points for wild type and hybrid infections. Data is pooled by bacterial type. Results of Mann-Whitney U-test or one-way ANOVA comparison between treatment medians or means are shown for percentage dead at each time point. N = 8 at each time point.

		Time point (hrs)	
Bacteria type	$40^{\dagger}$	44 <sup>‡</sup>	$48^{\dagger}$
pur1	15	67.5 (7.96)	95
yel3	0	61.2 (7.89)	90
P value	0.0274	0.587	0.3720
Test statistic (1 df)	W = 89.5	T = 0.56	W = 77.0

# 3.3.4 Survival of IJs in water

The *in vitro* survival in water of the two *H. downesi* colour phenotypes from North Bull Island was investigated as described in Section 0. There was a gradual decline in the numbers of IJs surviving, with most isolates showing a reduction in survival by the first time point (week two) (Figure 3.12). Infective juveniles of both colour phenotypes, PUR and YEL, survived in water at 20 °C for up to 14 weeks, at which point the experiment was terminated. No IJs of the PUR3 isolate remained alive at week 14 and the mean percentage survival for the PUR1 and PUR2 isolates was 0.2% and 0.6% respectively at week 14 while the mean percentage survival for the YEL isolates was 0.2 % (YEL2), 0.3 % (YEL3) and 0.4 % (YEL1). There was no overall trend for either colour phenotype to survive better over the fourteen week duration of the experiment.



**Figure 3.12:** Mean percentage ( $\pm$  SE) survival in water of (A) PUR and (B) YEL isolates from North Bull Island over a 14 week period at 20 °C. N = 5 at each time point.

In week 2 and week 4 of the experiment both the highest percentage survival and the lowest percentage survival occurred in the yellow phenotype (YEL1 and YEL2 respectively), whereas for week 8 to week 14 the highest percentage survival and the lowest percentage survival occurred in the purple phenotype (PUR2 and PUR3 respectively). In week 6 the mean percentage survival for the PUR IJs was 41.8%, 48.1% and 18.8% for PUR1, PUR2 and PUR3 respectively while mean percentage survival for the YEL IJs was 40.2%, 51.0% and 48.9% for YEL1, YEL2 and YEL3 respectively. Thus the trend in percentage survival was YEL2 > YEL3 > PUR2 >
PUR1 > YEL1 > PUR3. A Regression with Life Data analysis of the survival data in weeks 2 to 8 was carried out using week, colour phenotype and isolate as model terms. This showed no effect of colour phenotype or isolate on survival.

### 3.3.5 Bacterial quantification from individual infective juveniles

The experiment described in Section b) was carried out twice to determine the number of *P. temperata* colonising individual IJs of each isolate. The first experiment was carried out using the six isolates from North Bull Island only. The two hybrid combinations were included in the second experiment along with the six North Bull Island isolates. In each experiment the IJs were found to have variable numbers of bacteria, with 5 to 10% of IJs having no detectable bacteria, (PUR2 in experiment 1 and YEL2, YEL3 and YEL/*pur* in experiment 2 (Appendix Table A 14 )). Overall, bacteria were detected in 48/50 PUR IJs in experiment 1 and 56/60 PUR IJs in experiment 2 whereas the opposite trend was observed for the YEL IJs with bacteria being detected in 56/60 YEL IJs in experiment 1 and 59/60 YEL IJs in experiment 2. Bacteria were also detected in 19/20 PUR1/*yel3* hybrids and 20/20 YEL3/*pur1* hybrids in experiment 2. As there was no evidence of a colour bias no statistical tests were carried out on differences in the number of IJs having no detectable bacteria between the colour phenotypes.

In experiment 1 colonization levels ranged from 0 colony forming units (cfu) per IJ to 1746 cfu per IJ (Figure 3.13A). In experiment 2 colonization levels ranged from 0 cfu per IJ to 1154 cfu per IJ (Figure 3.13B). In order to compare the colonization level in the IJs which carried bacteria zeros and outliers were removed (Figure 3.14A-B). The data for both experiments were pooled and analysed using a fully nested general linear model with experiment, nematode type and isolate as factors. There was no effect of nematode type or isolate on the number of bacteria carried by

the IJs (P > 0.005 for both factors), however there was a highly significant effect of experiment (GLM,  $F_{1,195} = 11.71$ , P = 0.001). Since there was no evidence for either IJ phenotype, PUR or YEL to carry more bacteria in either experiment and there was a difference in the number of bacteria carried in each experiment no comparisons were carried out on the data pooled by nematode type.

For the wild type and hybrid strains in experiment 2 there was no significant difference between treatments when the number of cfu per IJ was compared within nematode phenotype (PUR or YEL) or within bacterial phenotype (*pur* or *yel*) (one-way ANOVA, P > 0.05).





<sup>1</sup> Natural associations are PUR1/*pur1*, PUR2/*pur2*, PUR3/*pur3*, YEL1/*yel1*, YEL2/*yel2*, YEL3/*yel3*.

<sup>2</sup> YEL3 nematode colonised with *pur1* bacteria.

<sup>3</sup> PUR1 nematode colonised with *yel3* bacteria.



**Figure 3.14:** Mean ( $\pm$  SE) number of bacterial colonies of *P. temperata* from (A) six *H. downesi* isolates from North Bull Island and (B) six *H. downesi* isolates from North Bull Island plus two hybrid strains. IJs containing no bacteria and outliers were removed prior to analysis and the data were square root transformed for analysis.

### 3.4 Discussion

The cross-compatibility experiment showed that both strains of *H. downesi*, PUR1 and YEL3, were able to invade cadavers previously killed by two different subspecies of *Photorhabdus temperata*, *P. temperata* subsp. *temperata* (*yel3*) and *P temperata* subsp. *cinerea* (*pur1*). The nematodes were able to develop and reproduce on both the homologous and novel bacteria and infective juveniles of each nematode strain retained the novel bacterial subspecies. This study showed that nematodes would form new combinations when they invaded insects already infected with a novel *Photorhabdus* subspecies. Hybrid combinations subsequently were stable over many cycles of laboratory culturing and were used in experiments in Chapter 4 and Chapter 5.

Other studies have shown that nematode/bacteria recombination was possible. For example, Gerritsen & Smits (1993) found that *H. megidis* strains NLH-E87.3 and NLH-F85 were able to recombine with each other's symbionts, *Photorhabdus* strains PE (XE) and PF (XF) respectively, when axenic IJs of each strain were placed on lawns of the other bacteria, which were serologically different (Gerritsen *et al.*, 1995) and had distinct restriction fragment length polymorphism (RFLP) patterns (Smits *et al.*, 1991). (The symbionts of *Heterorhabditis* spp. were initially classified as *Xenorhabdus luminescens* and subsequently renamed *Photorhabdus luminescens* (Boemare *et al.*, 1993b)). Separate species and subspecies within the genus were recognised later (see Section 1.2.4). The two *Photorhabdus* strains PE and PF were later identified as *P. temperata* subsp. *temperata* (Susurluk *et al.*, 2001; Maneesakorn *et al.*, 2011) using phylogenetic techniques. The novel symbionts supported the growth and reproduction of the host nematode and there was no difference in mortality of *G. mellonella* larvae exposed to natural or recombined

nematode/bacteria partnerships (Gerritsen & Smits, 1993). Gerritsen and Smits (1997) found that there was no difference in the number of IJs produced by nematodes cultured on their own or other bacteria for two strains of *H. megidis*, DH-SH1 and NLH-E87.3. On the other hand, the ability of the symbiont of one nematode species to successfully recombine with another nematode species is more limited. Gerritsen & Smits (1993) found that the two H. megidis symbionts PE and PF supported the growth and reproduction of axenic IJs of *H. bacteriophora* strains Hbac and Hmol *in vitro* but the novel combinations tended to be non-pathogenic to G. mellonella, suggesting that the symbiont was not carried by the IJs, confirmed by Gerritsen et al. (1998). The H. bacteriophora strain H06 was able to reproduce on the symbiont of *H. megidis*, HNA, both in vitro and in vivo, but not on the symbiont of *H. indica* (Han & Ehlers, 2001). It has been shown that heterologous bacteria may provide better growth conditions than the homologous bacteria (Han et al., 1991) where cultures of Heterorhabditis sp. H3 and HNZ both had higher yields of IJs when combined with non-homologous strains of P. luminescens. In an in vitro choice test using both homologous and heterologous bacterial lawns two H. megidis strains, HE and HSH IJs showed no preference for their own or the other bacteria (Gerritsen & Smits, 1997). Both strains were able to develop and reproduce on a combination of the two bacterial strains and both IJ strains cultured in this way were found to have both strains of bacteria in their gut. In an *in vitro* choice study of two H. bacteriophora strains from Italy Ith-CE1 and Ith-C6, which carried two different colour phenotypes of *Photorhabdus*, red and green respectively, IJs of both strains appeared to prefer the red bacterium (Tarasco et al., 2003). A 4th year under graduate project, which I co-supervised, was designed to test the preference of the Bull Island nematodes for their own or the other colour phenotype (Spring, 2009).

This showed that both nematode types PUR1 and YEL3 had a slight preference for the yel3 bacteria in choice experiments carried out an agar plates over 2.5 hour duration. These studies were looking at bacterial preference by IJs on artificial media. In the cross-compatibility experiment the bacteria are consumed by and colonise non-infective juveniles that develop into the next generation of IJs within the insect cadaver. The specificity of the nematode-bacterium complex is mediated by a number of factors including nutrition, surface structure recognition and gene regulation (reviewed in Chaston & Goodrich-Blair (2009) and Lewis & Clarke (2012)). In H. bacteriophora strain TT01 successful colonisation of the nematode by its bacterial symbiont, P. luminescens subsp. laumondii strain TT01 appears to involve the expression of specific cell surface characters such as lipopolysaccharide (LPS) (Easom et al., 2010) and filaments known as fimbriae (Somvanshi et al., 2010) by the bacteria. Other Heterorhabditis/Photorhabdus have not been studied in such detail and in Steinernema carpocapsae/Xenorhabdus nematophila complexes lectin-binding residues inside the specialised receptacle where the symbionts are carried in the nematode are involved (Martens & Goodrich-Blair, 2005). Until more detailed studies have been conducted on other EPN complexes it is uncertain how specific the symbiotic relationship is in each combination. Within the insect host the nematode is essentially a bacteria feeder and is probably not actively choosing which bacteria to eat. In the case of the North Bull Island purl and yel3 isolates both were able to colonise the *H. downesi* nematodes and it is probably the case that the relative proportion of each bacterial type present at a particular stage in the infection is determining the colonisation of the IJs in the *pur1/yel3* 1:1 treatment.

In Hungary both *H. downesi* and *H. megidis* have been found naturally associated with both *P. temperata* subsp. *temperata* and *P. temperata* subsp. *cinerea* at two

sites (Tóth & Lakatos, 2008; Tóth *et al.*, 2008; Tóth & Lakatos, 2009) and *H. bacteriophora* was found naturally associated with both *P. luminescens* subsp. *laumondii* at one site and with *P. luminescens* subsp. *thracensis* at another site (Tóth & Lakatos, 2009). In Turkey *H. bacteriophora* was found naturally associated with both *P. luminescens* subsp. *kayiii* at one site and with *P. luminescens* subsp. *thracensis* at another site (Hazir *et al.*, 2004). This is not unusual as *H. bacteriophora* is found associated with several species and even subspecies of *Photorhabdus* (Adams *et al.*, 2006; Lewis & Clarke, 2012), although there is some evidence that there may be different variants of *H. bacteriophora* as not all strains are reproductively compatible (Stack *et al.*, 2000).

I have already shown in Chapter 2 that the two isolates *purl* and *yel3* did not inhibit each other *in vitro* (Section 2.3.3.1) and so they would not be expected to inhibit each other *in vivo*. The results of the cross-compatibility experiment support this (Section 3.3.1.1). From the NA plates (Figure 3.2) inoculated from the *G. mellonella* homogenate it was clear that colonies of both bacterial phenotypes were present in similar numbers at 72 hours post injection and this was true for both experiment 1 and experiment 2. A number of studies have investigated interactions between different genera (Massey *et al.*, 2004), species (Hawlena *et al.*, 2010a; Bashey *et al.*, 2012a), and isolates (Hawlena *et al.*, 2010b), of entomopathogenic bacteria from a range of nematode hosts. Interactions between species can be mutualistic, selfish, altruistic or spiteful depending on the cost or benefit to the actor and the recipient (Gardner & West, 2004). In bacteria the interaction is to the detriment of one or even both bacterial species if they both produce toxins that can kill the other. These interactions are mediated in part by bacteriocins (xenorhabdicins (Thaler *et al.*, 1995) and lumicins (Sharma *et al.*, 2002) in *Xenorhabdus* spp. and *Photorhabdus*  spp., respectively). In a study of X. bovienii and X. koppenhoeferi (Bashey et al., 2011) found that where one bacterial species produced bacteriocins which inhibited the other this could give the host nematode a competitive advantage in mixed infections in G. mellonella. However, bacteriocins are costly to produce and may not always confer an advantage on the producer (Massey et al., 2004). Variation among bacteria in sensitivity to bacteriocins has been shown to influence the outcome of such interactions both in terms of growth rate of the bacteria and virulence against an insect host. The interactions between closely related bacterial species is termed spiteful when they result in lower virulence and lower within-host growth rates of both bacteria in co-infections (Gardner et al., 2004; Bashey et al., 2012a; Bashey et al., 2012b) or lower growth rates in co-cultures (Inglis et al., 2009; Inglis et al., 2011). In a study of P. asymbiotica, P. luminescens subsp. luminescens and X. nematophilus isolates Massey et al (2004) found that bacterial species which inhibited each other in vitro resulted in lower insect mortality when both were injected into G. mellonella compared to the mortality when each was used singly. The degree of relatedness may influence the bacteriocin-mediated interaction between the bacteria, with less inhibition occurring between genetically similar isolates of the same species compared to isolates of the same species that were genetically different (Vigneux et al., 2008; Hawlena et al., 2010b; Bashey et al., 2012b). Other antimicrobial compounds such as xenocoumacins, water-soluble peptides, are also involved in both interspecific and intraspecific competition (Park et al., 2009). In contrast to Massey et al. (2004), in the cross-compatibility experiment not only did neither subspecies inhibit the other, in both experiments there was a higher level of bacterial colonisation of the insects where both bacterial subspecies were injected than for either single subspecies, suggesting a possible

175

synergy between the two subspecies. It is also possible that differences in the metabolism of each isolate (Section 2.3.3.d) enabled both isolates to exploit the host to a greater extent than either isolate in single infections, contributing to an overall greater growth rate of the bacteria in the mixed infection.

In the cross compatibility experiments both the PUR1 and the YEL3 IJs successfully invaded cadavers which were already colonised by bacteria in all three bacterial treatments, purl alone, yel3 alone, and the purl/yel3 1:1 mixture, as well as the control (no bacteria) insects which were alive at the time of nematode infection. In the case of the PUR1 IJs the invasion rate was higher in all three bacterial treatments compared to the control insects whereas the converse was true for the YEL3 IJs (Figure 3.4). The nematodes were applied to the insects 72 hours after they had been injected with bacteria. In the field it would be usual for entomopathogenic nematodes to encounter dead as well as live hosts. In any event there is only one first invader and subsequent invading nematodes are faced with assessing the status of an already infected host. In studies with S. feltiae, S. affine and H. megidis it was found that species from both genera would invade both live and dead (freeze-killed) G. mellonella larvae (San-Blas & Gowen, 2008; San-Blas et al., 2008), although in the case of *H. megidis* the invasion rate in the dead insects was lower compared to live ones. Han & Ehlers (2001) found that H. bacteriophora strain HO6 infected G. mellonella larvae that had been injected with the homologous (P. luminescens HO6) or a heterologous (P. luminescens HNA, from H. megidis strain HNA) symbiont. Scavenging behaviour could be seen as an advantage for an entomopathogenic nematode as it might allow it to exploit hosts or life stages that would otherwise be unavailable. The chance of survival of the nematode and its symbiotic bacteria would also be increased if the nematode-bacterium complex didn't have to overcome

the host immune system in order to successfully colonise the host. The infection status of the cadaver would also be a factor; in the case of a cadaver already heavily infected with saprotrophic bacteria the symbiotic bacteria might not be able suppress them. This would be a disadvantage for EPN species that are dependent on their symbiont. However, invading cadavers already colonised by the symbiont does not have this disadvantage for the nematode. There is conflicting evidence as to the effect infection status of a host plays in subsequent invasion by conspecific and heterospecific entomopathogenic nematodes. Some studies have reported that hosts previously infected with Steinernema spp. were unattractive to conspecifics at 6-9 hours after infection (Glazer, 1997) or 48 hours post infection (Wang & Ishibashi, 1999). In contrast, Ramos-Rodriguez et al (2007) found that for S. glaseri there was no difference in invasion rate between infected and uninfected hosts. It should be remembered that in a successful entomopathogenic nematode infection there are in fact two colonisations occurring. There was a significant difference in the number of IJs invading between the first and second experiments (Figure 3.6). There was also a difference in the number of IJs invading the different treatments, with significantly more IJs invading the 1:1 pur1/yel3 treatment compared to the yel3 treatment in experiment two (Figure 3.4). This difference in invasion rate may be due to the level of bacterial colonisation as the number of colony forming units of bacteria recovered from G. mellonella was also significantly greater in the first experiment compared to the second (Figure 3.3), and the number of colony forming units was significantly greater in the *pur1/yel3* treatment compared to the *yel3* treatment in experiment two (Figure 3.2). It may be that the greater rate of nematode invasion can be explained in part by increased bacteria activity in the cadaver and that the bacterial feeding nematodes recognise the symbiont-colonised cadaver as a food source.

Bacterial type appeared to influence time of first emergence, the proportion of IJs emerging early on in the emergence period, and the number of IJs emerging. In both experiments the median time to first emergence of infective juveniles was longest in the purl treatment and shortest in the yel3 treatment. Nematode emergence was monitored over a period of 26 days with 24 hr harvests at days 4-5, 11-12 and 18-19 and a final harvest at day 26 (Section 3.2.1.6). Numbers of IJs emerging were assessed in three ways; estimation of total weekly harvest in each treatment (Section 3.3.1.4i), counts from individual White traps from the *pur1* and *yel3* treatments for day 4 and day 19 (section 3.3.1.4ii), and pooled counts of total emergence over a 26 day period from the *pur1* and *yel3* treatments (Section 3.3.1.4iii). There was a trend for first emergence of IJs to take longer from the *purl* treatment compared to the other bacterial treatments (Table 3.5) Individual counts for four cadavers of emergence in the first four days as a proportion of total harvest for days 1-4 and day 19 did not show a significant difference between *pur1* and *yel3* treatments although greater numbers of IJs did emerge from the *purl* treatment in both experiments (Figure 3.9). The estimated emergence in the first week of emergence as a proportion of total emergence was highest in the *pur1* treatment compared to the *pur1/yel3* and the *yel3* treatments (Table 3.6), and there was a greater proportion of IJs in the *pur1* treatment in the first week compared to the other treatments (Table 3.8). The overall emergence data showed that more IJs were recovered over a 26 day period from cadavers in the *pur1* treatment than the *yel3* treatment (Table 3.10). Compared to the bacterial treatments, emergence was highest in the no bacteria treatment in week 1 and declined over the three week period whereas for all other treatments except *pur1* in experiment 2 IJ emergence increased in week 2. The mean number of IJs emerging in experiment 1 was greater than in experiment 2 (Table 3.10). Studies

178

with Heterorhabditis megidis have shown that invasion rate, emergence time and reproduction rate are related to number of IJs applied (Boff et al., 2000; Ryder & Griffin, 2002). In both of these studies increasing densities of IJs were applied to live G. mellonella larvae. In the cross-compatibility experiment the larvae in the three bacterial treatments were already dead and colonised with either the homologous or the heterologous bacteria. The infection rate and the reproduction rate appear to be influenced by the bacterial load already established in the cadavers, as increasing numbers of IJs invaded with increasing bacterial load (Figure 3.6) and the overall yield was greater in experiment 1 than in experiment 2 (Table 3.10), suggesting that the readily available supply of nutrient in the cadavers was a factor. The bacteria also appear to have a role in the timing of emergence and this may indicate a difference in the quality of nutrients available. As was seen in Section 2.3.3.d there are differences in the metabolism of the two bacterial isolates, *pur1* and *yel3* and these differences may influence the timing of IJ emergence as bacterial type was found to be a highly significant factor influencing the proportion of IJs emerging early (Table 3.7). It has been shown that heterologous bacteria can support the growth and development of the non-host nematode in a number of *Heterorhabditis* spp. both in vitro and in vivo (Han et al., 1990; Gerritsen et al., 1998; Han & Ehlers, 1999). In these studies axenic IJs were grown on heterologous bacteria in vitro and subsequently those combinations that were successful were tested in G. mellonella for the ability of the bacterium to support growth and reproduction of the nematode.

The novel combinations obtained in the cross-compatibility experiment were assessed to determine the bacterium carried by the emerging IJs. In the case of the single bacterial treatments *pur1* and *yel3* all IJs tested using 1 on 1 bioassays were found to be carrying the bacteria which had been injected into the *G. mellonella* 

larva 3 days previously, even when this was different from the bacteria carried by the invading IJ. It is possible that if a larger number of IJs from each treatment were tested that some might have been found to be carrying the original bacteria. The level of bacterial colonisation of the injected G. mellonella was found to exceed 1 x  $10^6$  colony forming units in all cases and was nearly 1 x  $10^8$  colony forming units in the case of the *purl/yel3* treatment in experiment 1 (Figure 3.2). While it is likely that the homologous bacteria carried by the invading nematodes would grow, as *pur1* and yel3 were found not to inhibit each other's growth (see Chapter 2, Table 2.20) it seems that the already established bacteria would dominate and be more likely to be taken up by the developing nematodes. In the *pur1/yel3* treatment at both assessment times (days 1-4 and day 19 of emergence) more IJs carried the yel3 than pur1. There was no difference in the number of bacteria carried by the IJs in either the homologous or heterologous combinations (Figure 3.13B) and in all cases some IJs carried no bacteria, with no difference between the homologous or heterologous combinations in the proportion of IJs with no bacteria (Appendix Table A 14). There appeared to be a shift in the bacteria taken up by the nematodes in the cadaver as in all five cadavers which produced both *pur* and *yel* carrying IJs there was a switch from *pur* or *pur/yel* at day 4 to exclusively *yel* at day 19. It may be the case that in the dual bacterial colonisation the *pur* and *yel* colour phenotypes differ in their nutritional value and ability to successfully infect and colonise the maternal nematode. In *Heterorhabditis* spp. pre-IJs develop and hatch inside the maternal body, rupturing the uterus and feeding on the maternal body in a process known as endotokia matricida (Johnigk & Ehlers, 1999). In a study on H. bacteriophora Ciche et al. (2008) showed that the symbiotic bacteria first infected the maternal nematode by adhering to the maternal intestine and then invaded the maternal body cavity where they multiplied and then infected the pre-IJs developing inside the maternal body cavity. It is not clear if a single female can be infected by more than one bacterial type so in the IJs emerging from cadavers in the *pur/yel* bacterial treatment they may be the offspring of different females which were carrying the *pur* or *yel* subspecies or from females carrying both subspecies. However, if the latter were true I would have expected to detect both bacterial types in the IJs. It is possible that if a larger number of IJs were tested some might be found to carry both symbionts. The cross-compatibility experiment tested only one of nine possible pairings of PUR and YEL nematodes and their bacteria and it would be interesting to test other combinations in the same way to see if similar trends were observed.

The dynamics of the bacterial growth within the cadaver in terms of growth rate and metabolic capabilities may be having an effect on the availability of bacteria at different stages in the infection. A slower growth rate in the *pur1* isolate might explain why the time to first emergence of IJs in the *pur1* treatment was longer. Although there was no significant difference in growth rate between *pur1* and *yel3* in liquid cultures (Section 2.3.3b) the *pur1* isolate seemed to lag slightly in the initial stages of growth (see growth curves Appendix Figure A. 5). The differences in relative growth rate on different agars (Section 2.3.3c), and the metabolic capabilities (Section 2.3.3.d) of the two isolates might indicate different abilities to utilise resources within the cadaver and may allow that the *yel3* to outcompete the *pur1* isolate later in the infection.

The speed of kill and pathogenicity of the wild type and hybrid nematode/bacteria strains against *G. mellonella* was compared at three time points in order to assess the role of both the nematode and bacterial partner (Section 3.3.3). The greatest difference was seen at 40 hours. While the difference in the median number of

insects dead was not significant when comparing the four nematode/bacteria combinations, there was a significant difference depending on whether the nematodes carried the *pur1* or the *yel3* bacteria when data for the wild type and hybrid combinations were pooled (Table 3.18), with more dead in the purl treatment. This is similar to the result of tests using *Tipula oleracea* where Gerritsen (1998) found that the heterologous combinations of H. megidis/P. luminescens strains NLH-E87.3 and NLH-jun had a higher mortality rate than the homologous combinations. As discussed earlier, the *pur1* isolate appeared to grow slightly slower in the earlier stages of growth in nutrient broth compared to MacConkey broth, perhaps indicating differing growth requirements for the two isolates, however by 40 hours both isolates had reached stationary phase when grown in broth. The growth of the bacteria in an insect is likely to be different from the growth in liquid culture, as growth in any medium depends on type and level of availability of nutrients, culturing conditions and competition from other microorganisms. The initial inoculum of bacteria depends on the number of nematodes invading and the number of bacteria each nematode carries in to the insect. While these two factors were not assessed in the cross-compatibility experiment they would not be expected to differ based on the results from the bacterial quantification from individual IJs (Section 3.3.5). The ability of *Photorhabdus* spp. to kill the insect host depends on the production of a range of proteases and toxins by the bacteria (Bowen et al., 2000). Work by Crawford et al. (2010) suggested that a number of molecules linked to both virulence and antibiosis are upregulated upon recognition of the host by P. luminescens TTO1. The difference in mortality might be due to differences in growth rate and toxins produced by the two bacterial types. In antibiosis tests it was found that all three *pur* isolates had a greater inhibitory effect against a range of bacteria and fungi than did any of the three *yel* isolates (Section 2.3.3.1) and it may be the case that the *purl* isolate also produces a greater or more virulent range of compounds which are responsible for the greater rate of kill seen at the 40 hour time point.

The survival of IJs in water was found to vary both within and between bacterial subspecies (Section 3.3.4). The greatest rate of decline was seen between week 4 and week 10 for the PUR phenotype and between week 4 and week 8 for the YEL phenotype (Figure 3.11). While there were differences in percentage survival of the isolates at the various time points neither colour phenotype or isolate had a significant effect on the overall rate of survival from week 2 to week 8 of the experiment. The variation in survival rate was similar to that reported by Hass *et al.* (2001) for a number of *H. downesi* isolates from Ireland. Carrying the symbiotic bacteria appears to exert a cost on the nematode in terms of survival. Studies with *S. carpocapsae* found that IJs carrying their symbiotic bacteria had lower survival rates than IJs which had no symbionts (Mitani *et al.*, 2004; Emelianoff *et al.*, 2007). Similar studies do not appear to have been carried out with *Heterorhabditis* species. However, if there is a cost to *H. downesi* in carrying the symbiont it is the same for both subspecies of *P. temperata*.

The bacterial quantification (Section 3.3.5) showed that a small proportion of IJs carried no bacteria and this is consistent with the reported colonisation rate of over 90% of IJs (Lewis & Clarke, 2012). Overall there was no difference in the number of bacteria carried when PUR and YEL isolates were compared and in a subsequent experiment there was no difference in the number of bacteria carried by hybrid when compared to the wild type. These results are consistent with other studies, for example, Lewis *et al.* (1995) found that *H. bacteriophora* retained approximately

*P. luminescens* colony forming units during the first week of storage, with the number of bacteria retained by the IJs decreasing over time and the authors suggested that the bacteria may be utilised as a food source by the nematode. Studies with *H. bacteriophora* GPS11 and its symbiotic bacteria *P. temperata* showed that the number of bacterial cells in the nematode fell by approximately 50% in 30-day old IJs compared to freshly emerged IJs (An & Grewal, 2010a).

### Chapter 4 Infectivity and development of *Heterorhabditis downesi* associated with two subspecies of *Photorhabdus* against a range of hosts

### 4.1 Introduction

Heterorhabditis downesi was found on North Bull Island to be associated with P. temperata subsp. temperata (P.t temperata) and P. temperata subsp. cinerea (P.t cinerea) (yel and pur, respectively). The natural host of H. downesi on North Bull Island are unknown but since the two colour phenotypes PUR and YEL are spatially distributed within the dune system it is possible that the two nematode-bacteria complexes are adapted to different hosts. Because the natural host of a particular EPN species is usually unknown laboratory-based studies are used to determine the efficacy of a particular nematode species against a range of hosts and to look at differences in virulence between strains of the same species. Different strains of the same species from different geographic locations may become adapted to particular hosts or indeed to different abiotic conditions (Mráček et al., 1998). Both Tenebrio molitor and Galleria mellonella larvae are widely used in the laboratory to maintain cultures of nematodes. These insects are also widely used in bioassays to assess the ability of EPN species and strains to locate, infect and reproduce in insect hosts (Glazer & Lewis, 2000). Bioassays are also useful in determining the number of nematodes required to kill a particular host. It can also be useful to compare the infectivity of a particular EPN species against a range of life stages of a particular host or against hosts from different orders to gain a better understanding of the fundamental ecology of a particular species and to predict the suitability of a particular EPN species or strain for use against a range of insect pests in the field.

In Ireland H. downesi K122, an indigenous Irish strain, has been shown to kill pine weevil, H. abietis (Dillon et al., 2006; 2007), and black vine weevil larvae, Otiorhynchus sulcatus (Lola-Luz et al., 2005) both in the laboratory and in field trials and this strain has been commercialised. Larvae of the garden chafer, Phyllopertha horticola, infected by H. downesi have been recovered on dunes (C Griffin, pers communication). A Hungarian strain, H. downesi strain 267 was found to be effective against the European cockchafer, Melolontha melolontha in the laboratory (Lakatos & Tóth, 2006). In chapter 3 the wild type (Pp, Yy) and hybrid (Py, Yp) combinations of *H. downesi*, produced in the cross-compatibility experiment, were tested against G. mellonella larvae. Some differences were found in the time to death for larvae infected with PUR1 or YEL3 nematodes, irrespective of the bacteria they carried, purl or yel3 and there were also differences in time to death between insects infected with purl or yel3 bacteria, irrespective of the nematode type (Section 3.3.3), indicating that both nematode and bacteria are involved in the mortality in this particular insect. However, G. mellonella is a very susceptible insect and is easily killed by many EPN species. Moreover, it is not an insect that the soil-dwelling EPN are likely to encounter as it is a pest of beehives and thus is of limited use in predicting the efficacy of a particular species or strain of EPN in the field. However, G. mellonella is of use in bioassays designed to look for differences between particular species or strain of EPN. As mentioned above, H. downesi K122 is effective against H. abietis. The symbiotic partner of K122 is P. temperata subsp. temperata strain K122. In this chapter the role of the bacteria in

infection will be investigated in more detail by challenging insects from three insect orders Lepidoptera (G. mellonella) Coleoptera (T. molitor, H. Abietis) Diptera (*Coelopa* spp). Both G. mellonella larvae and T. molitor larvae are commonly used in bioassays, as mentioned above. The large pine weevil, H. abietis, is an important pest of forestry and *H. abietis* larvae have been shown to be susceptible to the *H.* downesi K122 nematode strain which carries P. temperata subsp. temperata as its symbiont and is used as a proxy for other beetles (Phyllopertha spp. and Otiorhynchus spp.) that occur in sand dunes but are less readily available in the numbers required for bioassays. Two species of kelp fly, Coelopa species, Coelopa frigida and C. pilipes are commonly found in beds of seaweed (mainly Laminaria spp. and Fucus spp.) on the foreshore (Dobson, 1974) and the larvae feed mainly on the bacteria that are abundantly found in these clumps (Edward et al., 2008). Coelopa frigida has been recorded on North Bull Island (Healy, 1975) and it is likely that C. pilipes occurs there also, since both species commonly occur together (Dobson, 1974), and might form part of the natural host range of *H. downesi* on North Bull Island. Since these dipterans occur in the strand line the PUR isolates might be expected to have a greater efficacy against these insects, given that the PUR isolates were found towards the front of the dune system.

The objective of this chapter is to investigate differences between *P.t. temperata* and *P.t. cinerea* in adaptation to different host species either in pathogenicity or in their ability to support *H. downesi* reproduction as evidenced by IJ production.

Specifically:

i) To test the wild type and hybrid nematode/bacterium combinations obtained in Chapter 3 against *G. mellonella* and *H. abietis*. ii) To test the three PUR and three YEL nematode/bacterium isolates (Chapter 2) against *G. mellonella*, *H. abietis*, *T. molitor* and *Coelopa* spp.

### 4.2 Materials and Methods

# 4.2.1 Tests of wild type and hybrid combinations against *G. mellonella* (infectivity and infective juvenile yields)

A continuous exposure experiment was set up with the wild type and hybrid nematode/bacteria combinations established in Section 3.2.3.1, Table 3.2, consisting of a single nematode applied per insect using final instar *G. mellonella* larvae. The experiment included four blocks consisting of one replicate 24-well plate for each combination Pp, Py, Yy, Yp, plus one control plate per block (= 96 insects total per combination). The 24-well plates were prepared and stored as described in Section 3.2.2.2. Plates were checked after 48 hours and every eight hours thereafter up to 120 hours (5 days). Plates were checked daily from day 5 up to day 14. Insect mortality due to *P. temperata* infection (see Section 2.2.1.3) was recorded. Five cadavers per replicate were randomly chosen to assess emergence (= 20 cadavers total per combination). Fourteen days after infection a single White trap was set up for each replicate (= 5 cadavers per White trap). White traps were stored at 20°C and each White trap was harvested twice at 14 day intervals following first emergence. All IJs were stored at 9°C prior to counting as described in Section 3.2.1.6. Both harvests from each White trap were pooled for counting.

# 4.2.2 Tests of wild type and hybrid combinations against *Hylobius abietis* (infectivity and infective juvenile yields)

Pine weevil larvae were collected in the field from pine tree stumps and larvae were placed individually in the wells of 24-well plates. To prevent pupation and desiccation the lid of each plate was lined with five layers of moist tissue paper and plates were stored at 9°C. The tissue paper was replaced weekly and larvae were used in experiments within four weeks of collection. All larvae were weighed prior to use and larvae in the range 110 mg – 173 mg were used in all experiments.

A single *H. abietis* larva was added to a well of 24-well plates lined with three filter papers. IJs were added to each well in a total volume of 50  $\mu$ l tap water at a range of application rates. Plates were randomly stacked in blocks consisting of one replicate of each application rate for each wild type or hybrid used, and one control plate, with three filter papers, 50  $\mu$ l of tap water and one *H. abietis* larva per well. One empty plate with filter papers and 50  $\mu$ l tap water only was placed on the top and bottom of each stack as a buffer. Plates were stored as described in Section 3.2.2.2 and checked daily and insect mortality recorded.

The experiment was run four times. In experiment 1 nematode treatments were Pp and Py, applied at a rate of 5, 10, 20 or 50 IJs per insect with 10 insects at each application rate. In experiment 2 nematode treatments were Pp and Py, applied at a rate of 5, 10, 20 or 50 IJs per insect with 7 insects at each application rate. In experiment 3 nematode treatments were Pp and Py, applied at a rate of 10, 20, 50 65 or 80 IJs with 10 insects at each application rate. In experiments were Yy and Yp, applied at a rate of 10, 20, 50 65 or 80 IJs with 10 insects at each application rate. In experiment 4 nematode treatments were Yy and Yp, applied at a rate of 10, 20, 50 65 or 80 IJs with 12 insects at each application rate and Pp and Py, applied at a rate of 50 IJs per insect with 12 insects per nematode treatment. All cadavers exhibiting evidence of death due to *P. temperata* infection (see Section 2.2.1.3) in the 50 IJ treatments for all nematode/bacterium combinations in experiments 2 - 4 were placed on individual White traps fourteen days after infection and each White trap was harvested

189

seventeen days after first emergence. All IJs were stored at 9°C prior to counting as described in Section 3.2.1.6.

### 4.2.3 Tests of PUR and YEL isolates against *G. mellonella* (infectivity and infective juvenile yields)

Infections were set up at an application rate of ten IJs per insect with six *H. downesi* nematode isolates from North Bull Island (PUR1, PUR2, PUR3, YEL1, YEL2, YEL3). Infections were set up as described in Section 3.2.2.2 with the following variations: three 1.5 cm filter papers were placed in each well; nematode suspensions were adjusted to 10 IJs/50  $\mu$ l and 10 IJs were added to each of four wells at a time using a multichannel pipette. Sample counts were regularly carried out to check the consistency of IJ delivery. Plates were randomly stacked in blocks consisting of one plate of each isolate and one control plate with three filter papers, 50  $\mu$ l of tap water and one *G. mellonella* larva per well. One plate with three filter papers and 50  $\mu$ l tap water only was placed on the top and bottom of each stack as a buffer. Each stack was wrapped in tinfoil to ensure constant light conditions and stored at 20°C.

Experiment 1 consisted of four blocks of one replicate 24-well plate for each isolate (= 96 insects total per isolate). Plates were checked after 48 hours and insect mortality was recorded. Tap water (50  $\mu$ l) was added to the wells as required to avoid the filter paper drying out. Insect mortality was recorded daily up to day 8, at which point the experiment was terminated. On days 2, 3 and 4 post infection up to three cadavers exhibiting signs of mortality due to EPN infection (Section 2.2.1.3) were chosen at random from each plate, stored for three days at 20°C and then frozen for dissection at a later date to assess the number of IJs invading.

In experiment 2 the procedure above was repeated with two replicate plates per isolate (= 48 insects total per isolate) and mortality was recorded after 29 hours and then at 12 hour (+/- 3 hrs) intervals up to 140 hours (day 6) post infection and daily up to 8 days post infection. At each time point on days 2, 3 and 4 of the experiment up to three cadavers exhibiting signs of mortality due to EPN infection were chosen at random from each plate and stored for dissection as described above. Fourteen days after infection all remaining cadavers exhibiting signs of mortality due to EPN infection were placed on individual White traps and stored at 20°C. All White traps were harvested (see Section 3.2.1.6) 17 days following first emergence of IJs and IJs were stored at 9°C prior to counting as described in Section 3.2.1.6. All IJs from each replication were pooled and counted.

## 4.2.4 Tests of PUR and YEL isolates against *H. abietis* (infectivity and infective juvenile yields)

Infections were set up at an application rate of 50 IJs per insect with six *H. downesi* nematode isolates from North Bull Island (PUR1, PUR2, PUR3, YEL1, YEL2, YEL3). Multiwell plates were prepared as described in Section 3.2.2.2, with the following variations: three filter papers were placed in each well; nematode suspensions were adjusted to 50 IJs/50  $\mu$ l and 50 IJs were added to each of four wells at a time using a multichannel pipette; a single *H. abietis* larva was added to each well.

There were two experiments with sixteen and twenty four insects per isolate, respectively. Plates were stored as described in Section 4.2.3 with one control plate with three filter papers, 50  $\mu$ l of tap water and one *H. abietis* larva per well. Plates were checked after 48 hours and insect mortality was recorded. Tap water (50  $\mu$ l) was added to the wells as required to avoid the filter paper drying out. Insect

mortality was recorded daily for fourteen days at which point the experiment was terminated. Fourteen days after infection cadavers exhibiting evidence of death due to *P. temperata* infection (Section 2.2.1.3) were transferred to individual White traps and stored at 20°C. White traps were harvested at day 17 following first emergence of IJs. All IJs were stored at 9°C prior to counting as described in Section 3.2.1.6. In each experiment IJs from all White traps per isolate were pooled before counting.

#### 4.2.5 Tests of PUR and YEL isolates against *T. molitor* (infectivity)

Infections were set up at an application rate of ten IJs per insect with six *H. downesi* nematode isolates from North Bull Island (PUR1, PUR2, PUR3, YEL1, YEL2, YEL3) as described in Section 3.2.2.2 with the following variations: 0.2 g moist coir (1:2.5 w/w coir:tap water) was added to each well and tamped down so that the well was half filled with coir; nematode suspensions were adjusted to 10 IJs/50  $\mu$ l and added as described in Section 4.2.3. The experiment consisted of two blocks of one replicate 24-well plate for each isolate (= 48 insects total per isolate). Plates were stored as described in Section 4.2.3 with the following variations: control plate had moist coir, 50  $\mu$ l of tap water and one *T. molitor* larva per well; buffer plates had moist coir, 50  $\mu$ l of tap water. Plates were checked after 27 hours and then every 12 hours (+/- 3hrs) up to day 6 and then every 24 hours up to day 8 and insect mortality was recorded. Tap water (50  $\mu$ l) was added to the wells as required to avoid the coir drying out. The experiment was terminated on day 8.

## 4.2.6 Tests of PUR and YEL isolates against Kelp fly (Infectivity and infective juvenile yields)

Kelp fly larvae (*Coelopa* sp.) were collected in the field from drifts of seaweed deposited by spring tides (wrack-beds) on Mullaghmore Strand, Co. Donegal. The larvae were stored in the dark in seaweed in 24 l plastic storage boxes with close

fitting lids (Tesco plc) at 9°C and used within one week of collection. A 48 hour infectivity test based on Peters (2005) was set up with six *H. downesi* nematode isolates from North Bull Island (PUR1, PUR2, PUR3, YEL1, YEL2, YEL3) as follows: one 9 cm filter paper was placed in the base of a 9 cm Petri dish, 0.5 ml of nematode suspension was pipetted evenly over the filter paper and ten kelp fly larvae were added to the dish. The larvae were covered with 12 g of air dried sterile sand (B & Q plc) mixed with 8% w/w <sup>1</sup>/<sub>4</sub> strength sterile Ringer's solution (Figure 4.). A further 0.5 ml nematode suspension was evenly pipetted over the sand and the lid was replaced.

There were two experiments, with a dose of 1000 IJs/dish and 5000 IJs/dish, respectively. Each experiment consisted of five blocks. Dishes were randomly stacked in blocks consisting of one dish of each isolate and one control consisting of a 9 cm Petri dish with filter paper, moist sand, 10 kelp fly larvae and 2 x 0.5 ml aliquots of tap water added as described above. Two 9 cm Petri dishes with filter paper, moist sand and 2 x 0.5 ml aliquots of tap water were placed on the top and bottom of each stack and the dishes were securely taped together. All five blocks were placed in a plastic food container surrounded by eight stacks of empty dishes. The food containers were placed in closed polystyrene containers to maintain constant light and ventilation conditions and stored at 20°C. After 48 hours all insects were removed from the sand, rinsed in sterile 1/4 strength Ringer's solution and transferred to 5 cm Petri dishes lined with a single filter paper moistened with 400 µl sterile <sup>1</sup>/<sub>4</sub> strength Ringer's solution. Insect mortality was recorded and insects were stored at 20°C. In experiment 1 (1000 IJs/dish) one block was chosen at random and set aside for dissection. Ten days after infection all remaining cadavers in both experiments exhibiting evidence of death due to P. temperata infection (Section 2.2.1.3) were transferred to individual White traps and stored at 20°C. Emerging IJs were allowed to accumulate in the White traps and a single harvest was carried out seventeen days following first emergence of infective juveniles. All IJs were stored at 9°C prior to counting as described in Section 3.2.1.6.



**Figure 4.1:** Kelp fly infections were set up in 9 cm Petri dishes, with 10 larvae per dish. Nematode concentration was 100 or 500 IJs/larva.

Numbers of IJs emerging were dealt with as follows:

- i) Individual counts of emergence per cadaver were carried out for a single replication in each experiment.
- ii) Counts of pooled total emergence for all remaining replications.

### 4.2.7 Statistical analysis

Routine statistical analysis, including tests for normality, was carried out using Minitab statistical software, version 16.1.1 (Minitab Inc., 2010). Normality of data

was tested using the Anderson-darling method. Analysis of variance was carried out using a one-way ANOVA or when groups had an uneven number of observations the general linear model with an unbalanced nested design with isolate nested within colour phenotype was used to compare isolate and colour phenotype. Analysis of variance was followed by pair-wise multiple comparisons using Tukey's test. In experiments where nematode type and bacterial type were factors wild type and hybrid results were analysed using a two-way ANOVA. Where no significant interactions occurred treatment means for each factor were compared using a oneway ANOVA and post-hoc Tukey's test for three or more treatments or 2-sample Ttest for two treatments. When data could not be normalised treatment medians were compared using a Kruskal-Wallis test, followed by multiple pairwise comparisons using a Mann-Whitney U-test with Bonferroni 95% confidence intervals, obtained by calculating  $(1 - \alpha / g)$  where  $\alpha$  is the level of significance and g is the number of pairwise comparisons carried out. Lethal concentrations were calculated using Probit analysis. In all tests a P value of < 0.05 was taken as significant unless indicated otherwise.

### 4.3 **Results**

#### 4.3.1 Tests of wild type and hybrid combinations against G. mellonella

### 4.3.1.1 Infectivity

In a single nematode continuous contact experiment using *G. mellonella* larvae the mortality was recorded at 8 hour intervals from 48 hours post infection to 120 hours post infection and at 24 hour intervals thereafter up to day 8 post infection. The data for the 48 hour, 56 hour and 80 hour time points were not normally distributed and could not be normalised. The mean or median percentage of insects dead at any time

point did not differ between the four strains tested (P > 0.05, Table 4.). Mortality increased from negligible levels at 48hrs to about 30% at 96hrs. Overall, the mean percentage mortality by day 8 of the experiment was approximately 35%. At each time point there was a higher percentage of insects dead in the *pur* treatments (Pp and Yp) than in the *yel* treatments (Yy and Py). This is dealt with below.

**Table 4.1:** Median<sup>†</sup> or mean<sup>‡</sup> ( $\pm$  SE) percentage of *G. mellonella* dead at eight hour intervals for wild type and hybrid combinations of nematode and bacteria. All insects were infected with a single IJ in a continuous contact exposure. n = 96 per treatment. Results of Kruskal-Wallis test or one-way ANOVA comparison between treatment medians or means are shown for proportion dead at each time point.

	Time point (hrs)						
Nematode/ bacteria combination	48 <sup>†</sup>	56 <sup>†</sup>	61 <sup>‡</sup>	72 <sup>‡</sup>	80 <sup>†</sup>	88 <sup>‡</sup>	96 <sup>‡</sup>
Pp	2.1	12.5	12.5 (2.95)	18.8 (2.69)	20.8	25.0 (2.95)	29.2 (4.50)
Py	0.0	4.2	8.3 (2.41)	12.5 (2.95)	18.8	22.9 (3.61)	26.0 (4.60)
Yp	0.0	12.5	17.7 (5.48)	25.0 (7.01)	27.1	30.2 (7.49)	31.3 (7.70)
Yy	2.1	8.3	13.5 (2.62)	19.8 (3.13)	27.1	26.0 (1.99)	28.1 (2.62)
P value	0.730	0.156	0.367	0.289	0.692	0.715	0.912
	H (3 df)	H (3 df)	F <sub>3,12</sub>	F <sub>3,12</sub>	H ( 3 df)	F <sub>3,12</sub>	F <sub>3,12</sub>
Test statistic	1.3	5.23	1.15	1.41	1.46	0.46	0.17

In order to explore the effect of the bacterial type more fully, the data were pooled by bacterial type and plotted. The data at each time point except 48 hours were normal. The data for the 48 hour time point were excluded and each of the remaining time points were analysed using a two-way ANOVA with nematode type and bacteria type as factors. The analysis showed there was no effect of nematode type or bacteria type at any time point at the 5% level, and no interaction between nematode type and bacteria type at any time point. However, bacteria type was significant at the 10% level at 56 hours post infection with more insects dead in the *pur* treatment (two-way ANOVA,  $F_{1,12}$  =3.87, P = 0.073) (Appendix Table A 15). There was no difference in the speed of kill between the *pur* and the *yel* bacterial treatments over the period from 48 hours post infection to 96 hours post infection. (Two-sample T-test on pairwise slopes, df = 7, T = -0.68, P = 0.516) (Figure 4.2).



**Figure 4.2:** Mean percentage  $(\pm SE)$  of *G. mellonella* dead for the period 48 hours to 96 hours post infection in a single nematode continuous exposure experiment. Data were pooled by bacterial treatment.

#### 4.3.1.2 Numbers of IJs emerging (G. mellonella)

A two-way ANOVA on number of IJs emerging with nematode type and bacterial type as factors showed that neither nematode type nor bacterial type had a significant effect on the total number of IJs emerging over the 28 day period (P > 0.05) (Table 4.2) and no further analysis was carried out on the emergence data..

**Table 4.2:** Mean number of IJs ( $\pm$ SE) emerging from *G. mellonella* over 28 days following an infection with a single wild type or hybrid IJ. Data shown is mean number of IJs emerging from a pooled harvest from five cadavers for each treatment, n = 4. Results of GLM analysis of data pooled by nematode type and by bacterial type are also given.

Nematode/bacteria combination	mean number of	TJs in 4 pooled harvests	
Рр	135767 (36	58)	
Ру	120780 (56	95)	
Үу	116580 (50	116580 (5026)	
Yp	108317 (134	35)	
Factor	Test statistic	Р	
Nematode type	$F_{1,12} = 0.10$	0.761	
Bacterial type	$F_{1,12} = 1.16$	0.302	
Interaction	$F_{1,12} = 2.15$	0.168	

### 4.3.2 Virulence of wild type and hybrid combinations against H. abietis

In both experiment 1 and 2 of the continuous exposure test a concentration of 5 IJs per insect showed little or no kill (Table 4.3), and this concentration was omitted in experiments 3 and 4. While there was some evidence of a dose response this was not consistent. The LD<sub>50</sub> was calculated for all four strains (using data from experiments 1-3 for Pp, Py, and from experiment 4 for Yy, Yp) (Table 4.4). There was a highly significant difference in the LD<sub>50</sub> values between the Yy and Yp strains (experiment 4) (Probit analysis, Z = -3.18, P = 0.001). There was no difference in the slopes between strains in either comparison (experiments 1-3,  $\chi^2 = 3.290$ , df = 1, P = 0.070; experiment 4,  $\chi^2 = 1.261$ , df = 1, P = 0.261).

		Experiment			
		1	2	3	4
			n		
nematode/bacteria combination	IJ concentration	10	7	10	12
	5	0	0		
	10	10	14.29	10	
Dn	20	20	14.29	30	
гр	50	50	57.14	50	83.33
	65			80	
	80			80	
	5	0	14.29		
	10	10	14.29	40	
D	20	20	14.29	30	
РУ	50	40	71.43	50	66.67
	65			80	
	80			40	
	10				16.67
	20				41.67
Yy	50				83.33
	65				75.00
	80				83.33
	10				33.33
	20				75.00
Yp	50				100
	65				100
	80				100

**Table 4.3** Percentage mortality of *H. abietis* larvae exposed to a range of concentrations of wild type and hybrid nematode/bacteria combinations in four experiments.

**Table 4.4:** Results of Probit analysis of mortality of *H. abietis* larvae exposed to a range of concentrations of wild type and hybrid nematode/bacteria combinations in four experiments. Day 8  $LD_{50}$  values and lower and upper Fiducial confidence interval (Fc) values for the number of IJs required for wild type and hybrid nematode/bacteria combinations applied to *H. abietis* larvae are shown. Strains which do not share the same letter in each experiment(s) were significantly different.

Nematode/bacteria combination	$LD_{50} (\pm SE)$	Lower Fc – Upper Fc
Pp (experiments 1-3)	57.13 (10.50)	40.92 - 88.67
Py (experiments 1-3)	60.24 (11.01)	43.23 - 93.26
Yy (experiment 4 <sup>†</sup> )	30.47 (5.45)a	19.07 - 41.39
Yp (experiment 4)	11.73 (2.67)b	6.50 - 17.42

<sup>†</sup> The Pp and Py data were not included in the analysis as there was only a single dose rate for each of these strains in experiment 4.

## 4.3.2.1 Numbers of IJs emerging (*H. abietis*) (fifty IJs per insect application rate)

In order to investigate the effect of nematode type and bacterial type on the number of IJs emerging two analyses were carried out. Firstly, the data from experiment 4 for all four strains (Pp, Py, Yy, Yp) were analysed using a GLM with nematode type and bacterial type as factors (Table 4.5). The analysis showed that neither nematode type nor bacterial type had a significant effect on the number of IJs emerging over the 17 day period, however there was a highly significant interaction effect, with the highest number of IJs per cadaver emerging from insects in the Yy treatment followed by the Pp treatment (Table 4.5) Secondly, the data from experiments 2 - 4for strains Pp and Py were combined and analysed using a GLM with bacterial type as factor (Table 4.6). This analysis also showed that bacterial type did not have a significant effect on the total number of IJs emerging over the 17 day period, but that experiment had a highly significant effect (Table 4.6). Overall the mean emergence per cadaver was 53551 IJs following infection with wild type or hybrid nematodes at dose rate of 50 IJs per insect (See Appendix Table A 16 for emergence data).

**Table 4.5:** Mean number of IJs ( $\pm$  SE) emerging from *H. abietis* over 17 days following infection with wild type or hybrid nematodes at an application rate of 50 IJs per insect. Data shown is for experiment 4. n = 5 for Pp and n = 6 for remaining treatments. Results of GLM analysis are also given.

Nematode/bacteria combination	Mean number of IJs pe	er cadaver
Рр	68510 <sub>(7362)</sub>	
Ру	58709 (10670)	
Үу	89086 (9432)	
Yp	48488 (7146)	
Factor	Test statistic	Р
Nematode type	$F_{1,19} = 0.80$	0.382
Bacterial type	$F_{1,19} = 1.63$	0.218
Interaction	$F_{1,19} = 9.04$	0.007

**Table 4.6:** Mean number of IJs ( $\pm$  SE) emerging from *H. abietis* over 17 days following infection with wild type or hybrid nematodes at an application rate of 50 IJs per insect. Data shown is combined emergence from experiments 2-4 from insects infected with wild type or hybrid nematodes at a dose of 50 IJs per insect. n = 5 for Pp and n = 6 for remaining treatments. Results of GLM analysis are also given.

Experiment	nematode/bacteria combination	Mean number of IJs per cadaver	n
2	Рр	36369 (9131)	3
	Ру	48316 (3947	4
3	Рр	23090 (1907)	4
	Ру	29437 (4901)	6
4	Рр	68510 <sub>(7362)</sub>	5
	Ру	58709 (10670)	6
Factor		Test statistic	Р
Experiment		$F_{2,23} = 7.76$	0.003
Bacterial type		$F_{1,23} = 0.01$	9.350

### 4.3.3 Test of PUR and YEL isolates against G. mellonella

#### 4.3.3.1 Overall mortality and time to death

In the assay with an application rate of ten IJs per insect with *G. mellonella* more insects died on day 3 of the experiment than on any other day in both experiment 1 and 2 (Appendix Table A 17). The proportion of insects dead due to EPN infection was tested at each time point for each experiment separately. The data were not normally distributed and were arcsine square root transformed for analysis. The data from each experiment were first analysed separately.

In experiment 1 there was a significant difference in insect mortality due to colour phenotypes at day 2, with a higher proportion of insects dead in the PUR treatment, (nested ANOVA with isolate nested within colour phenotype,  $F_{1,4} = 15.44$ , P = 0.017) and at days 5 – 8 the difference between colour phenotypes was significant
at the 10% level (P < 0.1). In experiment 2 there was no difference in mortality due to colour phenotype at any time point (Appendix Figure A. 8).

A nested ANOVA of the combined data with colour phenotype nested within experiment and isolate nested within colour phenotype showed that there was a difference in the insect mortality due to colour phenotype at the 10% level at day 2,  $(F_{2,8} = 3.588, P = 0.077)$ , however, there was an interaction between colour and experiment at each time point day 3–day 8 (Figure 4.4).



**Figure 4.3:** Mean proportion ( $\pm$  SE) of *G. mellonella* killed by PUR and YEL IJs over eight days at an application rate of ten IJs per insect. Data shown is from experiment 1 with three isolates of each colour phenotype in each trial. n per colour phenotype = 288. The proportion of insects killed by the PUR colour phenotype was significantly higher at day 2 post infection (nested ANOVA with isolate nested within phenotype and experiment nested within isolate, P = 0.017. Data was arcsine square root transformed for analysis)



**Figure 4.4:** Mean proportion ( $\pm$  SE) of *G. mellonella* killed by PUR and YEL IJs over eight days at an application rate of ten IJs per insect. Data shown is mean of two experiments with three isolates of each colour phenotype in each trial. Total n per colour phenotype = 432. The proportion of insects killed by the PUR colour phenotype was significantly higher at the 10% level at day 2 post infection (nested ANOVA with isolate nested within phenotype and experiment nested within isolate, P = 0.077. Data was arcsine square root transformed for analysis).

### 4.3.3.2 Invasion rate (G. mellonella)

In both experiment 1 and 2 the median number of adult nematodes present in insects which were dead at day 2, day 3 and day 4 ranged from 2 to 4 nematodes per insect (Appendix Table A 18). The data were not normally distributed and could not be normalised by transformation. For each experiment the data were pooled by colour phenotype and the number of IJs of each colour phenotype invading on each day was compared (Table 4.7). In experiment 1 the median number of adult nematodes present at each time point was 3 in all cases except for the YEL IJs at day 3, when the median number of adults nematodes present in insects was 4 and this difference was highly significant (Kruskal-Wallis test, df = 5, H = 15.78, P = 0.008). A series of multiple pairwise comparisons were carried out which showed that the number of YEL IJs invading on day 3 was significantly higher than the number of PUR IJs

invading on day 3 or day 4. (Mann-Whitney U-test, 95% Bonferroni-adjusted significance level of  $\alpha = 0.003$ , P = 0.0009, and P = 0.0030 respectively) (Appendix Table A 19). There were no other differences between the number of PUR and YEL IJs invading at any other time point. There was no difference in experiment 2 (Kruskal-Wallis test, df = 5, P > 0.05).

**Table 4.7:** Median number of adult nematodes of each colour phenotype present in *G. mellonella* cadavers at three time points following infection with PUR or YEL nematodes at an application rate of ten IJs per insect. At each time point in each experiment up to twelve insects per isolate were randomly chosen for dissection. In each experiment treatments which do not share the same Section 4.3.3.2letter were significantly different. (Mann-Whitney U-test with a 95% Bonferroni confidence interval) (Appendix Table A 19)

Experiment	Phenotype	Day		
		2	3	4
1	PUR	3a,b	3a	3a
	YEL	3a,b	4b	3a,b
2	PUR	3	3	3
	YEL	3	3	3

A similar trend was seen when the data from both experiments were pooled by colour phenotype (Kruskal-Wallis test df = 5, H = 16.20, P = 0.006). Multiple pairwise comparisons showed that the number of YEL IJs invading was significantly higher on day 3 compared to the number of PUR IJs invading on day3 or day 4 (Table 4.8).

**Table 4.8:** Results of pairwise comparisons of median number of adult nematodes per cadaver in *G. mellonella* larvae which were dead at three time points post infection for each of two colour phenotypes of *H. downesi* nematodes from North Bull Island. Nematodes were applied at a rate of ten IJs per insect. Treatments which do not share the same letter were significantly different. (Mann-Whitney U-test with a 95% Bonferroni confidence interval) (Appendix Table A 20).

Phenotype		Day	
	2	3	4
PUR	3a,b	3a	3a
YEL	3a,b	4b	3a,b

### **4.3.3.3** Time of infective juvenile first emergence (*G. mellonella*)

In the second experiment all the remaining cadavers were placed on White traps fourteen days after infection. There was 100% emergence in all treatments except for the PUR3 isolate where there was no emergence from one cadaver (= 95% emergence). Time to first emergence ranged from 16 to 23 days post infection for the PUR isolates and from 16 to 25 days post infection for the YEL isolates with a median time to first emergence of 17 to 19 days for both colour phenotypes (Appendix Table A 21). The data were not normally distributed and could not be normalised. The data were pooled by colour phenotype and time to first emergence analysed using a Mann-Whitney U-Test. There was no difference in time to first emergence between colour phenotypes (P > 0.05) (Appendix Table A 22). A regression analysis carried out on time to first emergence vs time to death showed there was a positive correlation between time to death and time to first emergence for both colour phenotypes (R<sup>2</sup> = 0.603 for PUR and R<sup>2</sup> = 0.487 for YEL (see Appendix Figure A. 9A and Appendix Figure A. 9B).

### 4.3.3.4 Numbers of IJs emerging (G. mellonella)

The IJ harvests (Table 4.9, see also Appendix Table A 23) were compared using a nested ANOVA with isolate nested within colour. The analysis showed that neither colour phenotype nor isolate had a significant effect on the mean number of IJs emerging over the 17 day period (P < 0.05) (Appendix Table A 24) and no further analysis was carried out on the emergence data.

**Table 4.9:** Mean ( $\pm$  SE) number of IJs emerging for 3 PUR and 3 YEL isolates for *G. mellonella* over a 17 day period. n = 2 with 7 – 12 insects per replicate.

Isolate	Mean emergence per cadaver ( $\pm$ SE)
PUR1	131685 (4194)
PUR2	89918 (34300)
PUR3	136996 <sub>(31437)</sub>
YEL1	108103 (11543)
YEL2	114662 (9238)
YEL3	141245 (4141)

### 4.3.4 Test of PUR and YEL isolates against H. abietis

### 4.3.4.1 Overall mortality and time to death

In both experiment 1 and experiment 2 more insects died on day 5 of the experiment than on any other day and the total mortality reached approximately 90% by day 14 (Appendix Figure A. 10A and Appendix Figure A. 10B). The proportion of insects dead due to EPN infection at each time point for day 4 to 14 was analysed. The data were not normally distributed and were arcsine squareroot transformed for analysis. There were no differences between colour phenotypes in the proportion of insects dead at any time point in either experiment (nested ANOVA with isolate nested within colour, P > 0.05). Similarly, when the data from both experiments were combined (Figure 4.5) there was no difference between colour phenotypes at any time point (nested ANOVA, P > 0.05), although the PUR 3 isolate killed significantly fewer insects by day 14 (one-way ANOVA  $F_{5,6} = 4.51$ , P = 0.047) (Appendix Table A 25).



**Figure 4.5:** Mean proportion ( $\pm$  SE) of *H. abietis* larvae killed by PUR and YEL IJs over fourteen days. Data shown is mean of two experiments with three isolates of each colour phenotype in each experiment. n = 120 insects per colour phenotype.

### **4.3.4.2** Time of infective juvenile first emergence (*H. abietis*)

The percentage of cadavers with emergence of IJs ranged from 80% to 100% (Appendix Table A 26) with no difference due to colour phenotype in either experiment or when the data for both experiments were combined, (2-Sample T-test, P > 0.05 in all cases) (Appendix Table A 27). While there was a difference in time to first emergence at isolate level in both experiments separately (Table 4.10)., there was no overall pattern due to phenotype. Time to first emergence ranged from 19 to 54 days post infection in experiment 1 and from 18 to 50 days post infection in experiment 2.

**Table 4.10:** Mean ( $\pm$  SE) or median time post infection to first emergence of infective juveniles for each of six isolates of *H. downesi* nematodes from North Bull Island in two experiments using *H. abietis*. Treatments which do not share the same letter were significantly different.

Isolate	mean <sup><math>+</math></sup> (± SE) or median	mean <sup>+</sup> ( $\pm$ SE) or median <sup>‡</sup> time to first emergence of IJs (days post infection)					
	Experiment 1 <sup>†</sup>	Experiment 2 <sup>‡</sup>					
PUR1	$26.0_{(1.25)}$ a b	18.0 a					
PUR2	24.8 $_{(0.61)}$ a b	23.0 b					
PUR3	$32.5_{(3.81)}$ b	20.0 a b					
YEL1	$23.2_{(1.01)}$ a	22.0 a b					
YEL2	$28.1_{(120)}$ a b	21.0 a b					
YEL3	27.8 $_{(0.90)}$ a b	21.5 b					

 $\dagger$  GLM, F<sub>4,69</sub> = 3.98, P = 0.006

‡ Mann-Whitney U-test with a 95% Bonferroni confidence interval, (see Appendix Table A 28 for results of pairwise comparisons.

As was the case in the bioassay with *G. mellonella* (Section 4.3.3.3), there was a positive correlation between time to death and time to first emergence for both colour phenotypes in both experiments and this relationship was stronger for the PUR colour phenotype than the YEL in both experiments ( $R^2 = 0.2335$  for PUR and  $R^2 = 0.1355$  for YEL in experiment 1,  $R^2 = 0.3375$  for PUR and  $R^2 = 0.0488$  for YEL in experiment 2, see Figure A. 11A-D).

### 4.3.4.3 Numbers of IJs emerging (*H. abietis*)

Numbers of IJs emerging from *H. abietis* larvae ranged from approximately 30,000 per insect (YEL3 in experiment 1) to over 60,000 (PUR3 in experiment 2) (Table 4.11) (See also Appendix Table A 26). There was no difference in the percentage of cadavers with emergence between colour phenotypes in either experiment or when both experiments were combined (Appendix Table A 27). There was also no

difference between colour phenotypes in the mean number of IJs emerging in either experiment (2-sample T-Test, experiment 1: df = 2, T = 0.60, P = 0.610; experiment 2: df =2, T = 0.89, P = 0.468). The data from experiment 1 and experiment 2 were combined and analysed using a nested ANOVA, with isolate nested within colour and colour nested within experiment. There were no differences between colour phenotypes in number of IJs emerging over the 17 day period (nested ANOVA, F<sub>2,8</sub> = 0.506, P = 0.621)

**Table 4.11:** Number of IJs emerging from *H. abietis* cadavers for three PUR and three YEL isolates over a 17 day period in each of two experiments. In each experiment one pooled count was carried out for each isolate with 12 - 14 cadavers per isolate in experiment 1 and 20 - 22 cadavers per isolate in experiment 2.

Experiment	Isolate	Mean emergence per cadaver
1	PUR1	47356
	PUR2	49977
	PUR3	45730
	YEL1	57755
	YEL2	45164
	YEL3	29528
2	PUR1	50305
	PUR2	45167
	PUR3	64285
	YEL1	50791
	YEL2	46545
	YEL3	46751

### **4.3.5** Host infectivity – *T. molitor*

In the assay with *T. molitor*, with an application rate of ten IJs per insect, more insects died on day 3 of the experiment than on any other day (Figure 4.6). The total mortality by day 8 ranged from 38% to 54%. The proportion of insects dead at each

time point was compared using a nested ANOVA with isolate nested within colour. The data were arcsine squareroot transformed for analysis. There was no difference in the mean proportion of insects dead for either factor at any time point.



**Figure 4.6:** Mean proportion ( $\pm$  SE) of *T. molitor* killed by PUR and YEL IJs over eight days in a continuous exposure experiment. Nematodes were applied at a rte of en IJs per insect. Data shown is mean of three isolates of each colour phenotype. n = 144 insects per colour phenotype

### **4.3.6 Host infectivity – Kelp fly**

### 4.3.6.1 Overall Mortality

There were two 48 hour exposure experiments, with a concentration of 1000 or 5000 IJs per 10 larva, respectively. In the 1000 IJ concentration experiment mortality due to EPN infection ranged from 60% to 100% after 48 hours and from 70% to 100% by day 6 (Appendix Table A 29). In the 5000 IJ concentration mortality due to EPN infection ranged from 20% to 80% after 48 hours and from 50

% to 100% by day 6 post infection. In each treatment a number of the larvae had pupated (Figure 4.7A) and in some instances adult flies had emerged. Nematodes developing within the larvae were easily observed (Figure 4.7B). There was no effect of colour phenotype or isolate on the proportion of insects dead for either concentration (nested ANOVA with isolate nested within phenotype, P > 0.05) (Figure 4.8). Interestingly, there was a lower mean mortality for each colour phenotype at the higher concentration (Figure 4.8B).



**Figure 4.7:** Kelp fly larvae infected with *H. downesi* isolates from North Bull Island. (A) Some larvae started to pupate prior to death. (B) Nematodes can be seen developing within the kelp fly larva. Image shown is a larva in the 1000 IJ /10 larvae treatment



**Figure 4.8:** Mean proportion ( $\pm$  SE) mortality of kelp fly larvae in a single 48 hour exposure with (A) 1000 IJs / 10 insects and (B) 5000 IJs / 10 insects. n = 15 insects per colour phenotype in each experiment.

## 4.3.6.2 Assessment of colonisation of kelp fly larva by the entomopathogenic nematodes

All ten insects for each isolate from one block in the 1000 IJ / 10 insect experiment were sampled destructively to assess the success of the infection (Table 4.12). In a number of cadavers the number of nematodes observed was low (3/27 and 12/30 cadavers for PUR and YEL, respectively), with only dead nematodes present in two of the PUR cadavers. In the YEL treatment no nematodes were observed in 10/30 cadavers, despite evidence of bacterial colonisation, whereas no nematodes were observed in only 1/27 cadavers in the PUR treatment. The colonisation appeared more advanced in the YEL treatment as most of the cadavers contained pre IJs with few of the previous life stages present, whereas for two of the three PUR isolates there were many worms in the intermediate life stages still present in the cadavers.

			Total number of EPN infections(/10)					
Isolate	C. frigida (/10)	none/dead*	adults	adults & juveniles	adults, juveniles & bagged worms†	bagged worms & juveniles	pre IJs	
PUR1	10	1	1		1	6		9
PUR2	8	2	1	1	2	1	1	8
PUR3	10						10	10
YEL1	10	1	1				8	10
YEL2	10	5		1			4	10
YEL3	9	6	1				3	10

**Table 4.12:** Colonisation of kelp fly by entomopathogenic nematode-bacterium complex. Ten insects were exposed to IJs for 48 hrs at a concentration of 1000 IJs/10 larvae and dissected fourteen days later.

\* none/dead was recorded where there was a large amount of bacterial biomass present, with either low numbers of dead nematodes or no nematodes.

<sup>†</sup> In a process known as *endotokia matricida*, in each generation some eggs hatch within the maternal body and consume it. These are referred to as bagged worms.

### 4.3.6.3 Numbers of IJs emerging (kelp fly)

In both experiments some emergence occurred in all treatments within 24 hrs of cadavers being placed on White traps. Emergence ranged from 37.5% to 100%, with no difference between isolates, colour phenotypes or IJ concentration (Appendix Table A 30A and Table A 30B).

i) Individual counts per cadaver

Emergence ranged from 304 IJs to 20832 IJs per cadaver. (See Appendix Table A 31 and Table A 32 for mean emergence per cadaver). The data for both concentrations were combined and analysed using a nested GLM with isolate nested within colour and concentration included as a factor. There was no difference in number of IJs emerging per cadaver between colour phenotypes and no effect of concentration on number of IJs emerging per cadaver (P > 0.05 for both factors), however there was a highly significant difference between isolates (P < 0.001) (Table 4.13).

**Table 4.13:** Mean emergence ( $\pm$  SE) over 17 days from kelp fly larvae. The data have been combined for two infection concentrations (1000 IJs and 5000 IJs) as there was no difference in number of IJs emerging due to concentration. n is the total number of cadavers with IJ emergence for each isolate. Isolates which do not share the same letter were significantly different (nested GLM,  $F_{4,70} = 5.86$ , P < 0.001)

Isolate	Mean number of IJs per isolate	n
PUR1	6273 <sub>(972)</sub> ab	16
PUR2	9960 (1789) a	12
PUR3	10326 <sub>(1760)</sub> a	10
YEL1	10664 <sub>(701)</sub> a	13
YEL2	4569 <sub>(812)</sub> b	16
YEL3	6817 (1053) ab	10

(ii) Counts of pooled emergence from all remaining cadavers

There was no difference in mean emergence per cadaver when the individual counts or the pooled counts were compared. All the emergence data for both concentrations was combined (Appendix Table A 33) and the mean emergence per cadaver was analysed using a GLM with isolate nested within colour and concentration included as a factor. There was no effect due to isolate, colour phenotype or concentration on the mean IJ emergence per cadaver (Table 4.14).

**Table 4.14:** Mean ( $\pm$  SE) emergence per cadaver over 17 days from kelp fly larvae infected with 1000 IJs per 10 larvae or 5000 IJs per 10 larvae. Nematodes used were three PUR and three YEL isolates from North Bull Island. Data shown is from pooled harvests for each isolate. There were three blocks in the 1000 IJ treatment and four blocks in the 5000 IJ treatment.

	Concer	ntration
Colour phenotype	1000	5000
PUR	6496 (821)	8075 (605)
YEL	5832 (1673)	7912 (706)

### 4.4 Discussion

At most, there were minor differences between *P.t. temperata* and *P.t. cinerea*. The ability of both PUR and YEL strains to utilise four species from three insect orders indicates that under laboratory conditions at least, nematode-bacterium complexes, *H. downesi-P.t. temperata* (YEL) and *H. downesi-P.t. cinerea* (PUR) have a broad host range as do many other EPN species (Peters, 1996). Mortality of the test insects ranged from 81% to 95% for *G. mellonella* larvae at an application rate of ten EPN per insect over an eight day period, 88% to 94% for *H. abietis* larvae at an application rate of fifty EPN per insect over a fourteen day period, 33% to 58% for *T. molitor* at an application rate of ten EPN per insect over an eight day period, and

70% to 100% at an application rate of 100 IJs per insect and 50% to 100% at an application rate of 500 IJs per insect for *Coelopa* spp. larvae over a six day period. The difference in the mortality rate of *G. mellonella* and *T. molitor*, a smaller host, at the same application rate and over the same time period is not surprising as *G. mellonella* are known to be very susceptible to EPN infection. On the other hand, coleopteran species are more difficult to kill and *H. abietis* larvae, which are similar in size to *G. mellonella* larvae, required a five-fold higher application rate and a longer time period to achieve the same mortality rate as seen in *G. mellonella*.

This is the first report of EPN being able to use *Coelopa* spp. as hosts. It is possible that the nematodes and Coelopa larvae on North Bull Island might encounter one another, particularly the H. downesi-P.t. cinerea (PUR) strains as these predominate at the front of the dunes system and C. frigida has been reported from the zone between the strandline and the embryonic dunes on North Bull Island (Healy, 1975). While there was no difference in mortality between P.t. cinerea and P.t. temperata subspecies, there were differences in the progression of the infection. The nematodes failed to develop in many cases in the P.t. temperata infections (Table 4.12), and where the IJs did develop there were with more pre-IJs present in cadavers infected with P.t. temperata (Table 4.12). This suggests that P.t. temperata is less able to utilise *Coelopa* cadavers, thus triggering a switch to production of infective juveniles earlier than is the case for the *P.t. cinerea* strain. However, there were no differences in overall IJ yields between the two subspecies. The characteristics of the bacteria were investigated in Chapter 2 and Chapter 3. There were differences in the growth of the *pur* phenotype but not the *yel* phenotype in nutrient broth and MacConkey broth, suggesting that the two phenotypes had different growth requirements (Section 2.3.3c). There were also some differences in the biochemical profiles of the two phenotypes, suggesting differences in their metabolism. The differences in the time to switching development to infective juvenile production within the host cadaver may be due to differences in the ability of the two bacterial phenotypes to utilise the nutrients in this particular host. The lower mortality rate seen with the higher application rate may be due to competition within the host between the developing nematodes or between the bacteria for scarce resources.

In tests with the wild type (Pp, Yy) and hybrid (Py, Yp) combinations of nematodebacterium against G. mellonella there was no significant difference between the combinations in the percentage of insects dead at any time point. However, when the data were compared by bacteria only the *pur1* strain of bacteria appeared to kill slightly quicker than the *yel3* strain, with higher percentage mortality in the *pur* treatments, irrespective of combination (Pp or Yp), although the difference was significant at the 10% level only and only at 56hrs post infection. These results confirm the effect seen in Chapter 3 (Section 3.3.3). In two tests against G. mellonella larvae with all six isolates, (three PUR and three YEL), this effect was also observed, with a higher mortality due to PUR nematodes at 48 hrs post infection, though again the effect was significant at the 10% level only. Similar results were obtained in further experiments with all six isolates (M Asaiyah, unpublished) and support the suggestion that the *pur* bacteria, *P.t. cinerea* kills *G*. mellonella larvae quicker than the yel bacteria, P.t. temperata. There was also some evidence that the *pur1* strain grew more quickly than the *yel3* strain in *G. mellonella* larvae (Section 3.3.1.1). It may be the case that early in the infection the growth rate of the *pur* bacteria is higher if this phenotype is better adapted to G. mellonella due to differences in the ability of the two phenotypes to utilise this particular host. Temperature is also a factor in speed of kill. In these experiments the insects were

incubated at 20°C following infection whereas the bacterial growth rate experiments were carried out at 28°C. The temperature profile for the growth of the two bacteria may be a factor in the quicker kill in the case of infection by the PUR nematode-bacterium combination.

In tests with the wild type and hybrid strains against *H. abietis* all strains were able to kill the larval stage at doses as low as 10 IJs per insect and in one experiment the *pur* bacteria was more virulent when a range of doses were compared (Section 4.3.2). However, this difference in speed of kill was not seen in tests with all six isolates against *H. abietis, T. molitor* or *Coelopa* spp. as there were no significant differences in percentage mortality at any stage in the infection between the colour phenotypes. The K122 strain of *H. downesi* has been shown to control *H. abietis* in forest situations (Dillon *et al.*, 2006; Williams *et al.*, 2013). This strain carries the *P.t. temperata* bacteria which is the subspecies carried by the *H. downesi* YEL strains in this study.

There is no evidence of differential adaptation of *P.t. temperata* and *P.t. cinerea* to different hosts. However, only four hosts were tested and none of these are known to be natural hosts of *H. downesi* in the sand dune system on North Bull Island, although *C. frigida* does occur in rotting clumps of seaweed deposited along the strandline.

## Chapter 5 Effect of host desiccation on IJ survival and emergence

### 5.1 Introduction

It is clear from the results presented in Chapter 2 that both *P. temperata* subsp. *cinerea* (*pur*) and *P. temperata* subsp. *temperata* (*yel*) have persisted in the *H. downesi* nematodes in the sand dunes on North Bull Island for at least 10 years. There is some evidence of niche separation between the two nematode/bacteria combinations, with the *pur* phenotype predominating in areas of low soil moisture. In 2002, 2005 and 2008 nematodes carrying the *pur* subspecies were more prevalent in the front portion of the dunes, Rolston (2005; 2006), and Section 2.3.1.2. One possible reason for this spatial distribution is that differences in desiccation tolerance between the two bacterial subspecies are driving niche separation. Soil moisture content is likely to be more variable in the front than in the rear of the dunes, since sandy soil loses moisture quickly in dry or hot conditions due to its unstructured nature and low humic content (Rodwell *et al.*, 2000; Verhoeven, 2002) (see also Section 2.1).

A number of mechanisms may be responsible for this spatial separation. Nematode population survival in the soil depends primarily on the availability of suitable hosts. Although nothing is known about the natural host range of *H. downesi* on North Bull Island there is no evidence that the three PUR or three YEL phenotypes differ in their ability to kill or reproduce in different hosts (Sections 4.3.3 - 4.3.6). While the IJ can persist for a considerable time in the soil, it is subject to a range of biotic and

220

abiotic conditions, some of which can make the environment unfavourable and thus limit its survival. Among the challenging abiotic conditions are periods of low water availability, fluctuations in soil temperature, UV radiation, as well as variations in pH and salinity (reviewed in Section 2.1).

It is reasonable to assume that EPN, like other organisms isolated from a particular environment have, over the course of evolutionary time, adapted to survive the range of environmental conditions that pertain in that particular habitat (Hominick, 2002). For example, EPN isolated from regions and altitudes where sub-zero soil temperatures occur are able to survive freezing, either through freeze tolerance (Brown & Gaugler, 1996) or freeze avoidance (Wharton & Surrey, 1994), although survival rate is low for IJs directly exposed to sub-zero temperatures. Similarly, EPN species isolated from warmer regions or which have an ambusher foraging strategy can survive temperatures above 30°C, although there is evidence that heat tolerance is not simply a question of adaptation to a geographic locality (reviewed in Grewal et al. (2006). A correlation has been observed between the heat or cold tolerance of the nematode and its symbiont, for example P. luminescens akhurstii, the symbiont of H. indica and X. poinarii, the symbiont of S. glaseri and S. cubanum, nematodes that are found in tropical and sub-tropical regions, have upper growth limits at 38-39°C and 40°C respectively, while P. luminescens, the symbiont of H. megidis, and X. bovienii, the symbiont of S. feltiae respectively, nematodes that are found in temperate regions, have upper growth limits at 34°C and 32°C respectively (Boemare, 2002).

A number of studies have looked at the role of the host cadaver in protecting the EPN from unfavourable environmental conditions over a prolonged time period. In a study using *S. carpocapsae* strain A10 Serwe-Rodriguez *et al.* (2004) found that IJs

emerged from G. mellonella that had been desiccated in air at 29% relative humidity (RH) for up to 56 days post-infection. The authors speculated that the A10 strain might be adapted to survive within a desiccated host as it was isolated from farmland in Wisconsin, USA, an area which can experience prolonged dry periods during the summer months, interspersed with heavy rainfall. Koppenhöfer et al. (1997) found that EPN (three Steinernema spp. and one Heterorhabditis sp.) could survive in G. mellonella cadavers stored in soil with 2% moisture for up to 91 days, with differences in survival between the *Steinernema* spp., possibly an adaptation due to their foraging strategy or to the habitat where they were isolated. The authors suggested two possible mechanisms for the survival of IJs within the desiccated host cadaver; first, that the drying of the host cuticle acts as a buffer against further moisture loss within the cadaver, and second, that the IJs become trapped within the hardened cuticle and are unable to emerge until the cadaver is rehydrated. This is similar to the views of Brown and Gaugler (1997), who found that while IJs (S. *carpocapsae*) could survive in G. mellonella cadavers for more than 40 days at  $15^{\circ}$ C and 75% RH, the nematodes could only emerge following rehydration of the cadavers. Spence et al., (2011) found that EPN (2 Steinernema spp. and 1 Heterorhabditis sp.) could survive within G. mellonella cadavers desiccated at 0% RH which had lost up to 70% of their body weight (desiccated for up to 15 days).

It is clear from the results presented in Chapter 2 (Section 2.3.1.2) that both *P. temperata* subsp. *cinerea* (*pur*) and *P. temperata* subsp. *temperata* (*yel*) (Chapter 2, Section 2.3.4) have persisted in the *H. downesi* nematodes in the sand dunes on North Bull Island for at least 10 years. There is some evidence of niche separation between the two nematode/bacteria combinations as in 2002, 2005 and 2008 nematodes carrying the *pur* subspecies were more prevalent in the front portion of

the dunes Rolston (2005; 2006) and Section 2.3.1.2, where soil moisture content is likely to be more variable than in the rear of the dunes, since sandy soil loses moisture quickly in dry or hot conditions due to its unstructured nature and low humic content (Rodwell *et al.*, 2000; Verhoeven, 2002) (see also Section 2.1).

In this chapter a hypothesis for the apparent spatial segregation of the two nematode/bacteria colour phenotypes from North Bull Island will be investigated through a series of experiments designed to test ability of the IJs carrying each bacterial subspecies to survive in desiccated hosts (*G. mellonella*). Using the wild and hybrid nematode/bacteria combinations produced in Chapter 3 (see Table 3.2) to focus on the role of the bacterial partner in IJ survival in hosts stored under desiccating conditions, I will test the hypothesis that the *pur* bacterial phenotype, *P. temperata* subsp. *cinerea*, protects nematodes in cadavers against desiccation better than the *yel* bacterial phenotype, *P. temperata* subsp. *temperata*.

The objectives of this chapter are:

- To determine the ability of the wild type and hybrid nematode/bacteria combinations produced in Chapter 3 to survive and emerge from desiccated hosts.
- ii) To look for differences between the *pur* and *yel* bacterial subspecies in terms of infective juvenile survival and number of IJs emerging.

### 5.2 Materials and Methods

All desiccation experiments were carried out using *G. mellonella* last instar larvae weighing between 201 and 366 mg as the host. Experiment 1 was set up to determine the response in cadavers stored in sand at a range of different moisture levels. Experiment 2 was set up to determine the response in cadavers stored at 0% relative

humidity (RH). Following on from these two initial experiments, experiments 3 and 4 examined in more detail nematode survival in cadavers stored for various periods in dry sand or at 0% RH. Table 5. lists details of percentage moisture (desiccation in sand) and duration of desiccation treatments for experiments 1 - 4. The experimental methods are described in more detail in Sections 5.2.1 and 5.2.2 below.

Experiment	Nematode/bacte	Desiccation method						
		S	Sand			Glass desiccator		
			(moisture content w/w)	ture nt Duration (days)		RH <sup>1</sup>	Duratio (days)	n
	W	Е	,	W	E		W	Е
1 <sup>†</sup>	Рр Ру 3	8	0% 1% 2% 3% 8%	6 9 12 15 18	14 53 61			
2 <sup>‡</sup>	Pp Py 5-8 Yy Yp	8				0%	3 6 9 28	3 6 9 28
3*	Pp Py Yy Yp	5	0%	28	14 28 42 56	0%	28	21 28 42 56
4+	Pp Py Yy Yp	25	0%	28	28 60	0%	28	28

**Table 5.1:** Experimental set up to investigate the effect of host desiccation on IJ survival and emergence (E) from *G. mellonella*. Cadaver weight loss (W) was also monitored.

1 All glass desiccators were at 0% relative humidity (RH).

### † Experiment 1

Three cadavers per nematode/bacteria combination and sand treatment were weighed (W) at each of five time points (6, 9, 12, 15, and 18 days).

Eight cadavers per nematode/bacteria combination and sand treatment were assessed for nematode survival by recording IJ emergence (E) after 14 days. A further eight cadavers in the 0% sand treatment were assessed after 53 or 61 days.

2 Experiment 2

(W) n = 5 (day 3), n = 6 (day 6, day 9), n = 8 (day 28) for cadaver weight. (E) On day 9 five cadavers per nematode/bacteria combination were returned to the experiment and reweighed before rehydrating and recording IJ emergence at day 28, along with a further three cadavers per nematode/bacteria combination.

\* Experiment 3

(E) Five cadavers per nematode/bacteria combination were weighed (W) at each time point in each desiccation treatment before rehydrating and recording IJ emergence. In the case of the 0 % RH treatments the cadavers were weighed at day 7, returned to the experiment and reweighed before rehydrating and recording IJ emergence at the subsequent time points.

+ Experiment 4

25 cadavers per nematode/bacteria combination were weighed at each time point in each desiccation treatment before rehydrating and recording IJ emergence.

### **5.2.1 Desiccation in sand**

An initial experiment was set up to examine the ability of the wild type and hybrid nematode/bacteria combinations produced in Chapter 3 to survive and emerge from desiccated cadavers. Host cadavers were stored for various time periods in sand at a range of moisture contents to determine the experimental conditions at which differences between the wild type and hybrid nematode/bacteria combinations might be apparent. Sand (Play-pit sand, B & Q Ireland Limited) was filter-washed to give a particle size in the range  $150 - 850 \mu m$ , allowed to air dry then heat sterilised over night at 120 °C (0% moisture). The moisture content of the sand was adjusted to 1%, 2%, 3% or 8% w/w by adding sterile water, mixing thoroughly and equilibrating for three days. Galleria mellonella were infected in multiwell plates as described in Section 3.2.2.2, with the following variations: nematode suspensions were adjusted to 1000 IJs/ml, 40 IJs were added to each well and the weight of each insect was recorded before adding it to the multiwell plate. Three days after initial exposure all insects exhibiting evidence of nematode infection (Section 2.2.1.3) were randomly assigned to sand pots as follows: 60 cm<sup>3</sup> polypropylene pots (4 cm x 4.4 cm, H x D) were filled to a depth of 2 cm with sterile sand and a single cadaver was placed in each pot (Figure 5.A, B). The pot was then filled with sterile sand (Figure 5.C) and a lid placed on top with the initial weight of the insect recorded on the lid. The pots were double bagged in zip lock polythene sandwich bags, placed in polystyrene boxes and stored at 20 °C.

Weight loss during desiccation was assessed at each of five time points (6, 9, 12, 15 or 18 days) by removing the cadaver from the sand and weighing it. Three cadavers from each moisture treatment (0%, 1%, 2%, 3% or 8% w/w water) were assessed in this manner. Nematode survival within the cadavers removed for weighing was then

assessed by dissecting the insects as described in Section 3.2.1.5 and noting the abundance of nematodes within the cadaver and the life stages present. Infective juvenile emergence was assessed at each of three time points (14, 53 or 61 days) by removing the cadavers from the sand and placing them individually on a 7 cm filter paper on a 3.5 cm Petri dish lid in a 9 cm Petri dish (White trap). The filter paper was moistened with 1 ml of tap water and the traps were stored at 20 °C on trays covered with tinfoil for 24 hrs to rehydrate the cadavers. Water was then added to the traps and they were monitored daily for IJ emergence for up to four weeks. White traps where IJ emergence occurred were harvested and counted (Sections 2.2.1.4 & 2.2.1.6). At each assessment point sand pots from which the cadaver had been removed were returned to the experimental arena for the remainder of the experiment to minimise disruption to the storage conditions.

Experiments 1, 3 and 4 included sand treatments (Table 5.). All timings are from the start of the desiccation period with day 0 being the day the cadavers were placed in the sand pots. Experiment 1 consisted of 32 blocks of ten sand pots with one pot per treatment, at each of two infection types (Pp or Py) for each of five moisture treatments (0, 1, 2, 3 & 8 %). The blocks were randomly assigned to be assessed as follows: three blocks at each of five time points (6, 9, 12, 15 or 18 days) to monitor weight loss during desiccation; eight blocks to assess nematode survival in desiccated cadavers; nine blocks to monitor IJ emergence into sand at each of 3 intervals (14, 53 & 60 days).

In experiment 3 cadavers were desiccated in dry sand only (0 % moisture). There were five blocks of four sand pots, with one pot for each of four infection types (Pp, Py, Yy or Yp), for each time point (14, 28, 42 or 56 days). Nematode emergence was

227

assessed for all time points. White traps were harvested and counted nine days following first emergence in any trap and a further seven days later in the 28 day treatment only. All traps were harvested and counted eight days following first emergence in any trap and a further seven days later as previously described.

In experiment 4 cadavers were desiccated in dry sand only (0% moisture). There were five replicates with 5 pots per replicate, for each treatment, in a randomised block design. Treatments were four infection types, (Pp, Py, Yy or Yp) for each of two time points (28 or 61 days). Nematode emergence was assessed as previously described. Traps were harvested ten days following first emergence in any trap and a further seven days later. Harvests for all five cadavers per replicate were pooled and a single count carried out.



**Figure 5.1:** *G. mellonella* cadavers showing the characteristic colouration of infection with (A) *pur* or (B) *yel* bacteria were placed in sand pots, (C) covered with sand and (D) double bagged and stored at 20°C for up to 61 days.

### **5.2.2 Desiccation at 0% relative humidity**

Desiccant cartridges containing silica gel beads (197 mm, DES 850-050G, Fisher Scientific) were dried at 50 °C overnight. The cartridges were placed in glass desiccators (200mm, FB35005, Fisher Scientific) which were allowed to equilibrate at 20 °C for three days before adding *G. mellonella* cadavers. Relative humidity (RH) was monitored using Humonitor<sup>®</sup> humidity indicator cards (Z163457, Sigma-Aldrich). *Galleria mellonella* were infected as described in Section 5.2.1. Three days after initial exposure all insects exhibiting evidence of nematode infection (Section 2.2.1.3) were randomly assigned to 4-well trays, made by cutting multiwell plates into segments (Figure 5.2A). Each well had three 3 mm diam. holes drilled in the base to allow for air movement. Cadavers were desiccated for up to 28 days at 0% RH in glass desiccators (Figure 5.2B). Weight loss and nematode emergence were assessed at intervals as described in Section 5.2.1. Experiments 2, 3 and 4 included desiccator treatments (Table 5.).

Experiment 2 consisted of five desiccators with four or five blocks, each block containing one cadaver of each infection type, (Pp, Py, Yy or Yp). At least one block per desiccator was removed at each of three time points (3, 6, or 9 days) and weight loss was assessed. The cadavers removed at day 3 and day 6 were stored on moist tissue paper and placed on White traps 14 days after the initial infection to assess nematode emergence. The cadavers removed on day 9 were returned to the desiccators and reweighed on day 28. All remaining cadavers were also removed on day 28 and nematode emergence was assessed. All traps in the 28 day treatment were harvested and counted eight days following first emergence in any trap and a further eight days later as previously described.

Experiment 3 consisted of five desiccators with five blocks, each containing one cadaver of each infection type, (Pp, Py, Yy or Yp), one block was removed at each of five time points (14, 21, 28, 42 or 56 days). All cadavers were weighed at day 7, returned to the desiccator and one block per desiccator was reweighed at the time points listed. Nematode emergence was assessed. The 28 day treatments were harvested and counted eight days following first emergence in any trap and a further seven days later as previously described.

Experiment 4 consisted of five desiccators with five replicate blocks, each containing one cadaver of each infection type, (Pp, Py, Yy or Yp). All cadavers were removed after 28 days and weight loss and nematode survival was assessed. White traps were harvested and counted eight days following first emergence in any trap and a further seven days later as previously described.



Figure 5.2: (A) *G. mellonella* cadavers infected with one of four nematode/bacteria combinations were placed in multiwell plate segments and (B) desiccated at 0% RH and  $20^{\circ}$ C for 28 days.

### 5.2.3 Statistical analysis

Routine statistical analysis, including tests for normality, was carried out using Minitab statistical software, version 16.1.1 (Minitab Inc., 2010). Normality of data was tested using the Anderson-Darling method. Where the data were not normally distributed comparisons between two groups were carried out using a Mann-Whitney U test. Analysis of variance was carried out using a one-way ANOVA or when groups had an uneven number of observations the general linear model (GLM) with an unbalanced nested design was used to compare isolate and colour phenotype. Analysis of variance was followed by pair-wise multiple comparisons using Tukey's test where appropriate. In experiments where nematode type and bacterial type were factors, results were analysed using a two-way ANOVA. Where no significant interactions occurred treatment means for each factor were compared using a oneway ANOVA and post-hoc Tukey's test for three or more treatments. When data could not be normalised treatment medians were compared using a Kruskal-Wallis test, followed if necessary by multiple pairwise comparisons using a Mann-Whitney U-test test with Bonferroni 95% confidence intervals, obtained by calculating  $(1 - \alpha / \alpha)$ g) where  $\alpha$  is the level of significance and g is the number of pairwise comparisons carried out. Proportional data were analysed using a Chi-square test. In all tests a P value of < 0.05 was taken as significant unless indicated otherwise.

### 5.3 **Results**

## **5.3.1 l** Nematode survival in cadavers stored at a range of moisture levels (experiment 1)

In experiment 1 three cadavers per infection type (Pp & Py) from each moisture treatment (sand at 0%, 1%, 2%, 3% or 8% moisture content (w/w)) were dissected at

each of five time points post desiccation (6, 9, 12, 15 & 18 days). Numbers of nematodes in the cadavers were assessed semi-quantitatively (Table 5.2) and there was no difference in the estimated number of nematodes per cadaver between infection types due to moisture level at any time point (Kruskal-Wallis test, P > 0.05at each time point). Both nematode types survived in cadavers in all moisture treatments and at all time points (Appendix Table A 34 - Table A 38). There was no difference between infection types in the number of cadavers with live nematodes. The estimated number of nematodes per cadaver ranged from less than 500 to more than 10000 by day 6 of the experiment. By day 18 of the experiment pre-infective juvenile nematodes were present in all cadavers. It was apparent during this initial experiment (experiment 1) that IJs were emerging into the sand in all treatments except 0% moisture and therefore in experiment 1 only cadavers from the dry sand treatment (0% moisture) were placed on White traps to assess IJ emergence. As the objectives of this chapter were to investigate differences between the *pur* and *yel* bacterial subspecies in terms of nematode survival and emergence from desiccated hosts subsequent experiments involving hosts desiccated in sand used only the dry sand treatment and the duration of the desiccation period was varied (experiment 3 and experiment 4).

**Table 5.2:** Estimated<sup>†</sup> median number of nematodes per *G. mellonella* cadaver dissected at each of five time points<sup>‡</sup>. Insects were infected with Pp or Py nematode/bacteria combinations and stored in sand at a range of different moisture levels prior to dissection. n = 3 for all treatments.

		Е	Estimated median number of nematodes per cadaver							
Nematode/bacteria	Moisture content	Day 6	Day 0	Doy 12	Doy 15	Day 19				
combination	(w/w)	Day 0	Day 9	Day 12	Day 13	Day 10				
	0%	500	10000	10000	10000	10000				
	1%	1000	10000	10000	10000	10000				
Рр	2%	10000	10000	10000	10000	10000				
	3%	1000	10000	10000	10000	10000				
	8%	10000	10000	10000	10000	10000				
	0%	500	1000	10000	1000	10000				
	1%	500	10000	10000	10000	10000				
Ру	2%	1000	10000	10000	10000	10000				
	3%	1000	10000	10000	10000	10000				
	8%	500	10000	10000	10000	1000				

The number of nematodes per cadaver was assessed semi-quantitatively as follows: 500 = up to 500 nematodes present; 1000 = 500 - 1000 nematodes present; 10000 = 1000 - 10000 nematodes present.

By day 18 IJs were emerging into the sand in all moisture treatments except 0% (dry sand).

### 5.3.2 Emergence of infective juveniles from cadavers stored in dry sand (experiments 1, 3 & 4)

There was IJ emergence from 100% of the cadavers stored for 14 days in dry sand in both experiment 1 (Pp & Py only) and experiment 3 (Pp, Py, Yy & Yp) (Table 5.3). IJ emergence from cadavers stored in dry sand for 28 days ranged from 60 to 100% and 72 to 80% respectively in experiment 3 and experiment 4 (Pp, Py, Yy & Yp) (Table 5.3). There was IJ emergence from cadavers store for up to 61 days in dry sand in experiment 1 (Table 5.3). There was no emergence from any cadavers in the 42 day treatment except those infected with the Pp nematode/bacteria combination in experiment 3 and no emergence from any treatment after 56 days in experiment 3 (Table 5.3), and in experiment 4 there was no emergence from any cadaver after 60 days in dry sand (Table 5.3).

There was no clear trend in the time to first emergence following rehydration of the cadavers (Table 5.3) and in most cases IJs emerged 2-3 days after rehydration from cadavers stored for 14 days (experiments 1 & 3), although for four cadavers in the Pp infections IJs emerged after 4-5 days. For cadavers stored for 28 days (experiments 3 & 4) IJs emerged after 1-3 days but here, for two cadavers in the Yy infection and three cadavers in the Py infection, first emergence occurred after 7-8 days and 4, 6, and 9 days, respectively. In experiment 1 the bacterial type had a significant effect on the time to first emergence of IJs following rehydration from cadavers stored for 14 days in dry sand (Mann-Whitney U test, W = 88.0, P = 0.0269) while in experiment 3 there was an effect of bacteria type at the 10% level and an effect due to nematode type at the 5% level in this treatment (Table 5.4). In both experiments, IJs took longer to emerge from *pur* than from *yel* cadavers.

The mean numbers of IJs emerging from cadavers in experiment 1 are shown in Table 5.5. There was a significant difference due to moisture content (two-way ANOVA,  $F_{1,28} = 4.85$ , P = 0.036) but not due to bacterial phenotype (two-way ANOVA, P > 0.05) and no interaction between the two factors (two-way ANOVA, P > 0.05) following storage for 14 days in dry sand or sand at 8% moisture. Fewer IJs emerged from the dry treatment (approximately 13,000 overall average of *pur* and *yel* in the dry sand treatment compared to approximately 18,000 overall average of *pur* and *yel* in the 8% moisture sand treatment). Following storage of the cadavers in dry sand for up to 61 days the number of IJs emerging in the Pp nematode/bacteria combination was between 1 and 1507. The cadavers with no

emergence were excluded and data for emergence at days 53 and 61 were combined for analysis. There was a significant difference in the number of IJs emerging due to bacterial phenotype between the two nematode/bacterial combinations (one-way ANOVA,  $F_{1,10} = 10.12$ , P = 0.010) (Table 5.5), with the number of IJs emerging from Pp cadavers two orders of magnitude higher than from Py cadavers.

As day 28 was common to both experiment 3 and experiment 4 it is treated separately below.

Experiment Nematode/		Percentage of White traps with emergence					Median time to first emergence (days post rehydration)				
	combination	d	lays in sand	at 0% m	oisture		days in sand at 0% moisture			sture	
		14	28	42	53•	$61^{\dagger}$	14	28	42*	53	61
1	Рр	100			100	100	3.5			4.5	1
	Ру	100			50	50	2.0			1	1
3	Рр	100	60	20	0		3.0	3.0	2		
	Yp	100	100	0	0		2.0	1.0			
	Yy	100	80	0	0		2.0	2.0			
	Ру	100	80	0	0		3.0	1.0			
$4^{\dagger}$	Рр		80 (12.6)			0		2.0			
	Yp		72 (12.0)			0		2.0			
	Yy		76 <sub>(16.0)</sub>			0		2.0			
	Ру		76 (16.0)			0		2.0			

**Table 5.3:** Percentage  $(\pm SE)^{\dagger}$  of White Traps with emergence and time to first emergence of IJs from *G. mellonella* cadavers following desiccation in dry sand for 14, 28, 42, 53 or 61 days.

n = 8 for each nematode/bacteria combination at 14 days, n = 2 for each nematode/bacteria combination at 53 days, n = 6 for each nematode/bacteria combination at 61 days in experiment 1.

n = 5 for each nematode/bacteria combination at each time point in experiment 3

n = 25 for each nematode/bacteria combination in experiment 4.

† In experiment 4 percentages of White traps with emergence is a mean of 25 for each nematode/bacteria combination. The second time point was 60 days.

\* In experiment 3 there was emergence from 1 cadaver only.

• In experiment 3 cadavers were removed from sand pots after 56 days.

**Table 5.4:** Results of Mann-Whitney U test on median time to first emergence of IJs from cadavers desiccated for 14 days in sand at 0% moisture. Data is from experiments 1 and 3 with n = 8 per bacterial type in experiment 1 and n = 10 per bacterial type or nematode type in experiment 3

		Median time to first IJ emergence (days)					
Experiment		Bacterial typ	e	Nematode type			
1		pur	3.5				
		yel	2.0				
	P value	P = 0.0269					
	test statistic	W = 88.0					
3		pur	4.5	PUR	3.0		
		yel	3.0	YEL	2.0		
	P value	P = 0.0625		P = 0.0318			
	test statistic	W = 128.5		W = 130			

**Table 5.5:** Mean ( $\pm$  SE) numbers of IJs emerging from *G. mellonella* cadavers following desiccation for 14 days in dry (0%) or moist (8%) sand or 53-61 days in dry (0%) sand. n = 8 for each nematode/bacteria combination in all treatments. Data shown is for experiment 1. Results of Two-way ANOVA (14 days 0% and 8%) and One-way ANOVA (53-61 days) analysis of number of IJs emerging are also given.

Nematode/bacteria combination	Sand treatment		
	0%	8%	0%
	Duration		
	14 days	14 days	53-61 days
Рр	143372 (38627)	197099 <sub>(24664)</sub>	98841 (21328)
Ру	115878 (11547)	139767 (12500)	509 (346)
	P value	P = 0.036	P value $P = 0.010$
	Test statistic	$F_{1,28} = 4.85$	Test statistic $F_{1,10} = 10.12$

# 5.3.3 Emergence of infective juveniles from cadavers stored for 28 days in dry sand (experiments 3 & 4)

Infective juvenile nematodes emerged from cadavers infected with each nematode/bacteria combination and stored for 28 days in dry sand in both experiments (Table 5.6). There was no effect of nematode type or bacteria on the percentage of cadavers with IJ emergence in either experiment (Chi-Square test, P>0.05 in all cases).

In experiment 3 the number of IJs emerging per cadaver over a 16 day period from first emergence was between 7000 and 126000 approximately, while in experiment 4 the mean number of IJs emerging over a 17 day period from first emergence was between 800 and 67000 approximately. The data from both experiments were combined and the mean numbers of IJs emerging per cadaver (cadavers with emergence only) were analyzed using a GLM with nematode and bacteria as fixed factors. A nematode by bacteria term was included in the model to test for interaction. The analysis showed that following desiccation of the cadavers for 28 days in dry sand more IJs emerged from insects infected with nematodes carrying the *pur* bacteria than from insects infected with nematodes carrying the *yel* bacteria and this difference was significant at the 10% level ( $F_{1,32} = 4.10$ , P = 0.051) (Figure 5.3). There was no effect of nematode on the numbers of IJs emerging and no evidence of any interaction (P > 0.05 in both cases).
**Table 5.6:** Mean emergence ( $\pm$ SE) per cadaver, percentage of cadavers with emergence, median time to first emergence of IJs following rehydration and mean emergence ( $\pm$ SE) per cadaver (cadavers with emergence only) from *G. mellonella* cadavers following desiccation for 28 days in dry sand in two experiments.

			All cadavers			Cadavers with emergence only				
Experiment	Nematode	Bacteria	Mean emergence per cadaver (±SE) n		Percentage of cadavers with emergence	Median time to firs emergence (days post rehydration)	t Mean em per cadav	ergence ver (±SE)	n	
3	Р	р	42238	(23010)	5	60	3.0	70396	(27816)	3
	Y	р	67321	(21714)	5	100	1.0	67321	(21714)	5
	Y	у	35793	(18940)	5	80	2.0	44742	(21551)	4
	Р	у	44291	(14674)	5	80	1.0	55364	12432)	4
4	Р	р	37273 <sup>†</sup>	(5356)	$5^{\dagger}$	80	2.0	47340 <sup>‡</sup>	(1569)	5 <sup>‡</sup>
	Y	р	$30203^{\dagger}$	(5058)	$5^{\dagger}$	72	2.0	44183 <sup>‡</sup>	(7304)	5 <sup>‡</sup>
	Y	у	$19724^{\dagger}$	(4869)	$5^{\dagger}$	76	2.0	$25860^{\ddagger}$	(2373)	5 <sup>‡</sup>
	Р	у	$20424^{\dagger}$	(7930)	$5^{\dagger}$	76	2.0	21019 <sup>‡</sup>	(7605)	$5^{\ddagger}$

Exp 3: 1 insect x 5 blocks, individual counts for each cadaver

Exp 4: 5 replicate insects x 5 blocks, 1 pooled count for each block

† mean emergence per cadaver is mean of 5 blocks (pooled count per block / 5 to give mean per cadaver for each block)

‡ mean emergence per cadaver is mean of 5 blocks (pooled count per block / n cadavers with emergence for each block)



**Figure 5.3:** Mean ( $\pm$  SE) number of IJs emerging from *G. mellonella* cadavers (A) for each nematode type and (B) for each bacteria type. Insects were infected with 40 IJs of one of four nematode/bacteria combinations (Pp, Py, Y, Yp) and cadavers were desiccated for 28 days in dry sand. Data shown is from experiments 3 and 4 (cadavers with emergence only). n for each type is displayed above the bar. In each chart types which do not share the same letter were significantly different (GLM  $F_{1,32} = 4.10$ , P = 0.051) for bacteria type.

#### 5.3.4 Emergence of infective juveniles from cadavers stored for 28 days at 0%

#### relative humidity (experiment 2, 3 & 4)

In experiment 2 IJs emerged from all eight cadavers for all five nematode/bacteria combination stored for 3 or 6 days at 0% RH. In the 28 day treatment there was no emergence from any cadavers in the Py nematode/bacteria combination and emergence from cadavers in the other nematode/bacteria combinations ranged from 37.5 to 62.5% (Table 5.7). In experiment 3 IJs emerged from some cadavers in all four nematode/bacteria combinations in the 21 day treatment, ranging from 60 to 100% (data not shown) and this fell to between 20 and 40% in the 28 day treatment (Table 5.7). There was no IJ emergence from any cadavers in the 42 or 56 day treatments. After 28 days in experiment 4 there was no IJ emergence of IJs occurred 32 days after the cadaver was rehydrated. There was an extremely wide

variation across the three experiments in the 28 day treatment in numbers of IJs emerging from individual cadavers, ranging from 18 IJs in a Yy infection in experiment 2 to 117491 IJs in a Yp infection, also in experiment 2 (Appendix Table A 39).

The data for 28 days from experiments 2 - 4 were combined for analysis. The combined emergence data were not normal since in many cases no infective juveniles emerged from individual cadavers. A Kruskal-Wallis test on combined data showed a highly significant difference in numbers of IJs emerging between the nematode/bacteria combinations (H = 11.82, DF = 3, P = 0.008). The data were pooled by nematode type and by bacteria type and compared using a Mann-Whitney test. This showed that there was a highly significant difference in the number of IJs emerging depending on the bacteria the nematodes carried (Mann-Whitney U-Test, W = 6389.5, P = 0.0017), with ten times more IJs emerging where the infecting nematodes carried the *pur* bacterial phenotype than the *yel* bacterial phenotype (Appendix Table A 40). There was no difference in numbers of IJs emerging when the data were pooled by nematode type (P > 0.05).

The combined data for the cadavers with emergence only were analyzed using a GLM with nematode and bacteria as fixed factors and experiment as a random factor. A nematode by bacteria term was included in the model to test for interaction. The analysis showed that there was an effect of bacteria at 10% level on the number of IJs emerging from the cadavers stored for 28 days at 0% RH with more IJs emerging from insects infected with nematodes carrying the *pur* bacteria than from insects infected with nematodes carrying the *yel* bacteria (Figure 5.4) (F<sub>1,22</sub> = 3.37, P = 0.080). There was no effect of nematode and no evidence of a nematode-bacteria

interaction (P > 0.05 in both cases), however, experiment was also significant at the 10% level.

When the number of cadavers with emergence were grouped by nematode type or by bacteria type there was a highly significant difference between the *pur* and *yel* phenotypes in the number of cadavers with emergence with more cadavers infected with the *pur* bacteria producing IJs, (21/76 *pur* compared to 7/76 *yel*) (Chi-square Test,  $\chi^2 = 8.581$ , DF = 1, P = 0.003) but there was no difference between the PUR and YEL nematodes in the number of cadavers with emergence (Chi-square Test, P > 0.05).

**Table 5.7:** Mean emergence ( $\pm$ SE) per cadaver, percentage of cadavers with emergence, mean time to first emergence of IJs following rehydration and mean emergence ( $\pm$ SE) per cadaver from *G. mellonella* cadavers following desiccation for 28 days at 0% RH in three experiments.

			All cadavers		Cadavers with emergence only				
Experiment	Nematode	Bacteria	Mean em	nergence ver (±SE)	n	Percentage of cadavers with emergence	Median time emergence (c post rehydrat	to first lays Mean emerg ion) per cadaver	gence (±SE) n
2	Р	р	37577	(13985)	8	62.5	ť	60124 (14	447) 5
	Y	р	47824	(18498)	8	50.0	ţ	95648 (84	89) 4
	Y	У	3239	(3065)	8	37.5	<b>†</b>	8637 (80	22) 3
	Р	у	0		8	0.0			
3	Р	р	9604	(8849)	5	40.0	10	24010 (20	911) 2
	Y	р	209	(209)	5	20.0	8	1045	1
	Y	У	12567	(8138)	5	40.0	2.5	31417 (83	68) 2
	Р	у	26	(26)	5	20.0	3	129	1
4	Р	р	6560	(3861)	25	16.0	10	41000 (16	571) 4
	Y	р	7725	(4012)	25	20.0	4	38627 (13	579) 5
	Y	У	747	(747)	25	4.0	4	18667	1
	Р	у	0		25	0.0			

<sup>†</sup> In experiment 2 exact times to first emergence of infective juveniles were not recorded. In the Pp nematode/bacteria combination IJ emergence had occurred in 2/5 cadavers by 3 days post rehydration and in the remaining 3/5 cadavers by 7 days post rehydration. In the Yp nematode/bacteria combination IJ emergence had occurred in 2/4 cadavers by 3 days post rehydration and in the remaining 2/4 cadavers by 7 days post rehydration. In the Yp nematode/bacteria combination IJ emergence had occurred in 1/3 cadavers by 3 days post rehydration and in the remaining 2/4 cadavers by 7 days post rehydration.



**Figure 5.4:** Mean ( $\pm$  SE) number of IJs emerging from *G. mellonella* cadavers (A) for each nematode type and (B) for each bacteria type. Insects were infected with 40 IJs of one of four nematode/bacteria combinations (Pp, Py, Yy, Yp) and cadavers were stored for 28 days at 0% RH. Data shown is from experiments 2 – 4. n for each type is displayed above the bar. (B) Bacterial types were significantly different at the 10% level (GLM F<sub>1,22</sub> = 3.37, P =0.080).

#### 5.3.5 Weight loss during desiccation

### 5.3.5.1 Desiccation in sand

In experiment 1 cadavers were weighed at each of five time points for each moisture treatment (Appendix Table A 41). In the dry sand treatment cadavers lost between 30% and 45% of their initial weight over a period of 18 days. There was no difference in the mean percentage weight loss between insects infected with either the wild type or hybrid nematode/bacteria combinations (Pp and Yp, respectively) at any time point (one-way ANOVA, P > 0.05 in all cases) (Figure 5.5).



**Figure 5.5:** Mean ( $\pm$  SE) weight loss from *G. mellonella* cadavers stored in dry sand over an 18 day period. n = 3 at each time point.

In experiment 3 the percentage weight loss per cadaver ranged from approximately 34 to 53% at day 14, 61 to 74% at day 28 and 63 to 74% at day 42 (data not shown). In experiment 4 the percentage weight loss per cadaver ranged from approximately 50 to 75% at day 28. The weight loss data from experiment 3 and experiment 4 for the 28 day dry sand treatment were not normal and could not be normalised. The data from both experiments were pooled by nematode type or by bacteria type and were analysed by Mann-Whitney U test. In each case there was no difference in the percentage weight loss between the colour phenotypes (Table 5.8).

**Table 5.8:** Median percentage change in weight of *G. mellonella* cadavers stored in dry sand for 28 days. The data were pooled by nematode type and by bacteria type and were combined for two experiments with n = 10 per nematode type or bacteria type in experiment 3 and n = 50 per nematode type or bacteria type in experiment 4.

	Median percentage weight loss ( $n = 60$ )
Nematode type	
PUR	-68.97
YEL	-68.62
Bacteria type	
pur	-68.96
yel	-68.62

#### 5.3.5.2 Desiccation at 0% RH

In experiment 2 weight loss was assessed at days 3, 6 and 9 (Figure 5.6). Percentage weight loss per cadaver ranged from approximately 7 – 12% at day 3, 15 – 27% at day 6 and 21 – 38% at day 9. The data at each time point were pooled by nematode type (PUR or YEL) or by bacteria type (*pur* or *yel*) and were tested using a two-way ANOVA of percentage weight loss per cadaver. There was no difference in the mean percentage weight loss between insects for either nematode type or bacteria type and no evidence of an interaction at any time point (two-way ANOVA, P > 0.05 in all cases).

The percentage weight loss per cadaver in the 28 day treatment ranged from approximately 60 - 77%, 61 - 78% and 50 - 81% in experiments 2, 3, and 4, respectively (Table 5.9). The data for the 28 day treatment from experiments 2 - 4 were combined for analysis and pooled by nematode type (PUR or YEL) or by bacteria type (*pur* or *yel*) and were tested using a two-way ANOVA of percentage

weight loss per cadaver. The analysis showed there was no effect of nematode or bacteria on the percentage weight loss per cadaver and no interaction effect and no further analysis was carried out on the data.



**Figure 5.6:** Mean (±SE) percentage weight loss from cadavers stored at 0% RH at 3 time points.

		Nematode/bacte	eria combination	
Experiment	Рр	Yp	Yy	Ру
2	-70.73 <sub>(1.968)</sub>	-71.53 (1.153)	-73.45 <sub>(0.481)</sub>	-68.45 <sub>(2.368)</sub>
3	-71.05 (2.63)	-74.79 <sub>(0.68)</sub>	-71.56 <sub>(0.86)</sub>	-75.21 <sub>(1.04)</sub>
4	-72.41 (0.74)	-71.97 (0.40)	-70.61 (0.44)	-71.04 (0.56)

**Table 5.9:** Mean ( $\pm$  SE) percentage weight loss from *G. mellonella* infected with one of four nematode /bacteria combinations and stored for 28 days at 0% RH.

## 5.4 **Discussion**

Cadavers with *P. temperata* subsp. *cinerea* (*pur1*) subjected to desiccation produce more IJs than those with P. temperata subsp. temperata (yel3) so the hypothesis that the pur bacterial phenotype, P. temperata subsp. cinerea, protects nematodes in cadavers against desiccation better than the yel bacterial phenotype, P. temperata subsp. temperata is plausible. Koppenhöfer et al. (1997), using G. mellonella larvae as the host, found that for three Steinernema spp. (S. carpocapsae All strain, S. glaseri NC strain, S. riobravis Texas strain and one Heterorhabditis sp. (H. *bacteriophora* NC1 strain) there was no emergence from cadavers stored for 14 days in loamy sand at 1% moisture (equivalent to soil water potential -500 mPa) whereas for cadavers stored for 14 days in loamy sand at 2% moisture (equivalent to soil water potential -40 mPa) emergence ranged from 75 to 100%. The cadavers were not rehydrated. In a second experiment with the same species the authors found that IJs of all four species emerged from rehydrated cadavers that had been stored for up to four weeks in loamy sand at 2% moisture and for two to three months in the case of H. bacteriophora and S. riobravis, respectively and emergence at both the 21 and 35 day time points was 100% for H. bacteriophora.

From my results it is clear that a substantial proportion of *H. downesi* nematodes carrying either bacterial subspecies, *P. temperata* subsp. *cinerea* (*pur1*) or *P. temperata* subsp. *temperata* (*yel3*), can survive for at least 28 day in host cadavers stored in dry sand or glass desiccators at 0% RH. In my experiments I found that IJs emerged from all rehydrated cadavers at the 14 day time point and from 60 to 100% of rehydrated cadavers at the 28 day time point in the dry sand treatment. In the 0% RH treatment emergence was more variable, ranging from 0 to over 60% after 28 days. If we compare this result to that obtained by Kung *et al* (1991) cited above it is

clear that the host cadaver provides considerable protection to IJs under severe desiccating conditions. Similarly to the results obtained by Koppenhöfer et al. (1997), IJs emerged from rehydrated cadavers that had been stored in dry sand for up to 61 days, although this result was obtained in experiment 1 only. Koppenhöfer et al. (1997) noted that the H. bacteriophora-infected G. mellonella cadavers had a "gummous consistency" and speculated that this might be a factor in helping the IJs to survive in the cadavers under desiccating conditions by helping to retain moisture in the cadavers. In my first experiment the cadavers which were removed from the sand treatments at the five time points for weight loss assessment were dissected to assess nematode survival and there was no difference in the number of cadavers with live nematodes (Appendix Table A 34 - Table A 38). I observed that the yel3infected cadavers had a stickier consistency than the *pur1*-infected ones, however, this gummy consistency of the yel3-infected cadavers does not appear to assist the IJ survival as there was a trend for greater emergence from *pur1*-infected cadavers. There was a difference in IJ emergence as early as the 14 day time point with fewer IJs emerging from *yel3*-infected cadavers compared to the *pur1*-infected ones and by days 53-61 emergence from cadavers in the Py treatment (yel3-infected cadavers) was approximately 0.5% of the emergence from cadavers in the Pp treatment (pur1infected cadavers). Similarly, in experiments 3 and 4, more IJs emerged from pur1infected cadavers than from *yel3*-infected ones following storage for 28 days in dry sand, although there was no difference in the number of cadavers with IJ emergence in any of these experiments at the same time points. Under severe conditions (0% RH) there was a highly significant difference in the IJ emergence from the cadavers following 28 days storage with emergence from more purl-infected cadavers in experiments 2 and 4, although in experiment 2 there was emergence from equal numbers of *pur1*-infected and *yel3*-infected cadavers. Here again, however, significantly more IJs emerged from *pur1*-infected cadavers than from *yel3*-infected ones.

In a study of *S. affine* in the Czech Republic Půža & Mráček (2005, 2007) observed a decrease in IJ numbers in the soil during dry periods followed by a rapid increase in IJ numbers in the soil as soil moisture increased following precipitation. The authors suggested IJ persistence in host cadavers in the soil during dry conditions as a possible reason for this observed increase and also speculated that this might be an adaptation for this particular EPN rather than a case of being trapped within the dried cuticle or having no choice as emergence would expose the IJs to low moisture conditions in the soil which they would be unlikely to survive for very long Koppenhöfer *et al.* (1997). Both groups have shown that IJs can persist for considerable lengths of time in cadavers under conditions that would be rapidly lethal to IJs in the soil. Neither the Koppenhöfer *et al.* (1997) study or the work by Půža & Mráček (2005, 2007) have however considered the role of the symbiotic bacteria in the IJ survival within the cadaver.

Across all infection types and both experiments the mean weight loss for cadavers stored in dry sand in experiments 3 and 4 was approximately 67%, while across all infection types and three experiments the mean weight loss for cadavers stored at 0% RH was approximately 72%. This is similar to the weight loss reported for *G. mellonella* infected with *S. carpocapsae* (All strain), *S. riobrave* Cabanillas, Poinar and Raulston, and *H. bacteriophora* HB1 strain and subsequently stored at 0% RH by Spence *et al.* (2011). Here the authors were able to show that IJs could persist in cadavers that had lost up to 70% of their original mass and it was noted that the number of cadavers with IJs emerging, and the number of IJs emerging per cadaver

were both species specific, with a greater proportion of emergence for both *Steinernema* spp. while the lowest IJ production was observed for *S. riobrave*. The Spence *et al.* (2011) study also noted that the rate of weight loss was slowest for cadavers infected with *S. carpocapsae* and suggested that differences in the rate of nematode development or in the activity of the bacterial symbionts on the host cuticle could be involved. From the results of my experiments with *H. downesi* nematodes it appears that the two symbiont subspecies differentially influence the survival of IJs within the host cadaver, although the precise mechanism is unknown. While there was a difference in the internal consistency of the EPN infected cadaver, the stickier condition of the *yel*-infected cadavers, which is presumably a function of differences in metabolism between the two subspecies (see Section 2.3.3.d), this does not seem to benefit the nematodes carrying this subspecies.

Soil-dwelling nematodes require a film of water of sufficient thickness (relative to their body size) or water filled pores that provide a continuous film of water within the soil for them to move freely. The thickness of the water film and its continuity will be related to the type and size of soil particle. The free draining nature and small particle size of sand means that sand at a low moisture content (<1% by weight) will probably lack sufficient moisture to allow a continuous film of water to be present (Molyneux & Bedding, 1984). On the other hand, soil with high moisture content will lack sufficient air filled pores for the nematodes and in addition a continuous gradient of carbon dioxide and other signal molecules that the nematodes follow are unlikely to form. In open dune habitat water availability can be quite low. For example, the water table is likely to be more than 2 m below the soil surface and in dry dune slacks it will typically be 1 to 2 m deep (Ranwell, 1972). In addition, the

moisture capacity of the soil can vary considerably from front to rear in a dune system with the moisture capacity of older dunes being as high as 33%, while that of younger dunes can be as low as 7%, with the very young fore dunes having a moisture capacity of less than 1% and soil moisture typically being around 4% by weight (McLachlan & Brown, 2006). This difference in moisture capacity is due to the higher organic content of the older dune soil. For sandy soils 1% moisture capacity represents a water potential of around -10kPa (Richards & Weaver, 1944). Molyneux and Bedding (1984) achieved water potentials of between -1kPa and -10.2 kPa for a fine sand mixture (25% coarse sand, 65% fine sand, 7% silt, 3% clay) with the addition of 13% - 1% water. They found that *H. bacteriophora* was able to infect L. cuprina larvae in fine sand within this range of water potentials, indicating that this species was able to move through the sand and locate the host insect at this range of moisture contents. Thus the emergence of nematodes from cadavers stored at the range of soil water contents (1 - 8%) in the first desiccation experiment is not surprising. Since the main aim of this chapter was to explore differences between the pur and yel bacterial subspecies in terms of nematode survival and emergence from desiccated hosts subsequent experiments involving hosts desiccated in sand used only the dry sand treatment and the duration of the desiccation period was varied (experiment 3 and experiment 4). In addition, nematode-infected cadavers were subjected to severe desiccation in the form of exposure to 0% RH as hosts infected at or near the soil surface might be expected to experience dry periods due to exposure to the sun or during periods of drought.

# Chapter 6 General Discussion

Heterorhabditis rely on their symbiont to kill the insect host and provide nutrients for growth and reproduction. Variations in the attributes of the bacteria affect the fitness of the associated nematode partner, which may in turn allow them to exploit different habitats. Given the data in this and previous studies (Rolston et al., 2005; Rolston et al., 2006) there is evidence of spatial separation of the two colour phenotypes of H. downesi within the dune habitat on North Bull Island. The tendency in some years is for the PUR colour phenotype to predominate in the front of the dunes. Six Heterorhabditis isolates from North Bull Island were identified as *H. downesi* on the basis of analysis of a portion of the ITS1 region of the ribosomal DNA (Section 2.3.2). This analysis showed that although there were seven single nucleotide polymorphisms along a 1031 nucleotide sequence there was no pattern of variation due to colour phenotype, as only a single isolate varied at each site and all six nematode isolates were identified as *H. downesi* on the basis of comparison with sequences from the *H. downesi* type strain, K122. On the other hand the *pur* and *yel* colour phenotypes of the symbiotic bacteria were identified as *P. temperata* subsp. cinerea (P.t. cinerea) and P. temperata subsp. temperata (P.t. temperata) respectively on the basis of analysis of a portion of the gyrB gene (Section 2.3.4).

In 2001 and 2002 Rolston *et al.* (2005) found that the range of the *H. downesi* colour phenotypes on North Bull Island overlapped, with both colour phenotypes found in both the front 50 m and the rear 50 m of the dune system. In both years the PUR phenotype was more common towards the front of the dune system and in 2002 this trend was significant. In a 2005 targeted resurvey of the section of dunes where *H*.

downesi had previously been recorded the PUR phenotype was again significantly more common than the YEL in the front 30 m of the dunes (Rolston *et al.*, 2006). In contrast, in this study, there was no overlap of range in 2008, with the PUR phenotype occurring at the front and the YEL phenotype occurring at the rear of the dunes only (Section 2.3.1.2). In 2012 a resurvey of a single transect from the 2008 survey (transect 1) showed that once again the range of the two colour phenotypes overlapped. Overall the distribution of the two phenotypes recorded in this study was significantly different, with the PUR phenotype occurring more frequently in the front section and the YEL phenotype occurring more frequently in the rear section of the dune system (Chapter 2, Table 2.17 & Table 2.18). The continued occurrence of the H. downesi associated with both P.t. cinerea and P.t. temperata on North Bull Island, as demonstrated by five surveys between 2001 and 2012 and the tendency for the *P.t. cinerea* symbiont to predominate towards the front of the dune system points to the stability of the association of the nematode with two subspecies of the symbiotic bacteria and also to the stability of the distribution of the both subspecies of the symbiont within the dune system.

The co-existence at a single site of two different species or subspecies of *Photorhabdus* with a single *Heterorhabditis* species is unusual and has only previously been reported from an oak forest and a nearby clearing in Hungary (Tóth & Lakatos, 2009). However, it is likely that the purple and yellow colour variants of the symbiont associated with *H. downesi* reported from a survey of sites in Ireland and Britain (Griffin *et al.*, 1994) are in fact *P.t. cinerea* and *P.t. temperata*, respectively. Furthermore, a targeted survey of sandy soils in Ireland, Britain, Denmark, Estonia and Hungary (Griffin *et al.*, 1999) reported the occurrence of *H. downesi* with purple and yellow variants of the *Photorhabdus* symbiont, and both

colour variants co-occurred in two sites in Ireland, neither of which were on North Bull Island, two sites in Wales and one site in Denmark (C. Griffin, pers. comm.), and it is likely that these are also *P.t. cinerea* and *P.t. temperata*, respectively. Thus, it would seem that the co-existence of these two symbionts within *H. downesi* is a general occurrence and occurs throughout the range of this nematode species.

The association of *H. downesi* and its *Photorhabdus temperata* symbiont with sandy soils in Ireland (Griffin *et al.*, 1994; Rolston *et al.*, 2002), Britain and Denmark and its absence from sandy coastal sites in Estonia (Griffin *et al.*, 1999), where winters tend to be colder and wetter, suggest that the nematode-bacterium complex is adapted to drier, warmer conditions in the more northerly section of its range. In the more southerly section of its range, in Hungary, the nematode-bacterium complex tends to occur in shaded areas, such as under trees in roadside verges (Griffin *et al.*, 1999), or in forest clearings (Tóth & Lakatos, 2009), where the nematode-bacterium complex complex would be shaded from the more intense effects of UV radiation and higher temperatures associated with a more southerly latitude.

For species to coexist they must use different resources or respond to biotic and abiotic stresses differently (Chase & Leibold, 2003). In the dune system on North Bull Island biotic and abiotic factors vary along a gradient from front to back of the dunes. The North Bull Island dune system is in fact comprised a number of microhabitats, with loosely compacted, free draining sand with little ground cover at the front of the dunes progressing through a build-up of organic content to a more moisture-retentive grass dominated soil with a large component of humic matter towards the rear of the dune system. Along this soil gradient there is also a decrease in both salinity and pH due to the leaching of minerals and the build-up of organic matter in the soil. These abiotic changes result in an increase in the range and diversity of both above and below ground fauna and plant communities from front to rear of the dune system. Competition from other biotic factors within the dune system such as *S. feltiae*, nematophagous fungi or scavengers will also play a role in the success of the two *H. downesi-Photorhabdus temperata* complexes. It is clear from the surveys carried out in 2008 and 2012 and from the earlier surveys (Rolston *et al.*, 2005; Rolston *et al.*, 2006) carried out that *S. feltiae* is much more widely distributed in the sand dune system on North Bull Island than is *H. downesi* (Section 2.3.1.1). Although nothing is known of the distribution of nematophagous fungi within the sand dune system on North Bull Island. There is some evidence that the second-stage cuticle that is retained by the IJs helps to protect *Heterorhabditis* spp. from nematophagous fungi (Timper & Kaya, 1989)

A number of factors could contribute to the observed distribution of the symbiont types associated with *H. downesi* on North Bull Island. Host specialisation is an obvious factor. While little is known about its natural host preferences, *H. downesi*-infected larvae of the garden chafer, *Phyllopertha horticola* have been found in modified dune grassland in Co. Clare by R. Enright (C. Griffin, pers. comm.). A wide range of invertebrate species have been recorded from the habitats in the sand dune system in North Bull Island (reviewed in Section 1.4). Host adaptation may be influenced by the antimicrobial activity of the bacterial symbiote as well as a range of other pathogenicity factors. In Chapter 2, *P.t. cinerea* was shown to be significantly better than *P.t. temperata* at inhibiting a range of bacteria and fungi, and this may be an important factor in the adaptation of *P.t. cinerea* to the front section of the dune system, by enhancing the ability of the nematode-bacterium to exploit a wide range of hosts and also to scavenge already dead hosts, as is the case

in some *Heterorhabditis* and *Steinernema* species (San-Blas & Gowen, 2008; San-Blas *et al.*, 2012), although in those studies the insects were freeze-killed. Scavenging on epn-killed cadavers may be a feature of *H. downesi* behaviour as nematodes carrying either symbiont infected cadavers killed by their own or the other symbiont (Section 3.3.1). On the other hand, *H. downesi* and *S. feltiae* may avoid cadavers already infected with the other nematode as no bait insects infected with both *H. downesi* and *S. feltiae* were recovered from any of the North Bull Island soil samples (Section 2.3.1.1). This may be due to the presence of the other species of symbiont rendering the cadaver unattractive to the later nematode, although *in vitro* tests showed that both *P.t. cinerea* and *P.t. temperata* inhibit the growth of *X. bovienii*, the symbiont of *S. feltiae* (Section 2.3.1.1), however, this wasn't tested *in vivo* in the current study.

Insects vary in the diversity of their intestinal microbial community (Broderick *et al.*, 2004; Egert *et al.*, 2005; van der Hoeven *et al.*, 2008; Wang *et al.*, 2011; Hirsch *et al.*, 2012) and the invading nematode-bacterium complex has to overcome not only the host insect immune system but also the gut microbiota resident in the insect and opportunistic soil-dwelling microorganisms which may also try to invade the insect cadaver. One source of this difference in antimicrobial activity may be the pigments produced by the bacteria. The two subspecies of the *H. downesi* symbiotic bacteria caused infected larvae (*G. mellonella, H. abietis* and *Coelopa* spp.) to turn grey/purple or straw to golden yellow and produced different coloured culture in broth and different coloured colonies on nutrient or LB Millar agar. These pigments have been identified as anthraquinones, secondary metabolites produced by all *Photorhabdus* spp. (Bode, 2009). Several anthraquinones including a red pigment from *P. luminescens* strain HK (Richardson *et al.*, 1988), two yellow pigments from

*P. luminescens* strain C9 (Li *et al.*, 1995), and two anthraquinones from *P. luminescens* TT01 (Brachmann *et al.*, 2007) were all shown to have antimicrobial properties and while the exact mode of action of this class of compounds is not fully understood they are believed to play a role in protecting the insect cadaver from competing microorganisms. Since *P.t. cinerea* and *P.t. temperata* have different colour phenotypes it may well be the case that they each produce a different range of these antimicrobial pigments. Preliminary HPLC analysis (data not shown) indicated that there were differences in the range of compounds secreted by both subspecies and these compounds may include antimicrobial products that allow the EPN to exploit a range of hosts (Dillon & Charnley, 1988), and so play a role in the distribution of *P.t. cinerea* and *P.t. temperata* within the dune system on North Bull Island.

In plant-feeding insects, variation in the symbiotic microbes can allow the colonisation of new host plants (Douglas, 2009), increase the fitness of the insect host to abiotic conditions such as increased temperature (Chen *et al.*, 2000), and assist in overcoming plant defences (Paine *et al.*, 1997). The differences in growth dynamics *in vitro* and *in vivo* (Sections 2.3.3 and 3.3.1.1) and the results of the biochemical characterisation (Section 2.3.3) support this with, for example only *P.t. temperata* isolates testing positive for tryptophan deaminase, suggesting that *P.t. cinerea* and *P.t. temperata* differ in their ability to utilise resources. The growth dynamics within the host are particularly interesting (Chapter 3 Figure 3.2) as more colony forming units were recovered from insects co-infected with both bacterial subspecies than from hosts with either type on its own. It is possible that differences in the metabolic profile of the two subspecies allow them to partition the resources within the insect, thus allowing both subspecies to multiply and producing an

overall greater level of colonisation of the insect host suggesting a synergy between the two subspecies.

The observed in vitro differences in antimicrobial activity and growth rate appear to have only a slight impact on the behaviour of the nematode-bacterium complexes in hosts under normal laboratory conditions. No differences were found in virulence against a range of hosts (Chapter 3) but these were not necessarily typical of the biota used in North Bull Island. Using hosts from three insect orders (Lepidoptera (1), Coleoptera (2), Diptera (1)), there was no difference in the ability of *H. downesi* carrying either P.t. cinerea or P.t. temperata to kill the insect host or for the nematode to successfully complete its development. Photorhabdus. t. cinerea killed G. mellonella larvae quicker than P.t. temperata (Sections 3.3.3 Table 3.18 and Section 4.3 Table 4.1 and Figure 4.2), but there was no difference between subspecies in overall mortality of the four species of hosts from three orders or in numbers of IJs emerging from these hosts. There was also no difference in the number of bacteria of either subspecies carried by the six isolates from North Bull Island or by the recombined lines or in the proportion of IJs that had no bacteria (Section 3.3.5). Differences in number of bacteria carried could be either good (enhanced virulence) or bad (reduced IJ survival) (Mitani et al., 2004; Emelianoff et al., 2008a).

A difference however was found in the proportion of IJs carrying each subspecies of bacteria from cadavers that had been co-infected with both subspecies (Section 3.3.2 table 3.12). The bacteria carried by the IJs were assessed 4 and 19 days after emergence began. At both time points more IJs carried *P.t. temperata* and at the second time point in the case of 5/19 of the cadavers there was a switch in the subspecies of bacteria colonising the IJs from *P.t. cinerea* (3/5) or a mixture of both

subspecies (2/5) to *P.t. temperata* only. It is unclear if this is a result of a preponderance of *P.t. temperata* late in the infection as bacterial quantification from infected larvae was only carried out at one time point, 72 hours post injection of the bacteria, and at that time point *P.t. cinerea* was more plentiful (Section 3.3.1.1). Another possibility is that *P.t. temperata* is better at colonising the nematodes late in the infection. This might arise due to metabolic differences between the two subspecies disadvantaging *P.t. cinerea* in its ability to utilise resources late in the infection and produce components necessary for successful colonisation of the nematode. The *pur1* and *yel3* isolates did not inhibit each other in the antibiosis assays carried out in Chapter 2 (Section 2.3.3.1). This suggests that interference competition is occurring for the resources within the host.

There is evidence that *P.t. cinerea* protects IJs in cadavers in dry sand. *Heterorhabditis downesi* associated with *P.t. cinerea* within insect cadavers survived dry conditions significantly better (assessed by number of IJs emerging) than *H. downesi* associated with *P.t. temperata*. While there was no difference in the number of IJs emerging from cadavers infected with either subspecies under normal conditions, significantly more IJs emerged from insects infected with *P.t. cinerea* and stored under desiccating conditions for an extended duration (Sections 5.3.1 – 5.3.3). As discussed in Chapter 2 (Section 2.1) and Chapter 5 (Section 5.1), sand dunes represent a challenging environment for the animals and plants which live there and this is particularly true for smaller organisms such as nematodes, (Koppenhöfer *et al.*, 1997). *Photorhabdus t. cinerea*, occurring at the front of the dune system, is most exposed to desiccation and appears to be adapted to these conditions as evidenced by the greater survival of IJs carrying this subspecies in dry

conditions. Differences between the symbiotic bacteria may alter the nematodes' stress tolerance. Under dry conditions a number of biochemical changes occur which allows the nematode to survive desiccation. For example, nematodes have been shown to accumulate sugars and polyols, as well as undergoing alterations in fatty acid composition and the synthesis of a number of low molecular weight proteins (Glazer, 2002; Grewal et al., 2006) and any of these features may be influenced by diet. Dietary factors can influence resistance to desiccation or thermal stress (Andersen et al., 2010; Sisodia & Singh, 2012; Colinet & Renault, 2014). In Drosophila melanogaster dietary live yeast altered protein synthesis as well as altering total lipids and metabolic profile of the insect and promoted thermal tolerance (Colinet & Renault, 2014). Heterorhabditis species have been shown to reproduce when fed on some Photorhabdus strains but not on others (Han et al., 1991; Gerritsen & Smits, 1993, 1997; Han & Ehlers, 1998), presumably due to differences in the nutritional factors provided by their own symbionts or that of another nematode, although the precise nature of these differences are unclear (Clarke, 2008). There is also some evidence that nematodes may take up potential protectant molecules across the cuticle (Qiu et al., 2000; Qiu & Bedding, 2002) and differences in the biochemical activity of P.t. temperata and P.t. cinerea could also be a factor in making potential protectants such as trehalose available within the insect host, resulting in differences in survival of the H. downesi nematode in desiccated insects infected with one or the other symbiont.

Whatever the precise mechanism for the differences in protection from desiccation provided by the two *P. temperata* subspecies to *H. downesi* within a cadaver, *P.t. cinerea* provides a clear advantage to its associated nematodes under dry conditions. Although it is unusual for extreme drought conditions to persist for a period of

several weeks in Ireland, the effects of drying on the cadaver and the nematodes within it may be intensified by other extremes such as high temperatures or fluctuations in salinity due to cycles of drought and re-wetting and diurnal fluctuations in temperature (Willis *et al.*, 1959). During a hot summer sand near the surface may reach temperatures as high as 40°C (Huiskes, 1979). This would have the effect of adding to the thermal stress of the nematode and its symbiont and increasing the rate of drying of the cadaver.

Bacterial symbionts are mediators of ecologically important traits for their hosts. Photorhabdus provide a range of services for the Heterorhabditis hosts including entomopathogenicity, essential nutrition, and suppression of the host immune system and of competing microbes. While the trait differences between P.t. cinerea and P.t. temperata have not been fully identified and characterised the two subspecies differ in their ability to protect nematodes in cadavers under desiccating conditions and in their antimicrobial activity. Trait differences between the two symbionts P.t. cinerea and P.t. temperata leading to niche separation may enable the two symbionts to coexist and allow H. downesi expands its niche by using two subspecies of P. temperata and may help to explain the continued co-existence of two distinct subspecies of symbiotic bacteria within the *H. downesi* population .in the sand dune system on North Bull Island. Heterorhabditis populations occur at low frequency and have a patchy distribution (Campbell et al., 1996). Similarly, the host insects are a patchy resource (Crossan et al., 2007) Co-infection of a host by H. downesi carrying the different bacterial symbionts will probably be infrequent due to the patchiness of the nematode and host populations, but may be facilitated by movement of the host prior to death, since it takes 72 hours or more for the nematode-bacterium complex to kill the host insect, depending on its size. Since *Heterorhabditis* IJs always develop into self-fertile hermaphrodites in the first generation the nematodes can always reproduce. This allows the possibility for cross-fertilisation to occur in the second generation of worms. Under laboratory conditions, nematodes emerging from a co-infected host were found to be able to carry either host (Section 3.3.2). From the experiments investigating the role of the symbiont in nematode survival within a host (Section 5.3), it appears that under normal conditions *P.t. temperata* out-competes *P.t. cinerea* in forming a symbiotic association with the nematode when both subspecies are present in the same insect host, resulting in the *H. downesi-P.t. temperata* association predominating in the rear section of the dunes, where the increased organic content of the soil favours moisture retention. On the other hand the drier conditions that predominate in the sandier soil towards the front of the dune system appears to favour the *H. downesi-P.t. cinerea* association. There is an apparent lack of specialisation by *H. downesi* toward either *P. temperata* subspecies and presumably it can pick up the other symbiont by invading an already colonised host (Chapter 3).

A number of issues that merit further research have arisen from the work presented here. Relatively little is known about the natural host range of EPN. This study has shown that *H. downesi* North Bull Island isolates can infect, kill, reproduce in and emerge from the kelp flies *C. frigida* and *C. pilipes* but there were no differences between *P.t. cinerea* and *P.t. temperata*. Sicard *et al.* (2008) found that EPN were pathogenic to terrestrial isopods and Mauleon *et al.* (2006) reported that EPN were pathogenic to beach associated terrestrial isopods and amphipods. On North Bull Island *C. frigida* occurs in seaweed in the strandline (Healy, 1975) and it is likely that amphipods occur there also. Terrestrial isopods occur in the zone between the strand line and the embryonic dunes (Healy, 1975; Speight, 1997). These may be a natural host of *P. t. cinerea* since they occur at the front of the dune system where *P.t.* cinerea predominates. Testing *P.t. temperata* and *P.t. cinerea* against invertebrate species known to occur in the dune grassland habit may reveal further differences between the two subspecies, in particular, the effect of the antimicrobial activity of the symbiont against the diversity of insect intestinal microbe communities, which can vary considerably (Dillon & Dillon, 2004). It would also be of interest to carry out a survey of microbiota associated with a range of natural hosts from North Bull Island and test the efficacy of the *P.t. temperata* and *P.t. cinerea* against these where possible. Antimicrobials produced by the symbiotic bacteria allow the EPN to exploit a range of hosts and HPLC analysis could be used to investigate the range of chemicals produced by the two subspecies and their antimicrobial effect could also be tested.

The growth dynamics within the host early and late in the infection may be a factor in the colonisation of the early and late emerging IJs. *Photorhabdus t. cinerea* appears to predominate in the insect cadaver early in the infection. Is this because it is better at colonising the insect? Is *P.t. temperata* dominating the cadaver later in the infection? If both subspecies are equally colonising the insect late in the infection what factors are favouring IJ colonisation by *P.t.* temperata?

Much of the work described in this study looked at a single pair of *P.t. temperata* and *P.t. cinerea* isolates and therefore further study needs to be carried out to determine whether the observed differences i.e. *P.t. temperata* outcompetes *P.t. cinerea* in co-infections in the same insect host while *P.t. cinerea* is more desiccation tolerant than *P.t. temperata*, is an isolate effect or applies at subspecies level. The promiscuous behaviour of *H. downesi* in relation to the two subspecies of

264

*Photorhabdus* from North Bull Island may be a survival strategy that enables *H*. *downesi* to persist.

Two Photorhabdus subspecies, P.t. cinerea and P.t. temperata have been shown to co-exist at a single site in association with *H. downesi* (this study and that of Toth & Lakatos (2009)), and in association with H. downesi and H. megidis (Tóth & Lakatos, 2009), suggesting that these two nematodes are promiscuous in their association with the bacterial partner and that the symbiosis is not under tight regulatory control. In this study the niche separation of the two Photorhabdus subspecies appears to involve as yet undiscovered traits related to nematode colonisation and resistance to abiotic stress and by switching bacterial partners H. Downesi may be able to extend its niche in the dune system. This co-existence of two bacterial subspecies with a nematode species is in contrast to Maneesakorn *et al.* (2011) (discussed in Chapter 1), where a study of five Heterorhabditis species, including H. megidis and H. downesi, and their associated symbiotic bacteria found evidence of cospeciation events between Heterorhabditis and Photorhabdus. However, that study did not include analysis of H. downesi - P.t. cinerea and H. megidis - P.t. cinerea associations. Within the sand dune system on North Bull Island it is likely that the *H. downesi – P.t. cinerea* and *H. downesi – P.t. temperata* nematode-bacteria complexes co-exist as a freely-interbreeding population, capable of exchanging symbionts whenever IJs carrying the different symbiont co-infect the same host. The work presented here suggests that the persistence and success of each symbiont in the different microhabitats within the North Bull Island dune system depends in part on the ability of the symbiont to protect its nematode host as well as its ability to utilise and defend the insect host cadaver.

# References

- Abu Hatab, M., Stuart, R. J. & Gaugler, R. 1998. Antibiotic resistance and protease production by Photorhabdus luminescens and Xenorhabdus poinarii bacteria symbiotic with entomopathogenic nematodes: variation among species and strains. *Soil Biology and Biochemistry* **30**, 1955-1961.
- Adams, A. & Nguyen, K. B. 2002. Taxonomy and Systematics. In *Entomopathogenic Nematology* (ed. R. Gaugler), pp. 1-33. Wallingford, UK: CABI Publishing UK.
- Adams, B. J., Burnell, A. M. & Powers, T. O. 1998. A phylogenetic analysis of Heterorhabditis (Nemata : Rhabditidae) based on internal transcribed spacer 1 DNA sequence data. *Journal of Nematology* **30**, 22-39.
- Adams, B. J., Fodor, A., Koppenhofer, H. S., Stackebrandt, E., Stock, S. P. & Klein, M. G. 2006. Biodiversity and systematics of nematode-bacterium entomopathogens. *Biological Control* 37, 32-49.
- Akhurst, R. J. 1982. Antibiotic Activity of Xenorhabdus spp., Bacteria Symbiotically Associated with Insect Pathogenic Nematodes of the Families Heterorhabditidae and Steinernematidae. *Journal of General Microbiology* 128, 3061-3065.
- Akhurst, R. J. 1986. Xenorhabdus nematophilus subsp. poinarii: Its Interaction with Insect Pathogenic Nematodes. Systematic and Applied Microbiology 8, 142-147.
- Akhurst, R. J., Boemare, N. E., Janssen, P. H., Peel, M. M., Alfredson, D. A. & Beard, C. E. 2004. Taxonomy of Australian clinical isolates of the genus *Photorhabdus* and proposal of *Photorhabdus asymbiotica* subsp. asymbiotica subsp. nov. and *P. asymbiotica* subsp. australis subsp. nov. International Journal of Systematic and Evolutionary Microbiology 54, 1301-1310.
- Akhurst, R. J., Mourant, R. G., Baud, L. & Boemare, N. E. 1996. Phenotypic and DNA relatedness between nematode symbionts and clinical strains of the genus Photorhabdus (Enterobacteriaceae). *International Journal of Systematic Bacteriology* **46**, 1034-1041.
- Altschul, S. F., Madden, T. L., Schaffer, A. A., Zhang, J. H., Zhang, Z., Miller, W. & Lipman, D. J. 1997. Gapped BLAST and PSI-BLAST: a new generation of protein database search programs. *Nucleic Acids Research* 25, 3389-3402.
- An, R. & Grewal, P. S. (2010a). Molecular mechanisms of persistence of mutualistic bacteria Photorhabdus in the entomopathogenic nematode host. PLoS One 5(10): e13154. doi: 10.1371/journal.pone.0013154

- An, R. & Grewal, P. S. 2010b. Photorhabdus temperata, subsp.stackebrandtii; subsp. nov. (Enterobacteriales: Enterobacteriaceae). Current Microbiology 61, 291-297.
- An, R. & Grewal, P. S. 2011a. *Photorhabdus luminescens* subsp. *kleinii* subsp. nov. (Enterobacteriales: Enterobacteriaceae). *Current Microbiology* **62**, 539-543.
- An, R. S. & Grewal, P. S. 2011b. purL gene expression affects biofilm formation and symbiotic persistence of Photorhabdus temperata in the nematode Heterorhabditis bacteriophora. *Microbiology-Sgm* 157, 2595-2603.
- Andersen, L. H., Kristensen, T. N., Loeschcke, V., Toft, S. & Mayntz, D. 2010. Protein and carbohydrate composition of larval food affects tolerance to thermal stress and desiccation in adult Drosophila melanogaster. *Journal of Insect Physiology* 56, 336-340.
- Ansari, M., Tirry, L. & Moens, M. 2005. Antagonism between entomopathogenic fungi and bacterial symbionts of entomopathogenic nematodes. *Biocontrol* 50, 465-475.
- Armer, C. A., Rao, S. & Berry, R. E. 2004. Insect cellular and chemical limitations to pathogen development: the Colorado potato beetle, the nematode Heterorhabditis marelatus, and its symbiotic bacteria. *Journal of Invertebrate Pathology* 87, 114-122.
- Bai, X., Adams, A., Ciche, T. A., Clifton, S., Gaugler, R., Kim, K., Spieth, J., Sternberg, P. W., Wilson, K. L. & Grewal, P. S. (2013). A Lover and a Fighter: The Genome Sequence of an Entomopathogenic Nematode Heterorhabditis bacteriophora. PLoS One 8(7): e69618. doi: 10.1371/journal.pone.0069618
- Baker, A. C. 2003. Flexibility and Specificity in Coral-Algal Symbiosis: Diversity, Ecology, and Biogeography of Symbiodinium. *Annual Review of Ecology, Evolution, and Systematics* **34**, 661-689.
- Baliadi, Y., Yoshiga, T. & Kondo, E. 2004. Infectivity and post-infection development of infective juveniles originating via endotokia matricida in entomopathogenic nematodes. *Applied entomology and zoology* **39**, 61-69.
- Barbosa Negrisoli, C. R. d. C., Negrisoli Júnior, A. S., Botton, M., Garcia, M. S. & Bernardi, D. 2013. Evaluation of efficacy of 18 strains of entomopathogenic nematodes (Rhabditida) against Planococcus citri (Risso, 1813) (Hemiptera: Pseudococcidae) under laboratory conditions. *Experimental Parasitology* 134, 295-298.
- Bashey, F., Hawlena, H. & Lively, C. M. 2012a. Alternative paths to success in a parasite community: within-host competition can favor higher virulence or direct interference. *Evolution*.

- Bashey, F., Reynolds, C., Sarin, T. & Young, S. K. 2011. Virulence and competitive ability in an obligately killing parasite. *Oikos* **120**, 1539-1545.
- Bashey, F., Young, S. K., Hawlena, H. & Lively, C. M. 2012b. Spiteful interactions between sympatric natural isolates of *Xenorhabdus bovienii* benefit kin and reduce virulence. *Journal of Evolutionary Biology* 25, 431-437.
- Bedding, R. & Akhurst, R. 1975. A simple technique for the detection of insect parasitic rhabditid nematodes in soil. *Nematologica* **21**, 109-110.
- Bedding, R. A., Molyneux, A. S. & Akhurst, R. J. 1983. *Heterorhabditis* spp., *Neoaplectana* spp., and *Steinernema kraussei*: Interspecific and intraspecific differences in infectivity for insects. *Experimental Parasitology* 55, 249-257.
- Bennett, H. P. J. & Clarke, D. J. 2005. The *pbgPE* operon in *Photorhabdus luminescens* is required for pathogenicity and symbiosis. *Journal of Bacteriology* **187**, 77-84.
- Blackshaw, R. 1988. A survey of insect parasitic nematodes in Northern Ireland. Annals of Applied Biology 113, 561-565.
- Blaxter, M. (2011). Nematodes: The Worm and Its Relatives. Plos Biology 9(4): e1001050. doi: 10.1371/journal.pbio.1001050
- Blaxter, M. L., De Ley, P., Garey, J. R., Liu, L. X., Scheldeman, P., Vierstraete, A., Vanfleteren, J. R., Mackey, L. Y., Dorris, M., Frisse, L. M., Vida, J. T. & Thomas, W. K. 1998. A molecular evolutionary framework for the phylum Nematoda. *Nature* **392**, 71-75.
- Bode, H. B. 2009. Entomopathogenic bacteria as a source of secondary metabolites. *Current opinion in chemical biology* **13**, 224-230.
- Boemare, N. 2002. Biology, taxonomy and systematics of Photorhabdus and Xenorhabdus. In *Entomopathogenic Nematology* (ed. R. Gaugler), pp. 35 56. Wallingford, UK: CAB International.
- Boemare, N., Akhurst, R. & Mourant, R. 1993a. DNA relatedness between *Xenorhabdus* spp.(Enterobacteriaceae), symbiotic bacteria of entomopathogenic nematodes, and a proposal to transfer *Xenorhabdus luminescens* to a new genus, *Photorhabdus* gen. nov. *International Journal of Systematic Bacteriology* 43, 249-255.
- Boemare, N., Givaudan, A., Brehelin, M. & Laumond, C. 1997. Symbiosis and pathogenicity of nematode-bacterium complexes. *Symbiosis* **22**, 21-45.
- Boemare, N. & Lanois, A. 1996. Training courses for characterisation of *Xenorhabdus* and *Photorhabdus* phase variation. In COST 819 Entomopathogenic nematodes - Symbiosis and pathogenicity of nematodebacterium complexes (ed. N. Boemare, R.-U. Ehlers, A. Fodor & A. Szentirmai), pp. 109-114. Brussels, Belgium: European Commision.

- Boemare, N., Laumond, C. & Mauleon, H. 1996. The entomopathogenic nematodebacterium complex: biology, life cycle and vertebrate safety. *Biocontrol Science and Technology* 6, 333-346.
- Boemare, N. & Tailliez, P. 2009. Molecular approaches and techniques for the study of entomopathogenic bacteria. In *Insect pathogens: molecular approaches and techniques* (ed. S. P. Stock, J. Vanderberg, I. Glazer & N. Boemare), pp. 32-49. Wallingford, UK: CAB International.
- Boemare, N. E. & Akhurst, R. J. 1988. Biochemical and physiological characterization of colony form variants in Xenorhabdus spp (Enterobacteriaceae). *Journal of General Microbiology* **134**, 751-761.
- Boemare, N. E., Akhurst, R. J. & Mourant, R. G. 1993b. DNA Relatedness between Xenorhabdus spp. (Enterobacteriaceae), Symbiotic Bacteria of Entomopathogenic Nematodes, and a Proposal To Transfer Xenorhabdus luminescens to a New Genus, Photorhabdus gen. nov. *International Journal* of Systematic Bacteriology 43, 249-255.
- Boff, M. I. C., Wiegers, G. L., Gerritsen, L. J. M. & Smits, P. H. 2000. Development of the entomopathogenic nematode Heterorhabditis megidis strain NLH-E 87.3 in Galleria mellonella. *Nematology* **2**, 303-308.
- Boff, M. I. C., Zoon, F. C. & Smits, P. H. 2001. Orientation of Heterorhabditis megidis to insect hosts and plant roots in a Y-tube sand olfactometer. *Entomologia Experimentalis Et Applicata* **98**, 329-337.
- Bowen, D., Blackburn, M., Rocheleau, T., Grutzmacher, C. & ffrench-Constant, R.
   H. 2000. Secreted proteases from Photorhabdus luminescens: separation of the extracellular proteases from the insecticidal Tc toxin complexes. *Insect Biochemistry and Molecular Biology* 30, 69-74.
- Bowen, D., Rocheleau, T. A., Blackburn, M., Andreev, O., Golubeva, E., Bhartia, R.
  & ffrench-Constant, R. H. 1998. Insecticidal Toxins from the Bacterium Photorhabdus luminescens. *Science* 280, 2129-2132.
- Bowen, D. J. & Ensign, J. C. 1998. Purification and characterization of a highmolecular-weight insecticidal protein complex produced by the entomopathogenic bacterium *Photorhabdus luminescens*. Applied and Environmental Microbiology 64, 3029-3035.
- Brachmann, A., Joyce, S., Jenke-Kodoma, H., Schwar, G., Clarke, D. & Bode, H. 2007. A type II polyketide synthase is responsible for anthraquinone biosynthesis in Photorhabdus luminescens. *Chembiochem* 8, 1721 - 1728.
- Bright, M. & Bulgheresi, S. 2010. A complex journey: transmission of microbial symbionts. *Nat Rev Micro* 8, 218-230.

- Brillard, J., Duchaud, E., Boemare, N., Kunst, F. & Givaudan, A. 2002. The PhIA hemolysin from the entomopathogenic bacterium Photorhabdus luminescens belongs to the two-partner secretion family of hemolysins. *Journal of Bacteriology* 184, 3871-3878.
- Brillard, J., Ribeiro, C., Boemare, N., Brehélin, M. & Givaudan, A. 2001. Two Distinct Hemolytic Activities in *Xenorhabdus nematophila* Are Active against Immunocompetent Insect Cells. *Applied and Environmental Microbiology* 67, 2515-2525.
- Brivio, M. F., Pagani, M. & Restelli, S. 2002. Immune suppression of *Galleria* mellonella (Insecta, Lepidoptera) humoral defenses induced by Steinernema feltiae (Nematoda, Rhabditida): involvement of the parasite cuticle. Experimental Parasitology 101, 149-156.
- Broderick, N. A., Raffa, K. F., Goodman, R. M. & Handelsman, J. 2004. Census of the Bacterial Community of the Gypsy Moth Larval Midgut by Using Culturing and Culture-Independent Methods. *Applied and Environmental Microbiology* 70, 293-300.
- Brown, I. M. & Gaugler, R. 1996. Cold tolerance of steinernematid and heterorhabditid nematodes. *Journal of Thermal Biology* **21**, 115-121.
- Brown, I. M. & Gaugler, R. 1997. Temperature and Humidity Influence Emergence and Survival of Entomopathogenic Nematoles. *Nematologica* **43**, 363-375.
- Brown, S. E., Cao, A. T., Hines, E. R., Akhurst, R. J. & East, P. D. 2004. A novel secreted protein toxin from the insect pathogenic bacterium Xenorhabdus nematophila. *Journal of Biological Chemistry* **279**, 14595-14601.
- Byers, J. & Poinar Jr, G. 1982. Location of insect hosts by the nematode, Neoaplectana carpocapsae, in response to temperature. *Behaviour*, 1-10.
- Caldas, C., Cherqui, A., Pereira, A. & Simoes, N. 2002. Purification and characterization of an extracellular protease from Xenorhabdus nematophila involved in insect immunosuppression. *Applied and Environmental Microbiology* 68, 1297-1304.
- Campbell, J. F. & Gaugler, R. 1993. Nictation behaviour and its ecological implications in the host search strategies of entomopathogenic nematodes (Heterorhabditidae and Steinernematidae). *Behaviour*, 155-169.
- Campbell, J. F. & Kaya, H. K. 2002. Variation in entomopathogenic nematode (Steinernematidae and Heterorhabditidae) infective-stage jumping behaviour. *Nematology* **4**, 471-482.
- Campbell, J. F., Lewis, E., Yoder, F. & Gaugler, R. 1996. Entomopathogenic nematode (Heterorhabditidae and Steinernematidae) spatial distribution in turfgrass. *Parasitology* 113, 473-482.

- Campbell, L. R. & Gaugler, R. 1991a. Mechanisms for exsheathment of entomopathogenic nematodes. *International Journal for Parasitology* **21**, 219-224.
- Campbell, L. R. & Gaugler, R. 1991b. Role of the Sheath in Desiccation Tolerance of Two Entomopathogenic Nematodes. *Nematologica*, *37* **1**, 324-332.
- Campos-Herrera, R., Escuer, M., Labrador, S., Robertson, L., Barrios, L. & Gutiérrez, C. 2007. Distribution of the entomopathogenic nematodes from La Rioja (Northern Spain). *Journal of Invertebrate Pathology* **95**, 125-139.
- Campos Herrera, R., Barbercheck, M., How, C. W. & Stock, S. P. 2012. Entomopathogenic Nematodes as a Model System for Advancing the Frontiers of Ecology. *Journal of Nematology* **44**, 162-176.
- Cash, H. L., Whitham, C. V., Behrendt, C. L. & Hooper, L. V. 2006. Symbiotic bacteria direct expression of an intestinal bactericidal lectin. *Science* **313**, 1126-1130.
- Chapuis, É., Emelianoff, V., Paulmier, V., Le Brun, N., Pagès, S., Sicard, M. & Ferdy, J. B. 2009. Manifold aspects of specificity in a nematode–bacterium mutualism, vol. 22, pp. 2104-2117: Blackwell Publishing Ltd.
- Chase, J. M. & Leibold, M. A. 2003. *Ecological niches: linking classical and contemporary approaches:* University of Chicago Press.
- Chaston, J. & Goodrich-Blair, H. 2009. Common trends in mutualism revealed by model associations between invertebrates and bacteria. *FEMS Microbiology Reviews* **34**, 41-58.
- Chaston, J. M., Suen, G., Tucker, S. L., Andersen, A. W., Bhasin, A., Bode, E., Bode, H. B., Brachmann, A. O., Cowles, C. E., Cowles, K. N., Darby, C., de Léon, L., Drace, K., Du, Z., Givaudan, A., Herbert Tran, E. E., Jewell, K. A., Knack, J. J., Krasomil-Osterfeld, K. C., Kukor, R., Lanois, A., Latreille, P., Leimgruber, N. K., Lipke, C. M., Liu, R., Lu, X., Martens, E. C., Marri, P. R., Médigue, C., Menard, M. L., Miller, N. M., Morales-Soto, N., Norton, S., Ogier, J.-C., Orchard, S. S., Park, D., Park, Y., Qurollo, B. A., Sugar, D. R., Richards, G. R., Rouy, Z., Slominski, B., Slominski, K., Snyder, H., Tjaden, B. C., van der Hoeven, R., Welch, R. D., Wheeler, C., Xiang, B., Barbazuk, B., Gaudriault, S., Goodner, B., Slater, S. C., Forst, S., Goldman, B. S. & Goodrich-Blair, H. (2011). The entomopathogenic bacterial endosymbionts *Xenorhabdus* and *Photorhabdus*: Convergent lifestyles from divergent genomes. PLoS One 6(11): e27909. doi: 10.1371/journal.pone.0027909
- Chen, D. Q., Montllor, C. B. & Purcell, A. H. 2000. Fitness effects of two facultative endosymbiotic bacteria on the pea aphid, *Acyrthosiphon pisum*, and the blue alfalfa aphid, *A. kondoi. Entomologia Experimentalis Et Applicata* **95**, 315-323.

- Ciche, T. & Ensign, J. 2003. For the insect pathogen Photorhabdus luminescens, which end of a nematode is out? *Appl Environ Microbiol* **69**, 1890 1897.
- Ciche, T. A., Darby, C., Ehlers, R. U., Forst, S. & Goodrich-Blair, H. 2006. Dangerous liaisons: The symbiosis of entomopathogenic nematodes and bacteria. *Biological Control* **38**, 22-46.
- Ciche, T. A., Kim, K. S., Kaufmann-Daszczuk, B., Nguyen, K. C. Q. & Hall, D. H. 2008. Cell invasion and matricide during *Photorhabdus luminescens* transmission by *Heterorhabditis bacteriophora* nematodes. *Applied and Environmental Microbiology* 74, 2275-2287.
- Clarke, D. J. 2008. Photorhabdus: a model for the analysis of pathogenicity and mutualism. *Cellular Microbiology* **10**, 2159-2167.
- Clarke, D. J. & Dowds, B. C. A. 1995. Virulence Mechanisms of Photorhabdus sp. Strain K122 toward Wax Moth Larvae. *Journal of Invertebrate Pathology* 66, 149-155.
- Colinet, H. & Renault, D. 2014. Dietary live yeast alters metabolic profiles, protein biosynthesis and thermal stress tolerance of Drosophila melanogaster. *Comparative Biochemistry and Physiology Part A: Molecular & Integrative Physiology* 170, 6-14.
- Collins, C. H. & Lyne, P. M. 1984. Identification Methods. In *Microbiological methods* (ed. C. H. Collins & P. M. Lyne), pp. 102 113. London: Butterworths.
- Cowles, C. E. & Goodrich-Blair, H. 2008. The Xenorhabdus nematophila nilABC genes confer the ability of Xenorhabdus spp. to colonize Steinernema carpocapsae nematodes. *Journal of Bacteriology* **190**, 4121-4128.
- Cowles, K. N. & Goodrich-Blair, H. 2005. Expression and activity of a *Xenorhabdus nematophila* haemolysin required for full virulence towards Manduca sexta insects. *Cellular Microbiology* **7**, 209-219.
- Crawford, J. M., Kontnik, R. & Clardy, J. 2010. Regulating alternative lifestyles in entomopathogenic bacteria. *Current Biology* **20**, 69-74.
- Crawford, J. M., Portmann, C., Zhang, X., Roeffaers, M. B. J. & Clardy, J. 2012. Small molecule perimeter defense in entomopathogenic bacteria. *Proceedings of the National Academy of Sciences* **109**, 10821-10826.
- Crossan, J., Paterson, S. & Fenton, A. 2007. Host availability and the evolution of parasite life-history strategies. *Evolution* **61**, 675-684.
- Daly, G. 2010. The effects of environmental factors on *Heterorhabditis downesi* carrying a symbiotic purple bacteria and *Heterorhabditis downesi* carrying a symbiotic yellow bacteria B.Sc. Thesis, National University of Ireland, Maynooth, Ireland.

- Dauga, C. 2002. Evolution of the gyrB gene and the molecular phylogeny of Enterobacteriaceae: a model molecule for molecular systematic studies. *Int J Syst Evol Microbiol* **52**, 531-547.
- Dazzo, F. B. & Hubbell, D. H. 1975. Cross-reactive antigens and lectin as determinants of symbiotic specificity in the Rhizobium-clover association. *Applied microbiology* **30**, 1017-1033.
- Derzelle, S., Duchaud, E., Kunst, F., Danchin, A. & Bertin, P. 2002. Identification, Characterization, and Regulation of a Cluster of Genes Involved in Carbapenem Biosynthesis in Photorhabdus luminescens. *Appl. Environ. Microbiol.* 68, 3780-3789.
- Dillon, A. 2003. Biological control of the large pine weevil, Hylobius abietis L.,(Coleoptera: Curculionidae) using entomopathogenic nematodes Ph. D Thesis, National University of Ireland, Maynooth, Ireland,
- Dillon, A. B., Downes, M. J., Ward, D. & Griffin, C. T. 2007. Optimizing application of entomopathogenic nematodes to manage large pine weevil, Hylobius abietis L. (Coleoptera : Curculionidae) populations developing in pine stumps, Pinus sylvestris. *Biological Control* 40, 253-263.
- Dillon, A. B., Ward, D., Downes, M. J. & Griffin, C. T. 2006. Suppression of the large pine weevil Hylobius abietis (L.) (Coleoptera: Curculionidae) in pine stumps by entomopathogenic nematodes with different foraging strategies. *Biological Control* 38, 217-226.
- Dillon, R. & Charnley, A. 1988. Inhibition of Metarhizium anisopliae by the gut bacterial flora of the desert locust: characterisation of antifungal toxins. *Canadian Journal of Microbiology* **34**, 1075-1082.
- Dillon, R. & Dillon, V. 2004. The gut bacteria of insects: nonpathogenic interactions. *Annual Reviews in Entomology* **49**, 71-92.
- Dobson, T. 1974. Studies on the biology of the kelp-fly Coelopa in Great Britain. Journal of Natural History 8, 155-177.
- Douglas, A. 2009. The microbial dimension in insect nutritional ecology. *Functional Ecology* **23**, 38-47.
- Dubilier, N., Bergin, C. & Lott, C. 2008. Symbiotic diversity in marine animals: the art of harnessing chemosynthesis. *Nature Reviews Microbiology* **6**, 725-740.
- Duchaud, E., Rusniok, C., Frangeul, L., Buchrieser, C., Givaudan, A., Taourit, S., Bocs, S., Boursaux-Eude, C., Chandler, M., Charles, J. F., Dassa, E., Derose, R., Derzelle, S., Freyssinet, G., Gaudriault, S., Medigue, C., Lanois, A., Powell, K., Siguier, P., Vincent, R., Wingate, V., Zouine, M., Glaser, P., Boemare, N., Danchin, A. & Kunst, F. 2003. The genome sequence of the entomopathogenic bacterium *Photorhabdus luminescens*. *Nature Biotechnology* 21, 1307-1313.

- Easom, C., Joyce, S. & Clarke, D. 2010. Identification of genes involved in the mutualistic colonization of the nematode Heterorhabditis bacteriophora by the bacterium Photorhabdus luminescens. *BMC Microbiol* **10**, 45.
- Edgington, S., Buddie, A., Moore, D., France, A., Merino, L. & Hunt, D. 2011. Heterorhabditis atacamensis n. sp.(Nematoda: Heterorhabditidae), a new entomopathogenic nematode from the Atacama Desert, Chile. *Journal of Helminthology* 85, 381-394.
- Edward, D. A., Newton, J. & Gilburn, A. S. 2008. Investigating dietary preferences in two competing dipterans, Coelopa frigida and Coelopa pilipes, using stable isotope ratios of carbon and nitrogen. *Entomologia Experimentalis Et Applicata* **127**, 169-175.
- Egert, M., Stingl, U., Dyhrberg Bruun, L., Pommerenke, B., Brune, A. & Friedrich, M. W. 2005. Structure and Topology of Microbial Communities in the Major Gut Compartments of Melolontha melolontha Larvae (Coleoptera: Scarabaeidae). *Applied and Environmental Microbiology* **71**, 4556-4566.
- Egert, M., Wagner, B., Lemke, T., Brune, A. & Friedrich, M. W. 2003. Microbial community structure in midgut and hindgut of the humus-feeding larva of Pachnoda ephippiata (Coleoptera: Scarabaeidae). *Applied and Environmental Microbiology* **69**, 6659-6668.
- Ehlers, R.-U. & Niemann, I. 1998. Molecular Identification of Photorhabdus luminescens strains by Amplification of Specific Fragments of the 16S Ribosomal DNA. *Systematic and Applied Microbiology* **21**, 509-519.
- Eleftherianos, I., Boundy, S., Joyce, S. A., Aslam, S., Marshall, J. W., Cox, R. J., Simpson, T. J., Clarke, D. J., ffrench-Constant, R. H. & Reynolds, S. E. 2007. An antibiotic produced by an insect-pathogenic bacterium suppresses host defenses through phenoloxidase inhibition. *Proceedings of the National Academy of Sciences* 104, 2419-2424.
- Eleftherianos, I., Joyce, S., Ffrench-Constant, R., Clarke, D. & Reynolds, S. 2010. Probing the tri-trophic interaction between insects, nematodes and Photorhabdus. *Parasitology*, 1-12.
- Emelianoff, V., Chapuis, E., Le Brun, N., Chiral, M., Moulia, C. & Ferdy, J. B. 2008a. A survival-reproduction trade-off in entomopathogenic nematodes mediated by their bacterial symbionts. *Evolution* **62**, 932-942.
- Emelianoff, V., Le Brun, N., Pagès, S., Stock, S. P., Tailliez, P., Moulia, C. & Sicard, M. 2008b. Isolation and identification of entomopathogenic nematodes and their symbiotic bacteria from Hérault and Gard (Southern France). *Journal of Invertebrate Pathology* **98**, 211-217.
- Emelianoff, V., Sicard, M., Le Brun, N., Moulia, C. & Ferdy, J. B. 2007. Effect of bacterial symbionts Xenorhabdus on mortality of infective juveniles of two Steinernema species. *Parasitology Research* 100, 657-659.
- Enright, M. R. & Griffin, C. T. 2004. Specificity of Association between Paenibacillus spp. and the Entomopathogenic Nematodes, Heterorhabditis spp. *Microbial Ecology* **48**.
- Erkosar, B., Storelli, G., Defaye, A. & Leulier, F. 2013. Host-Intestinal Microbiota Mutualism: "Learning on the Fly". *Cell Host & Microbe* **13**, 8-14.
- Feldhaar, H. 2011. Bacterial symbionts as mediators of ecologically important traits of insect hosts. *Ecological Entomology* **36**, 533-543.
- ffrench-Constant, R. H. & Bowen, D. J. 2000. Novel insecticidal toxins from nematode-symbiotic bacteria. *Cellular and Molecular Life Sciences* **57**, 828-833.
- Fischer-Le Saux, M., Viallard, V., Brunel, B., Normand, P. & Boemare, N. E. 1999.
  Polyphasic classification of the genus Photorhabdus and proposal of new taxa: P. luminescens subsp. luminescens subsp. nov., P. luminescens subsp. akhurstii subsp. nov., P. luminescens subsp. laumondii subsp. nov., P. temperata sp. nov., P. temperata subsp. nov. and P. asymbiotica sp. nov. *Int J Syst Bacteriol* 49, 1645-1656.
- Flood, D. T. 1977. Historical evidence for the growth of North Bull Island. In North Bull Island Dublin Bay - a modern coastal natural history. (ed. D. W. Jeffrey), pp. 9-12. Dublin: The Royal Dublin Society.
- Forst, S. & Clarke, D. 2002. Bacteria-Nematode Symbiosis. In *Entomopathogenic Nematology* (ed. R. Gaugler), pp. 57 77. Wallingford, UK: CAB International.
- Forst, S., Dowds, B., Boemare, N. & Stackebrandt, E. 1997. Xenorhabdus and Photorhabdus spp.: Bugs that kill bugs. *Annual Review of Microbiology* **51**, 47-72.
- Fossitt, J. A. 2000. *A guide to habitats in Ireland*. Kilkenny: Heritage Council/Chomhairle Oidhreachta.
- Gardner, A. & West, S. A. 2004. Spite and the scale of competition. *Journal of Evolutionary Biology* **17**, 1195-1203.
- Gardner, A., West, S. A. & Buckling, A. 2004. Bacteriocins, spite and virulence. *Proceedings of the Royal Society of London. Series B: Biological Sciences* **271**, 1529-1535.
- Gaugler, R., Lebeck, L., Nakagaki, B. & Boush, G. 1980. Orientation of the entomogenous nematode *Neoaplectana carpocapsae* to carbon dioxide. *Environmental Entomology* **9**, 649-652.

- Gaugler, R., Lewis, E. & Stuart, R. J. 1997. Ecology in the service of biological control: the case of entomopathogenic nematodes. *Oecologia* **109**, 483-489.
- Georgis, R., Koppenhöfer, A. M., Lacey, L. A., Bélair, G., Duncan, L. W., Grewal, P. S., Samish, M., Tan, L., Torr, P. & van Tol, R. W. H. M. 2006. Successes and failures in the use of parasitic nematodes for pest control. *Biological Control* 38, 103-123.
- Gerhardt, P. M., R. G. E.; Costilow, R. N.; Nester, E. W.; Wood, W. A.; Krieg, N. R.; Phillips, G. B. (ed.) 1981. *Manual of Methods for General Bacteriology*. Washington, DC: American Society for Microbiology.
- Gerritsen, L. J. M. & Smits, P. H. 1993. Variation in pathogenicity of recombinations of *Heterorhabditis* and *Xenorhabdus luminescens* strains. *Fundamental and Applied Nematology* 16, 367-373.
- Gerritsen, L. J. M. & Smits, P. H. 1997. The influence of *Photorhabdus luminescens* strains and form variants on the reproduction and bacterial retention of *Heterorhabditis megidis*. *Fundamental and Applied Nematology* **20**, 317-322.
- Gerritsen, L. J. M., Vanderwolf, J. M., Vanvuurde, J. W. L., Ehlers, R. U., Krasomilosterfeld, K. C. & Smits, P. H. 1995. Polyclonal antisera to distinguish strains and form variants of Photorhabdus (Xenorhabdus) luminescens. *Applied and Environmental Microbiology* **61**, 284-289.
- Gerritsen, L. J. M., Wiegers, G. L. & Smits, P. H. 1998. Pathogenicity of new combinations of Heterorhabditis spp. and Photorhabdus luminescens against Galleria mellonella and Tipula oleracea. *Biological Control* **13**, 9-15.
- Glare, T., Jackson, T. & Zimmermann, G. 1993. Occurrence of *Bacillus popilliae* and two nematode pathogens in populations of *Amphimallon solstitialis* (Col.: Scarabaeidae) near Darmstadt, Germany. *Entomophaga* **38**, 441-450.
- Glazer, I. 1997. Effects of infected insects on secondary invasion of steinernematid entomopathogenic nematodes. *Parasitology* **114**, 597-604.
- Glazer, I. 2002. Survival biology. In *Entomopathogenic Nematology* (ed. R. Gaugler), pp. 169-187. Wallingford UK: CABI.
- Glazer, I. & Lewis, E. 2000. Bioassays for entomopathogenic nematodes. In Bioassays of entomopathogenic microbes and nematodes (ed. A. Navon & K. R. S. Ascher), pp. 229-247: Cabi.
- Gotz, P., Boman, A. & Boman, H. G. 1981. Interactions between Insect Immunity and an Insect-Pathogenic Nematode with Symbiotic Bacteria. *Proceedings of the Royal Society of London. Series B. Biological Sciences* **212**, 333-350.

- Gouge, D. H. & Snyder, J. L. 2006. Temporal association of entomopathogenic nematodes (Rhabditida: Steinernematidae and Heterorhabditidae) and bacteria. *Journal of Invertebrate Pathology* **91**, 147-157.
- Gray, M. W. & Doolittle, W. F. 1982. Has the endosymbiont hypothesis been proven? *Microbiological Reviews* **46**, 1.
- Grewal, P. S. 2005. One-on-one filter-paper bioassay. In *Quality control of entomopathogenic nematodes* (ed. J. M. Grunder), pp. 134. Wädenswil, Switzerland.: AGROSCOPE FAW.
- Grewal, P. S., Bornstein-Forst, S., Burnell, A. M., Glazer, I. & Jagdale, G. B. 2006. Physiological, genetic, and molecular mechanisms of chemoreception, thermobiosis, and anhydrobiosis in entomopathogenic nematodes. *Biological Control* 38, 54-65.
- Grewal, P. S., Ehlers, R. U. & Shapiro-Ilan, D. I. 2005. Nematodes as Biocontrol Agents: CABI Pub.
- Grewal, P. S., Matsuura, M. & Converse, V. 1997. Mechanisms of specificity of association between the nematode Steinernema scapterisci and its symbiotic bacterium. *Parasitology* 114, 483-488.
- Grewal, P. S., Selvan, S., Lewis, E. E. & Gaugler, R. 1993. Male insect-parasitic nematodes: a colonizing sex. *Experientia* **49**, 605-608.
- Griffin, C., Boemare, N. & Lewis, E. 2005. Biology and behaviour. In *Nematodes As Biocontrol Agents* (ed. P. S. Grewal, R. U. Ehlers & D. I. Shapiro-Ilan), pp. 47-64. Wallingford, UK: CABI Pub.
- Griffin, C., O'CALLAGHAN, K. & Dix, I. 2001. A self-fertile species of Steinernema from Indonesia: further evidence of convergent evolution amongst entomopathogenic nematodes? *Parasitology* **122**, 181-186.
- Griffin, C. T., Chaerani, R., Fallon, D., Reid, A. P. & Downes, M. J. 2000. Occurrence and distribution of the entomopathogenic nematodes Steinernema spp. and Heterorhabditis indica in Indonesia. *Journal of Helminthology* 74, 143-150.
- Griffin, C. T., Dix, I., Joyce, S. A., Burnell, A. M. & Downes, M. J. 1999. Isolation and characterisation of Heterorhabditis spp. (Nematoda : Heterorhabditidae) from Hungary, Estonia and Denmark. *Nematology* **1**, 321-332.
- Griffin, C. T., Downes, M. & Block, W. 1990. Tests of Antarctic soils for insect parasitic nematodes. *Antarctic Science* **2**, 221-222.
- Griffin, C. T., Joyce, S. A., Dix, I., Burnell, A. M. & Downes, M. J. 1994. Characterisation of the entomopathogenic nematode *Heterorhabditis* (Nematoda: Heterorhabditidae) from Ireland and Britain by molecular and

cross-breeding techniques, and the occurrence of the genus in these islands. *Fundamental and Applied Nematology* **17**, 245-253.

- Griffin, C. T., Moore, J. F. & Downes, M. J. 1991. Occurrence of insect-parasitic nematodes (Steinernematidae, Heterorhabditidae) in the republic of Ireland. *Nematologica* 37, 92-100.
- Grobler, J. 1980. Host selection and species preference of the red-billed oxpecker Buphagus erythrorhynchus in the Kruger National Park. *Koedoe* 23, 89-97.
- Gualtieri, M., Aumelas, A. & Thaler, J.-O. 2009. Identification of a new antimicrobial lysine-rich cyclolipopeptide family from Xenorhabdus nematophila. *J Antibiot* **62**, 295-302.
- Hallem, E. A., Dillman, A. R., Hong, A. V., Zhang, Y., Yano, J. M., DeMarco, S. F. & Sternberg, P. W. 2011. A Sensory Code for Host Seeking in Parasitic Nematodes. *Current Biology* 21, 377-383.
- Han, R. & Ehlers, R.-U. 2001. Effect of *Photorhabdus luminescens* phase variants on the in vivo and in vitro development and reproduction of the entomopathogenic nematodes *Heterorhabditis bacteriophora* and *Steinernema carpocapsae. FEMS Microbiology Ecology* 35, 239-247.
- Han, R. & Ehlers, R. U. 1999. Nutritional significance of *Photorhabdus* bacteria for the growth and reproduction of *Heterorhabditis* nematodes. In *Taxonomy*, *Phylogeny and Gnotobiological studies of Entomopathogenic Nematode Bacterium Complexes* (ed. N. Boemare, P. Richardson & F. Coudert), pp. 89-96. Luxembourg: Office for Official Publications of the EU.
- Han, R., Wouts, W. & Li, L. 1991. Development and virulence of *Heterorhabditis* spp. strains associated with different *Xenorhabdus luminescens* isolates. *Journal of Invertebrate Pathology* 58, 27-32.
- Han, R., Wouts, W. M. & Li, L. 1990. Development of *Heterorhabditis* spp strains as characteristics of possible *Xenorhabdus luminescens* ssp. *Revue de Nematologie* 13, 411-416.
- Han, R. C. & Ehlers, R. U. 1998. Cultivation of axenic *Heterorhabditis* spp. dauer juveniles and their response to non-specific *Photorhabdus luminescens* food signals. *Nematologica* 44, 425-435.
- Han, R. C. & Ehlers, R. U. 2000. Pathogenicity, development, and reproduction of *Heterorhabditis bacteriophora* and *Steinernema carpocapsae* under axenic in vivo conditions. *Journal of Invertebrate Pathology* 75, 55-58.
- Hao, Y.-J., Montiel, R., Abubucker, S., Mitreva, M. & Simões, N. 2010. Transcripts analysis of the entomopathogenic nematode *Steinernema carpocapsae* induced *in vitro* with insect haemolymph. *Molecular and Biochemical Parasitology* 169, 79-86.

- Hao, Y. J., Montiel, R., Lucena, M. A., Costa, M. & Simoes, N. 2012. Genetic diversity and comparative analysis of gene expression between *Heterorhabditis bacteriophora* Az29 and Az36 isolates: Uncovering candidate genes involved in insect pathogenicity. *Experimental Parasitology* 130, 116-125.
- Harris, C. R. 1977. Sedimentology and Geomorphology. In North Bull Island Dublin Bay - a modern coastal natural history. (ed. D. W. Jeffrey), pp. 13-25. Dublin: The Royal Dublin Society.
- Hass, B., Downes, M. J. & Griffin, C. T. 2001. Correlation between survival in water and persistence of infectivity in soil of Heterorhabditis spp. isolates. *Nematology* **3**, 573-579.
- Hawlena, H., Bashey, F. & Lively, C. M. 2010a. The evolution of spite: population structure and bacteriocin-mediated antagonism in two natural populations of xenorhabdus bacteria. *Evolution* **64**, 3198-3204.
- Hawlena, H., Bashey, F., Mendes-Soares, H. & Lively, C. M. 2010b. Spiteful Interactions in a Natural Population of the Bacterium Xenorhabdus bovienii. *American Naturalist* **175**, 374-381.
- Hay, D. B. & Fenlon, J. S. 1995. A modified binomial model that describes the infection dynamics of the entomopathogenic nematode Steinernema feltiae (Steinernematidae; Nematoda). *Parasitology* **111**, 627-633.
- Hazir, S., Stackebrandt, E., Lang, E., Schumann, P., Ehlers, R.-U. & Keskin, N. 2004. Two new Subspecies of Photorhabdus luminescens, Isolated from Heterorhabditis bacteriophora (Nematoda: Heterorhabditidae): Photorhabdus luminescens subsp. kayaii subsp. nov. and Photorhabdus luminescens subsp. thracensis subsp. nov. Systematic and Applied Microbiology 27, 36-42.
- Healy, B. 1975. Fauna of salt-marsh, North Bull Island, Dublin. Proceedings of the Royal Irish Academy Section B-Biological Geological and Chemical Science 75, 225-244.
- Heungens, K., Cowles, C. E. & Goodrich-Blair, H. 2002. Identification of Xenorhabdus nematophila genes required for mutualistic colonization of Steinernema carpocapsae nematodes. *Molecular Microbiology* 45, 1337-1353.
- Hinchliffe, S. J., Hares, M. C. & Dowling, A. J. 2010. Insecticidal toxins from the *Photorhabdus* and *Xenorhabdus* bacteria. *Open Toxinology Journal* 3, 101-118.
- Hirsch, J., Strohmeier, S., Pfannkuchen, M. & Reineke, A. 2012. Assessment of bacterial endosymbiont diversity in Otiorhynchus spp.(Coleoptera: Curculionidae) larvae using a multitag 454 pyrosequencing approach. BMC Microbiology 12, S6.

- Hodgson, M. M., Day, B., White, D. J. & Tisa, L. S. 2003. Effect of growth conditions on the motility of Photorhabdus temperata. Archives of Microbiology 180, 17-24.
- Hodson, A., Friedman, M., Wu, L. & Lewis, E. 2011. European earwig (*Forficula auricularia*) as a novel host for the entomopathogenic nematode *Steinernema carpocapsae*. Journal of Invertebrate Pathology **107**, 60-64.
- Hominick, W. M. 2002. Biogeography. In *Entomopathogenic Nematology* (ed. R. Gaugler), pp. 115-143. Wallingford, UK: CABI Publishing UK.
- Hominick, W. M. & Briscoe, B. R. 1990. Occurrence of entomopathogenic nematodes (rhabditida, steinernematidae and heterorhabditidae) in british soils. *Parasitology* 100, 295-302.
- Hominick, W. M., Briscoe, B. R., del Pino, F. G., Heng, J., Hunt, D. J., Kozodoy, E., Mracek, Z., Nguyen, K. B., Reid, A. P., Spiridonov, S., Stock, P., Sturhan, D., Waturu, C. & Yoshida, M. 1997. Biosystematics of entomopathogenic nematodes: current status, protocols and definitions. *Journal of Helminthology* **71**, 271-298.
- Hominick, W. M., Reid, A. P. & Briscoe, B. R. 1995. Prevalence and habitat specificity of Steinernematid and Heterorhabditid nematodes isolated during soil surveys of the UK and the Netherlands. *Journal of Helminthology* 69, 27-32.
- Hoy, C. W., Grewal, P. S., Lawrence, J. L., Jagdale, G. & Acosta, N. 2008. Canonical correspondence analysis demonstrates unique soil conditions for entomopathogenic nematode species compared with other free-living nematode species. *Biological Control* 46, 371-379.
- Hu, K., Li, J., Wang, W., Wu, H., Lin, H. & Webster, J. M. 1998. Comparison of metabolites produced in vitro and in vivo by Photorhabdus luminescens, a bacterial symbiont of the entomopathogenic nematode Heterorhabditis megidis. *Canadian Journal of Microbiology* 44, 1072-1077.
- Hu, K., Li, J. & Webster, J. M. 1999. Nematicidal metabolites produced by Photorhabdus luminescens (Enterobacteriaceae), bacterial symbiont of entomopathogenic nematodes. *Nematology* 1, 457-469.
- Hu, K. & Webster, J. M. 2000. Antibiotic production in relation to bacterial growth and nematode development in *Photorhabdus-Heterorhabditis* infected *Galleria mellonella* larvae. *Fems Microbiology Letters* **189**, 219-223.
- Huang, X. Q. & Madan, A. 1999. CAP3: A DNA sequence assembly program. *Genome Research* 9, 868-877.
- Huiskes, A. H. L. 1979. Ammophila Arenaria (L.) Link (Psamma Arenaria (L.) Roem. et Schult.; Calamgrostis Arenaria (L.) Roth). *Journal of Ecology* 67, 363-382.

- Hurek, T. & Reinhold-Hurek, B. 2003. *Azoarcus* sp. strain BH72 as a model for nitrogen-fixing grass endophytes. *Journal of Biotechnology* **106**, 169-178.
- Inglis, R. F., Gardner, A., Cornelis, P. & Buckling, A. 2009. Spite and virulence in the bacterium *Pseudomonas aeruginosa*. *Proceedings of the National Academy of Sciences of the United States of America* **106**, 5703-5707.
- Inglis, R. F., Roberts, P. G., Gardner, A. & Buckling, A. 2011. Spite and the scale of competition in *Pseudomonas aeruginosa*. *American Naturalist* 178, 276-285.
- Jang, E. K., Ullah, I., Lim, J. H., Lee, I. J., Kim, J. G. & Shin, J. H. 2012. Physiological and molecular characterization of a newly identified entomopathogenic bacteria, *Photorhabdus temperata* M1021. *Journal of Microbiology and Biotechnology* 22, 1605-1612.
- Janson, E. M., Stireman, J. O., Singer, M. S. & Abbot, P. 2008. Phytophagous insect-microbe mutualisms and adaptive evolutionary diversification. *Evolution* **62**, 997-1012.
- Jarosz, J. 1996. Ecology of anti-microbials produced by bacterial associates of Steinernema carpocapsae and Heterorhabditis bacteriophora. *Parasitology* **112**, 545-552.
- Jarosz, J. 1998. Active resistance of entomophagous rhabditid Heterorhabditis bacteriophora to insect immunity. *Parasitology* **117**, 201-208.
- Jeffrey, D. W. 1977. Comparison of the habitats. In North Bull Island Dublin Bay a modern coastal natural history (ed. D. W. Jeffrey), pp. 26-31. Dublin: RDS.
- Johnigk, S.-A. & Ehlers, R.-U. 1999. Endotokia matricida in hermaphrodites of Heterorhabditis spp. and the effect of the food supply. *Nematology* **1**, 717-726.
- Jonasson, M. 1987. Fish cleaning behaviour of shrimp. *Journal of Zoology* **213**, 117-131.
- Joyce, S., Brachmann, A., Glazer, I., Lango, L., Schwar, G., Clarke, D. & Bode, H. 2008. Bacterial biosynthesis of a multipotent stilbene. *Angewandte Chemie* **47**, 1942 1945.
- Joyce, S. A., Burnell, A. M. & Powers, T. O. 1994. Characterization of Heterorhabditis isolates by pcr amplification of segments of mtDNA and rDNA genes. *Journal of Nematology* **26**, 260-270.
- Kaya, H. K., Aguillera, M. M., Alumai, A., Choo, H. Y., de la Torre, M., Fodor, A., Ganguly, S., HazIr, S., Lakatos, T., Pye, A., Wilson, M., Yamanaka, S., Yang, H. & Ehlers, R. U. 2006. Status of entomopathogenic nematodes and

their symbiotic bacteria from selected countries or regions of the world. *Biological Control* **38**, 134-155.

- Kaya, H. K. & Gaugler, R. 1993. Entomopathogenic Nematodes. Annual Review of Entomology 38, 181-206.
- Kaya, H. K. & Stock, P. S. 1997. Techniques in insect nematology. In *Manual of Techniques in Insect Pathology* (ed. L. A. Lacey), pp. 281-324. London: Academic Press.
- Kennedy, S. 2011. Niche separation of variants of the bacterial symbiont carried by an entomopathogenic nematode. B.Sc. Thesis, National University of Ireland, Maynooth, Ireland.
- Kim, Y., Ji, D., Cho, S. & Park, Y. 2005. Two groups of entomopathogenic bacteria, Photorhabdus and Xenorhabdus, share an inhibitory action against phospholipase A2 to induce host immunodepression. J Invertebr Pathol 89, 258 - 264.
- Kimura, M. 1980. A simple method for estimating evolutionary rates of base substitutions through comparative studies of nucleotide sequences. *Journal of Molecular Evolution* **16**, 111-120.
- Koppenhöfer, A. M., Baur, M. E., Stock, S. P., Choo, H. Y., Chinnasri, B. & Kaya, H. K. 1997. Survival of entomopathogenic nematodes within host cadavers in dry soil. *Applied Soil Ecology* 6, 231-240.
- Koppenhöfer, A. M. & Fuzy, E. M. 2006. Effect of soil type on infectivity and persistence of the entomopathogenic nematodes Steinernema scarabaei, Steinernema glaseri, Heterorhabditis zealandica, and Heterorhabditis bacteriophora. *Journal of Invertebrate Pathology* 92, 11-22.
- Koppenhöfer, A. M. & Fuzy, E. M. 2007. Soil moisture effects on infectivity and persistence of the entomopathogenic nematodes Steinernema scarabaei, Sglaseri, Heterorhabditis zealandica, and H-bacteriophora. *Applied Soil Ecology* 35, 128-139.
- Koppenhöfer, H. S. & Gaugler, R. 2009. Entomopathogenic nematode and bacteria mutualism. In *Defensive mutualism in microbial symbiosis (eds JF White & MS Torres)* (ed. J. F. White & M. S. Torres), pp. 99-116. Boca Raton, FI: CRC Press.
- Kung, S.-P., Gaugler, R. & Kaya, H. K. 1990. Soil type and entomopathogenic nematode persistence. *Journal of Invertebrate Pathology* **55**, 401-406.
- Kung, S.-P., Gaugler, R. & Kaya, H. K. 1991. Effects of soil temperature, moisture, and relative humidity on entomopathogenic nematode persistence. *Journal of Invertebrate Pathology* 57, 242-249.

- Kuwata, R., Yoshiga, T., Yoshida, M. & Kondo, E. 2008. Mutualistic association of *Photorhabdus asymbiotica* with Japanese heterorhabditid entomopathogenic nematodes. *Microbes and Infection* **10**, 734-741.
- Lakatos, T. & Tóth, T. 2006. Biological control of European cockchafer larvae (Melolontha melolontha L.)-preliminary results. *Journal of Fruit and Ornamental Plant Research* 14, 73.
- Lanois, A., Pages, S., Bourot, S., Canoy, A., Givaudan, A. & Gaudriault, S. 2010. Transcriptional analysis of a Photorhabdus variant reveals that phenotypic variation is under transcriptional control and that pathogenicity in insects is multifactorial. *Appl Environ Microbiol*.
- Larkin, M. A., Blackshields, G., Brown, N. P., Chenna, R., McGettigan, P. A., McWilliam, H., Valentin, F. & Higgins, D. G. 2007. Clustal W and Clustal X version 2.0 *Bioinformatics* 23, 2947-2948.
- Lawrence, J. L., Hoy, C. W. & Grewal, P. S. 2006. Spatial and temporal distribution of endemic entomopathogenic nematodes in a heterogeneous vegetable production landscape. *Biological Control* **37**, 247-255.
- Lee, M. M. & Stock, S. P. 2010. A multilocus approach to assessing co-evolutionary relationships between *Steinernema* spp. (Nematoda: Steinernematidae) and their bacterial symbionts *Xenorhabdus* spp. (γ–Proteobacteria: Enterobacteriaceae). *Systematic Parasitology* **77**, 1-12.
- Leung, T. & Poulin, R. 2008. Parasitism, commensalism, and mutualism: exploring the many shades of symbioses. *Vie et Milieu* **58**, 107.
- Lewin, R. A. 1982. Symbiosis and Parasitism: Definitions and Evaluations. *BioScience* **32**, 254-260.
- Lewis, E. 2002. Behavioural Ecology. In *Entomopathogenic nematology* (ed. R. Gaugler), pp. 205-223. Wallingford, UK: CABI Publ.
- Lewis, E. E., Campbell, J., Griffin, C., Kaya, H. & Peters, A. 2006. Behavioral ecology of entomopathogenic nematodes. *Biological Control* **38**, 66-79.
- Lewis, E. E. & Clarke, D. J. 2012. Nematode Parasites and Entomopathogens. In Insect Pathology (Second Edition) (ed. F. E. Vega & H. K. Kaya), pp. 395-424. San Diego: Academic Press.
- Lewis, E. E., Selvan, S., Campbell, J. F. & Gaugler, R. 1995. Changes in foraging behaviour during the infective stage of entomopathogenic nematodes. *Parasitology* 110, 583-590.
- Ley, R. E., Lozupone, C. A., Hamady, M., Knight, R. & Gordon, J. I. 2008. Worlds within worlds: evolution of the vertebrate gut microbiota. *Nature Reviews Microbiology* 6, 776-788.

- Li, J., Chen, G. & Webster, J. M. 1997. Nematophin, a novel antimicrobial substance produced by Xenorhabdus nematophilus (Enterobactereaceae). *Canadian Journal of Microbiology* 43, 770-773.
- Li, J., Chen, G., Wu, H. & Webster, J. M. 1995. Identification of two pigments and a hydroxystilbene antibiotic from Photorhabdus luminescens. *Applied and Environmental Microbiology* **61**, 4329-33.
- Lola-Luz, T., Downes, M. & Dunne, R. 2005. Control of Black Vine Weevil larvae Otiorhynchus sulcatus (Fabricius) (Coleoptera:Curculionidae) in grow bags outdoors with nematodes, vol. 7, pp. 121-126: Blackwell Science Ltd.
- Malan, A. P., Nguyen, K. B., De Waal, J. Y. & Tiedt, L. 2008. Heterorhabditis safricana n. sp (Rhabditida : Heterorhabditidae), a new entomopathogenic nematode from South Africa. *Nematology* **10**, 381-396.
- Maneesakorn, P., An, R. S., Daneshvar, H., Taylor, K., Bai, X. D., Adams, B. J., Grewal, P. S. & Chandrapatya, A. 2011. Phylogenetic and cophylogenetic relationships of entomopathogenic nematodes (*Heterorhabditis*: Rhabditida) and their symbiotic bacteria (*Photorhabdus*: Enterobacteriaceae). *Molecular Phylogenetics and Evolution* 59, 271-280.
- Martens, E. C. & Goodrich-Blair, H. 2005. The Steinernema carpocapsae intestinal vesicle contains a subcellular structure with which Xenorhabdus nematophila associates during colonization initiation. Cellular Microbiology 7, 1723-1735.
- Martens, E. C., Heungens, K. & Goodrich-Blair, H. 2003. Early Colonization Events in the Mutualistic Association between Steinernema carpocapsae Nematodes and Xenorhabdus nematophila Bacteria. *J. Bacteriol.* **185**, 3147-3154.
- Massey, R. C., Buckling, A. & ffrench-Contstant, R. 2004. Interference competition and parasite virulence. *Proceedings of the Royal Society B: Biological Sciences* 271, 785.
- Mauleon, H., Denon, D. & Briand, S. 2006. Spatial and temporal distribution of *Heterorhabditis indica* in their natural habitats of Guadeloupe. *Nematology* **8**, 603-617.
- McCorry, M. & Ryle, T. 2009. A management plan for North Bull Island. Dublin: Dublin City Council.
- McInerney, B. V., Taylor, W. C., Lacey, M. J., Akhurst, R. J. & Gregson, R. P. 1991. Biologically active metabolites from *Xenorhabdus* spp., Part 2. Benzopyran-1-one derivatives with gastroprotective activity. *Journal of Natural Products* 54, 785-795.
- McLachlan, A. & Brown, A. C. 2006. *The Ecology of Sandy Shores*. London, UK: Elsevier Science.

- Menti, H., Wright, D. J. & Perry, R. N. 1997. Desiccation survival of populations of the entomopathogenic nematodes *Steinernema feltiae* and *Heterorhabditis megidis* from Greece and the UK. *Journal of Helminthology* **71**, 41-46.
- Miduturi, J., Moens, M., Hominick, W., Briscoe, B. & Reid, A. 1996. Naturally occurring entomopathogenic nematodes in the province of West-Flanders, Belgium. *Journal of Helminthology* **70**, 319-327.
- Mitani, D. K., Kaya, H. K. & Goodrich-Blair, H. 2004. Comparative study of the entomopathogenic nematode, Steinernema carpocapsae, reared on mutant and wild-type Xenorhabdus nematophila. *Biological Control* **29**, 382-391.
- Molyneux, A. & Bedding, R. 1984. Influence of soil texture and moisture on the infectivity of Heterorhabditis sp. D1 and Steinernema glaseri for larvae of the sheep blowfly, Lucilia cuprina. *Nematologica* **30**, 358-365.
- Moore, J. J. 1977. Vegetation of the dune comlex. In *North Bull Island Dublin Bay a modern coastal natural history*. (ed. D. W. Jeffrey), pp. 104 106. Dublin: The Royal Dublin Society.
- Morales-Soto, N. & Forst, S. A. 2011. The xnp1 P2-like tail synthesis gene cluster encodes xenorhabdicin and is required for interspecies competition. J Bacteriol 193, 3624-32.
- Moran, N. A. 2006. Symbiosis. Current Biology 16, R866-R871.
- Moran, N. A. 2007. Symbiosis as an adaptive process and source of phenotypic complexity. *Proceedings of the National Academy of Sciences* **104**, 8627-8633.
- Morton, A. & García-del-Pino, F. 2009. Ecological characterization of entomopathogenic nematodes isolated in stone fruit orchard soils of Mediterranean areas. *Journal of Invertebrate Pathology* **102**, 203-213.
- Mrácek, Z., Becvár, S., Kindlmann, P. & Jersáková, J. 2005. Habitat preference for entomopathogenic nematodes, their insect hosts and new faunistic records for the Czech Republic. *Biological Control* 34, 27-37.
- Mráček, Z., Bečvář, S., Kindlmann, P. & Webster, J. M. 1998. Infectivity and Specificity of Canadian and Czech Isolates of Steinernema Kraussei (Steiner, 1923) To Some Insect Pests At Low Temperatures in the Laboratory. *Nematologica* 44, 437-448.
- Mracek, Z. & Webster, J. M. 1993. Survey of Heterorhabditidae and Steinernematidae (Rhabditida, Nematoda) in Western Canada. *Journal of Nematology* 25, 710.
- Mukuka, J., Strauch, O. & Ehlers, R.-U. 2010. Variability in desiccation tolerance among different strains of the entomopathogenic nematode Heterorhabditis bacteriophora. *Nematology* **12**, 711-720.

- Nguyen, K. B. & Hunt, D. J. 2007. Heterorhabditidae: species descriptions. In Entomopathogenic Nematodes: Systematics, Phylogeny and Bacterial Symbionts (ed. K. B. Nguyen & D. J. Hunt), pp. 611-692. Leiden, The Netherlands: Brill.
- Nguyen, K. B., Shapiro-Ilan, D. I. & Mbata, G. N. 2008. *Heterorhabditis georgiana* n. sp. (Rhabditida: Heterorhabditidae) from Georgia, USA. *Nematology* **10**, 433-448.
- Nguyen, K. B., Sharpiro-Ilan, D. I., Stuart, R. J., McCoy, C. W., James, R. R. & Adams, B. J. 2004. Heterorhabditis mexicana n. sp. (Rhabditida: Heterorhabditidae) from Tamaulipas, Mexico, and morphological studies of the bursa of Heterorhabditis spp. *Nematology* 6, 231-244.
- Nielsen, O. & Lübeck, P. S. 2002. Characterisation of symbionts of entomopathogenic nematodes by universally primed-PCR (UP-PCR) and UP-PCR product cross-hybridisation. *Fems Microbiology Letters* **215**, 63-68.
- Nordbringhertz, B. & Mattiasson, B. 1979. Action of a nematode-trapping fungus shows lectin-mediated host-microorganism interaction. *Nature* **281**, 477-479.
- Noujeim, E., Khater, C., Pages, S., Ogier, J.-C., Tailliez, P., Hamze, M. & Thaler, O. 2011. The first record of entomopathogenic nematodes (Rhabiditiae: Steinernematidae and Heterorhabditidae) in natural ecosystems in Lebanon: A biogeographic approach in the Mediterranean region. *Journal of Invertebrate Pathology* 107, 82-85.
- O'Connor, J. P. & Speight, M. C. D. 2002. Neoleria maritima (Villeneuve) (Dipt., Heleomyzidae) new to Ireland. *Entomologist's Monthly Magazine* **138**, 32.
- O'Leary, S., Power, A., Stack, C. & Burnell, A. 2001. Behavioural and physiological responses of infective juveniles of the entomopathogenic nematode Heterorhabditis to desiccation. *Biocontrol* **46**, 345-362.
- Oliver, K. M., Degnan, P. H., Burke, G. R. & Moran, N. A. 2010. Facultative Symbionts in Aphids and the Horizontal Transfer of Ecologically Important Traits. *Annual Review of Entomology* **55**, 247-266.
- Orozco, R. A. 2012. Characterization and Phylogenetic Relationships of Photorhabdus luminescens subsp. sonorensis (gamma-Proteobacteria: Enterobacteriaceae), the Bacterial Symbiont of the Entomopathogenic Nematode Heterorhabditis sonorensis (Nematoda: Heterorhabditidae). *Current Microbiology* **66**, 30-39.
- Paine, T., Raffa, K. & Harrington, T. 1997. Interactions among scolytid bark beetles, their associated fungi, and live host conifers. *Annual Review of Entomology* 42, 179-206.

- Pankewitz, F. & Hilker, M. 2008. Polyketides in insects: ecological role of these widespread chemicals and evolutionary aspects of their biogenesis. *Biological Reviews* 83, 209-226.
- Paracer, S. & Ahmadjian, V. 2000. Symbiosis: an introduction to biological associations. Cary, NC, USA: Oxford University Press.
- Park, D., Ciezki, K., Van Der Hoeven, R., Singh, S., Reimer, D., Bode, H. B. & Forst, S. 2009. Genetic analysis of xenocoumacin antibiotic production in the mutualistic bacterium Xenorhabdus nematophila. *Molecular Microbiology* 73, 938-949.
- Park, Y., Herbert, E. E., Cowles, C. E., Cowles, K. N., Menard, M. L., Orchard, S. S. & Goodrich†Blair, H. 2007. Clonal variation in Xenorhabdus nematophila virulence and suppression of Manduca sexta immunity. *Cellular Microbiology* 9, 645-656.
- Parniske, M. 2008. Arbuscular mycorrhiza: the mother of plant root endosymbioses. *Nature Reviews Microbiology* **6**, 763-775.
- Peat, S. M., ffrench-Constant, R. H., Waterfield, N. R., Marokházi, J., Fodor, A. & Adams, B. J. 2010. A robust phylogenetic framework for the bacterial genus Photorhabdus and its use in studying the evolution and maintenance of bioluminescence: A case for 16S, gyrB, and glnA. *Molecular Phylogenetics* and Evolution 57, 728-740.
- Peters, A. 1996. The natural host range of *Steinernema* and *Heterorhabditis* spp. and their impact on insect populations. *Biocontrol Science and Technology* **6**, 389-402.
- Peters, A. 2005. Single dose infectivity test using larvae of the lesser mealworm, *Tenebrio molitor*: Insects grouped in one arena. In *Quality control of entomopathogenic nematodes* (ed. J. M. Grunder). Wädenswil, Switzerland: AGROSCOPE FAW.
- Peters, A. & Ehlers, R.-U. 1994. Susceptibility of Leatherjackets (Tipula paludosa and Tipula oleracea; Tipulidae; Nematocera) to the Entomopathogenic Nematode Steinernema feltiae. *Journal of Invertebrate Pathology* 63, 163-171.
- Poinar, G. & Grewal, P. 2012. History of Entomopathogenic Nematology. *Journal of Nematology* 44, 153.
- Poinar, G. O. 1976. Description and biology of a new insect parasitic rhabditoid, Heterorhabditis bacteriophora n. gen., n. sp.(Rhabditida; Heterorhabditidae n. fam.). *Nematologica* **21**, 463-470.
- Poinar, G. O. 1979. Nematodes for Biological Control of Insects. Boca Raton, Fl: CRC Press, Inc.

- Poinar, G. O. 1993. Origins and phylogenetic relationships of the entomophilic rhabditis, *Heterorhabditis* and *Steinernema*. *Fundamental and Applied Nematology* **16**, 333-338.
- Powers, T. O., Todd, T. C., Burnell, A. M., Murray, P. C. B., Fleming, C. C., Szalanski, A. L., Adams, B. A. & Harris, T. S. 1997. The rDNA internal transcribed spacer region as a taxonomic marker for nematodes. *Journal of Nematology* 29, 441-450.
- Půža, V. & Mráček, Z. 2005. Seasonal dynamics of entomopathogenic nematodes of the genera Steinernema and Heterorhabditis as a response to abiotic factors and abundance of insect hosts. *Journal of Invertebrate Pathology* 89, 116-122.
- Půža, V. & Mráček, Z. 2007. Natural population dynamics of entomopathogenic nematode Steinernema affine (Steinernematidae) under dry conditions: Possible nematode persistence within host cadavers? *Journal of Invertebrate Pathology* 96, 89-92.
- Půža, V. & Mráček, Z. 2010. Does scavenging extend the host range of entomopathogenic nematodes (Nematoda: Steinernematidae)? *Journal of Invertebrate Pathology* **104**, 1-3.
- Qiu, L. & Bedding, R. A. 2002. Characteristics of protectant synthesis of infective juveniles of Steinernema carpocapsae and importance of glycerol as a protectant for survival of the nematodes during osmotic dehydration. *Comparative Biochemistry and Physiology Part B: Biochemistry and Molecular Biology* 131, 757-765.
- Qiu, L., Lacey, M. J. & Bedding, R. A. 2000. Permeability of the infective juveniles of Steinernema carpocapsae to glycerol during osmotic dehydration and its effect on biochemical adaptation and energy metabolism. *Comparative Biochemistry and Physiology Part B: Biochemistry and Molecular Biology* 125, 411-419.
- Qiu, X., Yan, X., Liu, M. & Han, R. (2012). Genetic and Proteomic Characterization of *rpoB* Mutations and Their Effect on Nematicidal Activity in *Photorhabdus luminescens* LN2. PLoS One 7(8): e43114. doi: 10.1371/journal.pone.0043114
- Ramos-Rodriguez, O., Campbell, J. F., Christen, J. M., Shapiro-Ilan, D. I., Lewis, E. E. & Ramaswamy, S. B. 2007. Attraction behaviour of three entomopathogenic nematode species towards infected and uninfected hosts. *Parasitology* 134, 729-738.
- Ranwell, D. S. 1972. *Ecology of salt marshes and sand dunes*: Chapman and Hall London.

- Rasmann, S., Köllner, T. G., Degenhardt, J., Hiltpold, I., Toepfer, S., Kuhlmann, U., Gershenzon, J. & Turlings, T. C. 2005. Recruitment of entomopathogenic nematodes by insect-damaged maize roots. *Nature* 434, 732-737.
- Regeai, S. O. M. & Burnell, A. M. 2006. Novel mitochondrial gene arrangement among some species of the Genus *Heterorhabditis*. unpublished.
- Richards, G. R. & Goodrich-Blair, H. 2010. Examination of Xenorhabdus nematophila lipases in pathogenic and mutualistic host interactions reveals a role for xlpA in nematode progeny production. *Applied and Environmental Microbiology* **76**, 221-229.
- Richards, L. & Weaver, L. 1944. Moisture retention by some irrigated soils as related to soil moisture tension. J. agric. Res 69, 215-235.
- Richardson, W. H., Schmidt, T. M. & Nealson, K. H. 1988. Identification of an anthraquinone pigment and a hydroxystilbene antibiotic from Xenorhabdus luminescens. *Applied and Environmental Microbiology* **54**, 1602-1605.
- Robertson, G. P., Sollins, P., Ellis, B. G. & Lajtha, K. 1999. Exchangeable ions, pH, and cation exchange capacity. In *Standard Soil Methods for Long-Term Ecological Research* (ed. G. P. robertson, D. C. Coleman, C. S. Bledsoe & P. Sollins), pp. 106-114. Cary, NC, USA: Oxford University Press, USA.
- Rodwell, J. S., Birks, H. J. B. & Malloch, A. J. C. 2000. Shingle, strandline and sand-dune communities. In *British Plant Communities*, vol. 5 (ed. J. S. Rodwell), pp. 113-250. Cambridge, U.K.: Cambridge University Press.
- Rolston, A., Downes, M. & Griffin, C. 2002. Entomopathogenic nematode occurrence in an Irish dune system. In *Nematology*, vol. 4, pp. 196-197. Tenerife.
- Rolston, A. N., Griffin, C. T. & Downes, M. J. 2005. Distribution of entomopathogenic nematodes in an Irish sand dune system. *Nematology* 7, 259-266.
- Rolston, A. N., Meade, C. & Downes, M. J. 2006. Analysis of populations of the entomopathogenic nematodes *steinernema feltiae* and *Heterorhabditis downesi* (Steinernematidae, Heterorhabditidae) from Bull Island, Ireland, using Amplified Fragment Length Polymorphism (AFLP). unpublished.
- Rosa, J., Bonifassi, E., Amaral, J., Lacey, L., Simoes, N. & Laumond, C. 2000. Natural occurrence of entomopathogenic nematodes (Rhabditida: Steinernema, Heterorhabditis) in the Azores. *Journal of Nematology* **32**, 215.
- Rosner, B. M., Ensign, J. C. & Schink, B. 1996. Anaerobic metabolism of primary and secondary forms of Photorhabdus luminescens. *Fems Microbiology Letters* 140, 227-232.

- Ryder, J. J. & Griffin, C. T. 2002. Density-dependent fecundity and infective juvenile production in the entomopathogenic nematode, Heterorhabditis megidis. *Parasitology* 125, 83-92.
- San-Blas, E. & Gowen, S. R. 2008. Facultative scavenging as a survival strategy of entomopathogenic nematodes. *International Journal for Parasitology* 38, 85-91.
- San-Blas, E., Gowen, S. R. & Pembroke, B. 2008. Scavenging or infection? Possible host choosing by entomopathogenic nematodes. *Nematology* **10**, 251-259.
- San-Blas, E., Pembroke, B. & Gowen, S. R. 2012. Scavenging and infection of different hosts by *Steinernema carpocapsae*. *Nematropica* **42**, 123-130.
- Sapp, J. 2004. The dynamics of symbiosis: an historical overview. In *Canadian Journal of Botany*, vol. 82, pp. 1046-1056: Canadian Science Publishing.
- Schmitz, R. P. H. & Galinski, E. A. 1996. Compatible solutes in luminescent bacteria of the genera Vibrio, Photobacterium and Photorhabdus (Xenorhabdus): Occurrence of ectoine, betaine and glutamate. *Fems Microbiology Letters* 142, 195-201.
- Serwe-Rodriguez, J., Sonnenberg, K., Appleman, B. & Bornstein-Forst, S. 2004. Effects of host desiccation on development, survival, and infectivity of entomopathogenic nematode Steinernema carpocapsae. *Journal of Invertebrate Pathology* 85, 175-181.
- Shapiro-Ilan, D. & Grewal, P. 2008. Entomopathogenic Nematodes and Insect Management. In *Encyclopedia of Entomology* (ed. J. L. Capinera), pp. 1336-1340: Springer Netherlands.
- Shapiro-Ilan, D. I., Stuart, R. & McCoy, C. W. 2003. Comparison of beneficial traits among strains of the entomopathogenic nematode, Steinernema carpocapsae, for control of Curculio caryae (Coleoptera: Curculionidae). *Biological Control* 28, 129-136.
- Sharma, S., Waterfield, N., Bowen, D., Rocheleau, T., Holland, L., James, R. & ffrench-Constant, R. 2002. The lumicins: novel bacteriocins from Photorhabdus luminescens with similarity to the uropathogenic-specific protein (USP) from uropathogenic Escherichia coli. *FEMS Microbiol Lett* 214, 241 - 249.
- Sharon, N. & Lis, H. 1989. Lectins as cell recognition molecules. *Science* **246**, 227-234.
- Sharon, N. & Lis, H. 2004. History of lectins: from hemagglutinins to biological recognition molecules. *Glycobiology* **14**, 53R-62R.
- Sicard, M., Ferdy, J. B., Pages, S., Le Brun, N., Godelle, B., Boemare, N. & Moulia, C. 2004. When mutualists are pathogens: an experimental study of the

symbioses between Steinernema (entomopathogenic nematodes) and Xenorhabdus (bacteria). *Journal of Evolutionary Biology* **17**, 985-993.

- Sicard, M., Le Brun, N., Pages, S., Godelle, B., Boemare, N. & Moulia, C. 2003. Effect of native Xenorhabdus on the fitness of their Steinernema hosts: contrasting types of interaction. *Parasitology Research* 91, 520-524.
- Sicard, M., Raimond, M., Prats, O., Lafitte, A. & Braquart-Varnier, C. 2008. Pathogenic effect of entomopathogenic nematode-bacterium complexes on terrestrial isopods. *Journal of Invertebrate Pathology* **99**, 20-27.
- Simões, N., Caldas, C., Rosa, J., Bonifassi, E. & Laumond, C. 2000. Pathogenicity Caused by High Virulent and Low Virulent Strains of Steinernema carpocapsae to Galleria mellonella. Journal of Invertebrate Pathology 75, 47-54.
- Simões, N. & Rosa, J. S. 1996. Pathogenicity and host specificity of entomopathogenic nematodes. *Biocontrol Science and Technology* 6, 403-412.
- Sisodia, S. & Singh, B. N. (2012). Experimental Evidence for Nutrition Regulated Stress Resistance in Drosophila ananassae. PLoS One 7(10): e46131. doi: doi:10.1371/journal.pone.0046131
- Smits, P. H., Groenen, J. T. M. & De Raay, G. 1991. Characterization Of Heterorhabditis Isolates Using Dna Restriction Fragment Length Polymorphism. *Revue de Nematologie* 14, 445-454.
- Snyder, H., He, H., Owen, H., Hanna, C. & Forst, S. 2011. Role of Mrx fimbriae of *Xenorhabdus nematophila* in competitive colonization of the nematode host. *Applied and Environmental Microbiology* **77**, 7247-7254.
- Snyder, H., Stock, S. P., Kim, S.-K., Flores-Lara, Y. & Forst, S. 2007. New Insights into the Colonization and Release Processes of Xenorhabdus nematophila and the Morphology and Ultrastructure of the Bacterial Receptacle of Its Nematode Host, Steinernema carpocapsae. *Applied and Environmental Microbiology* 73, 5338-5346.
- Somvanshi, V., Sloup, R., Crawford, J., Martin, A., Heidt, A., Kim, K., Clardy, J. & Ciche, T. 2012. A single promoter inversion switches Photorhabdus between pathogenic and mutualistic states. *Science* **337**, 88-93.
- Somvanshi, V. S., Kaufmann-Daszczuk, B., Kim, K.-s., Mallon, S. & Ciche, T. A. 2010. Photorhabdus phase variants express a novel fimbrial locus, mad, essential for symbiosis. *Molecular Microbiology* 77, 1021-1038.
- Speight, M. C., Chandler, P. & Nash, R. 1975. Irish Syrphidae (Diptera): notes on the species and an account of their known distribution. In *Proceedings of the Royal Irish Academy. Section B: Biological, geological, and chemical science*, pp. 1-80: JSTOR.

- Speight, M. C. D. 1997. Invertebrates of dune and grassland. In North Bull Island Dublin Bay - a modern coastal natural history. (ed. D. W. Jeffrey), pp. 107 -111. Dublin: The Royal Dublin Society.
- Spence, K. O., Stevens, G. N., Arimoto, H., Ruiz-Vega, J., Kaya, H. K. & Lewis, E. E. 2011. Effect of insect cadaver desiccation and soil water potential during rehydration on entomopathogenic nematode (Rhabditida: Steinernematidae and Heterorhabditidae) production and virulence. *Journal of Invertebrate Pathology* **106**, 268-273.
- Spiridonov, S. E., Moens, M. & Wilson, M. J. 2007. Fine scale spatial distributions of two entomopathogenic nematodes in a grassland soil. *Applied Soil Ecology* 37, 192-201.
- Spring, D. 2009. The role of nematode behaviour in maintaining a nematorebacterial symbiosis B. Sc. Thesis, National University of Ireland, Maynooth, Ireland.
- Stack, C. M., Easwaramoorthy, S. G., Metha, U. K., Downes, M. J., Griffin, C. T. & Burnell, A. M. 2000. Molecular characterisation of Heterorhabditis indica isolates from India, Kenya, Indonesia and Cuba. *Nematology* 2, 477-487.
- Stock, S. P. 2009. Molecular approaches and the taxonomy of insect-parasitic and pathogenic nematodes. In *Insect pathogens: molecular approaches and techniques* (ed. S. P. Stock, J. Vanderberg, I. Glazer & N. Boemare), pp. 71-100. Wallingford, UK: CAB International.
- Stock, S. P., Griffin, C. & Burnell, A. M. 2002. Morphological characterisation of three isolates of Heterorhabditis Poinar, 1976 from the 'Irish group' Nematoda:Rhabditida: Heterorhabditidae) and additional evidence supporting their recognition as a distinct species, H. downesi n. sp. Systematic Parasitology 51, 95-106.
- Stock, S. P., Lee, M. & Flores-Lara, Y. 2012. The rectal glands of Heterorhabditis bacteriophora (Rhabditida: Heterorhabditidae) hermaphrodites and their role in symbiont transmission. *J Invertebr Pathol*.
- Strauch, O. & Ehlers, R. U. 1998. Food signal production of *Photorhabdus luminescens* inducing the recovery of entomopathogenic nematodes *Heterorhabditis* spp. in liquid culture. *Applied Microbiology and Biotechnology* 50, 369-374.
- Stuart, R. J., Barbercheck, M. E., Grewal, P. S., Taylor, R. A. J. & Hoy, C. W. 2006. Population biology of entomopathogenic nematodes: Concepts, issues, and models. *Biological Control* 38, 80-102.
- Stuart, R. J. & Gaugler, R. 1994. Patchiness in populations of entomopathogenic nematodes. *Journal of Invertebrate Pathology* **64**, 39-45.

- Sturhan, D. 1999. Prevalence and habitat specificity of entomopathogenic nematodes in Germany. In COST 819 - Entomopathogenic Nematodes -Application and Persistence of Entomopathogenic Nematodes., vol. 819 (ed. R. l. Gwynn, P. H. Smits, C. Griffin, R.-U. Ehlers, N. Boemare & J.-P. Masson), pp. 123-132. Luxembourg, DG XII: European Commission.
- Susurluk, A., Dix, I., Stackebrandt, E., Strauch, O., Wyss, U. & Ehlers, R.-U. 2001. Identification and ecological characterisation of three entomopathogenic nematode-bacterium complexes from Turkey. *Nematology* 3, 833-841.
- Tailliez, P. & Boemare, N. 2009. Phylogenetic studies with entomopathogenic bacteria with special emphasis on symbionts of entomopathogenic nematodes. In *Insect pathogens: molecular approaches and techniques* (ed. S. P. Stock, J. Vanderberg, I. Glazer & N. Boemare), pp. 131-144. Wallingford, UK: CAB International.
- Tailliez, P., Laroui, C., Ginibre, N., Paule, A., Pages, S. & Boemare, N. 2010. Phylogeny of *Photorhabdus* and *Xenorhabdus* based on universally conserved protein-coding sequences and implications for the taxonomy of these two genera. Proposal of new taxa: X. vietnamensis sp nov., P. luminescens subsp caribbeanensis subsp nov., P. luminescens subsp hainanensis subsp nov., P. temperata subsp hanii subsp nov., P. temperata subsp tasmaniensis subsp nov., and the reclassification of P. luminescens subsp thracensis as P. temperata subsp thracensis comb. nov. International Journal of Systematic and Evolutionary Microbiology 60, 1921-1937.
- Tailliez, P., Pages, S., Ginibre, N. & Boemare, N. 2006. New insight into diversity in the genus Xenorhabdus, including the description of ten novel species. *International Journal of Systematic and Evolutionary Microbiology* 56, 2805-2818.
- Takatsuka, J. & Kunimi, Y. 2000. Intestinal Bacteria Affect Growth of Bacillus thuringiensis in Larvae of the Oriental Tea Tortrix, Homona magnanima Diakonoff (Lepidoptera: Tortricidae). Journal of Invertebrate Pathology 76, 222-226.
- Tamura, K., Peterson, D., Peterson, N., Stecher, G., Nei, M. & Kumar, S. 2011. MEGA5: Molecular Evolutionary Genetics Analysis Using Maximum Likelihood, Evolutionary Distance, and Maximum Parsimony Methods. *Molecular Biology and Evolution* 28, 2731-2739.
- Tarasco, E., Enright, M. R. & Griffin, C. T. 2003. Preferences of Italian Heterorhabditis for different Photorhabdus bacterial types. In *IOBC/wprs Meeting "Insect pathogens and insect parasitic nematodes"*, vol. 28(3), pp. 23-26. Kiel, Germany.
- Thaler, J. O., Baghdiguian, S. & Boemare, N. 1995. Purification and characterization of xenorhabdicin, a phage tail-like bacteriocin, from the lysogenic strain fl of *Xenorhabdus-nematophilus*. Applied and Environmental Microbiology 61, 2049-2052.

- Thompson, J. D., Higgins, D. G. & Gibson, T. J. 1994. CLUSTAL W: improving the sensitivity of progressive multiple sequence alignment through sequence weighting, position-specific gap penalties and weight matrix choice. *Nucleic Acids Res* 22, 4673-80.
- Timper, P. & Kaya, H. K. 1989. Role of the second-stage cuticle of entomogenous nematodes in preventing infection by nematophagous fungi. *Journal of Invertebrate Pathology* **54**, 314-321.
- Timper, P., Kaya, H. K. & Gaugler, R. 1988. Dispersal of the entomogenous nematode Steinernema-feltiae (Rhabditida, Steinernematidae) by infected adult insects. *Environmental Entomology* **17**, 546-550.
- Tóth, T. 2006. Collection of entomopathogenic nematodes for the biological control of insect pests. *Journal of Fruit and Ornamental Plant Research* 14 (Suppl 3), 225-230.
- Tóth, T. & Lakatos, T. 2008. *Photorhabdus temperata* subsp. *cinerea* subsp. nov., isolated from *Heterorhabditis* nematodes. *Int J Syst Evol Microbiol* **58**, 2579-2581.
- Tóth, T. & Lakatos, T. 2009. Two different bacterial symbionts of Heterorhabditis megidis and Heterorhabditis downesi inside one population. *IOBC/wprs Bulletin* **45**, 395-397.
- Tóth, T., Lakatos, T. & Kaskötő, Z. 2008. Identification and typing of *Photorhabdus* isolates from entomopathogenic nematodes in Hungarian soils. *IOBC/wprs Bulletin* **31**, 138-143.
- Valadas, V., Laranjo, M., Mota, M. & Oliveira, S. 2013. A survey of entomopathogenic nematode species in continental Portugal. *Journal of Helminthology* 88, 327-341.
- van der Hoeven, R., Betrabet, G. & Forst, S. 2008. Characterization of the gut bacterial community in Manduca sexta and effect of antibiotics on bacterial diversity and nematode reproduction. *Fems Microbiology Letters* **286**, 249-256.
- van Tol, R., van der Sommen, A. T. C., Boff, M. I. C., van Bezooijen, J., Sabelis, M. W. & Smits, P. H. 2001. Plants protect their roots by alerting the enemies of grubs. *Ecology Letters* 4, 292-294.
- Verhoeven, R. 2002. The structure of the microtrophic system in a development series of dune soils. *Pedobiologia* **46**, 75-89.
- Vigneux, F., Bashey, F., Sicard, M. & Lively, C. M. 2008. Low migration decreases interference competition among parasites and increases virulence. *Journal of Evolutionary Biology* **21**, 1245-1251.

- Viney, M. E., Thompson, F. J. & Crook, M. 2005. TGF-β and the evolution of nematode parasitism. *International Journal for Parasitology* **35**, 1473-1475.
- Vrain, T. C., Wakarchuk, D. A., Levesque, A. C. & Hamilton, R. I. 1992. Intraspecific rDNA restriction-fragment-length-polymorphism in the *Xiphinema-americanum* group. *Fundamental and Applied Nematology* 15, 563-573.
- Walsh, K. T. & Webster, J. M. 2003. Interaction of microbial populations in Steinernema (Steinernematidae, Nematoda) infected Galleria mellonella larvae. *Journal of Invertebrate Pathology* 83, 118-126.
- Wang, H., Jin, L. & Zhang, H. 2011. Comparison of the diversity of the bacterial communities in the intestinal tract of adult *Bactrocera dorsalis* from three different populations. *Journal of Applied Microbiology* **110**, 1390-1401.
- Wang, X. & Ishibashi, N. 1999. Infection of the entomopathogenic nematode, *Steinernema carpocapsae*, as affected by the presence of *Steinernema glaseri*. Journal of Nematology **31**, 207.
- Wang, Y. & Gaugler, R. 1998. Host and Penetration Site Location by Entomopathogenic Nematodes against Japanese Beetle Larvae. *Journal of Invertebrate Pathology* 72, 313-318.
- Waterfield, N., Bowen, D., Fetherston, J., Perry, R. & ffrench-Constant, R. 2001. The tc genes of Photorhabdus: a growing family. *Trends Microbiol* 9, 185 - 191.
- Waterfield, N. R., Ciche, T. & Clarke, D. 2009. *Photorhabdus* and a host of hosts. *Annual Review of Microbiology* **63**, 557-574.
- Watson, R. J., Joyce, S. A., Spencer, G. V. & Clarke, D. J. 2005. The exbD gene of Photorhabdus temperata is required for full virulence in insects and symbiosis with the nematode Heterorhabditis. *Molecular Microbiology* 56, 763-773.
- Watson, R. J., Millichap, P., Joyce, S. A., Reynolds, S. & Clarke, D. J. 2010. The role of iron uptake in pathogenicity and symbiosis in Photorhabdus luminescens TT01. *BMC Microbiology* **10**, e177.
- Webster, J. M., Chen, G., Hu, K. & Li, J. 2002. Bacterial metabolites. In *Entomopathogenic Nematology* (ed. R. Gaugler), pp. 99-113. Wallingford, UK: CABI Publishing.
- Weeks, A. R., Turelli, M., Harcombe, W. R., Reynolds, K. T. & Hoffmann, A. A. (2007). From parasite to mutualist: rapid evolution of Wolbachia in natural populations of Drosophila. Plos Biology 5(5): e114. doi: 10.1371/journal.pbio.0050114

- Wharton, D. & Surrey, M. 1994. Cold tolerance mechanisms of the infective larvae of the insect parasitic nematode, Heterorhabditis zealandica Poinar. *Cryo letters* **15**, 353-360.
- Williams, C. D., Dillon, A. B., Girling, R. D. & Griffin, C. T. 2013. Organic soils promote the efficacy of entomopathogenic nematodes, with different foraging strategies, in the control of a major forest pest: A meta-analysis of field trial data. *Biological Control* 65, 357-364.
- Willis, A., Folkes, B., Hope-Simpson, J. & Yemm, E. 1959. Braunton Burrows: the dune system and its vegetation. *The Journal of Ecology* **47**, 249-288.
- Wyatt, G. R. 1961. The Biochemistry of Insect Hemolymph. Annual Review of Entomology 6, 75-102.
- Wyatt, G. R. & Kalf, G. F. 1957. The chemistry of insect hemolymph .2. Trehalose and other carbohydrates.
- . Journal of General Physiology 40, 833-847.
- Yamamoto, S. & Harayama, S. 1995. PCR amplification and direct sequencing of gyrB genes with universal primers and their application to the detection and taxonomic analysis of Pseudomonas putida strains [published erratum appears in Appl Environ Microbiol 1995 Oct;61(10):3768]. Applied and Environmental Microbiology 61, 1104-1109.
- Yamanaka, S., Hagiwara, A., Nishimura, Y., Tanabe, H. & Ishibashi, N. 1992. Biochemical and physiological characteristics of Xenorhabdus species, symbiotically associated with entomopathogenic nematodes including Steinernema kushidai and their pathogenicity against Spodoptera litura (Lepidoptera: Noctuidae). Archives of Microbiology 158, 387-393.

## Appendix



**Figure A. 1:** Overlap of sampling areas on North Bull Island, this study (Transect 1, 2, 3) and Rolston et al (2005) 2001 - 2002. (See Section 2.2.1.1, Figure 2.1).

1 x Tris-Acetate (TAE) Buffer	2 L
50 x TAE	40 ml
adjust volume to 2 L with $dH_2O$	
50 x Tris-Acetate (TAE) Buffer	1 L
Trizma base (Sigma-Aldrich)	242 g
acetic acid (glacial) (Merck chemicals)	57.1 ml
0.5M EDTA (pH 8.0)	100 ml
adjust volume to 1 L with dH <sub>2</sub> O	
Do not autoclave. Store at room temperature.	
0.5M EDTA	1 L
EDTA	146.1 g
dH <sub>2</sub> O	800 ml
pH adjusted to 8.0 with 25g - 35 g NaOH pellets	
adjust volume to 1 L with dH <sub>2</sub> O	
Sterilise by autoclaving and store at room ten	nperature

## Table A 1: TAE Buffer (See Section 2.2.2.2)

**Table A 2:** Parameters used for the Maximum Likelihood method for calculating the phylogeny of *Heterorhabditis downesi* isolates from North Bull Island (see Section 2.2.2.6 & Section 2.3.2, Figure 2.9). Phylogenetic analysis was carried out in MEGA5.

Analysis
Analysis Phylogeny Reconstruction
Statistical Method Maximum Likelihood
Phylogeny Test
Test of Phylogeny Bootstrap method
No. of Bootstrap Replications 1000
Substitution Model
Substitutions Type Nucleotide
Model/Method Kimura 2-parameter model
Rates and Patterns
Rates among Sites Gamma Distributed (G)
No of Discrete Gamma Categories 5
Data Subset to Use
Gaps/Missing Data Treatment Use all sites
Tree Inference Options
ML Heuristic Method Nearest-Neighbor-Interchange (NNI)
Initial Tree for ML Make initial tree automatically
Codons Included1st+2nd+3rd+Non-Coding
Number of Sequences: 29
Number of Sites: 1012

A	Nutrient agar	g/l
	Lab-Lemco powder	1
	Yeast extract	2
	Peptone	5
	Sodium chloride	5
	Agar	15

**Table A 3:** Culture media for *Photorhabdus temperata* subsp. (See Section 2.2.3.1)

B	Nutrient agar plus bromothymol blue and triphenyltetrazolium chloride	g/l
	Lab-Lemco powder	1
	Yeast extract	2
	Peptone	5
	Sodium chloride	5
	Agar	15
	Bromothymol blue	0.025
	triphenyltetrazolium chloride	0.04

MacConkey agar	g/l
Peptone	20
Lactose	10
Bile salts	5
Sodium chloride	5
Neutral red	0.075
Agar	12

D	Luria broth †	g/l
	Tryptone	10
	Yeast extract	5

*†* extra salt was added to Luria broth for differential growth investigation.

## A.1 Generation of standard curve for *Photorhabdus temperata pur1* and *yel3* isolates

Three replicate liquid cultures were set up in NB as described in Section 2.2.3.2 and incubated for 45 hr at 27° C with shaking at 200 rpm. A dilution series in <sup>1</sup>/<sub>4</sub> strength sterile Ringer's solution was made from each culture. Initial serial dilutions of 1 in 10 and 1 in 5 yielded too few data points to construct a standard curve therefore a 1 in 2 (*pur1*) or a 1 in 3 (*yel*) serial dilution was used. The optical density of each dilution was read at 660 nm (OD<sub>660</sub>). Direct cell counts were carried out in triplicate as described in Section 2.2.3.3b for each replicate culture using a suitable dilution and used to calculate the concentration of each dilution to generate the standard curves were then used to calculate concentration of bacterial cells for DNA extraction (Section 2.2.4.1) and for the cross-compatibility experiment described in Section 3.2.1.

**Figure A. 2:** Standard curve for *pur1* (A) and *yel3* (B) *Photorhabdus temperata* isolates from North Bull Island (See section 2.2.4.1 and Section 3.2.1.1).



**Table A 4:** Parameters used for the Maximum Likelihood method for calculating the phylogeny of *Photorhabdus* isolates from North Bull Island (See Section 2.2.4.3 & Section 2.3.4, Figure 2.17). Phylogenetic analysis was carried out in MEGA5.

Analysis
Analysis Phylogeny Reconstruction
Statistical Method Maximum Likelihood
Phylogeny Test
Test of Phylogeny Bootstrap method
No. of Bootstrap Replications1000
Substitution Model
Substitutions Type Nucleotide
Model/Method Kimura 2-parameter model
Rates and Patterns
Rates among Sites Gamma Distributed (G)
No of Discrete Gamma Categories 5
Data Subset to Use
Gaps/Missing Data Treatment Use all sites
Tree Inference Options
ML Heuristic Method Nearest-Neighbor-Interchange (NNI)
Initial Tree for ML Make initial tree automatically
Codons Included 1st+2nd+3rd+Non-Coding
Number of Sequences: 33
Number of Sites: 847

	Transect 1			Transect 2		Transect 3			
Distance (m)	Habitat type <sup>†</sup>	soil temp °C (8 cm depth)	pН	Habitat type	soil temp °C (8 cm depth)	рН	Habitat type	soil temp °C (8 cm depth)	pН
0	Е	15	8.38	Е	20	8.17	Е	20	8.28
10	E/M/SP	18	8.23	M/SFX/SP	15	8.17	E/M	15	8.37
20	M/SFX/SP	17	8.04	M/SFX/F/SP	14	8.19	M/SFX/SP	14	8.24
30	M/SFX/SP	16	7.89	M/SFX/F/SP	14	7.94	M/SFX/SP	14	8.11
40	M/SFX/SP	15	7.95	M/SFX/F	17	7.75	M/SFX/FX	17	8.02
50	M/SFX/SP	21	8.08	M/SFX/F/GP	14	7.89	M/SFX/SP	14	8.14
60	BO/M/SFX/F/ SP	17	7.95	M/SFX/F/GP	14	8.04	M/SFX/FX/GP	14	8.13
70	M/SFX/F/SP	15	8.23	M/SFX/F/GP	16	8.12	M/SFX/FX/GP	16	8.19
80	M/SFX/F/SP	13	7.85	M/SFX/F/GP	18	7.90	M/SFX/SP/GP	18	7.95
90	M/SFX/FX	14	7.99	M/SFX/F/SP/G P	15	7.87	M/SFX/FX/SP/ GP	15	7.90
100	SFX/FX	14	7.80	M/SFX/F/GP	14	7.87	M/SFX/SP/GP	14	7.89
110	SFX/FX/GP	13	7.75	M/SFX/F/GP	14	7.87	M/SFX/GP	14	7.91
120	SFX/FX/GP	16	7.82	M/SFX/F/SP	13	7.98	M/SFX/FX/GP	13	7.91
130	SFX/FX	16	7.81	M/SFX/F/GP	14	7.93	M/SFX//GP	14	7.91

Table A 5: The flora, soil temperature and pH recorded from three transects on North Bull Island in 2008 (see Section 2.3.1.5)

<sup>†</sup> BO = Blowout, E = embryonic dunes, FX = fixed dunes, GP = grassy path, M = Marram dunes, SFX = Semi-fixed dunes, SP = sandy path (compacted soil)

PUR1	TTGATTACGTCCCTGCCCTTTGTACACCCCCCGTCGCTGTCCGGG	47
PUR2	TTGATTACGTCCCTGCCCTTTGTACACCCCCGTCGCTGTCCGGG	47
PUR3	GGAATTCGCCCTTTTGATTACGTCCCTGCCCTTTGTACACACCGCCCGTCGCTGTCCGGG	60
YEL1	CACCCCCCCCCCCCCCCCCCCCCCCCCCCCCCC	46
VEI 2		10
		17
IELS	11GATTACGTCCCTGCCTTTGTACACCCGCCGTCGCTGTCCGGG	4/
PUR1	ACTGAGCTGTTTCGAGAAGAGTGGAGACTGCTGTATCGGGGCTTTCGGGCTCTGGTATGA	107
PUR2	ACTGAGCTGTTTCGAGAAGAGTGGAGACTGCTGTATCGGGGCTTTCGGGCTCTGGTATGA	107
PUR3	ACTGAGCTGTTTCGAGAAGAGTGGAGACTGCTGTATCGGGGCTTTCGGGCTCTGGTATGA	120
YEL1	ACTGAGCTGTTTCGAGAAGAGTGGAGACTGCTGTATCGGGGCCTTTCCGGGCTCTGGTATGA	106
VEL2		107
VET 2		107
	*****	107
PUR1	TGGAAACCATTTTAATCGCAATGGCTTGAACCGGGCAAAAGTCGTAACAAGGTATCTGTA	167
PUR2	TGGAAACCATTTTTAATCGCAATGGCTTGAACCGGGCAAAAGTCGTAACAAGGTATCTGTA	T 6./
PUR3	TGGAAACCATTTTAATCGCAATGGCTTGAACCGGGCAAAAGTCGTAACAAGGTATCTGTA	180
YEL1	TGGAAACCATTTTAATCGCAATGGCTTGAACCGGGCAAAAGTCGTAACAAGGTATCTGTA	166
YEL2	TGGAAACCATTTTAATCGCAATGGCTTGAACCGGGCAAAAGTCGTAACAAGGTATCTGTA	167
YEL3	TGGAAACCATTTTAATCGCAATGGCTTGAACCGGGCAAAAGTCGTAACAAGGTATCTGTG *********************************	167
PUR1	GGTGAACCTGCAGATGGATCATCGCTGAAAACCTTTATGGTTATGCTTTGGTCACGAGAG	227
PUKZ	GGTGAACCTGCAGATGGATGATGATGGTTGGTCACGTGAGAGACC-TTATGGTTATGCTTTGGTCACGTGAG	226
PUR3	GGTGAACCTGCAGATGGATCATCGCTGAAAACC-TTATGGTTATGCTTTGGTCACGAGAG	239
YELL	GGTGAACCTGCAGATGGATCATCGCTGAAAACT-TTATGGTTATGCTTTGGTCACGAGAG	225
YEL2	GGTGAACCTGCAGATGGATCATCGCTGAAAACT-TTATGGTTATGCTTTGGTCACGAGAG	226
YEL3	GGTGAACCTGCAGATGGATCATCGCTGAAAACTTATGGTTATGCTTTTGGTCACGAGAG *******************************	225
_		
PUR1	ATCGGTGCTACTGGAATCAGGCTTGCTCTTGATTTCAATCGGTATCTCACCCCATCTAAG	287
PUR2	ATCGGTGCTACTGGAATCAGGCTTGCTCTTGATTTCAATCGGTATCTCACCCCCATCTAAG	286
PUR3	ATCGGTGCTACTGGAATCAGGCTTGCTCTTGATTTCAATCGGTATCTCACCCCATCTAAG	299
YEL1	ATCGGTGCTACTGGAATCAGGCTTGCTCTTGATTTCAATCGGTATCTCACCCCATCTAAG	285
YEL2	ATCGGTGCTACTGGAATCAGGCTTGCTCTTGATTTCAATCGGTATCTCACCCCATCTAAG	286
YEL3	ATCGGTGCTACTGGAATCAGGCTTGCTCTTGATTTCAATCGGTATCTCACCCCATCTAAG ***********************************	285
PUR1	CTCTCGGAGAGGTGTCTAATCGCAATCGGAGTCGCTTTGAGTGACGGCAATGAAGATTGT	347
PUR2	CTCTCGGAGAGGTGTCTAATCGCAATCGGAGTCGCTTTGAGTGACGGCAATGAAGATTGT	346
PUR 3	CTCTCGGAGAGGTGTCTAATCGCAATCGGAGTCGCTTTGAGTGACGGCAATGAAGATTGT	359
VFL1		345
VEL 2		346
YEL3	CTCTCGGAGAGGTGTCTAATCGCAATCGGAGTCGCTTTGAGTGACGGCAATGAAGATTGT	345
DTID 1		407
PURI	GTGTGTTCCCCGTAAGGGTAGAGCATAGACTTTAAGAACAGTGCTGGGACTGTCCCCCTCA	407
PUR2	GTGTGTTCCCCGTAAGGGTAGAGCATAGACTTTAAGAACAGTGCTGGGACTGTCCCCCTCA	406
PUR3	GTGTGTTCCCCGTAAGGGTAGAGCATAGACTTTAAGAACAGTGCTGGGACTGTCCCCTCA	419
YEL1	GTGTGTTCCCCGTAAGGGTAGAGCATAGACTTTAAGAACAGTGCTGGGACTGTCCCCTCA	405
YEL2	GTGTGTTCCCCGTAAGGGTAGAGCATAGACTTTAAGAACAGTGCTGGGACTGTCCCCTCA	406
YEL3	GTGTGTTCCCCGTAAGGGTAGAGCATAGACTTTAAGAACAGTGCTGGGACTGTCCCCTCA	405
PUR1	CCAACCGTCGATTACTGGTGGCTATGAGTGACTTCGGTCACTCGAGATCTGCTACGCAGA	467
PUR2	CCAACCGTCGATTACTGGTGGCTATGAGTGACTTCGGTCACTCCAGATCTGCTACCCAGA	466
PIIR 3	CCAACCGTCGATTACTGGTCGCTATGAGTGACTTCGGTCACTCGGAGATCTGCTACGCAGA	479
VET 1		165
IELL		405
IELZ	CCAACCGICGAITACIGGIGGCIAIGAGIGACITCGGICACICGAGAICIGCIACGCAGA	400
ХЕГЗ	CCAACCGTCGATTACTGGTGGCTATGAGTGACTTCGGTCACTCGAGATCTGCTACGCAGA	465
1 קווס	ᠿᢧᢗᡗᡣᢉ᠇ᡏᠿᢧ᠋᠋ᡆᡏᠿᢩᢧᠿᡎᡎᠿᡎᡎᢙᡘᠯᢙ᠕ᡞᢕᢙᢕᢕ᠘᠋ᡆᠬᡊᡆᠴᠬ᠕᠕᠉ᡎᡎᡎᡎᡎᡢᡢᠬ᠉ᡎᠬ᠉᠉ᡨᠬ	507
		521 576
		520
LOUG VET 1		539
IELL VELO	GAGULICAAIGAGIIGIICGICCAACCGCCGGTGTCGATAGAAATTTTTTTCCTATTAACT	5∠5
YEL2	GAGCCTCAATGAGTTGTTCGTCCAACCGCCGGTGTCGATAGAAATTTTTTCCTATTAACT	526
YEL3	GAGCCTCAATGAGTTGTTCGTCCAACCGCCGGTGTCGATAGAAATTTTTTCCTATTAACT ********************	525
PIIR1	<b>Ͳ</b> ᢙᢆ᠋ᡎᡎᡊᡎ᠋ᡎᡊᢙᡎᠿᡎᡀ᠋᠋ᢧᢧ᠗ᢙᢧᡎᡎᡎᡊ᠘ᢙ᠔᠕ᡊ᠘᠉ᡎᡘᡎ᠕ᡎᠬ᠕᠘ᢙᡊ᠕ᡎᢙ᠉ᡎᢙ᠉	587
PUR2		
	Ͳ;;ͲͲ;;ϫͷϫͷϫͷϫͷϫͷϫͷϫͷϫͷϫͷϫͷϫͷϫͷϫͷϫͷϫͷϫͷϫ	586
DITRR	TGTTTCTTGATTCGTGTTAATACATTTTGGCACAATGTATTAGCTTCAGCGATGGATCGG TGTTTCTTGATTCGTGTTAATACATTTTGGCACAATGTATTAGCTTCAGCGATGGATCGA	586
PUR3	TGTTTCTTGATTCGTGTTAATACATTTTGGCACAATGTATTAGCTTCAGCGATGGATCGG TGTTTCTTGATTCGTGTTAATACATTTTGGCACAATGTATTAGCTTCAGCGATGGATCGG TCTTTCTTCATTCATTCGTGTTAATACATTTTGGCACAATGTATTAGCTTCAGCGATGGATCGG	586 599
PUR3 YEL1	TGTTTCTTGATTCGTGTTAATACATTTTGGCACAATGTATTAGCTTCAGCGATGGATCGG TGTTTCTTGATTCGTGTTAATACATTTTGGCACAATGTATTAGCTTCAGCGATGGATCGG TGTTTCTTGATTCGTGTTAATACATTTTGGCACAATGTATTAGCTTCAGCGATGGATCGG	586 599 585

YEL3	TGTTTCTTGATTCGTGTTAATACATTTTGGCACAATGTATTAGCTTCAGCGATGGATCGG ***********************************	585
1 מוזמ		617
PURI		646
PURZ	TTGATTCGCGTATCGATGAAAAACGCAGCAGCTGCGTTATTTACCACGAATTGCAGACG	646
PUR3	TTGATTCGCGTATCGATGAAAAACGCAGCAAGCTGCGTTATTTACCACGAATTGCAGACG	659
YEL1	TTGATTCGCGTATCGATGAAAAACGCAGCAAGCTGCGTTATTTACCACGAATTGCAGACG	645
YEL2	TTGATTCGCGTATCGATGAAAAAACGCAGCAAGCTGCGTTATTTACCACGAATTGCTGACG	646
YEL3	TTGATTCGCGTATCGATGAAAAACGCAGCAAGCTGCGTTATTTACCACGAATTGCAGACG	645
	***************************************	
PUR1	CTTAGGGTTGGTGAAGTTTTGAACGCACAGCGCCGTTGGGTTTTCCCTTCGGCACGTCTG	707
PUR2	CTTAGAGT-GGTGAAGTTTTGAACGCACAGCGCCGTTGGGTTTTCCCTTCGGCACGTCTG	705
PUR3	CTTAGAGT-GGTGAAGTTTTGAACGCACAGCGCCGTTGGGTTTTCCCTTCGGCACGTCTG	718
YEL1	CTTAGAGT-GGTGAAGTTTTGAACGCACAGCGCCGTTGGGTTTTCCCCTTCGGCACGTCTG	704
VFL2		705
VET 2		703
16112	***** ** ******************************	704
DITD 1		
PURI		107
PUR2	GCTCAGGGTTGTTTAAAGTACTGCGGTGTTGCTGCGAAAGTAGCAACACCGAATGTCGAA	765
PUR3	GCTCAGGGTTGTTTAAAGTACTGCGGTGTTGCTGCGAAAGTAGCAACACCGAATGTCGAA	778
YEL1	GCTCAGGGTTGTTTAAAGTACTGCGGTGTTGCTGCGAAAGTAGCAACACCGAATGTCGAA	764
YEL2	GCTCAGGGTTGTTTAAAGTACTGCGGTGTTGCTGCGAAAGTAGCAACACCGAATGTCGAA	765
YEL3	GCTCAGGGTTGTTTAAAGTACTGCGGTGTTGCTGCGAAAGTAGCAACACCGAATGTCGAA	764
	***************************************	
PIIR1	СССССТТАСТАСТАСТАСТССССССТТСТАСТТССААТАТТСССААСАСАТСТСТТСТ	827
	CCCCCTTACTACTACTACTCCCCCCTTCTACTACTACCCAATATTCCCCAACACCTTCTT	825
		838
YDI 1		000
IELL	CGGCGTTAGTGTTACTGCCCCCGTTCTAGTCCGAATATTGGCAACATGTCTTCTTC	824
YELZ	CGGCGTTAGTGTTACTGCCCCCGTTCTAGTTCCGAATATTGGCAACATGTCTTCTTC	825
ХЕГЗ	CGGCGTTAGTGTTACTACTGGCCCCGTTCTAGTTCCGAATATTGGCAACATGTCTTCTTC *****	824
1 מוז מ	QTQ 3 3 Q3 QQQQT3 Q3 Q3 Q3 QT 3 7 3 2 QTQTQTA QQT3 TQQ 3 TQTQTQQQQT3 TQ3 3 3 T 3 TQ 3	007
PURI	GIGAAGACGCCGIAGAGAGIAIAAGICIGIACCIAIGGAIGICICGCGIAIGAAAIAIGA	00/
PUR2	GTGAAGACGCCGTAGAGAGTATAAGTCTGTACCTATGGATGTGTCGCGTATGAAATATGA	885
PUR3	GTGAAGACGCCGTAGAGAGTATAAGTCTGTACCTATGGATGTGTCGCGTATGAAATATGA	898
YEL1	GTGAAGACGCCGTAGAGAGTATAAGTCTGTACCTATGGATGTGTCGCGTATGAAATATGA	884
YEL2	GTGAAGACGCCGTAGAGAGTATAAGTCTGTACCTATGGATGTGTCGCGTATGAAATATGA	885
YEL3	GTGAAGACGCCGTAGAGAGTATAAGTCTGTACCTATGGATGTGTCGCGTATGAAATATGA	884
	***************************************	
PUR1	TGCTTCCCATACATAGCGAGGAGGTGCTTCTTATGCTATCTTGCTTATGCAACCTGAGCT	947
PUR2	TGCTTCCCATACATAGCGAGGAGGTGCTTCTTATGCTATCTTGCTTATGCAACCTGAGCT	945
PUR3	TGCTTCCCATACATAGCGAGGAGGTGCTTCTTATGCTATCTTGCTTATGCAACCTGAGCT	958
YEL1	TGCTTCCCATACATAGCGAGGAGGTGCTTCTTATGCTATCTTGCTTATGCAACCTGAGCT	944
VEL2	ͲϾϹͲͲϾϹϹϿͲϿϹϿͲϿϾϹϾϿϾϾϿϾϾϽͲϾϹͲͲϾͲͲϿͲϾϹͲϿͲϾϹ	945
VEL3	TCCTTCCCATACATACCACCACCACCTCCTTCTTATCCTATCCTTATCCAACCAACCTCACCT	944
	***************************************	<i>J</i> 1 1
1 קוזס	C 1 C T C C T C 1 C C C C C C C C C C C	1007
		1007
PURZ	CAGICGIGAITACCCGCCGAACITAAGCATATCATTCAGCGGAGGAAAGAAACTAACTA	1005
PUR3	CAGTCGTGATTACCCGCCGAACTTAAGCATATCATTCAGCGGAGGAAAAGAAACTAACT	1018
YEL1	CAGTCGTGATTACCCGCCGAACTTAAGCATATCATTCAGCGGAGGAAAAGAAACTAACT	1004
YEL2	CAGTCGTGATTACCCGCCGAACTTAAGCATATCATTCAGCGGAGGAAAAGAAACTAACT	1005
YEL3	CAGTCGTGATTACCCGCCGAACTTAAGCATATCATTCAGCGGAGGAAAAGAAACTAACT	1004
PUR1	GGATTCCCTTAGTAACGGCGAGTGAAA 1034	
PUR2	GGATTCCCTTAGTAACGGCGAGTGAAA 1032	
PUR3	GGATTCCCTTAGTAACGGCGAGTGAAAAAGGGCGAATT 1056	
YEL1	GGATTCCCTTAGTAACGGCGAGTGAAA 1031	
YEL2	GGATTCCCTTAGTAACGGCGAGTGAAA 1032	
VEL3		
	***************************************	

**Figure A. 3:** ClustalW Multiple Sequence Alignment of a portion (1056 bases with 1031 shared positions) of the ITS1 region from three PUR and three YEL *Heterorhabditis downesi* isolates from North Bull Island (see Section 2.3.2).

Asterisk (\*) denotes conserved sites Spaces denote nucleotide polymorphism.

2	Λ	0
3	υ	0

PUR1	AGAGATCGGTGCTACTGGAATCA-GGCTTGCTCTT-GA	TTTCA	41
2 סוזס	<u> </u>	T	41
		mmmaa	41
PUR3	AGAGATCGGTGCTACTGGAATCA-GGCTTGCTCTT-GA	·1·1·1·CA	41
YEL1	AGAGATCGGTGCTACTGGAATCA-GGCTTGCTCTT-GA	TTTCA	41
YEL2	AGAGATCGGTGCTACTGGAATCA-GGCTTGCTCTT-GA	TTTCA	41
YEL3	AGAGATCGGTGCTACTGGAATCA-GGCTTGCTCTT-GA		41
D000			11
2000	AGAGAIC GGIGCI ACIGGAAICA GGC TIGCICII GA	TITCA	41
H3107	AGAGATCGGTGCTACTGGAATCA-GGCTTGCTCTT-GA	-1-1-1-CA	4⊥
Н3173	AGAGATCGGTGCTACTGGATTCA-GGCTTGCTCTT-GA	TTTCA	41
AY321482	AGAGATCGGTGCTACTGGAATCA-GGCTTGCTCTT-GA	TTTCA	41
к122	AGAGATCGGTGCTACTGGAATCA-GGCTTGCTCTT-GA	TTTCA	41
OH10		TTTTTCA	41
51110		mmmaa	41
EF043441	AGAGAICGGIGCIACIGGAAICA-GGCIIGCICII-GA	TTTCA	41
AGC	AGAGATCGGTGCTACCGGAATCA-GGCTTGCTCCTCGA	TTTCG	42
Andong	AGAGATCGGTGCTACCGGAATCA-GGCTTGCTCCT-GA	TTTCG	41
H3014	AGAGATCGGTGCTACCGGAATCA-GGCTTGCTCCTCGA	TTTCG	42
HF85		TTTCG	42
NTUUS_VUUOE			/ 1
NIHHS-KHHUS	AGAGAICGGIGCIACCGGAAICA-GGCIIGCICCI-GA	- IIICG	41
N-UK76	AGAGATCGGTGCTACCGGAATCA-GGCTTGCTCCTCGA	-1-1-1-CG	42
КЈ938577	AGAGATCGGTGCTACCGGAATCA-GGCTTGCTCCTCGA	TTTCG	42
SF52	AGAGATCGGTGCTACTGGAATCG-GGCTTGCTCTT-GA	TTTCA	41
SF670	AGAGATCGGTGCTACTGGAATCG-GGCTTGCTCTT-GA	TTTCA	41
SF724	AGAGATCGGTGCTACTGGAATCG-GGCTTGCTCTT-GA	TTTCA	41
EF488006			41
Florida		 	11
127		mmmaa	41
037	AGCIAICGGIGCICAIGGAAICA-GGCICACGIII-GA	TITCA	41
NZH3	AGCTATCGGTGCTCATGGAATCA-GGCTCACGTTT-GA	-1-1-1-CA	41
X1	AGCTATCGGTGCTCATGGAATCA-GGCTCACGTTT-GA	TTTCA	41
X03680	AAACCTTAAGATGCTCGACTGGCTTCACGGTCAGTTGAGTGTCGAAATGTCAACG	TTCCA	60
	* * **** ** ** * * * **	** *	
PIIR 1		CGG-A	94
			01
		aga a	04
PURS		A-DDJ.	94
XETT -	ATCGGTATCTCACCCC-ATCTAAGCTCTCGGAGAGG-TGTCTAATCGCAAT-	CGG-A	94
YEL2	ATCGGTATCTCACCCC-ATCTAAGCTCTCGGAGAGG-TGTCTAATCGCAAT-	CGG-A	94
YEL3	ATCGGTATCTCACCCC-ATCTAAGCTCTCGGAGAGG-TGTCTAATCGCAAT-	CGG-A	94
D099	ATCGGTATCTCACCCC-ATCTAAGCTCTCGGAGAGG-TGTCTAATCTCAAT-	TGG-A	94
H3107	ATCGGTATCTCACCCC-ATCTAAGCTCTCGGAGAGG-TGTCTAATCGCAAT-	CGG-A	94
H3173	ATCGGTATCTCACCCC-ATCTAAGCTCTCGGAGAGG-TGTCTAATCGCAAT-	CGG-A	94
AV321482		CCC-A	94
V100			01
KIZZ		-CGG-A	94
OHIU	ATCGGTATCTCACCTC-ATCTAAGCTCTCGGAGAGG-TGTCTAATCCCAAT-	-TGG-A	94
EF043441	ATCGGTATCTCACCCC-ATCTAAGCTCTCGGAGAGG-TGTCTAATCCCAAT-	TGG-A	94
AGC	ATCGGTATCTCACCCCCATCTAAGCTCTCGGAGAGG-TGTCTAATCCCAAT-	CGG-A	96
Andong	ATCGGTGTCTCACCCCCATCTAAGCTCTCGGAGAGGGTGTCTAATCCCAAT-	CGGGA	97
H3014	ATCGGTATCTCACCCCCATCTAAGCTCTCGGAGAGG-TGTCTAATCCCAAT-	CGG-A	96
HF85	ΔΨĊĠĠͲϪͲĊͲĊϪĊĊĊĊĊϪͲĊͲϪϪĠĊͲĊͲĊĠĠϪĠϪĠĠ_ͲĠͲĊͲϪϪͲĊĊĊΔϪͲ_	CGG-A	96
NTUUS_VUUOE			05
NTIHIS RIHOS		aga a	00
N-UK76	ATCGGTATCTCACCCCCATCTAAGCTCTCGGAGAGG-TGTCTAATCCCAAT-	CGG-A	90
KJ938577	ATCGGTATCTCACCCCCATCTAAGCTCTCGGAGAGG-TGTCTAATCCCAAT-	CGG-A	96
SF52	ATCGGTATCTCACCCC-ATCTAAGCTCTCGGAGAGG-TGTCTAATCCCAAT-	TGG-A	94
SF670	ATCGGTATCTCACCCC-ATCTAAGCTCTCGGAGAGG-TGTCTAATCCCAAT-	TGG-A	94
SF724	ATCGGTATCTCACCCC-ATCTAAGCTCTCGGAGAGG-TGTCTAATCCCAAT-	TGG-A	94
EF488006	ATCGGTATCTCACCCC-ATCTAAGCTCTCGGAGAGG-TGTCTAATCCCAAT-	TGG-A	94
Florida		TGG-A	94
	λΨĊĊĊͲλĊĊͲĊλĊĊĊĊ_λΨĊͲλλĊĊΨĊΨͲĊĊͲĊλĊĊ_ΨĊΨĊͲλΨͲĊĊĊλλΨ_	TCC-N	94
		TOO A	01
NZH3		TGG-A	94
X1	ATCGGTAGCTCACCCC-ATCTAAGCTCTTGGTGAGG-TGTCTATTCCCAAT-	-TGG-A	94
X03680	GTTGAGATGCCACAAC-AACTG-GCAAGAGTAGTGACTGTCCGACCCATGGAACG	TGGCA	118
	* * * * * * * * * * * * * * * * * * * *	** *	
PUR1	GTCGCTTTGAGTGACGGCAATGAAGATTGTGT-GTGTT	CCCCG	136
PUR2	GTCGCTTTGAGTGACGGCAATGAAGATTGTGT-GTGTT	CCCCG	136
PUR3	GTCGCTTTGAGTGACGGCAATGAAGATTGTGT-GTGTT	CCCCG	136
VET.1	CTCCCTTTCLCTCLCCCCLLTCLLCCCCLTCLLCCCCC	CCCCC	136
VEL 2			126
		addad	1 2 0
1000 1000	GILGLIIIGAGIGACGGC-AATGAAGATTGTGT-GTGT-	agaaa	130
D099	GTCGCTTTGAGTGACGGCAATGAAGGTTGGGT-GTGTT	CCCCG	136
H3107	GTCGCTTTGAGTGACGGCAATGAAGATTGTGT-GTGTT	CCCCG	136
Н3173	GTCGCTTTGAGTGACGGCAATGAAGATTGTGT-GTGTT	CCCCG	136
AY321482	GTCGCTTTGAGTGACGGCAATGAAGATTGTGT-GTGTT	CCCCG	136
К122	GTCGCTTTGAGTGACGGCAATGAAGATTGTGT-GTGTT	CCCCG	136
OH10	GTCGCT	CCCCC	126
FF043441		.00000	126
DI O IO ITI		CCCCG	1/1
AGC		agaa-	141
Andong	GTCGCTTTGAGTGACGGCAATGAGGATTGGGT-GTGCCATA	CCCCA	142

Н3014	GTCGCTTTGAGTGACGGCAATGAGGATTGGGT-GTGCCATACCCCA	141
HF85	GTCGCTTTGAGTGACGGCAATGAGGATTGGGT-GTGCCATACCCCA	141
NIHHS-KHH05	GTCGCTTTGAGTGACGGCAATGAGGATTGGGT-GTGCCATACCCCA	140
N-UK76	GTCGCTTTGAGTGACGGCAATGAGGATTGGGT-GTGCCATACCCCA	141
к.т938577	GTCGCTTTGAGTGACGGCAATGAGGATTGGGT-GTGCCATACCCCA	141
SE2	GTCGCGTTG1GTG1CGCC-D1TG1GCGT-GTGTTCCCCG	136
SE 52		126
SF070		1 2 6
SF/24	GTCGCCGTTGAGTGACGGC-AATGAAGGTTGGGT-GTGTTCCCCCG	136
EF488006	GTCGCGTTGAGTGACGGCAATGAAGGTTGGGT-GTGTTCCCCG	136
Florida	GTCGCTTGGAGTGACGGCAGTGATGATTGGGT-GTGTTGCCCG	136
J37	GTCGCTTGGAGTGACGGCAGTGATGATTGGGT-GTGTTGCCCG	136
NZH3	GTCGCTTGGAGTGACGGCAGTGATGATTGGGT-GTGTTGCCCG	136
x1	GTCGCTTGG2GTG2CGC2GTG2TG2TG2TG2TGCT-GTGTTGCCCG	136
x03680		175
X03000		1/5
-		
PUR1	TAAGGGTAGAGCATAGACTTTAAGAACAGTGCTGGGACTGTCCCCTCA-C-CAA-	188
PUR2	TAAGGGTAGAGCATAGACTTTAAGAACAGTGCTGGGACTGTCCCCCTCA-C-CAA-	188
PUR3	TAAGGGTAGAGCATAGACTTTAAGAACAGTGCTGGGACTGTCCCCTCA-C-CAA-	188
YEL1	TAAGGGTAGAGCATAGACTTTAAGAACAGTGCTGGGACTGTCCCCTCA-C-CAA-	188
YEL2	TAAGGGTAGAGCATAGACTTTAAGAACAGTGCTGGGACTGTCCCCTCA-C-CAA-	188
VEL3		188
D000		100
12107		100
H3107		100
H3173	TAAGGGTAGAGCATAGACTTTAAGAACAGTGCTGGGACTGTCCCCTCA-C-CAA-	T88
AY321482	TAAGGGTAGAGCATAGACTTTAAGAACAGTGCTGGGACTGTCCCCTCA-C-CAA-	188
K122	TAAGGGTAGAGCATAGACTTTAAGAACAGTGCTGGGACTGTCCCCTCA-C-CAA-	188
OH10	TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCTCA-C-CAAA	189
EF043441	TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCTCA-C-CAAA	189
AGC	TACGTGGGGTAGAGCATAGACGTTAGGAACAGTACTCGGACTGTCGCCTCA-C-CAA-	196
Andong	$T_{2} = C_{2} = C_{2$	197
112014		106
13014		100
HF85	TACGTGGGGTAGAGCATAGACGTTAGGAACAGTACTCGGACTGTCGCCTCA-C-CAA-	196
NIHHS-KHH05	TACGTGGGGTAGAGCATAGACGTTAAGAACAGTGCTCGGACTGTCGCCTCA-C-CAA-	195
N-UK76	TACGTGGGGTAGAGCATAGACGTTAGGAACAGTACTCGGACTGTCGCCTCA-C-CAA-	196
KJ938577	TACGTGGGGTAGAGCATAGACGTTAGGAACAGTGCTCGGACTGTCGCCTCA-C-CAA-	196
SF52	TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCTCA-C-CAAA	189
SE670		189
51070	IAAGGGIAGAGCAIAGACIIIAIGAACAGIGCIGGGACIGICGCCICA-C-CAAA	102
SF724	TAAGGGTAGAGCATAGACTITATGAACAGTGCTGGGACTGTCGCCTCA-C-CAAA T	189
SF724 EF488006	TA = -AGGGTAGAGCATAGACTTATGAACAGTGCTG = -GGACTGTCGCCTCA - C - CAAA = TA = -AGGGTAGAGCATAGACTTTATGAACAGTGCTG = -GGACTGTCGCCTCA - C - C - C - C - C - C - C - C - C -	189 189
SF724 EF488006 Elorida	TAAGGGTAGAGCATAGACTTATGAACAGTGCTGGGACTGTCGCCTCA-C-CAAA TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCCCA-C-CAAA TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCCCAC-C-CAAA	189 189 189
SF724 EF488006 Florida	TAAGGGTAGAGCATAGACTTATGAACAGTGCIGGGACTGTCGCCTCA-C-CAAA TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCTCA-C-CAAA TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCTCA-C-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA	189 189 189 192
SF724 EF488006 Florida J37	TAAGGGTAGAGCATAGACTTTATGAACAGTGCIGGGACTGTCGCCTCA-C-CAAA TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCTCA-C-CAAA TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCTCAGC-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA	189 189 192 192
SF724 EF488006 Florida J37 NZH3	TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCTCA-C-CAAA TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCTCA-C-CAAA TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCTCAGC-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCCGGACTGTTGCCTCAGC-CAAA	189 189 192 192 192
SF724 SF724 EF488006 Florida J37 NZH3 X1	TA = -AGGGTAGAGCATAGACTTATTATGAACAGTGCTG = -GGACTGTCGCCTCA - C - CAAA = TA AGGGTAGAGCATAGACTTTATGAACAGTGCTG GGACTGTCGCCTCA - C - C - C - C - C - C - C - C - C -	189 189 192 192 192 192
SF724 SF724 EF488006 Florida J37 NZH3 X1 X03680	TA = -AGGGTAGAGCATAGACTTATGAACAGTGCIG = -GGACTGTCGCCTCA - C - CAAA = TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCTCA - C - C - C - C - C - C - C - C - C -	189 189 192 192 192 192 215
SF724 SF724 EF488006 Florida J37 NZH3 X1 X03680	TAAGGGTAGAGCATAGACTTATGAACAGTGCIGGGACTGTCGCCTCA-C-CAAA TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCTCA-C-CAAA TATGGGTGGCGCATAGACTTTATGAACAGTGCTCTCGGACTGTTGCCCCAGC-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA	189 189 192 192 192 192 215
SF724 EF488006 Florida J37 NZH3 X1 X03680	TAAGGGTAGAGCATAGACTTATGAACAGTGCIGGGACTGTCGCCTCA-C-CAAA TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCTCA-C-CAAA TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTTGCCCCAGC-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA	189 189 192 192 192 192 215
SF724 EF488006 Florida J37 NZH3 X1 X03680 PUR1	TAAGGGTAGAGCATAGACTTATGAACAGTGCIGGGACTGTCGCCTCA-C-CAAA TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCTCA-C-CAAA TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCCCAC-C-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2	189 189 192 192 192 192 215 234
SF724 SF724 EF488006 Florida J37 NZH3 X1 X03680 PUR1 PUR2	TAAGGGTAGAGCATAGACTTATAGAACAGTGCIGGGACTGTCGCCTCA-C-CAAA TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCTCA-C-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA * * * ***** * ** ** ** ** *** CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2	189 189 192 192 192 215 234 234
SF724 SF724 EF488006 Florida J37 NZH3 X1 X03680 PUR1 PUR2 PUR3	TAAGGGTAGAGCATAGACTTATGAACAGTGCTGGGACTGTCGCCTCA-C-CAAA TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCTCA-C-CAAA TATGGGTGGGCGCATAGACTTTATGAACAGTGCTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA CCCGT-CGATACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2	189 189 192 192 192 215 234 234 234
SF724 SF724 EF488006 Florida J37 NZH3 X1 X03680 PUR1 PUR2 PUR2 PUR3 YEL1	TAAGGGTAGAGCATAGACTTATGAACAGTGCIGGGACTGTCGCCTCA-C-CAAA TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCTCA-C-CAAA TATGGGTGGGCGCATAGACTTTATGAACAGTGCTCGGACTGTCGCCTCAGC-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA CCCGT-CGATACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGACTC-2	189 189 192 192 192 215 234 234 234 234
SF724 SF724 EF488006 Florida J37 NZH3 X1 X03680 PUR1 PUR2 PUR2 PUR3 YEL1 YEL2	TAAGGGTAGAGCATAGACTTATGAACAGTGCTGGGACTGTGGCCTCA-C-CAAA TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCTCA-C-CAAA TATGGGTGGCGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCCCACC-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATCGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA CCGT-CGATACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2	189 189 192 192 192 215 234 234 234 234 234
SF774 SF724 EF488006 Florida J37 NZH3 X1 X03680 PUR1 PUR2 PUR3 YEL1 YEL2 YEL3	TAAGGGTAGAGCATAGACTTATGACAGTGCIGGGACTGTGCGCCTCA-C-CAAA TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCTCA-C-CAAA TATGGGTGGCGCATAGACTTTATGAACAGTGCTGGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TTCTGCAGTAGACGTTGTTCACGAGTCGTCTCAACACAAC * * * ***** * ** ** ** ** ** ** *******	189 189 192 192 192 215 234 234 234 234 234 234
SF724 SF724 EF488006 Florida J37 NZH3 X1 X03680 PUR1 PUR2 PUR3 YEL1 YEL2 YEL3 P0090	TAAGGGTAGAGCATAGACTTATAGAACAGTGCIGGGACTGTGCGCCTCA-C-CAAA TAAGGGTAGAGCATAGACTTTATGAACAGTGCIGGGACTGTCGCCTCA-C-CAAA TATGGGTGGGCGCATAGACTTTATGAACAGTGCTCGGACTGTCGCCTCA-C-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATCGGTGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA * * * ***** * ** ** ** ** ** *** **** CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGCCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGCCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGCCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGCCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTCGCCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 -	189 189 192 192 192 215 234 234 234 234 234 234
SF724 SF724 EF488006 Florida J37 NZH3 X1 X03680 PUR1 PUR2 PUR3 YEL1 YEL2 YEL2 YEL3 D099 U2107	TAAGGGTAGAGCATAGACTTATAGAACAGTGCIGGGACTGTGCGCCTCA-C-CAAA TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCTCA-C-CAAA TATGGGTGGGCGCATAGACTTTATGAACAGTGCTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TCCTGCAGTAGACGTTGTTCACGAGTCGTCTCAACACACAC * * * ***** * ** * ** * ** ** ** ** ****	189 189 192 192 192 215 234 234 234 234 234 234 234 234
SF070 SF724 EF488006 Florida J37 NZH3 X1 X03680 PUR1 PUR2 PUR3 YEL1 YEL2 YEL3 D099 H3107	TAAGGGTAGAGCATAGACTTATAGAACAGTGCTGGGACTGTCGCCTCA-C-CAAA TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCTCA-C-CAAA TATGGGTGGGCGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCTCAC-C-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATCGGTGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATCGGTGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA * * * ***** * ** ** ** ** ** *** ******	189 189 192 192 192 215 234 234 234 234 234 234 234 234 234 234
SF724 SF724 EF488006 Florida J37 NZH3 X1 X03680 PUR1 PUR2 PUR3 YEL1 YEL2 YEL3 D099 H3107 H3173	TAAGGGTAGAGCATAGACTTATAGAACTTATGAACAGTGCTGGGACTGTGGCGTCA-C-CAAA TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCTCA-C-CAAA TATGGGTGGGCGCATAGACTTTATGAACAGTGCTGTGGACTGTCGCCCCAC-C-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATCGGT-GCATACTGGTGGCTATG-A-GTGTTCACGAGTCGTCTCAACACACAC * * * ***** * ** ** ** ** ** *** ******	189 189 192 192 192 2192 2192 234 234 234 234 234 234 234 234 234 23
F10710 SF724 EF488006 Florida J37 NZH3 X1 X03680 PUR1 PUR2 PUR3 YEL1 YEL2 YEL3 D099 H3107 H3173 AY321482	TAAGGGTAGAGCATAGACTTATATGAACAGTGCTGGGACTGTGCGCTCA-C-CAAA TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCTCA-C-CAAA TATGGGTGGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2	189 189 192 192 192 2192 2192 234 234 234 234 234 234 234 234 234 23
SF724 SF724 EF488006 Florida J37 NZH3 X1 X03680 PUR1 PUR2 PUR3 YEL1 YEL2 YEL3 D099 H3107 H3173 AY321482 K122	TAAGGGTAGAGCATAGACTTATAGAACAGTGCTGGGACTGTGCGCCTCA-C-CAAA TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCTCA-C-CAAA TATGGGTGGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATCGGTGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATCGGTGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA * * * ***** * ** ** ** ** ** ** *******	189 189 192 192 192 215 234 234 234 234 234 234 234 234 234 234
SF724 SF724 EF488006 Florida J37 NZH3 X1 X03680 PUR1 PUR2 PUR3 YEL1 YEL2 YEL3 D099 H3107 H3173 AY321482 K122 OH10	TA = -AGGGTAGAGCATAGACTTATAGAACTTATGAACAGTGCTG = -GGACTGTCGCCTCA - C - CAAA = TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCTCA - C - CAAA = TAGGGTGGCGCCATAGACTTAATGAACAGTACTCTCCGGACTGTTGCCCCAGC - C - CAA = TAGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC - CAAA = TAGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC - CAAA = TAGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC - CAAA = TAGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC - CAAA = TAGGGTGGCCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC - CAAA = TA	189 189 192 192 192 215 234 234 234 234 234 234 234 234 234 2234 2234 2234 2234 2234 2234
SF070 SF724 EF488006 Florida J37 NZH3 X1 X03680 PUR1 PUR2 PUR3 YEL1 YEL2 YEL3 D099 H3107 H3173 AY321482 K122 OH10 EF043441	TA = -AGGGTAGAGCATAGACTTATATGAACAGTGCTG = -GGACTGTCGCCTCA - C-CAAA 1 TA = -AGGGTAGAGCATAGACTTTATGAACAGTGCTG = -GGACTGTCGCCTCA-C-CAAA 1 TAT = -GGGTGGCGCATAGACTTTATGAACAGTACTCTCGGACTGTTGCCTCAGC - CAAA 1 TAT = -GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC - CAAA 1 TTC =TGCAG =TAGACGTT = -GTTCAC = -GA = -GTCGTCCAACACACAC 2 * * * * ***** * ** * * * * * * * * *	189         189         192         192         192         192         234          234          234          234          234          234          234          234          234          234          234          234          234     <
SF724 SF	TA = -AGGGTAGAGCATAGACTTATAGAACTTATGAACAGTGCTG = -GGACTGTCGCCTCA - C = CAAA = TA = -AGGGTAGAGCATAGACTTTATGAACAGTGCTG = -GGACTGTCGCCTCA - C = C = AAA = TA = - AGGGTAGAGCATAGACTTTATGAACAGTGCTG = -GGACTGTCGCCCCAC - C = C = AAA = TA =GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC = CAAA = TA =GGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC = C = C =	189         189         192         192         192         192         234         2
SF724 SF	TA = -AGGGTAGAGCATAGACTTATATGAACAGTGCTG = -GGACTGTGCGCCTCA - C-CAAA = TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCTCA - C-CAAA = TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC - CAAA = TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC - CAAA = TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC - CAAA = TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTGCCTCAGC - CAAA = TATGGGTGCGCCATAGACTTAATGAACAGTACTCTCGGTCAC - TCGAGATCT = -CCGT - CGATTACTGGTGGCTATG - A-GTGACTTCGGTCAC - TCGAGATCT =CCGT - CGATTACTGGTGGCTATG - A-GTGACTTCGGTCAC - CGAGATCT =CCGT - CGATTACTGGTGGCTATG - A-GTGACTTCGGTCAC - TCGAGATCT =CCGT - CGATTACTGGTGGCTATG - A-GTGACTTCGGTCAC - TCGAGATCT =CCGT - CGATTACTGGTGGCTATG - A-GTGACTTCGGTCAC - CGAGATCT =CCGT - CGATTACTGGTGGCTATG - A-GTGACTTCGGTCAC - CGAGATCT =CCGT - CGATTACTGGTGGCTATG - A-GTGACTTCGGTCAC - CGAGATCT =CCGT - CGATTACTGGTGGCTATG - A-GTGACTTCGGTCGCACCGATCACCGTAGATCC =AGATCTCCGTGATTACTGGTGGCTATG - A-GTGACTTCGGTCGCATCGAGATCC =CCGT - CGATTACTGGTGGCTATG - A-GTGACTTCGGTCGCACCAGATCC =CCGTCCGT-CGATTACTGGTGGCTATG - A-GTGACTTCGGTCGCACCGATCC =CCGTCCGTGACTTCGGTCGCACC - CCGAGATCC =CCGTCCGTCCGTACGTGCCATCGAGATCC =CCGT	189 189 192 192 215 234 234 234 234 234 234 234 234 234 234
SF724 SF	TA = -AGGGTAGAGCATAGACTTATATGAACAGTGCTG = -GGACTGTGCGCCTCA - C-CAAA = TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCTCA - C-CAAA = TATGGGTGGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC - CAAA = TATGGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC - CAAA = TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC - CAAA = TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC - CAAA = TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTGCCTCAGC - CAAA = TATGGGTGCGCCATAGACTTAATGAACAGTACTCTCGGTCAC - TCGAGATCT = -CCGT - CGATTACTGGTGGCTATG - A-GTGACTTCGGTCAC - TCGAGATCT =CCGT - CGATTACTGGTGGCTATG - A-GTGACTTCGGTCAC - CCGAGATCT =CCGT - CGATTACTGGTGGCTATG - A-GTGACTTCGGTCAC - CCGAGATCT =CCGT - CGATTACTGGTGGCTATG - A-GTGACTTCGGTCGCAC - CCGAGATCT =CCGT - CGATTACTGGTGGCTATG - A-GTGACTTCGGTCGCACGACTC =CCGT - CGATTACTGGTGGCTATG - A-GTGACTTCGGTCGCACAGATCC =CCGT - CGATTACTGGTGGCTATG - A-GTGACTTCGGTCGCAGATCC =CCGT - CGATTACTGGTGGCTATG - A-GTGACTTCGGTCGCAGATCC =CCGT - CGATTACTGGTGGCTATG - A-GTGACTTCGGTCGCAGATCC =CCGT - CGATTACTGGT	189 189 192 192 215 234 234 234 234 234 234 234 234 234 234
SF724 SF	TAAGGGTAGAGCATAGACTTATATGAACAGTGCTGGGACTGTGCGCTCA-C-CAAA TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCTCA-C-CAAA TATGGGTGGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TCTGCAGTAGACGTTGTTCACGAGTCGTCTCAACACACAC * * * * ***** * ** * * * * * * * * * *	189 189 192 192 2192 2192 234 234 234 234 234 234 234 234 234 23
SF724 SF	$TA = -AGGGTAGAGCATAGACTTATATGAACAGTGCTG = -GGACTGTCGCCTCA - C-CAAA \\ TA = -AGGGTAGAGCATAGACTTTATGAACAGTGCTG = -GGACTGTCGCCTCA-C-CAAA \\ TAT = -GGGTGGCGCCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC - CAAA \\ TAT = -GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC - CAAA \\ TAT = -GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTGCCTCAGC - CAAA \\ TAT = -GGGTGCGCCATAGACTTAATGAACAGTACTCTCGGACACTGCACCACACACA$	189 192 192 2192 2192 2192 2215 234 234 234 234 234 234 234 234 234 234
SF070 SF724 EF488006 Florida J37 NZH3 X1 X03680 PUR1 PUR2 PUR3 YEL1 YEL2 YEL3 D099 H3107 H3173 AY321482 K122 OH10 EF043441 AGC Andong H3014 HF85 NIHHS-KHH05 NIHS-KH05	TA = -AGGGTAGAGCATAGACTTATATGAACAGTGCTG = -GGACTGTCGCCTCA - C-CAAA = TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCTCA - C-CAAA = TA AGGGTAGAGCATAGACTTTATGAACAGTGCTCGGACTGTCGCCTCA-C - CAAA = TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC - CAAA = TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTGCCTCAGC - CAAA = TATGGGTGCGCCATAGACTTAATGAACAGTACTCTCGGACCGCTCAGC - CAAA =	189           189           192           192           192           231           2234           2242           242           242           242           242           242           242           242           242           242           242           242           242           242           242           242           242           242           242
SF724 SF	TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCTCA-C-CAAA TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCTCA-C-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTGCTGGGACTGTCGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACCTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGCCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2	189 189 192 192 215 234 234 234 234 234 234 234 234 234 234
SF724 SF	TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCTCA-C-CAAA TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCTCA-C-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTGCTGGGACTGTCGCCTCAGC-C-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACGTGTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACGTGTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCCGGACCTCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 AACTCAT-CGATAACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATAACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATAACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATAACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATAACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATAACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATAACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-2 -	189         189         192         192         192         215         234         242         242         242         242         242         242         242         242         242         242         242         242         242         242         242         242          242          242          242
SF724 SF744 SF744 SF744 SF744 SF744 SF744 SF744 SF744 SF744 SF744 SF744 SF744 SF744 SF744 SF744 SF744 SF744 SF744 SF	TA AGGGTAGAGCATAGACTTTATGAACAGTGCTG GGACTGTCGCCTCA-C-CAAA TAAGGGTAGAGCATAGACTTTATGAACAGTGCTG GGACTGTCGCCTCA-C-CAAA TAT GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TAT GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TAT GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TAT GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TAT GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TAT GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TAT GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TAT GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTGTGCCTCAGC-CAAA TAT GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTGTGCCTCAGC-CAAA TAT GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGTCAC-TCGAGATCT- C CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT-	189 192 192 215 234 234 234 234 234 234 234 234 234 234
SF070 SF724 EF488006 Florida J37 NZH3 X1 X03680 PUR1 PUR2 PUR3 YEL1 YEL2 YEL3 D099 H3107 H3173 AY321482 K122 OH10 EF043441 AGC Andong H3014 HF85 NIHHS-KH405 N-UK76 KJ938577 SF52 SF670	TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCCCA-C-CAAA TAAGGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCCCA-C-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACCGTGTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACCGTGTGCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGTCAC-TCGAGATCT- CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATTA	189           189           192           192           192           231           2234           2242           242           242           242           242           242           242           242           242           242           242           242           242           242           242           242           242           242           242           242
SF070 SF724 EF488006 Florida J37 NZH3 X1 X03680 PUR1 PUR2 PUR3 YEL1 YEL2 YEL3 D099 H3107 H3173 AY321482 K122 OH10 EF043441 AGC Andong H3014 HF85 NIHHS-KHH05 N-UK76 KJ938577 SF52 SF670 SF724	TA AGGGTAGAGCATAGACTTTATGAACAGTGCTG GGACTGTCGCCTCA-C-CAA TA AGGGTAGAGCATAGACTTTATGAACAGTGCTG GGACTGTCGCCTCA-C-CAA TA AGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TAT GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TAT GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TAT GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TTC TGCAG TAGACGTT GTTCAC GA GTCGTCTCAACACACAC * * * * ***** * ** ** ** ** ** ** ** **	189           189           189           192           192           192           234           242           242           242           242           242           242           242           242           242           242           242           242           242           242      240      2
SF070 SF724 EF488006 Florida J37 NZH3 X1 X03680 PUR1 PUR2 PUR3 YEL1 YEL2 YEL3 D099 H3107 H3173 AY321482 K122 OH10 EF043441 AGC Andong H3014 HF85 NIHHS-KHH05 N-UK76 KJ938577 SF52 SF670 SF724 EF488006	TA AGGGTAGAGCATAGACTTTATGAACAGTGCTG GGACTGTCGCCTCA-C-CAAA TA AGGGTAGAGCATAGACTTTATGAACAGTGCTG GGACTGTCGCCTCAC-C-CAAA TA GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TA GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TA GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TA GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TA GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TA GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TA GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TA GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTGTCCCAGC-CAAA TA GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTGCCTCAGC-CAACACACAC * * * * ***** * ** * ** * **	189           189           192           192           192           234           2234           242           242           242           242           242           242           242           242           242           242           242           242           242           242           242           242
SF724 SF724 SF724 SF724 SF724 SF724 SF724 SF724 SF724 SF724 SF724 SF728 SF728 SF728 SF670 SF724 SF724 SF670 SF724 SF73 SF724 SF73 SF7	TA AGGGTAGAGCATAGACTTTATGAACAGTGCTG GGACTGTGCGCTCA-C-CAAA TA AGGGTAGAGCATAGACTTTATGAACAGTGCTG GGACTGTCGCCTCAC-C-CAAA TAT GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TAT GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TAT GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TAT GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TAT GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TAT GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TAT GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TAT GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TTCTGCAGTAGACGTTGTTCACGAGTCGGTCAC-TCGAGATCT- CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATTACTGGTGGCTATG-A-GT	189         189         192         192         192         234         242         242         242         242         242         242         242         242         242         242         240         2240         2240         2240         2240         240
SF070 SF724 EF488006 Florida J37 NZH3 X1 X03680 PUR1 PUR2 PUR3 YEL1 YEL2 YEL3 D099 H3107 H307 H307 H307 H307 H307 H307 H307 H3	TA AGGGTAGAGCATAGACTTTATGAACAGTGCTG GGACTGTGCGCTCA-C-CAAA TA AGGGTAGAGCATAGACTTTATGAACAGTGCTG GGACTGTCGCCTCAGC-C-CAAA TAT GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TAT GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TAT GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TAT GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TAT GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TAT GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TAT GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TAT GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCTCAGC-CAAA TAT GGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACATTGCCTCAGC-CAAA TAT GCGTGGCGCATAGCA-GT GACTTCGGTCAC-TCG AGATCT- CCGT -CGATTACTGGTGGCTATG -A-GT GACTTCGGTCACATCA AGATCT- CCGT -CGATTACTGGTGGCTATG -A-GT GACTTCGGTCACATCA AGATCT- CCGT -CGATTACTGGTGGCTATG -A-GT GACTTCGGTCACATC	189           189           192           192           192           215           234           240           240           240           240
SF070 SF724 EF488006 Florida J37 NZH3 X1 X03680 PUR1 PUR2 PUR3 YEL1 YEL2 YEL3 D099 H3107 H3173 AY321482 K122 OH10 EF043441 AGC Andong H3014 HF85 NIHHS-KH405 N-UK76 KJ938577 SF52 SF670 SF724 EF488006 Florida J37 NZH3	TA AGGTAGACATAGACTITATGAACAGTGCTGGGACTGTCGCCTCA-C-CAAA TAAGGTAGACATAGACTITATGAACAGTGCTGGGACTGTCGCCCCA-C-CAAA TAGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCCCCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCCCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCCCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCCCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCCCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCCCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCCCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCCCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCCCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACAGCTGTGCCCCAGC-CAAA TCTGCAGTAGACGTTGTCTCCGGTCAC-TCGAGATCT- CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCC- CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCACATCAAGATCT- AACTCAT-CGATAACTGGTGGCTATG-A-GTGACTTCGGTCACATCAAGATCT- AACTCAT-CGATAACTGGTGGCTATG-A-GTGACTTCGGTCACATCAAGATCT- CTAT-CGTGGACCGGTGGCTATG-A-TGTGAC	189           189           192           192           192           234           2242           242           242           242           242           242           242           242           242           242           242           242           242           240           240           240           240           240
SF724 SF724 SF724 SF724 SF724 SF724 SF724 SF724 SF724 SF724 SF724 SF724 SF724 SF724 SF724 SF670 SF724 SF724 SF670 SF724 SF	TA AGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCTCA-C-CAAA TAAGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCCCA-C-CAAA TAGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCCTCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCCCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCCCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCCCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCCCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCCCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACGACTGTGCCCCAGC-CAAA TATGCGTCGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATAACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATAACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATAACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCC- CCGT-CGATAACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATAACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATAACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATAACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATAACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATAACTGGTGGCTATG-A-GTGACTTCGGTCACATCAAGATCT- AACTCAT-CGATAACTGGTGGCTATG-A-GTGACTTCGGTCACATCAAGATCT- CCAT-CGTGAACCTGGTGGCTATG-A-GTGACTTCGGTCACATCAAGATCT- CCAT-CGTGGACCGTGGCTAT	189           189           189           192           192           192           234           240           240           240           240           240           240           240           240
SF724 SF724 SF724 SF724 SF724 SF724 SF724 SF724 SF724 SF724 SF724 SF724 SF06 SF724 SF670 SF724 SF070 SF724 SF070 SF724 SF070 SF724 SF070 SF724 SF070 SF724 SF070 SF724 SF070 SF724 SF070 SF724 SF070 SF724 SF070 SF724 SF070 SF724 SF070 SF724 SF070 SF724 SF070 SF724 SF070 SF070 SF724 SF070 SF724 SF070 SF724 SF070 SF724 SF070 SF0	TA AGGTAGAGCATAGACTITATGAACAGTGCTGGGACTGTCGCCCCA-C-CAAA TAAGGTAGAGCATAGACTTTATGAACAGTGCTGGGACTGTCGCCCCA-C-CAAA TAGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCCCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCCCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCCCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCCCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCCCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCCCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACTGTTGCCCCAGC-CAAA TATGGGTGGCGCATAGACTTAATGAACAGTACTCTCGGACATGTGCCCCAGC-CAAA TATGGGTGGCGCATGGACTATGAACGTACTCCGGACGTGTGCCCCAGC-CAAA TCCCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCAC-TCGAGATCT- CCGT-CGATTACTGGTGGCTATG-A-GTGACTTCGGTCACATCAAGATCT- CCGT-CGATAACTGGTGGCTATG-A-GTGACTTCGGTCACATCAAGATCT- CTAT-CGTGGACCGTGGCTATG-A-GTGA	189           189           192           192           192           234           240           240           240           240           240           240           240           240           240           240           240           240           240           240           240

2	1	Δ
J	T	υ

PUR1	GCTACGCAGAGAGCCTCAATG	AGTT-	-GTTCGTCCAACC-	GCCGGTG	278
PUR2	GCTACGCAGAGAGCCTCAATG	AGTT-	-GTTCGTCCAACC-	GCCGGTG	278
S ALID	CCTA CCCACACACCCCTCAATC	a		GCCGGTG	278
VDI 1				accourd	270
TELT	GCIACGCAGAGAGCCICAAIG	AGII-	-GIICGICCAACC-	GCCGGIG	2/8
YEL2	GCTACGCAGAGAGCCTCAATG	AGTT-	-GTTCGTCCAACC-	GCCGGTG	278
YEL3	GCTACGCAGAGAGCCTCAATG	AGTT-	-GTTCGTCCAACC-	GCCGGTG	278
D099	GCTATGCAGAGAGCCTCAATG	AGTT-	-GTTTGTTCAACC-	GCCGGTG	284
U2107		۸ (۱۳۳		CCCCCTC	270
115107			-GI-ICGICCAACC-	GCCGGIG	270
H3173	GCTACGCAGAGAGCCTCAATG	AG'I''I'-	-GIICGICCAACC-	GCCGGTG	278
AY321482	GCTACGCAGAGAGCCTCAATG	AGTT-	-GTTCGTCCAACC-	GCCGGTG	278
K122	GCTACGCAGAGAGCCTCAATG	AGTT-	-GTTCGTCCAACC-	GCCGGTG	278
OH10	COTTTCCCCCTACCCTCATC	a		GCCGGTG	284
50110					201
EF043441	GCTTTGCGGCTAGCCTCAATG	AG1"1'-	-GII'I'GCI'CAACC-	GCCGGTG	284
AGC	GCTATGCAGAGAGTCTCAATG	AGTT-	-GTGTTCGTCCAACC-	GCCGGTG	288
Andong	GCTATGCAGAGAGTCTCAATG	AGTT-	-GTTCGTCCAACC-	GCCGGTG	287
H3014	GCTATGCAGAGAGTCTCAATG	AGTT-	-GTGTTCGTCCAACC-	GCCGGTG	288
UF 85		 λ			288
		A GII	-GIGIICGICCAACC-	GCCGGIG	200
NIHHS-KHHU5	GCTATGCAGAGAGTCTCAATG	AG1"1'-	-GIICGICCAACC-	GCCGGTG	285
N-UK76	GCTATGCAGAGAGTCTCAATG	AGTT-	-GTGTTCGTCCAACC-	GCCGGTG	288
KJ938577	GCTATGCAGAGAGTCTCAATG	AGTT-	-GTGTTCGTCCAACC-	GCCGGTG	288
SF52	CCTA TCCACACACCCCTCAATC	∆GTT_			284
SE 52				CCCCCTC	201
5F670	GCIAIGCAGAGAGCCICAAIG	AGII-	GIIIGIICAACC-	GCCGGIG	204
SF724	GCTATGCAGAGAGCCTCAATG	AG'I''I'-	-GTTTGTTCAACC-	GCCGGTG	284
EF488006	GCTATGCAGAGAGCCTCAATG	AGTT-	-GTTTGTTCAACC-	GCCGGTG	284
Florida	GCTATGCGGAGAGCCTCAATG	AGTT-	-GTTCC-GTCAA-CC-	GCCTT	282
.T37		λC		<u>-</u> CCC <u>-</u> TT	282
037			-GIICC-GICAA-CC-	GCC II	202
NZH3	GCTATGCGGAGAGCCTCAATG	AG1"1'-	-GINCC-GICAA-CC-	GCC1"I"	282
Xl	GTTTTGCGGAGAGCCTCAATG	AGTT-	-GTTCCCGTCAAACC-	GCCTT	284
X03680	GTTAGTTGACGATAGTCTTGATG	ATCGTGGTTC	GGTTTCGGCCGAGCCA	AGGCGCCAAAA	334
	* * * * * * * * * * *	* ***	** * * **	* * *	
_					
PURI	TCG-ATAG-AAATTTTTT	-TCCTATTA-	ACTTGTT	TCTTG	313
PUR2	TCG-ATAG-AAATTTTT	-TCCTATTA-	ACTTGTT	TCTTG	313
PUR3	TCG-ATAG-AAATTTTT	-TCCTATTA-	ACTTGTT	TCTTG	313
VEL1	ͲϹϹ–ΔͲΔϹ–ΔΔΔͲͲͲͲͲ––––––	-тсстатта-	<u>A</u> CTTGTT		313
VEL 0			ACTICIT	TOTTO	212
IELZ	ICG-AIAG-AAAIIIII	-ICCIAIIA-	ACIIGII	ICIIG	515
YEL3	TCG-ATAG-AAATTTTT	-TCCTATTA-	ACTTGTT	TCTTG	313
D099	TCG-ATAAATCTTAT	-CTCAATTA-	ACTTGTT	TCTTG	318
H3107	TCG-ATAG-AAATTTTT	-ТССТАТТА-	ACTTGTT	TCTTG	313
цз173	<u></u> <u>тсс_лтлс_лллтттт</u>				313
115175		TCCIAIIA	ACTIGIT	TCTTG	212
AY321482	TCG-ATAG-AAATTTTTT	-TCCTATTA-	AC1"I'G1"I'	TCTTG	313
K122	TCG-ATAG-AAATTTTT	-TCCTATTA-	ACTTGTT	TCTTG	313
OH10	TCG-ATAAATCTTAT	-CTCAATTA-	ACTTGTT	TCTTG	318
EF043441	ΤΟΟ-ΔΤΔΔΔΤΟΤΤΔΤ	-стсаатта-	<u>A</u> CTTGTT		318
20010111				TCCTC	222
AGC		-ICCIATIA-	ACTIGIT	ICCIG	223
Andong	TCG-ATAG-AAATTTTTTT	-TCCTATTA-	AC'1"I'G'1"I'	TCCTG	322
Н3014	TCG-ATAG-AAATTTTT	-TCCTATTA-	ACTTGTT	TCCTG	323
HF85	TCG-ATAG-AAATTTTT	-TCCTATTA-	ACTTGTT	TCCTG	323
NTHHS-KHH05	ΤĊĠ-ΑͲΑĠ-ΑΑΑͲͲͲͲͲ	-ТССТАТТА-	ACTTGTT	TCCTG	320
N_UK76		TCCTATCA		TCCTC	222
N=OK76		-ICCIAICA-	ACTIGIT	ICCIG	223
KJ938577	TCG-ATAG-AAATTTTTTT	-TCCTATTA-	AC'1"I'G'1"I'	TCCTG	323
SF52	TCG-ATAAATCTTAT	-CTCAATTA-	ACTTGTT	TCTTG	318
SF670	TCG-ATAAATCTTAT	-CTCAATTA-	ACTTGTT	TCTTG	318
SF724	ΤϹϹ–ΑΤΑ––ΑΑΤϹΤΤΑΤ–––––	-СТСААТТА-	ACTTGTT	TCTTG	318
FF488006	TCC_NTNNNTCTTNT				318
EF408000			ACTIGIT	ICIIG	210
Florida	GCG-ATATTTGATTTTTGGGTAC	GTATCGTTA-		TCTTGG	326
J37	GCG-ATATTTGATTTTTGGGTAC	GTATCGTTA-	TCTTGTC	TCTTGG	326
NZH3	GCG-ATATTTGATTTTTGGGTAC	GTATCGTTA-	TCTTGTC	TCTTGG	326
X1	GCG-ATATTTGATTTTTGGGTAC	GTATCGTTA-	TCTTGTC	TCTTGG	328
V02690					202
X03080	ACACATAAGCACCTITICATAT		FIACACIGATIGCCAAA	AGICIICACGG	394
	· · · · ·	~ ^	~~ ~	~ ~ ~	
PUR1	ATTCGT-GTTAA-	TACATTT	TGGCACAATGTATTAG	CTTCAGCGATG	358
PUR2	ATTCGT-GTTAA-	TACATTT	TGGCACAATGTATTAG	CTTCAGCGATG	358
					250
PORS			TGGCACAAIGIAIIAG	CIICAGCGAIG	350
терт	ATTCGT-GTTAA-	IACATTT	IGGCACAATGTATTAG	CITCAGCGATG	358
YEL2	ATTCGT-GTTAA-	TACATTT	TGGCACAATGTATTAG	CTTCAGCGATG	358
YEL3	<u>-</u>	TACATTT	TGGCACAATGTATTAG	CTTCAGCGATG	358
0000	ALICGI GIIAA				363
D099	AATCGT-GTTAA-	TACATTT	TGGCACAATGTATTAG	CTTCAGCGATG	
H3107	AATCGT-GTTAA-	TACATTT	TGGCACAATGTATTAG	CTTCAGCGATG	358
H3107	AATCGT-GTTAA-	TACATTI	TGGCACAATGTATTAG TGGCACAATGTATTAG	CTTCAGCGATG CTTCAGCGATG	358
H3107 H3173	ATCGT-GTTAA- ATTCGT-GTTAA- ATTCGT-GTTAA-	TACATTI TACATTI TACATTI	TGGCACAATGTATTAG TGGCACAATGTATTAG TGGCACAATGTATTAG	CTTCAGCGATG CTTCAGCGATG CTTCAGCGATG	358 358
H3107 H3173 AY321482	ATTCGT GTTAA ATTCGT-GTTAA ATTCGT-GTTAA ATTCGT-GTTAA	TACATTT TACATTT TACATTT TACATTT	'TGGCACAATGTATTAG 'TGGCACAATGTATTAG 'TGGCACAATGTATTAG 'TGGCACAATGTATTAG	CTTCAGCGATG CTTCAGCGATG CTTCAGCGATG CTTCAGCGATG	358 358 358
H3107 H3173 AY321482 K122	ATTCGT-GTTAA ATTCGT-GTTAA- ATTCGT-GTTAA- ATTCGT-GTTAA- ATTCGT-GTTAA-	TACATTT TACATTT TACATTT TACATTT TACATTT TACATTT	'TGGCACAATGTATTAG 'TGGCACAATGTATTAG 'TGGCACAATGTATTAG 'TGGCACAATGTATTAG 'TGGCACAATGTATTAG	CTTCAGCGATG CTTCAGCGATG CTTCAGCGATG CTTCAGCGATG CTTCAGCGATG	358 358 358 358
H3107 H3173 AY321482 K122 OH10	ATTCGT-GTTAA- ATTCGT-GTTAA- ATTCGT-GTTAA- ATTCGT-GTTAA- ATTCGT-GTTAA- ATTCGT-GTTAA-	TACATTT TACATTT TACATTT TACATTT TACATTT TACATTT	TGGCACAATGTATTAG TGGCACAATGTATTAG TGGCACAATGTATTAG TGGCACAATGTATTAG TGGCACAATGTATTAG TGGCACAATGTATTAG	CTTCAGCGATG CTTCAGCGATG CTTCAGCGATG CTTCAGCGATG CTTCAGCGATG CTTCAGCGATG	358 358 358 358 358 363
H3107 H3173 AY321482 K122 OH10 EF043441	ATTCGT GTTAA ATTCGT GTTAA ATTCGT GTTAA 	TACATTT TACATTT TACATTT TACATTT TACATTT TACATTT TACATTT	TGGCACAATGTATTAG TGGCACAATGTATTAG TGGCACAATGTATTAG TGGCACAATGTATTAG TGGCACAATGTATTAG TGGCACAATGTATTAG	CTTCAGCGATG CTTCAGCGATG CTTCAGCGATG CTTCAGCGATG CTTCAGCGATG CTTCAGCGATG	358 358 358 358 358 363 363
H3107 H3173 AY321482 K122 OH10 EF043441	ATTCGT GTTAA ATTCGT-GTTAA ATTCGT-GTTAA ATTCGT-GTTAA ATTCGT-GTTAA ATTCGT-GTTAA ATTCGT-GTTAA ATTCGT-GTTAA	TACATTI TACATTI TACATTI TACATTI TACATTI TACATTI TACATTI	TGGCACAATGTATTAG TGGCACAATGTATTAG TGGCACAATGTATTAG TGGCACAATGTATTAG TGGCACAATGTATTAG TGGCACAATGTATTAG TGGCACAATGTATTAG	CTTCAGCGATG CTTCAGCGATG CTTCAGCGATG CTTCAGCGATG CTTCAGCGATG CTTCAGCGATG CTTCAGCGATG	358 358 358 358 363 363 363
H3107 H3173 AY321482 K122 OH10 EF043441 AGC	ATTCGT GTTAA ATTCGT-GTTAA ATTCGT-GTTAA ATTCGT-GTTAA ATTCGT-GTTAA ATTCGT-GTTAA ATTCGT-GTTAA ATTCGT-GTTAA	TACATTT TACATTT TACATTT TACATTT TACATTT TACATTT TACATTT TACATTT	TGGCACAATGTATTAG TGGCACAATGTATTAG TGGCACAATGTATTAG TGGCACAATGTATTAG TGGCACAATGTATTAG TGGCACAATGTATTAG TGGCACAATGTATTAG TGGCACAATGTATTAG	CTTCAGCGATG CTTCAGCGATG CTTCAGCGATG CTTCAGCGATG CTTCAGCGATG CTTCAGCGATG CTTCAGCGATG CTTCAGCGATG	358 358 358 358 363 363 363 363
H3014	ATTCGT-GTTAATACATTTTGGCACAATGTATTAGCTTCAGCGATG	368			
-----------------	---	-----			
HF85	ATTCGT-GTTAATACATTTTGGCACAATGTGTTAGCTTCAGCGATG	368			
NIHHS-KHH05	ATTCGT-GTTAATACATTTTGGCACAATGTATTAGCTTCAGCGATG	365			
N-UK76	ATTCGT-GTTAATACATTTTGGCACAATGTATTAGCTTCAGCGATG	368			
КЛ938577	ATTCGT-GTTAATACATTTTGGCACAATGTATTAGCTTCAGCGATG	368			
SF52		363			
SF 52 SF 670		262			
SF070		202			
SF/24	ATTCGT-GTTAATACATTTTTGGCACAATGTATTAGCTTCAGCGATG	363			
EF488006	ATTCGT-GTTAATACATTTTGGCACAATGTATTAGCTTCAGCGATG	363			
Florida	ATACGT-GTTAATACATTTTGGCACAATGTATTAGCTTCAGCGATG	371			
J37	ATACGT-GTTAATACATTTTGGCACAATGTATTAGCTTCAGCGATG	371			
NZH3	ATACGT-GTTAATACATTTTGGCACAATGTATTAGCTTCAGCGATG	371			
X1	ATACGT-GTTAATACATTTTGGCACAATGTATTAGCTTCAGCGATG	373			
x03680	acarcccccrcarrerrerrerreracarcarcaraaraaceracer	452			
	* ** *** * * *** * * * ***				
PUR1	GATCGGTTGATTCGCGTATCGATGAAAAACGCAGCAAGCTGCGTTATTTACCACGAATTG	418			
PUR2	GATCGGTTGATTCGCCGTATCGATGAAAAACGCAGCAAGCTGCGTTATTTACCACGAATTG	418			
2 קווס	GATCGGTTGATTCGCGTATCGATGAAAAACCGCAGCAACCTGCGTTATTACCACGAATTG	418			
VET 1		110			
IELI	GATCGGTTGATTCGCGTATCGATGATAAAACGCAGCAGCAGCTGCGTTATTTACCACGAATTG	410			
YELZ	GATCGGTTGATTCGCGTATCGATGAAAAACGCAGCAAGCTGCGTTATTTACCACGAATTG	418			
YEL3	GATCGGTTGATTCGCGTATCGATGAAAAACGCAGCAAGCTGCGTTATTTACCACGAATTG	418			
D099	GATCGGTTGATTCGCGTATCGATGAAAAACGCAGCAAGCTGCGTTATTTACCACGAATTG	423			
H3107	GATCGGTTGATTCGCGTATCGATGAAAAACGCAGCAAGCTGCGTTATTTACCACGAATTG	418			
H3173	GATCGGTTGATTCGCGTATCGATGAAAAACGCAGCAAGCTGCGTTATTTACCACGAATTG	418			
AY321482	GATCGGTTGATTCGCGTATCGATGAAAAACGCAGCAAGCTGCGTTATTTACCACGAATTG	418			
К122	GATCGGTTGATTCGCGTATCGATGAAAAACGCAGCAAGCTGCGTTATTTACCACGAATTG	418			
OH10	GATCGGTTGATTCGCGTATCGATGAAAAACCGCAGCAAGCTGCGTTATTTACCACGAATTG	423			
FF043441		423			
PLOIDIT		120			
AGC	GATCGGTTGATTCGCGTATCGATGATAAAACGCAGCAGCAGCTGCGTTATTTACCACGAATTG	420			
Andong	GATCGGTAGATTCGCGTATCGATGAAAAACGCAGCAAGCTGCGTTATTTACCACGAATTG	427			
H3014	GATCGGTTGATTCGCGTATCGATGAAAAACGCAGCAAGCTGCGTTATTTACCACGAATTG	428			
HF85	GATCGGTTGATTCGCGTATCGATGAAAAACGCAGCAAGCTGCGTTATTTACCACGAATTG	428			
NIHHS-KHH05	GATCGGTTGATTCGCGTATCGATGAAAAACGCAGCAAGCTGCGTTATTTACCACGAATTG	425			
N-UK76	GATCGGTTGATTCGCGTATCGATGAAAAACGCAGCAAGCTGCGTTATTTACCACGAATTG	428			
КЛ938577	GATCGGTTGATTCGCGTATCGATGAAAAACGCAGCAAGCTGCGTTATTTACCACGAATTG	428			
SE2	GATCGGTTGATTCGCGTATCGATGAAAAACCGCAGCAACCTGCGTTATTACCACGAATTG	423			
SF 52 SF 670		122			
0070		400			
DF / 24		400			
EF488006	GATCGGTTGATTCGCGTATCGATGAAAAACGCAGCAAGCTGCGTTATTTACCACGAATTG	423			
Florida	GATCGGTTGATTCGCGTATCGATGAAAAACGCAGCTAGCT	431			
J37	GATCGGTTGATTCGCGTATCGATGAAAAACGCAGCTAGCT	431			
NZH3	GATCGGTTGATTCGCGTATCGATGAAAAACGCAGCTAGCT	431			
Xl	GATCGGTTGATTCGCGTATCGATGAAAAACGCAGCTAGCT	433			
X03680	GATCGGTTGCATCGAGTATCGATGAAGAACGCAGCTTGCTGCGTTACTTAC	512			
	****** * *** ********* ****** ******				
PUR1	CAGACGCTTAGGGTTGGTGAAGTTTTGAACGCACAGCGCCGTTGGGTTTTCCCTTCGGCA	478			
PUR2	CAGACGCTTAGAGT-GGTGAAGTTTTGAACGCACAGCGCCGTTGGGTTTTCCCTTCGGCA	477			
PUR 3	CAGACGCTTAGAGT-GGTGAAGTTTTGAACGCACACGCCGTTGGGTTTTCCCTTCGGCA	477			
VFT.1		477			
VEL 0		177			
IELZ		4//			
YEL3	CAGACGCTTAGGGT-GGTGAAGTTTTGAACGCACAGCGCCGTTGGGTTTTCCCTTCGGCA	4//			
D099	CAGACGCTTAGAGT-GGTGAAGTTTTTGAACGCACAGCGCCGTTGGGTTTTCCCTTCGGCA	482			
H3107	CAGACGCTTAGAGT-GGTGAAGTTTTGAACGCACAGCGCCGTTGGGTTTTCCCTTCGGCA	477			
H3173	CAGACGCTTAGAGT-GGTGAAGTTTTGAACGCACAGCGCCGTTGGGTTTTCCCTTCGGCA	477			
AY321482	CAGACGCTTAGAGT-GGTGAAGTTTTGAACGCACAGCGCCGTTGGGTTTTCCCTTCGGCA	477			
K122	CAGACGCTTAGAGT-GGTGAAGTTTTGAACGCACAGCGCCGTTGGGTTTTCCCTTCGGCA	477			
OH10	CAGACGCTTAGAGT-GGTGAAGTTTTGAACGCACAGCGCCGTTGGGTTTTCCCTTCGGCA	482			
FF043441		482			
ACC		487			
AGC		107			
Andong		400			
H3U14	CAGACGC TTAGAGT-GGTGAAGTTTTTGAACGCACAGCGCCGTTGGGTTTCCCCCTTCGGCA	487			
HF.82	CAGACGCTTAGAGT-GGTGAAGTTTTGAACGCACAGCGCCGTTGGGTTTTCCCTTCGGCA	487			
NIHHS-KHH05	CAGACGCTTAGAGT-GGTGAAGTTTTGAACGCACAGCGCCGTCGGGTTTTCCCTTCGGCA	484			
N-UK76	CAGACGCTTAGAGT-GGTGAAGTTTTGAACGCACAGCGCCGTTGGGTTTTCCCTTCGGCA	487			
КЈ938577	CAGACGCTTAGAGT-GGTGAAGTTTTGAACGCACAGCGCCGTTGGGTTTTCCCTTCGGCA	487			
SF52	CAGACGCTTAGAGT-GGTGAAGTTTTGAACGCACAGCGCCGTTGGGTTTTCCCTTCGGCA	482			
SF670	CAGACGCTTAGAGT-GGTGAAGTTTTGAACGCACAGCGCCCGTTGGGTTTTCCCTTCGGCA	482			
SF724	CAGACGCTTAGAGT-GGTGAAGTTTTGAACGCACAGCCCCCGTTGGGCTTTCCCCTTCCCCCA	482			
/ FF488006		400			
EF TOOUUD		102			
riorida		490			
157	CAGACGCTTAGAGT-GGTGAAGTTTTGAACGCACAGCGCCGTTGGGTTTTCCCTTCGGCA	490			
NZH3	CAGACGC'ITAGAGT-GGTAAAGTTTTGAACGCACAGCGCCGTTGGGTTTTCCCTTCGGCA	490			
Xl	CAGACGCTTAGAGT-GGTGAAGTTTTGAACGCACAGCGCCGTTGGGTTTTCCCTTCGGCA	492			
X03680	CAGACGCTTAGAGT-GGTGAAATTTCGAACGCATAGCACCAACTGGGCCTCCAGTTGGTA	571			

PUR1	CGTCTGGCTCAGGGTTGTTTA	AAGTACTGCGGTGTTGCTG	CG-AAAGTAGCAACAC	533
PUR2	CGTCTGGCTCAGGGTTGTTTA	AAGTACTGCGGTGTTGCTG	CG-AAAGTAGCAACAC	532
PUR3	CGTCTGGCTCAGGGTTGTTTA	-AAGTACTGCGGTGTTGCTG	CG-AAAGTAGCAACAC	532
YEL1	CGTCTGGCTCAGGGTTGTTTA	-AAGTACTGCGGTGTTGCTG	CG-AAAGTAGCAACAC	532
YEL2	CGTCTGGCTCAGGGTTGTTTA	-AAGTACTGCGGTGTTGCTG	CG-AAAGTAGCAACAC	532
YEL3	CGTCTGGCTCAGGGTTGTTTA	-AAGTACTGCGGTGTTGCTG	CG-AAAGTAGCAACAC	532
D099	CGTCTGGCTCAGGGTTGTTTG	-AAGTACTACGGTGTTGCTG	CG-AAAGTAGCAACAT	537
H3LU/		-AAGTACTGCGGTGTTGCTGC		532 E22
TJ1/5 NV201/00				532
k100		-AAGIACIGCGGIGIIGCIGC	C-AAAGIAGCAACAC	532
0H10	CGTCTGGCTCAGGGTTGTTTA	-AAGTACTGCGGTGTTGCTG	C-AAAGTAGCAACAC	537
EF043441	CGTCTGGCTCAGGGTTGTTTA	-AAGTACTGCGGTGTTGCTG	G-AAAGTGGCAACAC	537
AGC	CGTCTGGCTCAGGGTTGTATATA	-GAGTACTGCGGTGTTGCTC	CG-AGAGTAGCAATGT	544
Andong	CGTCTGGCTCAGGGTTGTTTA	-GAGTACTGCGGTGTTGCTG	CG-AAAGTAGCAATGT	541
H3014	CGTCTGGCTCAGGGTTGTCTATA	-GAGTACTGCGGTGTTGCTC	CG-AGAGTAGCAATGT	544
HF85	CGTCTGGCTCAGGGTTGTTTATATA	AGAGTACTGCGGTGTTGCTC	CG-AGAGTAGCAATGT	546
NIHHS-KHH05	CGTCTGGCTCAGGGTTGTTTA	-GAGTACTGCGGTGTTGCTG	CG-AAAGTAGCAATGT	539
N-UK76	CGTCTGGCTCAGGGTTGTCTATA	-GAGTACTGCGGTGTTGCTC	CG-AGAGTAGCAATGT	544
KJ938577	CGTCTGGCTCAGGGTTGTTTATATA	AGAGTACTGCGGTGTTGCTC	CG-AGAGTAGCAATGT	546
SF52	CGTCTGGCTCAGGGTTGTTTA	-AAGTACTGCGGTGTTGCTG	CG-AAAGTAGCATCAT	537
SF670	CGTCTGGCTCAGGGTTGTTTA	-AAGTACTGCGGTGTTGCTG	CG-AAAGTAGCATCAT	537
SF724	CGTCTGGCTCAGGGTTGTTTA	-AAGTACTGCGGTGTTGCTGC	CG-AAAGTAGCATCAT	537
EF488006	CG1111GGG1CAGGG11G111A	-AAGTACTGCGGTGTTGCTGC	CG-AAAGTAGCATCAT	537
Florida		-AAGTACTGCGGTGTTGCTGC		545
N2H3				545
N2H3 X1		-AAGIACIGCGGIGIIGCIGC	CG-AAAGIGGCAACGC	545
X03680	CGTCTGGTTCAGGGTTGTTTAACTC	AAGIACIGCCGIGIIGCIGC	GGAGAGTCTT	626
100000	*** *** ******* *	* ** ** * **	* * ***	020
PUR1	CGA-ATGTCGAACGGCGTTAGTG	TT-AC	-TACTGGCCCCGTTCT	574
PUR2	CGA-ATGTCGAACGGCGTTAGTG	TT-AC	-TACTGGCCCCGTTCT	573
PUR3	CGA-ATGTCGAACGGCGTTAGTG	TT-AC	-TACTGGCCCCGTTCT	573
YEL1	CGA-ATGTCGAACGGCGTTAGTG	TT-AC	-TACTGGCCCCGTTCT	573
YEL2	CGA-ATGTCGAACGGCGTTAGTG	TT-AC	-TACTGGCCCCGTTCT	573
YEL3	CGA-ATGTCGAACGGCGTTAGTG	TT-AC	-TACTGGCCCCGTTCT	573
D099	CGA-ATGTCGAGCGGCGTTAGTG	TT-AC	-TACTGGCCCCGTTCT	578
H3107	CGA-ATGTCGAACGGCGTTAGTG	TT-AC	-TACTGGCCCCGTTCT	573
H3173	CGA-ATGTCGAACGGCGTTAGTG	TT-AC	TACTGGCCCCGTTCT	573
AY321482	CGA-ATGTCGAACGGCGTTAGTG		-TACTGGCCCCGTTCT	573
AIZZ OH10		II-AC		5/3
FF043441	CGA-TTGTCGAACGGCGTTAGTG	TT-AC		578
AGC	CGA-GTGTCGAACGCCGTTAGTG	TTGGT	-TACTGGCCCCGTTCT	586
Andong	CGA-GTGTCGAACGGCGTTAGTG	TTTGC	-TACTGGCCCCGTTCT	583
H3014	CGA-GTGTCGAACGGCGTTAGTG	 	-TACTGGCCCCGTTCT	586
HF85	CGA-GTGTCGAACGGCGTTAGTG	TTGGC	-TACTGGCCCCGTTCT	588
NIHHS-KHH05	CGA-GTGTCGAACGGCGTTAGTG	TTTGC	-TACTGGCCCCGTTCT	581
N-UK76	CGA-GTGTCGAACGGCGTTAGTG	TTGGT	-TACTGGCCCCGTTCT	586
KJ938577	CGA-GTGTCGAACGGCGTTAGTG	TTGGC	-TACTGGCCCCGTTCT	588
SF52	CGA-ATGTCGAACGGCGTTAGTG	TT-AC	-TACTGGCCCCGTTCT	578
SF670	CGA-ATGTCGAACGGCGTTAGTG	TT-AC	-TACTGGCCCCGTTCT	578
SF724	CGA-ATGTCGAACGGCGTTAGTG	TT-AC	-TACTGGCCCCGTTCT	578
EF488006	CGA-ATGTCGAACGGCGTTAGTG	TT-AC	TACTGGCCCCGTTCT	578
Florida	CGA-ATGTCGAGCGGCGTTAGTG		-TACTGGCCCCGTTCT	586
U S / N7U2				500
V1	CGA-ATGICGAGCGGCGTTAGIG-	TT-AC		588
X03680	CCCCTTCTCCCCCCACATTACTCAC			686
105000	** **** * * *****	**	***** * ** **	000
PUR1	AGTTCCGAA			583
PUR2	AGTTCCGAA			582
PUR3	AGTTCCGAA			582
YEL1	AGTTCCGAA			582
YEL2	AGTTCCGAA			582
YEL3	AGTTCCGAA			582
D099	AGTTCCGAA			587
H3107	AG'ITCCGAA			582
H31/3	AGTTCCGAA			582
AI321482 V100				502
0410				50∠ 527
EF043441	AGTTCCGAA			587
				201

Andong	AGTTCCGAA		5	92
H3014	AGTTCCGAA		5	95
HF85	AGTTCCGAA		5	597
NIHHS-KHH05	AGTTCCGAA		5	590
N-IIK76	AGTTCCGAA		5	95
VT020577	ACTTCCCAA		5	:07
0950577	AGTICCGAA		J	- 07
SF5Z	AGTICCGAA		5	187
SF670	AGTTCCGAA		5	·8./
SF724	AGTTCCGAA		5	87
EF488006	AGTTCCGAA		5	87
Florida	AGTTCCGAA		5	95
J37	AGTTCCGAA		5	595
NZH3			5	95
v1	λατταααλ		5	:07
X02600				116
X03080	AGICCAGAAGCAICACAAG	ICAAGACAGAGCIICACCGIICIGICIIG	TAAIGIGACGCG /	40
	····			
PURI	TATTGG	CAACATGTCTTCT	TCGTGAAG 6	10
PUR2	TATTGG	CAACATGTCTTCT	TCGTGAAG 6	,09
PUR3	TATTGG	CAACATGTCTTCT	TCGTGAAG 6	609
YEL1	TATTGG	CAACATGTCTTCT	TCGTGAAG 6	09
YEL2	TATTGG	CAACATGTCTTCT	TCGTGAAG 6	509
YEL3	TATTGG	CAACATGTCTTCT	TCGTGAAG 6	509
099	TATTGG	СААСАТСТСТТС Т	TCGTGAAG 6	514
H3107	TATTGG	TAACATGTCTTCT	TCGTGAAG 6	:09
u2172			TCCTCAAC 6	:00
TJ1/J	IATIGG		ICGIGAAG 0	109
AY321482	IAIIGG		ICGIGAAG 6	109
K122	TATTGG	CAACATGTCTTCT	ACGTGAAG 6	,09
OH10	TATCGG	CAACATGTCTTCT	TCGTGAAG 6	514
EF043441	TATCGG	CAACATGTCTTCT	TCGTGAAG 6	514
AGC	TATTGG	CAACATGTCATCACGTGT	TCGTGACG 6	527
Andong	TATTGG	CAACATGTCGTCACGTCT	TCGTGACG 6	524
н3014	TATTGG		TCGTGACG 6	27
HF85	TATTGG		TCGTGACG 6	:29
NTUUC_VUU^E	TATIOO	CARCATOTCATCACOTCT	TCOTOACO O	:22
NING-KHOS	IAIIGG		magmanag (	- 22
N-UK/6	IAIIGG		ICGIGACG 6	121
KJ938577	TATTGG	CAACATGTCATCACGTCT	TCGTGACG 6	,29
SF52	TATTGG	CAACATGTCTTCT	TCGTGAAG 6	514
SF670	TATTGG	CAACATGTCTTCT	TCGTGAAG 6	514
SF724	TATTGG	CAACATGTCTTCT	TCGTGAAG 6	514
EF488006	TATTGG	CAACATGTCTTCT	TCGTGAAG 6	514
Florida	TATTGG	CAACATGTCTTCT	TCGTGAAG 6	522
	TATTGG	@ A @ A # @ # @ # @ # # # # # # # # # #	TCGTGAAG 6	:22
N7U2	TATTCC		TCCTCAAC 6	:22
V1	TATIGG		TOTTONA C	22
AT X02C00			CTTCGIGAAG 0	000
XU3080	CIACICIIAIAICGGAGGAG	GCIGCCAAAIGGCGICCGACIIGACICGG	CIGAICAICAAG 8	00
_				
PUR1	ACGCCGTAG	AGAGTAT-AAGTCTGTACCTATG	GATGTGTCG- 6	,50
PUR2	ACGCCGTAG	AGAGTAT-AAGTCTGTACCTATG	GATGTGTCG- 6	549
PUR3	ACGCCGTAG	AGAGTAT-AAGTCTGTACCTATG	GATGTGTCG- 6	649
YEL1	ACGCCGTAG	AGAGTAT-AAGTCTGTACCTATG	GATGTGTCG- 6	549
YEL2	ACGCCGTAG	AGAGTAT-AAGTCTGTACCTATG	GATGTGTCG- 6	549
YEL3	ACGCCGTAG	AGAGTAT-AAGTCTGTACCTATG	GATGTGTCG- 6	549
0990	ACGCCGTGG-		GATGTGTCG- 6	54
H3107	ACCCCCTAC		GATGTGTCC- 6	349
113107	ACCC CGIAG	AGAGIAI AAGI CIGIACCIAIG	GAIGIGICG- 0	: 10
H31/3	ACGCCGIAG		GAIGIGICG- 6	- 40
AY321482	ACGCCGIAG	AGAGIAI-AAGICIGIACCIAIG	GAIGIGICG- 6	,49
K122	ACGCCGTAG	AGAGTAT-AAGTCTGTACCTATG	GATGTGTCG- 6	,49
OH10	ACGCCGTGG	AGAGTAT-AAGTCTGTACCTGTG	GATGTGTCG- 6	,54
EF043441	ACGCCGTGG	AGAGTAT-AAGTCTGTACCTGTG	GATGTGTCG- 6	54
AGC	ACGCCGTAG	AGAGTAT-AGGTCTGTACCTGTG	GATGTGTAG- 6	67
Andong	ACGCCGTAG	AGAGTAT-AAGTCTGTACCTGTG	GATGTGTAG- 6	64
Н3014	ACGCCGTAG	AGAGTAT-AGGTCTGTACCTGTG	GATGTGTAG- 6	67
HF85	ACGCCGTAG	AGAGTAT-AAGTCTGTACCTGTG	GATGTGTAG- 6	66
NTHHS-KHH05	ACCCCGTAG-		GATGTGTAG- 6	62
N_TIK76				67
N-OU/0	ACGCCGIAG		- GAIGIGIAG- 0	101
V0 2220 / /	ACGCCGTAG	AGAGIAI-AAGTCTGTACCTGTG	GAIGIGIAG- 6	109
SF52	ACGCCGTGG	AGAGTAT-AAGTCTGTACCTATG	GATGTGTCG- 6	,54
SF670	ACGCCGTGG	AGAGTAT-AAGTCTGTACCTATG	GATGTGTCG- 6	,54
SF724	ACGCCGTGG	AGAGTAT-AAGTCTGTACCTATG	GATGTGTCG- 6	54
EF488006	ACGCCGTGG	AGAGTAT-AAGTCTGTACCTATG	GATGTGTCG- 6	54
Florida	ACGCCGTAG	AGAGTAT-AAGTCTGTACCTATG	GATGTGTCG- 6	62
J37	ACGCCGTAG	AGAGTAT-AAGTCTGTACCTATG	GATGTGTCG- 6	62
NZH3	ACCCCCTAC		GATGTGTCC- 6	62
v1				61
AT X02600			GAIGIGICG- 0	04
V03000	**		****** *	00

PUR1		CGTATGAA	- ATAT -	GATGCT	668
PUR2		CGTATGAA	- ATAT-	GATGCT	667
PUR3		CGTATGAA	- ATAT-	GATGCT	667
YEL1		CGTATGAA	- ATAT-	GATGCT	667
YEL2		CGTATGAA	- ATAT-	GATGCT	667
YEL3		CGTATGAA	- ATAT-	GATGCT	667
 D099		ССТАТСАА			672
H3107		CGTATGAA		GATGCT	667
u2172		CGTATGAA		-CAT CCT	667
AV201/00		COTATOAA		CAT CCT	667
AI321402		CCTATGAA	- AIAI -	GAIGCI	667
NIZZ OW10		CGIAIGAA	- AIAI-	GAIGCI	607
OHIO		CGTATGAA	A'I'A'I'-	GATGCT	672
EF043441		CGTATGAA	- – A.I.A.I	GATGCT	672
AGC		CGTATGAA	ATAT-	GATGCT	685
Andong		CGTATGAA	ATAT-	GATGCT	682
Н3014		CGTATGAA	ATAT-	GATGCT	685
HF85		CGTATGAA	ATAT-	GATGCT	687
NIHHS-KHH05		CGTATGAA	ATAT-	GATGCT	680
N-UK76		CGTATGAG	- ATAT-	GATGCT	685
KJ938577		CGTATGAA	- ATAT-	GATGCT	687
SF52		CGTATGAA	- ATAT-	GATGCT	672
SF670		CGTATGAA	- ATAT-	GATGCT	672
SF724		CGTATGAA	- ATAT-	GATGCT	672
EF488006		CGTATGAA	ATAT-	GATGCT	672
Florida		CGTATGAA	- – АТАТ-	GATGCT	680
.T37		CGTATGAA		GATGCT	680
N7U3		CGTATGAA		-CAT CCT	680
v1		CGTATGAA		-CAT CCT	682
X02600		CGIAIGAA TTCCTCTACTCCTCTCTCTCTCTCTCTCTCTCTCTCT			002
X03000	CGCICCIIICGAAIACIGGGA	*** **	++++	-++ +	920
				~~ ~	
D				<b></b>	
PUR1	TCCCATACATAGCG	AGGAGGTGCTTCTTATGCT	TAT-CT	705	
PUR1 PUR2	TCCCATACATAGCG TCCCATACATAGCG	AGGAGGTGCTTCTTATGCT	TAT-CT TAT-CT	705 704	
PUR1 PUR2 PUR3	TCCCATACATAGCG TCCCATACATAGCG TCCCATACATAGCG	AGGAGGTGCTTCTTATGC1 AGGAGGTGCTTCTTATGC1 AGGAGGTGCTTCTTATGC1	FAT-CT FAT-CT FAT-CT	705 704 704	
PUR1 PUR2 PUR3 YEL1	TCCCATACATAGCG TCCCATACATAGCG TCCCATACATAGCG TCCCATACATAGCG	AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT	FAT-CT FAT-CT FAT-CT FAT-CT	705 704 704 704	
PUR1 PUR2 PUR3 YEL1 YEL2	TCCCAT-ACATAGCG TCCCAT-ACATAGCG TCCCAT-ACATAGCG TCCCAT-ACATAGCG TCCCAT-ACATAGCG	AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT	TAT-CT TAT-CT TAT-CT TAT-CT TAT-CT	705 704 704 704 704	
PUR1 PUR2 PUR3 YEL1 YEL2 YEL3	TCCCAT-ACATAGCG TCCCAT-ACATAGCG TCCCAT-ACATAGCG TCCCAT-ACATAGCG TCCCAT-ACATAGCG TCCCAT-ACATAGCG	AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT	FAT-CT FAT-CT FAT-CT FAT-CT FAT-CT FAT-CT	705 704 704 704 704 704 704	
PUR1 PUR2 PUR3 YEL1 YEL2 YEL3 D099	TCCCAT-ACATAGCG TCCCAT-ACATAGCG TCCCAT-ACATAGCG TCCCAT-ACATAGCG TCCCAT-ACATAGCG TCCCAT-ACATAGCG TCCCAT-ACATAGCG	AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT	FAT-CT FAT-CT FAT-CT FAT-CT FAT-CT FAT-CT FAT-CT F-ATCT	705 704 704 704 704 704 704 709	
PUR1 PUR2 PUR3 YEL1 YEL2 YEL3 D099 H3107	TCCCAT - ACATAGCG TCCCAT - ACATAGCG TCCCAT - ACATAGCG TCCCAT - ACATAGCG TCCCAT - ACATAGCG TCCCAT - ACATAGCG TCCCAT - ACATAGCG	AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATACT AGGAGGTGCTTCTTATGCG	FAT-CT FAT-CT FAT-CT FAT-CT FAT-CT FAT-CT F-ATCT GAAACT	705 704 704 704 704 704 709 705	
PUR1 PUR2 PUR3 YEL1 YEL2 YEL3 D099 H3107 H3173	TCCCAT - ACATAGCG TCCCAT - ACATAGCG	AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGGG AGGAGGTGCTTCTTATGGG	FAT-CT FAT-CT FAT-CT FAT-CT FAT-CT FAT-CT F-ATCT GAAACT GAAACT	705 704 704 704 704 704 704 709 705 705	
PUR1 PUR2 PUR3 YEL1 YEL2 YEL3 D099 H3107 H3173 AY321482	TCCCAT - ACATAGCG TCCCAT - ACATAGCG	AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGGG AGGAGGTGCTTCTTATGGG AGGAGGTGCTTCTTATGGG	FAT-CT FAT-CT FAT-CT FAT-CT FAT-CT FAT-CT F-ATCT GAAACT GAAACT FAT-CT	705 704 704 704 704 704 704 705 705 705 705	
PUR1 PUR2 PUR3 YEL1 YEL2 YEL3 D099 H3107 H3173 AY321482 K122	TCCCAT - ACATAGCG TCCCAT - ACATAGCG	AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGGC AGGAGGTGCTTCTTATGGC AGGAGGTGCTTCTTATGGC	FAT-CT FAT-CT FAT-CT FAT-CT FAT-CT FAT-CT FAT-CT GAAACT FAT-CT FAT-CT	705 704 704 704 704 704 709 705 705 705 704 704	
PUR1 PUR2 PUR3 YEL1 YEL2 YEL3 D099 H3107 H3173 AY321482 K122 OH10	TCCCAT - ACATAGCG TCCCAT - ACATAGCG	AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCC AGGAGGTGCTTCTTATGGC AGGAGGTGCTTCTTATGCC AGGAGGTGCTTCTTATGCC AGGAGGTGCTTCTTATGCC	FAT-CT FAT-CT FAT-CT FAT-CT FAT-CT F-ATCT JAAACT FAT-CT FAT-CT F-ATCT	705 704 704 704 704 709 705 705 705 705 704 704 709	
PUR1 PUR2 PUR3 YEL1 YEL2 YEL3 D099 H3107 H3173 AY321482 K122 OH10 EF043441	TCCCAT - ACATAGCG TCCCAT - ACATAGCG TCACAT - ACATGGCG	AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCG AGGAGGTGCTTCTTATGCG AGGAGGTGCTTCTTATGCG AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT	FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           JAAACT           JAAACT           FAT-CT           FAT-CT           JAAACT           FAT-CT           FA	705 704 704 704 704 709 705 705 705 705 704 709 709 709	
PUR1 PUR2 PUR3 YEL1 YEL2 YEL3 D099 H3107 H3173 AY321482 K122 OH10 EF043441 AGC	TCCCAT - ACATAGCG TCCCAT - ACATAGCG TCACAT - ACATAGCG TCACAT - ACATAGCG	AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGGC AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTGTACT AGGAGGTGCTTCTTGTACT	FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           JAAACT           JAAACT           FAT-CT           FAT-CT           IAAACT           JAAACT           FAT-CT           FAT-CT           IAACT           IAACT           IAACT           IAACT	705 704 704 704 704 709 705 705 705 705 704 704 709 709 723	
PUR1 PUR2 PUR3 YEL1 YEL2 YEL3 D099 H3107 H3173 AY321482 K122 OH10 EF043441 AGC Drdorg	TCCCAT - ACATAGCG         TCACAT - ACATAGCG         TCCCAT - ACATAGCG         TCCCAT - ACATAGCG         TCCCAT - ACATAGCG         TCCCAT - ACATAGCG	AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGGC AGGAGGTGCTTCTTATGGC AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTGTACT AGGAGGTGCTTCTTGTACT AGGAGGTGCTTCTTGTACT	FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           JAAACT           JAAACT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           C-ATCT           JTATCT	705 704 704 704 704 709 705 705 705 705 705 704 709 709 709 709 709	
PUR1 PUR2 PUR3 YEL1 YEL2 YEL3 D099 H3107 H3173 AY321482 K122 OH10 EF043441 AGC Andong U2014	TCCCAT - ACATAGCG	AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGGC AGGAGGTGCTTCTTATGGC AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTGTACT AGGAGGTGCTTCTTGTACC AGGAGGTGCTCTTATACC AGGAGGTGCTCTTATACC	TAT-CT TAT-CT TAT-CT TAT-CT TAT-CT TAT-CT T-ATCT JAAACT JAAACT TAT-CT T-ATCT T-ATCT T-ATCT T-ATCT T-ATCT	705 704 704 704 704 705 705 705 705 705 704 709 709 709 723 719 722	
PUR1 PUR2 PUR3 YEL1 YEL2 YEL3 D099 H3107 H3173 AY321482 K122 OH10 EF043441 AGC Andong H3014	TCCCAT - ACATAGCG TCCCAT - ACATAGCG TCACAT - ACATAGCG TCCCAT - ACATAGCG TCCCAT - ACATAGCG TCCCAT - ACATAGCG	AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGGC AGGAGGTGCTTCTTATGGC AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTGTACT AGGAGGTGCTTCTTGTACT AGGAGGTGCTTCTTGTACT AGGAGGTGCTTCTTATACC AGGAGGTGCTTCTTATACC	FAT-CT FAT-CT FAT-CT FAT-CT FAT-CT FAT-CT F-ATCT FAACT FAACT FAT-CT F-ATCT F-ATCT F-ATCT F-ATCT F-ATCT F-ATCT F-ATCT	705 704 704 704 704 709 705 705 705 705 704 704 709 709 723 719 722	
PUR1 PUR2 PUR3 YEL1 YEL2 YEL3 D099 H3107 H3173 AY321482 K122 OH10 EF043441 AGC Andong H3014 HF85	TCCCAT - ACATAGCG	AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCC AGGAGGTGCTTCTTATGGC AGGAGGTGCTTCTTATGGC AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTGTACT AGGAGGTGCTTCTTGTACT AGGAGGTGCTTCTTATACC AGGAGGTGCTTCTTATACC AGGAGGTGCTTCTTATACC AGGAGGTGCTCTTATACC	FAT-CT         FAT-CT         FAT-CT         FAT-CT         FAT-CT         FAT-CT         SAAACT         SAAACT         FAT-CT         FAT-CT         FAT-CT         FAT-CT         FAT-CT         FAT-CT         FAT-CT         FAT-CT         F-ATCT         F-ATCT         F-ATCT         F-ATCT         F-ATCT         F-ATCT	705 704 704 704 704 709 705 705 705 704 704 709 709 723 719 722 724	
PUR1 PUR2 PUR3 YEL1 YEL2 YEL3 D099 H3107 H3173 AY321482 K122 OH10 EF043441 AGC Andong H3014 HF85 NIHHS-KHH05	TCCCAT - ACATAGCG	AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCC AGGAGGTGCTTCTTATGGC AGGAGGTGCTTCTTATGGC AGGAGGTGCTTCTTATGCC AGGAGGTGCTTCTTATGCC AGGAGGTGCTTCTTGTACT AGGAGGTGCTTCTTATACC AGGAGGTGCTTCTTATACC AGGAGGTGCTTCTTATACC AGGAGGTGCTTCTTATACC AGGAGGTGCCTCTTATACC	FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           SAAACT           SAAACT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FATCT           FATCT           FATCT           FATCT           FATCT	705 704 704 704 709 705 705 705 705 704 709 709 709 709 723 719 722 724 717	
PUR1 PUR2 PUR3 YEL1 YEL2 VEL3 D099 H3107 H3173 AY321482 K122 OH10 EF043441 AGC Andong H3014 HF85 NIHHS-KHH05 N-UK76	TCCCAT - ACATAGCG	AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCG AGGAGGTGCTTCTTATGCC AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTGTACT AGGAGGTGCTTCTTATACC AGGAGGTGCTTCTTATACC AGGAGGTGCCTCTTATACC AGGAGGTGCCTCTTATACC AGGAGGTGCCTCTTATACC AGGAGGTGCCTCTTATACC	FAT-CT FAT-CT FAT-CT FAT-CT FAT-CT F-ATCT F-ATCT FAT-CT F-ATCT F-ATCT F-ATCT F-ATCT F-ATCT F-ATCT F-ATCT F-ATCT	705 704 704 704 704 709 705 705 705 705 704 709 709 709 709 723 719 722 724 717	
PUR1 PUR2 PUR3 YEL1 YEL2 VEL3 D099 H3107 H3173 AY321482 K122 OH10 EF043441 AGC Andong H3014 HF85 NIHHS-KHH05 N-UK76 KJ938577	TCCCAT - ACATAGCG	AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGGC AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTGTACT AGGAGGTGCTTCTTATACC AGGAGGTGCTTCTTATACC AGGAGGTGCTTCTTATACC AGGAGGTGCTCTTATACC AGGAGGTGCTCTTATACC AGGAGGTGCTCTTATACC AGGAGGTGCTCTTATACC AGGAGGTGCTCTTATACC AGGAGGTGCTCTTATACC	FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           JAACT           JAAACT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           F-ATCT           J-AACT           J-ACT           F-ATCT           F-ATCT           F-ATCT           F-ATCT           F-ATCT           F-ATCT	705 704 704 704 704 709 705 705 705 705 704 709 709 709 709 709 723 719 722 724 717 722 724	
PUR1 PUR2 PUR3 YEL1 YEL2 VEL3 D099 H3107 H3173 AY321482 K122 OH10 EF043441 AGC Andong H3014 HF85 NIHHS-KHH05 N-UK76 KJ938577 SF52	TCCCAT - ACATAGCG	AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGGC AGGAGGTGCTTCTTATGGC AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTGTACT AGGAGGTGCTTCTTGTACT AGGAGGTGCTTCTTATACC AGGAGGTGCCTCTTATACC AGGAGGTGCCTCTTATACC AGGAGGTGCTTCTTATACT AGGAGGTGCTTCTTATACT AGGAGGTGCTTCTTATACT AGGAGGTGCTCTTATACC AGGAGGTGCTTCTTATACT AGGAGGTGCTCTTATACC	FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           SAAACT           SAAACT           SAAACT           FAT-CT           FAT-CT           CAT-CT           FAT-CT           FAT-CT           CAT-CT           SAAACT           F-ATCT           S-AACT           F-ATCT           S-AACT           F-ATCT           F-ATCT           F-ATCT           F-ATCT           F-ATCT           F-ATCT           F-ATCT	705 704 704 704 705 705 705 705 704 709 709 723 719 722 724 717 722 724 709	
PUR1 PUR2 PUR3 YEL1 YEL2 YEL3 D099 H3107 H3173 AY321482 K122 OH10 EF043441 AGC Andong H3014 HF85 NIHHS-KHH05 N-UK76 KJ938577 SF52 SF670	TCCCAT - ACATAGCG	AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGGC AGGAGGTGCTTCTTATGGC AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTGTACT AGGAGGTGCTTCTTGTACT AGGAGGTGCTTCTTATACC AGGAGGTGCTCTTATACC AGGAGGTGCCTCTTATACC AGGAGGTGCCTCTTATACC AGGAGGTGCCTCTTATACC AGGAGGTGCCTCTTATACC AGGAGGTGCCTCTTATACC AGGAGGTGCCTCTTATACC AGGAGGTGCCTCTTATACC AGGAGGTGCCTCTTATACC AGGAGGTGCCTCTTATACC AGGAGGTGCCTCTTATACC	FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           SAAACT           SAAACT           SAAACT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FATCT           F-ATCT	705 704 704 704 704 709 705 705 705 705 704 709 709 723 719 722 724 717 722 724 717 722 724 709 709	
PUR1 PUR2 PUR3 YEL1 YEL2 YEL3 D099 H3107 H3173 AY321482 K122 OH10 EF043441 AGC Andong H3014 HF85 NIHHS-KHH05 N-UK76 KJ938577 SF52 SF670 SF724	TCCCAT - ACATAGCG	AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGGC AGGAGGTGCTTCTTATGGC AGGAGGTGCTTCTTATGGC AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTGTACT AGGAGGTGCTTCTTATACC AGGAGGTGCTTCTTATACC AGGAGGTGCTCTTATACC AGGAGGTGCTCTTATACC AGGAGGTGCTCTTATACC 	FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           JAAACT           JAAACT           JAAACT           FAT-CT           CAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           F-ATCT	705 704 704 704 705 705 705 705 704 704 709 709 723 719 722 724 717 722 724 717 722 724 709 709 709	
PUR1 PUR2 PUR3 YEL1 YEL2 YEL3 D099 H3107 H3173 AY321482 K122 OH10 EF043441 AGC Andong H3014 HF85 NIHHS-KHH05 N-UK76 KJ938577 SF52 SF670 SF724 EF488006	TCCCAT - ACATAGCG	<ul> <li>AGGAGGTGCTTCTTATGCT</li> <li>AGGAGGTGCTTCTTATGCT</li> <li>AGGAGGTGCTTCTTATGCT</li> <li>AGGAGGTGCTTCTTATGCT</li> <li>AGGAGGTGCTTCTTATGCT</li> <li>AGGAGGTGCTTCTTATGCT</li> <li>AGGAGGTGCTTCTTATGCT</li> <li>AGGAGGTGCTTCTTATGGCT</li> <li>AGGAGGTGCTTCTTATGGCT</li> <li>AGGAGGTGCTTCTTATGGCT</li> <li>AGGAGGTGCTTCTTATGGCT</li> <li>AGGAGGTGCTTCTTATGGCT</li> <li>AGGAGGTGCTTCTTATGCT</li> <li>AGGAGGTGCTTCTTATGCT</li> <li>AGGAGGTGCTTCTTATGCT</li> <li>AGGAGGTGCTTCTTATGCT</li> <li>AGGAGGTGCTTCTTATACCT</li> <li>AGGAGGTGCTTCTTATACCT</li> <li>AGGAGGTGCTCTTATACCT</li> <li>AGGAGGTGCTCTTATACCT</li> <li>AGGAGGTGCCTCTTATACCT</li> <li>AGGAGGTGCCTCTTATACT</li> <li>AGGAGGTGCCTCTTATACT</li> <li>AGGAGGTGCCTCTTATACT</li> <li>AGGAGGTGCTCTTATACT</li> <li>AGGAGGTGCTCTTATACT</li> <li>AGGAGGTGCTTCTTATACT</li> </ul>	FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           SAAACT           SAAACT           FAT-CT           FATCT           F-ATCT	705 704 704 704 709 705 705 705 704 709 709 723 719 722 724 717 722 724 717 722 724 709 709 709 709	
PUR1 PUR2 PUR3 YEL1 YEL2 D099 H3107 H3173 AY321482 K122 OH10 EF043441 AGC Andong H3014 HF85 NIHHS-KHH05 N-UK76 KJ938577 SF52 SF670 SF724 EF488006 Florida	TCCCAT - ACATAGCG	AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCG AGGAGGTGCTTCTTATGCG AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATGCT AGGAGGTGCTTCTTATACT AGGAGGTGCTTCTTATACT AGGAGGTGCTTCTTATACT AGGAGGTGCTCTTATACT AGGAGGTGCTCTTATACT AGGAGGTGCTCTTATACT AGGAGGTGCTTCTTATACT AGGAGGTGCTTCTTATACT AGGAGGTGCTTCTTATACT AGGAGGTGCTTCTTATACT AGGAGGTGCTTCTTATACT AGGAGGTGCTTCTTATACT AGGAGGTGCTTCTTATACT AGGAGGTGCTTCTTATACT 	FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           JAACT           JAAACT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           F-ATCT	705 704 704 704 704 705 705 705 705 704 709 709 709 723 719 722 724 717 722 724 717 722 724 709 709 709 709 709 709	
PUR1 PUR2 PUR3 YEL1 YEL2 VEL3 D099 H3107 H3173 AY321482 K122 OH10 EF043441 AGC Andong H3014 HF85 NIHHS-KHH05 N-UK76 KJ938577 SF52 SF670 SF724 EF488006 Florida J37	TCCCAT - ACATAGCG         TCCCAT - ACATAGCG <td><ul> <li>AGGAGGTGCTTCTTATGCT</li> <li>AGGAGGTGCTTCTTATACT</li> <li>AGGAGGTGCTTCTTATACT</li> <li>AGGAGGTGCTTCTTATACT</li> <li>AGGAGGTGCTTCTTATACT</li> <li>AGGAGGTGCTCTTATACT</li> <li>AGGAGGTGCTCTTATACT</li> <li>AGGAGGTGCTCTTATACT</li> <li>AGGAGGTGCTCTTATACT</li> <li>AGGAGGTGCTCTTATACT</li> <li>AGGAGGTGCTCTTATACT</li> <li>AGGAGGTGCTTCTTATACT</li> </ul></td> <td>FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           JAACT           JAAACT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           F-ATCT           J-AACT           F-ATCT           F-ATCT</td> <td>705 704 704 704 704 709 705 705 705 705 704 709 709 723 719 722 724 717 722 724 717 722 724 709 709 709 709 709 709 709 717</td> <td></td>	<ul> <li>AGGAGGTGCTTCTTATGCT</li> <li>AGGAGGTGCTTCTTATACT</li> <li>AGGAGGTGCTTCTTATACT</li> <li>AGGAGGTGCTTCTTATACT</li> <li>AGGAGGTGCTTCTTATACT</li> <li>AGGAGGTGCTCTTATACT</li> <li>AGGAGGTGCTCTTATACT</li> <li>AGGAGGTGCTCTTATACT</li> <li>AGGAGGTGCTCTTATACT</li> <li>AGGAGGTGCTCTTATACT</li> <li>AGGAGGTGCTCTTATACT</li> <li>AGGAGGTGCTTCTTATACT</li> </ul>	FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           JAACT           JAAACT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           F-ATCT           J-AACT           F-ATCT	705 704 704 704 704 709 705 705 705 705 704 709 709 723 719 722 724 717 722 724 717 722 724 709 709 709 709 709 709 709 717	
PUR1 PUR2 PUR3 YEL1 YEL2 VEL3 D099 H3107 H3173 AY321482 K122 OH10 EF043441 AGC Andong H3014 HF85 NIHHS-KHH05 N-UK76 KJ938577 SF52 SF670 SF724 EF488006 Florida J37 NZH3	$\begin{tabular}{lllllllllllllllllllllllllllllllllll$	<ul> <li>AGGAGGTGCTTCTTATGCT</li> <li>AGGAGGTGCTTCTTATGCT</li> <li>AGGAGGTGCTTCTTATGCT</li> <li>AGGAGGTGCTTCTTATGCT</li> <li>AGGAGGTGCTTCTTATGCT</li> <li>AGGAGGTGCTTCTTATGCT</li> <li>AGGAGGTGCTTCTTATGCT</li> <li>AGGAGGTGCTTCTTATGGC</li> <li>AGGAGGTGCTTCTTATGGC</li> <li>AGGAGGTGCTTCTTATGCT</li> <li>AGGAGGTGCTTCTTATGCT</li> <li>AGGAGGTGCTTCTTATGCT</li> <li>AGGAGGTGCTTCTTATGCT</li> <li>AGGAGGTGCTTCTTATGCT</li> <li>AGGAGGTGCTTCTTATACT</li> <li>AGGAGGTGCTTCTTATACT</li> <li>AGGAGGTGCTTCTTATACT</li> <li>AGGAGGTGCTTCTTATACT</li> <li>AGGAGGTGCTCTTATACT</li> <li>AGGAGGTGCTCTTATACT</li> <li>AGGAGGTGCTCTTATACT</li> <li>AGGAGGTGCTCTTATACT</li> <li>AGGAGGTGCTCTTATACT</li> <li>AGGAGGTGCTTCTTATACT</li> </ul>	FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           SAAACT           SAAACT           SAAACT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FATCT           F-ATCT	705 704 704 704 704 709 705 705 705 704 709 709 709 723 719 722 724 717 722 724 717 722 724 709 709 709 709 709 709 709 709 717 717	
PUR1 PUR2 PUR3 YEL1 YEL2 YEL3 D099 H3107 H3173 AY321482 K122 OH10 EF043441 AGC Andong H3014 HF85 NIHHS-KHH05 N-UK76 KJ938577 SF52 SF670 SF724 EF488006 Florida J37 NZH3 X1	$\begin{tabular}{lllllllllllllllllllllllllllllllllll$	<ul> <li>AGGAGGTGCTTCTTATGCT</li> <li>AGGAGGTGCTTCTTATACT</li> <li>AGGAGGTGCTTCTTATACT</li> <li>AGGAGGTGCTCTTATACT</li> <li>AGGAGGTGCTCTTATACT</li> <li>AGGAGGTGCTCTTATACT</li> <li>AGGAGGTGCTCTTATACT</li> <li>AGGAGGTGCTCTTATACT</li> <li>AGGAGGTGCTCTTATACT</li> <li>AGGAGGTGCTCTTATACT</li> <li>AGGAGGTGCTTCTTATACT</li> <li>AGGAGGTGCTTCTTATGCT</li> <li>AGGAGGTGCTTCTTATGCT</li> <li>AGGAGGTGCTTCTTATGCT</li> <li>AGGAGGTGCTTCTTATGCT</li> <li>AGGAGGTGCTTCTTATGCT</li> </ul>	FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           SAAACT           SAAACT           SAAACT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FATCT           F-ATCT	705 704 704 704 704 709 705 705 705 704 709 709 723 719 722 724 717 722 724 709 709 709 709 709 709 709 709 709 709	
PUR1 PUR2 PUR3 YEL1 YEL2 YEL3 D099 H3107 H3173 AY321482 K122 OH10 EF043441 AGC Andong H3014 HF85 NIHHS-KHH05 N-UK76 KJ938577 SF52 SF670 SF724 EF488006 Florida J37 NZH3 X1 X03680	TCCCAT - ACATAGCG	<ul> <li>AGGAGGTGCTTCTTATGCT</li> <li>AGGAGGTGCTTCTTGTACT</li> <li>AGGAGGTGCTTCTTGTACT</li> <li>AGGAGGTGCTTCTTATACT</li> <li>AGGAGGTGCTCTTATACT</li> <li>AGGAGGTGCTCTTATACT</li> <li>AGGAGGTGCTCTTATACT</li> <li>AGGAGGTGCTCTTATACT</li> <li>AGGAGGTGCTCTTATACT</li> <li>AGGAGGTGCTCTTATACT</li> <li>AGGAGGTGCTTCTTATACT</li> <li>AGGAGGTGCTTCTTATGCT</li> <li>AGGAGGTGCTTCTTATGCT</li> <li>AGGAGGTGCTTCTTATGCT</li> </ul>	FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           JAAACT           JAAACT           JAAACT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FAT-CT           FATCT           F-ATCT	705 704 704 704 704 705 705 705 705 704 704 709 709 723 719 722 724 717 722 724 709 709 709 709 709 709 709 709 709 709	

**Figure A. 4:** ClustalW Multiple Sequence Alignment of a portion of the ITS region from three PUR and three YEL *Heterorhabditis* isolates from North Bull Island, a selection of *Heterorhabditis* species and 1 outgroup (*C. elegans* strain XO3680). Asterisk (\*) denotes conserved sites (see Section 2.3.2 Figure 2.9).

Spaces denote nucleotide polymorphism

## A.2 Generation of growth curve for *P. temperata pur1* and *yel3* isolates

Liquid cultures were set up by inoculating single colonies from 36 - 48 hr plate cultures into 50 ml nutrient broth (NB) or MacConkey broth (MB) in a 250 ml flask and growing for 24 - 48 hr in the dark with shaking at 200 rpm at 27 °C. At a number of time points 3 x 1 ml aliquots were drawn from the flask and the optical density (OD) at 660 nm of liquid cultures was recorded as required using a Shimadzu UV mini 1240 Spectrophotometer (Fischer Scientific). The growth curve (Figure A. 5) was generated by plotting log of absorbance (OD<sub>660</sub>) vs time.



**Figure A. 5:** Growth curve for *pur1* and *yel3 Photorhabdus temperata* isolates from North Bull Island. Isolates were grown in MacConkey broth (See Section 2.3.3, & Section 3.4).

purl	ATGACTGTGCTGCATGCGGGGGGGGGAAATTCGATGATAACTCTTATAAAGTCTCCGGCGGT	60
pur2	ATGACTGTGCTGCATGCGGGGGGGGAAAATTCGATGATAACTCTTATAAAGTCTCCGGCGGT	60
pur3	ATGACTGTGCTGCATGCGGGGGGGGGAAATTCGATGATAACTCTTATAAAGTCTCCGGCGGT	60
yell	ATGACTGTGCTGCATGCGGGGGGGGGAAATTCGATGATAACTCTTATAAAGTTTCCGGCGGT	60
yel2	ATGACTGTGCTGCATGCGGGGGGGGAAATTCGATGATAACTCTTATAAAGTTTCCGGCGGT	60
yel3	ATGACTGTGCTGCATGCGGGGGGGGTAAATTCGATGATAACTCTTATAAAGTTTCCGGCGGT	60
	***************************************	
nurl		120
pur?		120
pur?		120
pul 3		120
yell		120
yel2		120
yer3	11GCACGGCG1AGGGG111C1G11G11AACGCC11G1C1GAAAAGC1GGAAC1GG11A1C	120
		100
puri		180
pur2	CGCCGTGACGGCAAAGTTCATGAACAGACTTATCACCTTGGTGTGCCGCAAAGCCCACTA	180
pur3	CGCCGTGACGGCAAAGTTCATGAACAGACTTATCACCTTGGTGTGCCGCAAAGCCCACTA	180
yell	CGTCGTGACGGCAAAGTTCATGAACAGACTTATCACCTTGGTGTGCCACAAAGCCCACTA	180
yel2	CGTCGTGACGGCAAAGTTCATGAACAGACTTATCACCTTGGTGTGCCACAAAGCCCACTA	180
yel3	CGTCGTGACGGCAAAGTTCATGAACAGACTTATCACCTTGGTGTGCCACAAAGCCCACTA	180
	_	
purl	AAAGTTGTTGGTGAAACCGAACAGACAGGTACCCGTGTCCGCTTCTGGCCAAGCATGGAT	240
pur2	AAAGTTGTTGGTGAAACCGAACAGACAGGTACCCGTGTCCGCTTCTGGCCAAGCATGGAT	240
pur3	AAAGTTGTTGGTGAAACCGAACAGACAGGTACCCGTGTCCGCTTCTGGCCAAGCATGGAT	240
yell	AAAGTCGTCGGTGAAACGGAGCAGACAGGTACCCGTGTCCGCTTCTGGCCGAGCATGGAC	240
yel2	AAAGTCGTCGGTGAAACGGAGCAGACAGGTACCCGTGTCCGCTTCTGGCCGAGCATGGAC	240
vel3	AAAGTCGTCGGTGAAACGGAGCAGACAGGTACCCGTGTCCGCTTCTGGCCGAGCATGGAC	240
-	*****	
purl	ACTITICCGTAACAACACTGAATTCCAGCACGACATTCTGGCTAAACGTCTGCGTGAATTG	300
pur2	ACTITICCGTAACAACACTGAATTCCAGCACGACATTCTGGCTAAACGTCTGCGTGAATTG	300
pur3	ACTTTCCGTAACAACACTGAATTCCAGCACGACATTCTGGCTAAACGTCTGCGTGAATTG	300
yell	ACTTTCAGTAACAACACTGAATTCCAGCACGACATTCTGGCTAAACGTCTGCGTGAATTG	300
yel2	ACTTTCAGTAACAACACTGAATTCCAGCACGACATTCTGGCTAAACGTCTGCGTGAATTG	300
yel3	ACTTTCAGTAACAACACTGAATTCCAGCACGACATTCTGGCTAAACGTCTGCGTGAATTG	300
purl	TCCTTTCTGAACTCAGGTGTATCTATCCGCCTGCTTGATAAACGCACCAATATTGAAGAC	360
pur2	TCCTTTCTGAACTCAGGTGTATCTATCCGCCTGCTTGATAAACGCACCAATATTGAAGAC	360
pur3	TCCTTTCTGAACTCAGGTGTATCTATCCGCCTGCTTGATAAACGCACCAATATTGAAGAC	360
vel1	TCCTTTCTGAACTCAGGTGTATCTATCCGCCTGCTTGATAAACGCACCAATATTGAAGAC	360
vel2	TCCTTTCTGAACTCAGGTGTATCTATCCGCCTCGCTTGATAAACGCACCAATATTGAAGAC	360
vel3		360
y C 1 3	*****	500
~~~1		420
Pur J		+2U
pur 2		420
pur's		420
yell		420
yel2		420
yel3	CATTTCCATTAGAAGGTGGCATTAAAGCGTTTGTCGAATTCTTAAATAAA	420
1		405
purl	CCAATCCATCCCAATGTTTTCTATTTCTCTACAGAAAAAGATGGCATCGGTGTTGAAGCC	480
pur2	CCAATCCATCCCAATGTTTTCTATTTCTCTACAGAAAAAGATGGCATCGGTGTTGAAGCC	480
pur3	CCAATCCATCCCAATGTTTTCTATTTCTCTCTACAGAAAAAGATGGCATCGGTGTTGAAGCC	480
yell	CCAATCCATCCCAATGTTTTCTATTTCTCTACAGAAAAAGATAGCATCGGTGTTGAAGTC	480
yel2	CCAATCCATCCCAATGTTTTCTATTTCTCTACAGAAAAAGATAGCATCGGTGTTGAAGTC	480
yel3	CCAATCCATCCCAATGTTTTCTATTTCTCTCTACAGAAAAAGATAGCATCGGTGTTGAAGTC	480
pur1	GCATTGCAATGGAACGATGGTTTCCAGGAAAATATTTACTGCTTTACCAACAACATCCCG	540
pur2	GCATTGCAATGGAACGATGGTTTCCAGGAAAATATTTACTGCTTTACCAACAACATCCCG	540
pur3	GCATTGCAATGGAACGATGGTTTCCAGGAAAATATTTACTGCTTTACCAACAACATCCCCG	540
- yell	GCATTGCAGTGGAATGATGGTTTCCAGGAAAATATTTACTGCTTTACCAACAACATCCCCG	540
vel2	GCATTGCAGTGGAATGATGGTTTCCAGGAAAATATTTACTGCTTTACCAACAACATCCCG	540
vel3	GCATTGCAGTGGAATGATGGTTTCCAGGAAAATATTTACTCCTTTACCAACAACAACATCCCC	540
1010	******* ***** ************************	JIU
Dur1		600
Pur J		600
PULZ		600
Purs		600
yell		000
yeız	CAACGCGACGGCGGTACTCACTTAGTTGGTTTCCGTACTGCGATGACCCCGTACCCTCAAC	600

yel3	CAACGCGACGGCGGTACTCACTTAGTTGGTTTCCGTACTGCGATGACCCCGTACCCTCAAC	600
	**** **********************************	
	3 COm3 C3 mCC3 m3 3 3 C3 CCCCm3 C3 3 C3	660
purr		600
purz	AGCTACATGGATAAAGAGGGGTACAACAAGAAATCCAAAGTCAACGCTACTGGTGATGAT	660
pur3	AGCTACATGGATAAAGAGGGGTACAACAAGAAATCCAAAGTCAGCGCTACTGGTGATGAT	660
yell	AGCTATATGGACAAAGAAGGGTACAACAAGAAATCCAAAGTCAGCGCCACCGGTGATGAT	660
yel2	AGCTATATGGACAAAGAAGGGTACAACAAGAAATCCAAAGTCAGCGCCACCGGTGATGAT	660
yel3	AGCTATATGGACAAAGAAGGGTACAACAAGAAATCCAAAGTCAGCGCCACCGGTGATGAT	660
	***** ***** ***** *********************	
		720
purr		720
purz		720
pur3		720
yell		720
yelz		720
yels	***** ** *****************************	120
purl	CAGACTAAAGATAAACTAGTCTCTTCCGAAGTGAAAACCGCAGTTGAAACACTGATGAAT	780
pur2	CAGACTAAAGATAAACTAGTCTCTTCCGAAGTGAAAACCGCAGTTGAAACACTGATGAAT	780
pur3	CAGACTAAAGATAAACTAGTCTCTTCCGAAGTGAAAACCGCAGTTGAAACACTGATGAAT	780
yel1	CAGACCAAAGACAAACTGGTTTCTTCCGAGGTGAAAACGGCGGTTGAAACACTGATGAAC	780
yel2	CAGACCAAAGACAAACTGGTTTCTTCCGAGGTGAAAACGGCGGTTGAAACACTGATGAAC	780
vel3	CAGACCAAAGACAAACTGGTTTCTTCCGAGGTGAAAACGGCGGTTGAAACACTGATGAAC	780
-	**** ***** ***** ** ****** ******* *****	
		0.4.0
purl	GAGAAGCTGGTGGAATATCTGCTGGAAAACCCAAATGACGCCAAAACAGTCGTCAGTAAA	840
pur2	GAGAAGCTGGTGGAATATCTGCTGGAAAACCCAAATGACGCCAAAACAGTCGTCAGTAAA	840
pur3	GAGAAGCTGGTGGAATATCTGCTGGAAAACCCAAATGACGCCAAAACAGTCGTCAGTAAA	840
yell	GAGAAGCTGGTGGAATACCTGCTGGAAAACCCCGAATGACGCTAAAACCGTCGTCGGTAAA	840
yel2	GAGAAGCTGGTGGAATACCTGCTGGAAAACCCGAATGACGCTAAAACCGTCGTCGGTAAA	840
yel3	GAGAAGCTGGTGGAATACCTGCTGGAAAACCCGAATGACGCTAAAACCGTCGTC	840
	***************************************	
1 וויי	ϪͲͲϪͲͲϹϪͲϾϲͲϾϲͲϹϲϲϲϲͲϹϪϪϾϲϪϾϲϲϲϲͲϪϪϪϾϲϪϲϾϲϲϲϪϪϪͲϹϪϲϪϲϭϲϲϲϲ	900
pur?		900
pur2		900
pul 5 woll		900
yell wol2		900
yel2		900
yer?	***** ** ***** ** ********************	900
purl	AAAGGGGCTCTTGATCTGGCTGGCTTACCAGGCAAACTAGCTGATTGTCAGGAACGTGAC	960
pur2	AAAGGGGCTCTTGATCTGGCTGGCTTACCAGGCAAACTAGCTGATTGTCAGGAACGTGAC	960
pur3	AAAGGGGCTCTTGATCTGGCTGGCTTACCAGGCAAACTAGCTGATTGTCAGGAACGTGAC	960
yel1	AAAGGGGCTCTTGATCTGGCTGGCCTGCCAGGCAAACTGGCGGATTGTCAGGAACGTGAT	960
yel2	AAAGGGGCTCTTGATCTGGCTGGCCTGCCAGGCAAACTGGCGGATTGTCAGGAACGTGAT	960
yel3	AAAGGGGCTCTTGATCTGGCTGGCCTGCCAGGCAAACTGGCGGATTGTCAGGAACGTGAT	960
	*******************	
1 וויי	იივიაუულიივაალოაიუუალივაავივიალივივილივიიაააააავი	1020
pur?	CCCCCATTCTCCCAACTCTACTTACTCCAAACCCCACTCCCCCC	1020
pur2		1020
pul 3		1020
yell		1020
yelz		1020
yel3	****** *******************************	1020
purl	GGCCGTAACCGCAAAAATCAGGCAATCTTGCCATTGAAAGGTAAAATCCTAAACGTTGAA	1080
pur2	GGCCGTAACCGCAAAAATCAGGCAATCTTGCCATTGAAAGGTAAAATCCTAAACGTTGAA	1080
pur3	GGCCGTAACCGCAAAAATCAGGCAATCTTGCCATTGAAAGGTAAAATCCTAAACGTTGAA	1080
yell	GGGCGTAACCGCAAAAATCAGGCAATTTTGCCACTGAAAGGTAAAATCCTGAACGTTGAA	1080
yel2	GGGCGTAACCGCAAAAATCAGGCAATTTTGCCACTGAAAGGTAAAATCCTGAACGTTGAA	1080
vel3	GGGCGTAACCGCAAAAATCAGGCAATTTTGCCACTGAAAGGTAAAATCCTGAACGTTGAA	1080
▲	** ************************************	
1		1140
purt	AAAGCGCGTIIICGAIAAAAIGCTCTCTTCTCAGGAAGTCGCAACACTGATCACTGCATTG	1140
pur2		1140
pur3	AAAGCGCGTTTTCGATAAAATGCTCTCTTCTCAGGAAGTCGCAACACTGATCACTGCATTG	1140
yell	AAAGCGCGTTTCGATAAAATGCTCTCTCTCAGGAAGTGGCAACCCTGATCACTGCGTTG	1140
yel2	AAAGCGCGTTTCGATAAAATGCTCTCTCTCAGGAAGTGGCAACCCTGATCACTGCGTTG	1140
yel3	AAAGCGCGTTTCGATAAAATGCTCTCTCTCAGGAAGTGGCAACCCTGATCACTGCGTTG	1140
purl	GGTTGTGGTATTGGCCGTGACGAATACAACCCGGATAAGCTGCGTTATCACAGTATCATT	1200
pur2	GGTTGTGGTATTGGCCGTGACGAATACAACCCGGATAAGCTGCGTTATCACAGTATCATT	1200

pur3	GGTTGTGGTATTGGCCGTGACGAATACAACCCGGATAAGCTGCGTTATCACAGTATCATT 1	1200
yel1	GGTTGCGGTATTGGCCGTGACGAATACAACCCGGATAAGCTGCGTTATCACAGTATTATT 1	1200
yel2	GGTTGCGGTATTGGCCGTGACGAATACAACCCGGATAAGCTGCGTTATCACAGTATTATT 1	1200
yel3	GGTTGCGGTATTGGCCGTGACGAATACAACCCGGATAAGCTGCGTTATCACAGTATTATT 1	1200
-	**** **********************************	
purl	ATCATGACAGATGCGGACGTTGATGGTTCTCATATCCGTACTCTATTGTTAACTTTTTTC 1	1260
pur2	ATCATGACAGATGCGGACGTTGATGGTTCTCATATCCGTACTCTATTGTTAACTTTTTTC 1	1260
pur3	ATCATGACAGATGCGGACGTTGATGGTTCTCATATCCGTACTCTATTGTTAACTTTTTTC 1	1260
yel1	ATCATGACAGATGCGGACGTCGATGGTTCTCATATCCGTACTCTGTTGTTGACTTTCTTC 1	1260
yel2	ATCATGACAGATGCGGACGTCGATGGTTCTCATATCCGTACTCTGTTGTTGACTTTCTTC 1	1260
yel3	ATCATGACAGATGCGGACGTCGATGGTTCTCATATCCGTACTCTGTTGTTGACTTTCTTC 1	1260
	******************	
purl	TATCGTCAAATGCCAGAAATCATTGAACGTGGTCACGTGTTTATAGCTCAGCCACCTCTG 1	1320
pur2	TATCGTCAAATGCCAGAAATCATTGAACGTGGTCACGTGTTTATAGCTCAGCCACCTCTG 1	1320
pur3	TATCGTCAAATGCCAGAAATCATTGAACGTGGTCACGTGTTTATAGCTCAGCCACCTCTG 1	1320
yel1	TATCGTCAAATGCCAGAAATCATTGAACGTGGTCATGTGTTTATTGCACAGCCGCCACTG 1	1320
yel2	TATCGTCAAATGCCAGAAATCATTGAACGTGGTCATGTGTTTATTGCACAGCCGCCACTG 1	1320
yel3	TATCGTCAAATGCCAGAAATCATTGAACGTGGTCATGTGTTTATTGCACAGCCGCCACTG 1	1320
-	***************************************	
purl	TATAAAGTGAAAAAAGGCAAACAGGAACAGTATATTAAAGATGATGAAGCAATGGATGAG 1	1380
pur2	TATAAAGTGAAAAAAGGCAAACAGGAACAGTATATTAAAGATGATGAAGCAATGGATGAG 1	1380
pur3	TATAAAGTGAAAAAAGGCAAACAGGAACAGTATATTAAAGATGATGAAGCAATGGATGAG 1	1380
yel1	TATAAAGTGAAAAAAGGCAAGCAGGAACAATATATTAAAGATGATGAAGCAATGGATGAG 1	1380
yel2	TATAAAGTGAAAAAAGGCAAGCAGGAACAATATATTAAAGATGATGAAGCAATGGATGAG 1	1380
yel3	TATAAAGTGAAAAAAGGCAAGCAGGAACAATATATTAAAGATGATGAAGCAATGGATGAG 1	1380
	****************	
purl	TA 1382	
pur2	TA 1382	
pur3	TA 1382	
yell	TA 1382	
yel2	TA 1382	
yel3	TA 1382	
	* *	

**Figure A. 6:** ClustalW Multiple Sequence Alignment of a portion (1382 bases) of the gyrB gene from three *pur* and three *yel Photorhabdus temperata* isolates from North Bull Island. Six single nucleotide polymorphisms resulting in non-synonymous amino acids in the *pur* and yel isolates have been outlined in red (See Section 2.3.4).

Asterisk (\*) denotes conserved sites Spaces denote nucleotide polymorphism.

H3107	<u>ΣΟΥΤΑΤΟΣΟΟΤΤΑΤΑΤΑΤΑΤΑΤΑΤΑΣ ΣΤΟ ΤΟ ΣΤΟ ΤΟ Τ</u>	60
u3240		60
H3014	ACTTATCACCTTGCTGCCGCAAAGCCCCACTAAAAGTTGTTGCTGAAACCGAACAGACA	60
purl	ACTTATCACCTTGGTGTGCCGCAAAGCCCACTAAAAGTTGTTGGTGAAACCGAACAGACA	60
pur2	ACTTATCACCTTGGTGTGCCGCAAAGCCCACTAAAAGTTGTTGGTGAAACCGAACAGACA	60
pur3	ACTTATCACCTTGGTGTGCCGCAAAGCCCACTAAAAGTTGTTGGTGAAACCGAACAGACA	60
yell	ACTTATCACCTTGGTGTGCCACAAAGCCCACTAAAAGTCGTCGGTGAAACGGAGCAGACA	60
yel2	ACTTATCACCTTGGTGTGCCACAAAGCCCACTAAAAGTCGTCGGTGAAACGGAGCAGACA	60
yel3	ACTTATCACCTTGGTGTGCCACAAAGCCCACTAAAAGTCGTCGGTGAAACGGAGCAGACA	60
K122	ACTTATCACCTTGGTGTGCCACAAAGCCCACTAAAAGTCGTCGGTGAAACGGAGCAGACA	60
HF85	ACTTATCACCTTGGTGTGCCACAAAGCCCACTAAAAGTCGTCGGTGAAACGGAGCAGACA	60
HL81	ACTTATCACCTTGGTGTGCCACAAAGCCCACTAAAAGTCGTCGGTGAAACGGAGCAGACA	60
HW79	ACTTATCACCTTGGTGTGCCACAAAGCCCACTAAAAGTCGTCGGTGAAACGGAGCAGACA	60
XlLit	ACTTATCACCTTGGTGTGCCACAAAGCCCACTAAAAGTCGTCGGTGAAACGGAGCAGACA	60
XlNach	ACTTATCACCTTGGTGTGCCACAAAGCCCACTAAAAGTCGTCGGTGAAACGGAGCAGACA	60
H3016	ACTTATCACCTTGGTGTGCCACAAAGCCCACTAAAAGTCGTCGGTGAAACGGAGCAGACA	60
H31/3		60
H31/9		60
n3102 C1		60
Mea		60
Habana	ACTTATCACCTTGGTGTGCCACAAAGCCCCACTAAAAGTCGTTGGTGAAACCGAACAGACC	60
NZH3	ACTTATCACCTTGGTGTGCCGCAAAGCCCACTAAAAGTTGTCGGTGAAACCGAACAGACT	60
T327	ACTTATCACCTTGGTGTGCCGCAAAGCCCACTAAAAGTTGTCGGTGAAACCGAACAGACT	60
USCA01	ACTTATCACCTTGGTGTGCCGCAAAGCCCACTAAAAGTTGTCGGTGAAACCGAACAGACT	60
H3210	ACTTATCACCTTGGTGTGCCGCAAAACCCACTAAAAGTTGTCGGTGAAACCGAACAGACA	60
CIP108426	ACTTATCACCTTGGTGTGCCGCAAAAACCCACTAAAAGTTGTCGGTGAAACCGAACAGACA	60
FR32	ACTTATCACCTTGGTGTGCCGCAAAACCCACTAAAAGTTGTCGGTGAAACCGAACAGACA	60
H111	ACTTATCACCTTGGTGTGCCACAAAGCCCACTAAAAGTCGTCGGTGAAACGGAGCAGACA	60
H267	ACTTATCACCTTGGTGTGCCGCAAAGCCCACTAAAAGTTGTTGGTGAAACCGAACAGACA	60
H295	ACTTATCACCTTGGTGTGCCACAAAGCCCACTAAAAGTCGTCGGTGAAACGGAGCAGACA	60
H3086	ACTTATCACCTTGGTGTGCCGCAAAGCTCACCCAAAGTTGTTGGTGAAACCGAACAGACA	60
AN6	ACTTACCATCATGGTGTTCCACAGTCTCCGCTGAACATTGTCGGTGATACCGAGCAGACA	60
	***** ** * ****** ** ** ** ** ** ** **	
u2107		120
u3240	CCTACCCGTCTCCCCCTCTCCCCCAACCACCCCCCCCCC	120
H3014	GGTACCCGTGTCCCGCTTCTGGCCAAGCATGGATACTTTCCGTAACAACACTGAATTCCAG	120
purl	GGTACCCGTGTCCCGCTTCTGGCCAAGCATGGATACTTTCCGTAACAACACTGAATTCCAG	120
pur2	GGTACCCGTGTCCGCTTCTGGCCAAGCATGGATACTTTCCGTAACAACACTGAATTCCAG	120
pur3	GGTACCCGTGTCCGCTTCTGGCCAAGCATGGATACTTTCCGTAACAACACTGAATTCCAG	120
yell	GGTACCCGTGTCCGCTTCTGGCCGAGCATGGACACTTTCAGTAACAACACTGAATTCCAG	120
yel2	GGTACCCGTGTCCGCTTCTGGCCGAGCATGGACACTTTCAGTAACAACACTGAATTCCAG	120
yel3	GGTACCCGTGTCCGCTTCTGGCCGAGCATGGACACTTTCAGTAACAACACTGAATTCCAG	120
K122	GGTACCCGTGTCCGCTTCTGGCCGAGCATGGACACTTTCAGTAACAACACTGAATTCCAG	120
HF85	GGGACCCGTGTCCGCTTCTGGCCGAGCATGGACACTTTCAGTAACAACACTGAATTCCAG	120
HL81	GGGACCCGTGTCCGCTTCTGGCCGAGCATGGACACTTTCAGTAACAACACTGAATTCCAG	120
HW79	GGGACCCGTGTCCGCTTCTGGCCGAGCATGGACACTTTCAGTAACAACACTGAATTCCAG	120
XlLit	GGGACCCGTGTCCGCTTCTGGCCGAGCATGGACACTTTCAGTAACAACACTGAATTCCAG	120
XINach	GGGACCCGTGTCCGCCTTCTGGCCGAGCATGGACACTTTCAGTAACAACACTGAATTCCAG	120
H3U10		120
H3170		120
u2182	CCTACCCGTCTCCCCCTCTCCCCCACCACCACCACCTCCACCACCCCCC	120
C1	GGTACCCGTGTCCCGCTTTTGGCCCGAGCATGGATACTTTCAGTAACAACACTGAATTCCAG	120
Meq	GGTACCCGTGTCCGCCTTTTGGCCCGAGCATGGATACTTTCAGTAACAACACTGAATTCCAG	120
Habana	GGTACCCGTGTCCGCTTTTGGCCGAGCATGGATACTTTCAGTAACAACACTGAATTCCAG	120
NZH3	GGTACTCGTGTCCGCTTTTGGCCGAGCATGGATACTTTCAGTAACAACACTGAATTCCAG	120
Т327	GGTACTCGTGTCCGCTTTTGGCCGAGCATGGATACTTTCAGTAACAACACTGAATTCCAG	120
USCA01	GGTACTCGTGTCCGCTTTTGGCCGAGCATGGATACTTTCAGTAACAACACTGAATTCCAG	120
H3210	GGTACCCGTGTCCGCTTCTGGCCAAGCATGGATACTTTCCGTGACAACACTGAATTCCAG	120
CIP108426	GGTACCCGTGTCCGCCTTCTGGCCAAGCATGGATACTTTCCGTGACAACACTGAATTCCAG	120
FR32	GGTACCCGTGTCCGCTTCTGGCCAAGCATGGATACTTTCCGTGACAACACTGAATTCCAG	120
H111	GGTACCCGTGTCCGCCTTCTGGCCGAGCATGGACACTTTCAGTAACAACACTGAATTCCAG	120
H267	GGTACCCGTGTCCGGCTTCTGGCCAAGCATGGATACTTTCCGTAACAACACTGAATTCCAG	120
H295	GGGACCCGTGTCCGCTTCTGGCCGAGCATGGACACTTTCCAGTAACAACACTGAATTCCAG	120
HJUOD	GGIACCCGTGTCCGCCTTCTGGCCAAGCATGGATACTTTCCCGTAACAACACTGAATTCCAG	120
0 1114	GGAACCACGGIICGIIICIGGCCAAGCAIGGAIACGIICAAGATTAACACTGAATTCGAA           **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         **         ** <td>⊥∠U</td>	⊥∠U
Н3107	CACGACATTCTGGCTAAACGTCTGCGTGAATTGTCCTTTCTGAACTCAGGTGTATCTATC	180
H3240	CACGACATTCTGGCTAAACGTCTGCGTGAATTGTCCTTTCTGAACTCAGGTGTATCTATC	180
H3014	${\tt Cacgacattctggctaaacgtctgcgtgaattgtcctttctgaactcaggtgtatctatc$	180
purl	${\tt Cacgacattctggctaaacgtctgcgtgaattgtcctttctgaactcaggtgtatctatc$	180
pur2	${\tt Cacgacattctggctaaacgtctgcgtgaattgtcctttctgaactcaggtgtatctatc$	180
pur3	CACGACATTCTGGCTAAACGTCTGCGTGAATTGTCCTTTCTGAACTCAGGTGTATCTATC	180
yell	CACGACATTCTGGCTAAACGTCTGCGTGAATTGTCCTTTCTGAACTCAGGTGTATCTATC	180

yel2	CACGACATTCTGGCTAAACGTCTGCGTGAATTGTCCTTTCTGAACTCAGGTGTATCTATC	180
yel3	CACGACATTCTGGCTAAACGTCTGCGTGAATTGTCCTTTCTGAACTCAGGTGTATCTATC	180
K122	CACGACATTCTGGCTAAACGTCTGCGTGAATTGTCCTTTCTGAACTCAGGTGTATCTATC	180
HF85	CACGACATTCTGGCTAAACGTCTGCGTGAATTGTCCTTTCTGAACTCAGGTGTATCTATC	180
HL81	CACGACATTCTGGCTAAACGTCTGCGTGAATTGTCCTTTCTGAACTCAGGTGTATCTATC	180
HW79	CACGACATTCTGGCTAAACGTCTGCGTGAATTGTCCTTTCTGAACTCAGGTGTATCTATC	180
XlLit	CACGACATTCTGGCTAAACGTCTGCGTGAATTGTCCTTTCTGAACTCAGGTGTATCTATC	180
XlNach	CACGACATTCTGGCTAAACGTCTGCGTGAATTGTCCTTTCTGAACTCAGGTGTATCTATC	180
H3016	CACGACATTCTGGCTAAACGTCTGCGTGAATTGTCCTTTCTGAACTCAGGTGTATCTATC	180
H3173	CACGACATTCTGGCTAAACGTCTGCGTGAATTGTCCTTTCTGAACTCAGGTGTATCTATC	180
H3179	CACGACATTCTGGCTAAACGTCTGCGTGAATTGTCCTTTCTGAACTCAGGTGTATCTATC	180
H3182	CACGACATTCTGGCTAAACGTCTGCGTGAATTGTCCTTTCTGAACTCAGGTGTATCTATC	180
CI	CACGACATTCTGGCTAAACGTCTGCGTGAGCTGTCCTTTCTGAACTCAGGTGTATCTATC	180
Meg		100
Habana		100
NZH3 T227		100
1327		100
U3CAU1 U2210		100
CTD108426		180
ED30		180
H111		180
H267	CACGACATTCTGGCTAAACGTCTGCGTGAATTGTCCTTTCTGAACTCAGGTGTATCTATC	180
H295	CACGACATTCTGGCTAAACGTCTGCGTGCATGAATTCTCTGCTGCACGTCAGGTGTATCTATC	180
H3086	CACGACATTCTGGCTAAACGTCTGCGTGAATTGTCCTTTCTGAACTCAGGTGTATCTATC	180
ANG	TACGATATTCTGGCTAAGCGCCTGCGTGAATTATCATTCCTGAATTCCGGTGTTTCTATT	180
	**** ********* ** ******* * ** ** ******	
Н3107	CGCCTGCTTGATAAACGCACCAATATTGAAGACCATTTCCATTATGAAGGTGGTATTAAG	240
H3240	CGCCTGCTTGATAAACGCACCAATATTGAAGACCATTTCCATTATGAAGGTGGTATTAAG	240
H3014	CGCCTGCTTGATAAACGCACCAATATTGAAGACCATTTCCATTATGAAGGTGGTATTAAG	240
purl	CGCCTGCTTGATAAACGCACCAATATTGAAGACCATTTCCATTATGAAGGTGGTATTAAG	240
pur2	CGCCTGCTTGATAAACGCACCAATATTGAAGACCATTTCCATTATGAAGGTGGTATTAAG	240
pur3	CGCCTGCTTGATAAACGCACCAATATTGAAGACCATTTCCATTATGAAGGTGGTATTAAG	240
yell	CGCCTGCTTGATAAACGCACCAATATTGAAGACCATTTCCATTACAAAGGTGGCATTAAA	240
yel2	CGCCTGCTTGATAAACGCACCAATATTGAAGACCATTTCCATTACAAAGGTGGCATTAAA	240
yel3	CGCCTGCTTGATAAACGCACCAATATTGAAGACCATTTCCATTACAAAGGTGGCATTAAA	240
K122	CGCCTGCTTGATAAACGCACCAATATTGAAGACCATTTCCATTACAAAGGTGGCATTAAA	240
HF85	CGCCTGCTTGATAAACGCACCAATATTGAAGACCATTTCCATTACAAAGGTGGCATTAAA	240
HL81	CGCCTGCTTGATAAACGCACCAATATTGAAGACCATTTCCATTACAAAGGTGGCATTAAA	240
HW79	CGCCTGCTTGATAAACGCACCAATATTGAAGACCATTTCCATTACAAAGGTGGCATTAAA	240
XlLit	CGCCTGCTTGATAAACGCACCAATATTGAAGACCATTTCCATTACAAAGGTGGCATTAAA	240
XlNach	CGCCTGCTTGATAAACGCACCAATATTGAAGACCATTTCCATTACAAAGGTGGCATTAAA	240
H3016	CGCCTGCTTGATAAACTCACCAATATTGAAGACCATTTCCATTACAAAGGTGGCATTAAA	240
H3173	CGCCTGCTTGATAAACGCACCAATATTGAAGACCATTTCCATTACAAAGGTGGCATTAAA	240
H3179	CGCCTGCTTGATAAACGCACCAATATTGAAGACCATTTCCATTACAAAGGTGGCATTAAA	240
H3182	CGCCTGCTTGATAAACGCACCAATATTGAAGACCATTTCCATTACAAAGGTGGCATTAAA	240
C1	CGCCTGCTTGATAAACGCACCAATATTGAAGACCATTTCCATTATAAAGGTGGTATTAAG	240
Meg	CGCCTGCTTGATAAACGCACCAATATTGAAGACCATTTCCATTATAAAGGTGGTATTAAG	240
Habana	CGCCTGCTTGATAAACGCACCAATATTGAAGACCATTTCCATTATAAAGGTGGTATTAAG	240
NZH3		240
132/		240
USCAUL		240
113210 CTD108496		⊿±0 24∩
ED30		240
H111	CCCCTCCTTCATAAACGCACCAATATTCAACACCATTTCCATTACAAAGGIGGIAITAAG	240
H267	CGCCTGCTTGATAAACGCACCAATATTGAAGACCATTTCCATTACCAACGCGCACCAATAT	240
H295	CGCCTGCTTGATAAACGCACCAATATTGAAGACCATTTCCATTACAAAGGTGGCATTAAA	240
H3086	CGCCTGCTTGATAAACGCACCAATATTGAAGACCATTTCCATTATGAAGGTGGTATTAAG	240
ANG	CGTTTAGTTGATAAACGCACCAATGCCGAAGATCTCTTCCACTATAAAGGCGGTATCAAA	240
	** * ****** * ****** ***** * ***** ** *	
H3107	GCGTTTGTCGAATTCTTAAATAAAAAAAAAAACAACAATCCATCC	300
H3240	GCGTTTGTCGAATTCTTAAATAAAAAAAAAAACAACAATCCATCC	300
H3014	GCGTTTGTCGAATTCTTAAATAAAAACAAAACACCAATCCATCC	300
purl	GCGTTTGTCGAATTCTTAAATAAAAACAAAACAACAACCAATCCATCC	300
pur2	GCGTTTGTCGAATTCTTAAATAAAAACAAAACAACAACCAATCCATCC	300
pur3	GCGTTTGTCGAATTCTTAAATAAAAACAAAACAACAACCAATCCATCC	300
yell	GCGTTTGTCGAATTCTTAAATAAAAACAAAACAACAACCAATCCATCC	300
yel2	GCGTTTGTCGAATTCTTAAATAAAAACAAAACAACAACCAATCCATCC	300
yel3	GCGTTTGTCGAATTCTTAAATAAAAACAAAACAACAACCAATCCATCC	300
K122	GCGTTTGTCGAATTCTTAAATAAAAACAAAACACCAATCCATCC	300
HF85	GCGTTTTGTCGAATTCTTAAATAAAAACAAAACACCAATCCATCC	300
HL81	GUGTTTGTUGAATTCTTTAAATAAAAACAAAACACCAATCCCAATGTTTTCTATTTC	300
HW/Y	GCG111GTCGAATTCTTAAATAAAAACACAAAACACCAATCCCATCCCAATGTTTTCTATTTC	300
AILIT	GUGIIIGTUGAATTUTTAAATAAAAACAAAACAUCAATCCATCCCAATGTTTTCTATTTC	300

XlNach	GCGTTTGTCGAATTCTTAAATAAAAACAAAACACCAATCCATCC	300
Н3016	GCGTTTGTCGAATTCTTAAATAAAAACAAAACACCAATCCATCC	300
Н3173	GCGTTTGTCGAATTCTTAAATAAAAACAAAACACCAATCCATCC	300
H3179	<u> </u>	300
H3182		300
C1		300
Mea		300
Habana		300
		200
m2113 m207		200
1327		200
USCAUL U2010		300
GTD100406		200
		300
FR32		300
HIII		300
H267		300
H295		300
H3086		300
AN6	GCGTTTGTTGAGTTCCTCAGCCGCAATAAAAACCCCGATTCACCCAAATGTTTTTTATTTC	300
	******* ** *** * * ** *** ** ** ** ** *	
<b>U3107</b>	ͲʹͲͳΛʹϤλʹϨλλλλλαλͲϲϨϨϽͲϲϨϨͲϨͲͲϨλλϨϲϲϨϤλͲͲϨϨλλͲϨϨλλϨϨλͲϨϨͲͲͲϲϤλϨ	360
нз240		360
u3014		360
num1		260
puri		300
purz		300
pur3		360
yell		300
yel2		300
yei3 v100		300
		300
		300
		300
		300
XINach		360
u2016		360
H3173		360
н3179	TCTACAGAAAAAGATACCATCCCTCTTCAACTCCCATTCCACTCCACTCCACTCCACTCCACT	360
H3182	TCTACAGAAAAAGATACCATCCCTCTTCAAGTCCCCATTCCAAGTCCAGTCCCAGTCCCAGTCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCAGTCCCCCCCC	360
C1	TCTACAGAAAAAAGCATAGCATCCGCTGTTGAAGTCCCCATTGCAATGGAACGATGGCTTTCCAG	360
Mea	TCTACACAAAAAACATACCATCCCTCTTCAACTCCCATTCCAATCCAATCCAATCCACACCATTCCAC	360
Habana	TCTACAGAAAAAAGCATAGCATCCGCTGTTGAAGTCCCCATTGCAATGGAACGATGGCTTTCCAG	360
NZH3	TCAACAGAAAAAAGCATAGCATCGGTGTTGAAGTTGCCCTTGCAGTGGAACGATGGTTTCCAG	360
т327	TCAACAGAAAAAAGCATAGCATCCGCTGCTTGCAAGTTGCCCTTGCAGCGAACGATGGCTTTCCAG	360
USCA01	TCAACAGAAAAAGATAGCATCGGTGTTGAAGTTGCCTTGCAGTGGAACGATGGTTTCCCAG	360
H3210	TCTACAGAAAAAGATAGCATCGGTGTTGAAGTCGCATTGCAATGGAACGATGGTTTCCAG	360
CIP108426	TCTACAGAAAAAGATAGCATCGGTGTTGAAGTCGCATTGCAATGGAACGATGGTTTCCAG	360
FR32	TCTACAGAAAAAGATAGCATCGGTGTTGAAGTCGCATTGCAATGGAACGATGGTTTCCAG	360
H111	TCTACAGAAAAAGATAGCATCGGTGTTGAAGTCGCATTGCAGTGGAATGATGGTTTCCAG	360
H267	TCTACAGAAAAAGATGGCATCGGTGTTGAAGCCGCATTGCAATGGAACGATGGTTTCCAG	360
Н295	TCTACAGAAAAAGATAGCATCGGTGTTGAAGTCGCATTGCAATGGAATGATGGTTTCCAG	360
Н3086	TCTACAGAAAAAGATGGCATTGGTGTTGAAGCCGCATTGCAATGGAACGATGGTTTCCAG	360
AN6	TCCACAGAAAAAGACGGTATTGGTGTTGAAATTTCCATGCAATGGAATGATGGTTTTCAG	360
	** ******** * ** ******* * ************	
Н3107	GAAAATATTTACTGCTTTACCAACAACATCCCGCAACGTGACGGCGGTACTCACTTAGTG	420
Н3240	GAAAATATTTACTGCTTTACCAACAACATCCCGCAACGTGACGGCGGTACTCACTTAGTG	420
H3014	GAAAATATTTACTGCTTTACCAACAACATCCCGCAACGTGACGGCGGTACTCACTTAGTG	420
purl	GAAAATATTTACTGCTTTACCAACAACATCCCGCAACGTGACGGCGGTACTCACTTAGTG	420
pur2	GAAAATATTTACTGCTTTACCAACAACATCCCGCAACGTGACGGCGGTACTCACTTAGTG	420
pur3	GAAAATATTTACTGCTTTACCAACAACATCCCGCAACGTGACGGCGGTACTCACTTAGTG	420
yell	GAAAATATTTACTGCTTTACCAACAACATCCCCGCAACGCGACGGCGGTACTCACTTAGTT	420
yel2	GAAAATATTTACTGCTTTACCAACAACATCCCGCAACGCGGCGGTACTCACTTAGTT	420
yel3	GAAAATATTTACTGCTTTACCAACAACATCCCGCAACGCGGCGGTACTCACTTAGTT	420
VT77	GAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAA	4∠U 400
пгор ut 01	GAAAAIAIIIIACIGCIIIIACCAACAICCCGCAACGTGACGGCGGTACTCACTTAGTG	4∠U ∕120
11101 UW70	GAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAA	±∠∪ 400
vin.i+	GAAAATATTTACIGUTTTACCAACAACATCCCGCAAACGTGACGGGGGGAACTCACTC	420
VINach		∠∪ //20
H3016	GAAAAIAIIIIACIGCIIIACCAACAACAICCCGCAACGIGACGCGGIACICACIIACIACIIACGACGCGCAACAICCCGCAACAICCCGCAACAICCCGCGAACAICCGCGCAACAICCCGCGAACAICCCGCGAACAICCCGCGAACAICCGCGCGAACAICCGCGGAACAICCCGCGAACAICCCGCGAACAICCGCGGAACAICCGGGAACAICCGGGAACAICCGGAACAICCGGAACAICCGGAACAICCGGAACAICCGGAACAICCGGAACAICCGGAACAICCGGAACAICCGGAACAICCGGAACAICGGCGGAACAICGGAACAICCGGAACAICGGAACAICGGAACAICGGAACAICGGAACAICGGAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICGACAACAICAACAICAICAACAACAICAACAACAICGACAACAICAACAACAACAACAICAACAACAAACA	±∠∪ 420
H3173		∠∪ 420
H3179	GAAAATTTTACIGCIIIACCAACAICCCCCAACGIGACGCGCIACICACIIAGIG GAAAATTTTACTGCIIIACCAACAICCCCCCAACGIGACGCGCIACICACIIAGIG	±20 42∩
H3182	GAAAATATTTATTGCTTTACCAACAACATCCCGCAACGCGGCGCGCGC	420
C1	GAAAATATTTACTGCTTTACCAACAACAACATCCCCGCAACGTGGTGGTGGTGCTCACTCA	420
Мед	GAAAATATTTACTGCTTTACCAACAACATCCCGCAACGTGATGGTGGTACTCACTTAGTT	420

Habana	${\tt GAAAATATTTACTGCTTTACCAACAACATCCCGCAACGTGATGGTGGTACTCACTTAGTT}$	420
NZH3	GAAAATATTTACTGCTTTACCAACAACATCCCGCAACGTGATGGCGGTACTCACTTAGTG	420
Т327	GAAAATATTTACTGCTTTACCAACAACATCCCGCAACGTGATGGCGGTACTCACTTAGTG	420
USCA01	GAAAATATTTACTGCTTTACCAACAACATCCCGCAACGTGATGGCGGTACTCACTTAGTG	420
H3210	GAAAATATTTACTGCTTTACCAACAACATCCCGCAACGTGACGGTGGTACTCACTTAGTG	420
CIP108426	GAAAATATTTACTGCTTTACCAACAACATCCCGCAACGTGACGGTGGTACTCACTTAGTG	420
FR32	GAAAATATTTACTGCTTTACCAACAACATCCCGCAACGTGACGGTGGTACTCACTTAGTG	420
H111	GAAAATATTTACTGCTTTACCAACAACATCCCGCAACGCGACGGCGGTACTCACTTAGTT	420
H267	GAAAATATTTACTGCTTTTACCAACAACATCCCGCAACGTGACGGCGGTACTCACTTAGTG	420
H295	GAAAATATTTACTGCTTTTACCAACAACATCCCGCAACGTGACGGCGGTACTCACTTAGTG	420
H3086	GAAAATATTTACTGCTTTACCAACAACATCCCCGCAACGTGACGGCGGTACTCACTTAGTG	420
ANG	GAAAACATTTACTGCTTCACCAACAATATTCCTCAGCGCGATGGCGGAACGCACTTAGTG	420
	***** ***** ***** ******* ** ** ** ** *	
H3107	GGTTTCCGTACCGCGATGACTCGTACCCTTAATAGCTACATGGATAAAGAGGGGTACAAC	480
H3240	GGTTTCCGTACCGCGATGACTCGTACCCTTAATAGCTACATGGATAAAGAGGGGGTACAAC	480
H3014	GGTTTCCGTACCGCGATGACTCGTACCCTTAATAGCTACATGGATAAAGAGGGGGTACAAC	480
purl	GGTTTCCGTACCGCGATGACTCGTACCCTTAATAGCTACATGGATAAAGAGGGGGTACAAC	480
pur2	GGTTTCCGTACCGCGATGACTCGTACCCTTAATAGCTACATGGATAAAGAGGGGGTACAAC	480
pur3	GGTTTCCGTACCGCGATGACTCGTACCCTTAATAGCTACATGGATAAAGAGGGGGTACAAC	480
yell	GGTTTCCGTACTGCGATGACCCGTACCCTCAACAGCTATATGGACAAAGAAGGGTACAAC	480
yel2	GGTTTCCGTACTGCGATGACCCGTACCCTCAACAGCTATATGGACAAAGAAGGGTACAAC	480
yel3	GGTTTCCGTACTGCGATGACCCGTACCCTCAACAGCTATATGGACAAAGAAGGGTACAAC	480
K122	GGTTTCCGTACTGCGATGACCCGTACCCTCAACAGCTATATGGACAAAGAAGGGTACAAC	480
HF85	GGTTTCCGTACCGCGATGACTCGTACCCTCAACAGCTATATGGACAAAGAAGGGTCCACC	480
HL81	GGTTTCCGTACCGCGATGACTCGTACCCTCAACAGCTATATGGACAAAGAAGGGTACAAC	480
HW79	GGTTTCCGTACCGCGATGACTCGTACCCTCAACAGCTATATGGACAAAGAAGGGTACAAC	480
XILit	GGTTTCCGTACCGCGATGACTCGTACCCTCAACAGCTATATGGACAAAGAAGGGTACAAC	480
XINach	GGTTTCCGTACCGCGATGACTCGTACCCTCAACAGCTATATGGACAAAGAAGGGTCCCACC	480
H3016	GGTTTCCGTACCGCGATGACTCGTACCCTCAACAGCTATATGGACAAAGAAGGGTACAAC	480
H3173	GGTTTCCGTACCGCGATGACTCGTACCCTCAACAGCTATATGGACAAAGAAGGGTACAAC	480
H3179	GGTTTCCGTACTGCGATGACCCGTACCCTCAACAGCTATATGGACAAAGAAGGGTACAAC	480
H3182	GGTTTCCGTACTGCGATGACCCGTACCCTCAACAGCTATATGGACAAAGAAGGGTACAAC	480
C1	GGTTTCCGTACTGCGATGACCCGTACTCTCAATAGCTATATGGACAAAGAAGGGTACAAC	480
Meg	GGTTTCCGTACTGCGATGACCCGTACTCTCAATAGCTATATGGACAAAGAAGGGTACAAC	480
Habana	GGTTTCCGTACTGCGATGACCCGTACTCTCAATAGCTATATGGACAAAGAAGGGTACAAC	480
NZH3	GGTTTCCGTACCGCAATGACCCGTACCCTCAACAGCTATATGGACAAAGAAGGGTATACC	480
T327	GGTTTCCGTACCGCAATGACCCGTACCCTCAACAGCTATATGGACAAAGAAGGGTATAAC	480
USCAUL	GGTTTCCGTACCGCAATGACCCGTACCCTCAACAGCTATATGGACAAAGAAGGGTATAAC	480
H3210	GGTTTCCGTACCGCGATGACCCGTACCCTCAACAGCTATATGGACAAAGAAGGGTACAAC	480
CIP108426	GGTTTCCGTACCGCGATGACCCGTACCCTCAACAGCTATATGGACAAAGAAGGGTACAAC	480
FR32	GGTTTCCGTACCGCGATGACCCGTACCCTCAACAGCTATATGGACAAAGAAGGGTACAAC	480
HIII	GGTTTTCCGTACTGCGATGACCCCGTACCCTCAACAGCTATATGGACAAAGAAGGGTACAAC	480
H267	GGTTTCCGTACCGCGATGACTCGTACCCTTAATAGCTACATGGATAAAGAGGGGGTACAAC	480
H295		480
H3086	GGTTTTCCGTACCGCGATGACTCGTACCCTTAATAGCTACATGGATAAAGAGGGGGTACAAC	480
ANO	GGIIICCGIACCCGAGCCCICAAIAGCIACAIGGAIAAAGAGGGCIACAAC	480
H3107	AAGAAATCCAAAGTCAGCGCTACTGGTGATGATGCCCGTGAAGGTTTAGTCGCCGTCATT	540
H3240	AAGAAATCCAAAGTCAGCGCTACTGGTGATGATGCCCCGTGAAGGTTTAGTCGCCGTCATT	540
H3014	AAGAAATCCAAAGTCAGCGCTACTGGTGATGATGCCCCGTGAAGGTTTAGTCGCCGTCATT	540
purl	AAGAAATCCAAAGTCAGCGCTACTGGTGATGATGCCCCGTGAAGGTTTAGTCGCCGTCATT	540
pur2	AAGAAATCCAAAGTCAGCGCTACTGGTGATGATGCCCCGTGAAGGTTTAGTCGCCGTCATT	540
pur3	AAGAAATCCAAAGTCAGCGCTACTGGTGATGATGCCCGTGAAGGTTTAGTCGCCGTCATT	540
yell	AAGAAATCCAAAGTCAGCGCCACCGGTGATGATGCCCGCGAGGGTTTAGTCGCCGTCATT	540
yel2	AAGAAATCCAAAGTCAGCGCCACCGGTGATGATGCCCGCGAGGGTTTAGTCGCCGTCATT	540
yel3	AAGAAATCCAAAGTCAGCGCCACCGGTGATGATGCCCGCGAGGGTTTAGTCGCCGTCATT	540
K122	AAGAAATCCAAAGTCAGCGCCACCGGTGATGATGCCCGCGAGGGTTTAGTCGCCGTCATT	540
HF85	AAGAAATCTAAAGTCAGCGCCACCGGTGATGATGCCCGCGAGGGTTTAGTCGCCGTCATT	540
HL81	AAGAAATCTAAAGACAGCGCCACCGGTGATGATGCCCGCGAGGGTTTAGTCGCCGTCATT	540
HW79	AAGAAATCTAAAGTCAGCGCCACCGGTGATGATGCCCGCGAGGGTTTAGTCGCCGTCATT	540
XlLit	AAGAAATCTAAAGTCAGCGCCACCGGTGATGATGCCCGCCAGGGTTTAGTCGCCGTCATT	540
XlNach	AAGAAATCTAAAGTCAGCGCCACCGGTGATGATGCCCGCCAGGGTTTAGTCGCCGTCATT	540
H3016	AAGAAATCTAAAGTCAGCGCCACCGGTGATGATGCCCGCGAGGGTTTAGTCGCCGTCATT	540
Н3173	AAGAAATCTAAAGTCAGCGCCACCGGTGATGATGCCCGCGAGGGTTTAGTCGCCGTCATT	540
Н3179	AAGAAATCCAAAGTCAGCGCCACCGGTGATGATGCCCGCGGGGTTTAGTCGCCGTCATT	540
H3182	AAGAAATCCAAAGTCAGCGCCACCGGTGATGATGCCCGCGGGGGTTTAGTCGCCGTCATT	540
CL	AAGAAATCTAAAGTCAGCGCCACCGGTGATGATGCCCGTGAAGGCTTGGTCGCAGTTATT	540
Meg	AAGAAATUTAAAGTCAGCGCCACCGGTGATGATGCCCGTGAAGGCTTGGTCGCAGTTATT	540
Hapana	AAGAAA ICTAAAGTCAGCGCCACCGGTGATGATGCCCCGTGAAGGCTTGGTCGCAGTTATT	540
N2H3 T227	AAGAAAIUIAAAGTCAGTGCCACCGGTGATGATGCCCCGTGAGGGTTTGGTCGCCGCGTTATT	54U
1327		54U
USCAUL US210		540 540
CTD108426		540
EB30	AGAZZZALAZA ZACAGOGOGA OGOGOGAZZA ZACOGOGAZZZZOGOGAZZZZACOGOGAZZZZZCZ ZACOGOGAZZZZZZCZZCZCZCZCZZZZZZZZZZZZZZZZ	540
1.1.7.7.	ANONANICIAAAGICACCACCAGIGAIGAIGCCCGIGAAGGCIIAGICGCCGIIAII	740

H111	AAGAAATCCAAAGTCAGCGCCACCGGTGATGATGCCCGCGAGGGTTTAGTCGCCGTCATT	540
H267	AAGAAATCCAAAGTCAGCGCTACTGGTGATGATGCCCGTGAAGGTTTAGTCGCCGTCATT	540
H295	AAGAAATCTAAAGTCAGCGCCACCGGTGATGATGCCCGCGAGGGTTTAGTCGCCGTCATT	540
H3086	A A G A A A TCC A A A G TC A G C G C T A C TG G TG A TG A	540
ANG		510
ANO	AAAAAAICCAAAGICAGCGCCACCGGIGAIGAIGCCCGIGAAGGIIIGAIIGCAGIIAII	540
H3107	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACTAAAGATAAACTAGTCTCTTCC	600
H3240	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACTAAAGATAAACTAGTCTCTTCC	600
H3014	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACTAAAGATAAACTAGTCTCTTCC	600
purl	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACTAAAGATAAACTAGTCTCTTCC	600
rur?	TCCGTTAAAGTACCTGATCCAAAATTCTCTCTCAGACTAAAGATAAACTAGTCTCTCC	600
Fur 2	ͲϹϹϹͲͲϪϪϪϾͲϪϹϹͲϹϪͲϹϹϪϪϪϪϪͲͲϹͲϹͲϹͲϹͲϹϪϹϪϹͲϪϪϪϹϫϪϪϪϹͲϪ	600
purs		600
yell	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACCAAAGACAAACTGGTTTCTTCC	600
yel2	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACCAAAGACAAACTGGTTTCTTCC	600
yel3	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACCAAAGACAAACTGGTTTCTTCC	600
K122	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACCAAAGACAAACTGGTTTCTTCC	600
HF85	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACCAAAGACAAACTGGTTTCTTCC	600
HI.81	TCCGTTAAAGTACCTGATCCAAAATTCTCTCAGACCAAAGACAAACTGGTTTCTTCC	600
11201 11201		600
		C 0 0
AILLL		600
XINach	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACCAAAGACAAACTGGTTTCTTCC	600
Н3016	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACCAAAGACAAACTGGTTTCTTCC	600
H3173	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACCAAAGACAAACTGGTTTCTTCC	600
Н3179	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACCAAAGACAAACTGGTTTCTTCC	600
H3182	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACCAAAGACAAACTGGTTTCTTCC	600
C1	TCCGTTAAAGTACCTGATCCAAAATTCTCTCAGACCAAAGACAAACTGGTTTCTTCC	600
Meg		600
Nebana		600
naballa		600
NZH3	TCCGTTAAAGTCCCTGATCCAAAATTCTCTTCTCAGACCAAAGACAAACTGGTTTCTTCC	600
Т327	TCCGTTAAAGTCCCTGATCCAAAATTCTCTTCTCAGACCAAAGACAAACTGGTTTCTTCC	600
USCA01	${\tt TCCGTTAAAGTCCCTGATCCAAAATTCTCTTCTCAGACCAAAGACAAACTGGTTTCTTCC}$	600
H3210	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACCAAAGACAAACTGGTTTCTTCC	600
CIP108426	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACCAAAGACAAACTGGTTTCTTCC	600
FR32	TCCGTTAAAGTACCTGATCCAAAATTCTCTCAGACCAAAGACAAACTGGTTTCTTCC	600
н111	ͲϹϹϾͲͲϿϿϿϾͲϿϹϹͲϾϿͲϹϹϿϿϿϿͲͲϹͲϹͲϹͲϹϿϾϿϹϨϿϿϿϾϿϘϿϿϿϹͲϾϾͲͲͲϹͲͲϹϹ	600
11267		600
		000
		600
H207 H295	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACTAAAGTAAACTAGTCTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTC	600
H207 H295 H3086	TCCGTTAAAGTACCTGATCCAAAATTCTCTCTCCAGACTAAAGTAAACTAGTCTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTC	600 600
H295 H3086 AN6	TCCGTTAAAGTACCTGATCCAAAATTCTCTCTCCAGACTAAAGTAAACTAGTCTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTTCTCAGACCAAAGACAAACTGGTTTCTTCC TCCGTTAAAGTACCTGATCCCAAAATTCTCTTTCTCAGACTAAAGATAAACTAGTCTCTTCC TCCGTTAAAGTACCTGATCCTAAATTCTCTTTCCCAGACGAAAGACAAACTGGTTTCTTCC	600 600 600
H295 H3086 AN6	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACTAAAGTAAACTAGTCTTCTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTC	600 600 600
H295 H3086 AN6	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACTAAAGTAAACTAGTCTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTCTTCTCAGACCAAAGACAAACTGGTTTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTC	600 600 600
H297 H295 H3086 AN6 H3107	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACTAAAGTAAACTAGTCTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTC	600 600 600
H207 H295 H3086 AN6 H3107 H3240	CCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACTAAAGTAAACTAGTCTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTC	600 600 600 660 660
H207 H295 H3086 AN6 H3107 H3240 H3014	CCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACTAAAGTAAACTAGTCTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTC	600 600 600 660 660
H295 H3086 AN6 H3107 H3240 H3014 Dur1	CCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACTAAAGTAAACTAGTCTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTC	600 600 600 660 660 660
H295 H3086 AN6 H3107 H3240 H3014 pur1 pur2	CCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACTAAAGTAAACTAGTCTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTC	600 600 600 660 660 660 660
H295 H3086 AN6 H3107 H3240 H3014 pur1 pur2	CCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACTAAAGTAAACTAGTCTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTC	600 600 660 660 660 660 660
H295 H3086 AN6 H3107 H3240 H3014 pur1 pur2 pur3	CCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACTAAAGTAAACTAGTCTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTC	600 600 660 660 660 660 660
H295 H3086 AN6 H3107 H3240 H3014 pur1 pur2 pur3 yel1	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACTAAAGTAAACTAGTCTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTC	600 600 660 660 660 660 660 660 660
H295 H3086 AN6 H3107 H3240 H3014 pur1 pur2 pur3 yel1 yel2	CCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACTAAAGTAAACTAGTCTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTC	600 600 660 660 660 660 660 660 660
H295 H3086 AN6 H3107 H3240 H3014 pur1 pur2 pur3 yel1 yel2 yel3	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACTAAAGTAAACTAGTCTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTC	600 600 660 660 660 660 660 660 660 660
H295 H3086 AN6 H3107 H3240 H3014 pur1 pur2 pur3 yel1 yel2 yel3 K122	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACTAAAGTAAACTAGTCTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTC	600 600 660 660 660 660 660 660 660 660
H207 H295 H3086 AN6 H3107 H3240 H3014 pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACTAAAGTAAACTAGTCTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTC	600 600 600 660 660 660 660 660 660 660
H295 H3086 AN6 H3107 H3240 H3014 pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACTAAAGTAAACTAGTCTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTC	600 600 600 660 660 660 660 660 660 660
H295 H3086 AN6 H3107 H3240 H3014 pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACTAAAGTAAACTAGTCTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTC	600 600 660 660 660 660 660 660 660 660
H295 H3086 AN6 H3107 H3240 H3014 pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACTAAAGTAAACTAGTCTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTC	600 600 660 660 660 660 660 660 660 660
H295 H3086 AN6 H3107 H3240 H3014 pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 XlLit	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACTAAAGTAAACTAGTCTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTC	600 600 660 660 660 660 660 660 660 660
H295 H3086 AN6 H3107 H3240 H3014 pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 XlLit XlNach	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACTAAAGTAAACTAGTCTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTC	600 600 660 660 660 660 660 660 660 660
H207 H295 H3086 AN6 H3107 H3240 H3014 pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 XlLit XlNach H3016	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACTAAAGTAAACTAGTCTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTC	600 600 600 660 660 660 660 660 660 660
H295 H3086 AN6 H3107 H3240 H3014 pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 X1Lit X1Nach H3016 H3173	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACTAAAGTAAACTAGTCTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTC	600 600 600 660 660 660 660 660 660 660
H295 H3086 AN6 H3107 H3240 H3014 pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 X1Lit X1Nach H3016 H3173 H3179	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACTAAAGTAAACTAGTCTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTC	600 600 600 660 660 660 660 660 660 660
H295 H3086 AN6 H3107 H3240 H3014 pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 XlLit XlNach H3016 H3173 H3179 H3182	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACTAAAGTAAACTAGTCTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTC	600 600 600 660 660 660 660 660 660 660
H295 H3086 AN6 H3107 H3240 H3014 pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 X1Lit X1Nach H3016 H3173 H3179 H3182 C1	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACTAAAGTAAACTAGTCTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTTCTCAGACTAAAGACAAACTAGTCTTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTTCTCAGACTAAAGATAAACTAGTCTTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTTCTCAGACGAAGACAAACTGGTTTCTTCC ***************************	600 600 600 660 660 660 660 660 660 660
H295 H3086 AN6 H3107 H3240 H3014 pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 XlLit XlNach H3016 H3173 H3179 H3182 C1 Meg	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACTAAAGTAAACTAGTCTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTTCTCAGACTAAAGATAAACTAGTCTTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTTCTCAGACTAAAGATAAACTAGTCTTTCC *******************************	600 600 600 660 660 660 660 660 660 660
H295 H3086 AN6 H3107 H3240 H3014 pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 X1Lit X1Nach H3016 H3173 H3179 H3182 C1 Meg Habana	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACTAAAGTAAACTAGTCTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTTCTCAGACTAAAGATAAACTAGTCTTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTTCTCAGACTAAAGATAAACTAGTCTTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTTCCCAGACGAAGACAAACTGGTTTCTTCC ***************************	600 600 600 660 660 660 660 660 660 660
H295 H3086 AN6 H3107 H3240 H3014 pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 X1Lit X1Nach H3016 H3173 H3179 H3182 C1 Meg Habana NZH3	TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACCAAAGACAAACTGGTTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTC	600 600 600 660 660 660 660 660 660 660
H295 H3086 AN6 H3107 H3240 H3014 pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 X1Lit X1Nach H3016 H3173 H3179 H3182 C1 Meg Habana NZH3 T327	ICCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACTAAAGACAAACTGGTTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTC	600 600 600 660 660 660 660 660 660 660
H295 H3086 AN6 H3107 H3240 H3014 pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 X1Lit X1Nach H3016 H3173 H3179 H3182 C1 Meg Habana NZH3 T327 USC201	ICCGITAAAGTACCIGATCCAAAATTCICTICICAGACIAAAGTAAACTAGTCICTICC TCCGTTAAAGTACCIGATCCAAAATTCICTCTCTCAGACCAAAGACAAACTGGTTTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCICTCTCTCAGACTAAAGATAAACTAGTCTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCICTCTCTCCAGACTAAAGATAAACTAGTCTCTTCC ******************************	600 600 600 660 660 660 660 660 660 660
H207 H295 H3086 AN6 H3107 H3240 H3014 pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 X1Lit X1Nach H3016 H3173 H3179 H3182 C1 Meg Habana NZH3 T327 USCA01 W3210	ICCGITAAAGTACCIGATCCAAAATTCTCTTCICAGACIAAAGTAAACTAGTCTCTC TCCGTTAAAGTACCIGATCCAAAATTCTCTTCTCAGACCAAAGACAAACTGGTTTCTTC TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTC	600 600 600 660 660 660 660 660 660 660
H207 H295 H3086 AN6 H3107 H3240 H3014 pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 X1Lit X1Nach H3016 H3173 H3179 H3182 C1 Meg Habana NZH3 T327 USCA01 H3210	ICCGTTAAAGTACCTGATCCAAAATTCTCTTCTCAGACTAAAGAAACTAGTCTCTTC TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTC	600 600 600 660 660 660 660 660 660 660
H295 H3086 AN6 H3107 H3240 H3014 pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 X1Lit X1Nach H3016 H3173 H3179 H3182 C1 Meg Habana NZH3 T327 USCA01 H3210 CIP108426	ICCGITAAAGTACCTGATCCAAAATTCTCTTTCTCAGACCAAAGACAAACTGGTTTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTTCTCAGACCAAAGACAAACTGGTTTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTTCTCAGACCAAAGACAAACTGGTTTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTTCTCAGACCAAAGACAAACTGGTTTCTTCC ********** ******* ************	600 600 600 660 660 660 660 660 660 660
H295 H3086 AN6 H3107 H3240 H3014 pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 X1Lit X1Nach H3016 H3173 H3179 H3182 C1 Meg Habana NZH3 T327 USCA01 H3210 CIP108426 FR32	ICCGITAAAGTACCIGATCCAAAATTCCTCTTCTCAGACCAAAGACAAACTGGTTTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCCTCTTCTCAGACCAAAGACAAACTGGTTTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTTCTCAGACCAAAGACAAACTGGTTTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTTCTCAGACCAAAGAAAACTGGTTTCTTCC ********** ******* ************	600 600 600 660 660 660 660 660 660 660
H207 H295 H3086 AN6 H3107 H3240 H3014 pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 XlLit XlNach H3016 H3173 H3179 H3182 C1 Meg Habana NZH3 T327 USCA01 H3210 CIP108426 FR32 H111	ICCGITAAAGTACCIGATCCAAAATTCTCTTTCTCAGACCAAAGACAAACTGGTTTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTTCTCAGACCAAAGACAAACTGGTTTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTTCTCAGACCAAAGACAAACTGGTTTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTTCTCAGACCAAAGACAAACTGGTTTCTTCC ********** ******* ************	600 600 600 660 660 660 660 660 660 660
H207 H295 H3086 AN6 H3107 H3240 H3014 pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 X1Lit X1Nach H3016 H3173 H3179 H3182 C1 Meg Habana NZH3 T327 USCA01 H3210 CIP108426 FR32 H111 H267	ICCGITAAAGTACCIGATCCAAAATTCICTTCICAGACCAAAGCAAACTGGTTTCITC TCCGTTAAAGTACCTGATCCAAAATTCICTTCICAGACCAAAGCAAACTGGTTTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCICTTCICAGACCAAAGCAAACTGGTTTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCICTTCCCAGACGAAAGCAAACTGGTTTCTTCC ***************************	600 600 600 660 660 660 660 660 660 660
H207 H295 H3086 AN6 H3107 H3240 H3014 pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 X1Lit X1Nach H3016 H3173 H3179 H3182 C1 Meg Habana NZH3 T327 USCA01 H3210 CIP108426 FR32 H111 H267 H295	ICCGITAAAGTACCIGATCCAAAATTCICTITCICAGACIAAAGATAAACTAGTCITTCIC TCCGTTAAAGTACCIGATCCAAAATTCICTTCTCAGACCAAAGCAAACTAGTTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCICTTCTCCAGACCAAAGCAAACTAGTTCTTCC ********** ******** ************	600 600 600 660 660 660 660 660 660 660
H207 H295 H3086 AN6 H3107 H3240 H3014 pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 X1Lit X1Nach H3016 H3173 H3179 H3182 C1 Meg Habana NZH3 T327 USCA01 H3210 CIP108426 FR32 H111 H267 H295 H3086	ICCGITAAAGTACCIGATCCAAAATTCICTTCTCAGACCAAAGCAAACTGGTTTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTC	600 600 600 660 660 660 660 660 660 660
H207 H295 H3086 AN6 H3107 H3240 H3014 pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 X1Lit X1Nach H3016 H3173 H3179 H3182 C1 Meg Habana NZH3 T327 USCA01 H3210 CIP108426 FR32 H111 H267 H295 H3086 AN6	TCCGTTAAAGTACTGATCCAAAATTCTCTTCTCAGACTAAAGATAAACTGGTTTCTTCC TCCGTTAAAGTACCTGATCCAAAATTCTCTTCTC	600 600 600 660 660 660 660 660 660 660

H3107	AACCCAAATGACGCCAAAACAGTCGTCAGTAAAATTATTGATGCTGCTCGTGCCCGTGAA	720
Н3240	AACCCAAATGACGCCAAAACAGTCGTCAGTAAAATTATTGATGCTGCTCGTGCCCGTGAA	720
H3014	AACCCAAATGACGCCAAAACAGTCGTCAGTAAAATTATTGATGCTGCTCGTGCCCGTGAA	720
purl	AACCCAAATGACGCCAAAACAGTCGTCAGTAAAATTATTGATGCTGCTCGTGCCCGTGAA	720
pur2	AACCCAAATGACGCCAAAACAGTCGTCAGTAAAATTATTGATGCTGCTCGTGCCCGTGAA	720
pur3	AACCCAAATGACGCCAAAACAGTCGTCAGTAAAATTATTGATGCTGCTCGTGCCCGTGAA	720
yell	AACCCGAATGACGCTAAAACCGTCGTCGGTAAAATTATCGACGCTGCCCGCGCCCGTGAA	720
yel2	AACCCGAATGACGCTAAAAACCGTCGTCGGTAAAATTATCGACGCTGCCCGCGCCCGTGAA	720
yel3		720
KIZZ UROF		720
HF 85		720
HL81		720
HW/9		720
XILLIL XINoch		720
u2016		720
H3010 H2172		720
H3179	ACCCGATGACGCTAAAACCGTCGTCGCGTAAATTATCGACGCTGCCCGCGCCCGCGCGCG	720
H3182	A A CCCCA A TCA CCCTA A A A CCCTCCTCCCCCCC CCCCCCCC	720
C1	ACCCGAATGACGCTAAAACCGTCGTCGGTAAATTATCGACGCTGCCCGCGCCCGCGCCGCGCGCG	720
Mea	AACCCGAATGACGCTAAAACCGTCGTCGGTAAAATTATCGACGCCCGCC	720
Habana		720
NZH3		720
T327	AATCCAAATGACGCCAAAAACCGTCGTCGGCAAAATTATCGACGCGCCCCGCGCTCGTGAA	720
USCA01	AATCCAAATGACGCCAAAAACCGTCGTCGGCCAAAATTATCGACGCGCCCGCC	720
H3210	AACCCGAATGACGCTAAAAACCGTCGTCGGTAAAATTATCGATGCTGCCCGCGCGCCCGCGAA	720
CIP108426	AACCCGAATGACGCTAAAACCGTCGTCGGTAAAATTATCGATGCTGCCCGCGCCCGCGAA	720
FR32	AACCCGAATGACGCTAAAACCGTCGTCGGTAAAATTATCGATGCTGCCCGCGCGCCCGCGAA	720
H111	AACCCGAATGACGCTAAAAACCGTCGTCGGTAAAATTATCGACGCTGCCCGCGCCCGTGAA	720
Н267	AACCCAAATGACGCCAAAACAGTCGTCAGTAAAATTATTGATGCTGCTCGTGCCCGTGAA	720
Н295	AACCCGAATGACGCTAAAACCGTCGTCGGTAAAATTATCGACGCTGCCCGCGCCCGTGAA	720
Н3086	AACCCAAATGACGCCAAAACAGTCGTCAGTAAAATTATTGATGCTGCTCGTGCCCGTGAA	720
AN6	AACCCGAACGACGCCAAAACGGTTGTCGGCAAAATCATTGATGCGGGCTCGCGCCCGTGAA ** ** ** ***** ***** ** *** ** *** *	720
Н3107	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCAAAGGGGGCTCTTGATCTGGCTGG	780
Н3240	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCAAAGGGGGCTCTTGATCTGGCTGG	780
H3014	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCAAAGGGGGCTCTTGATCTGGCTGG	780
purl	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCAAAGGGGCTCTTGATCTGGCTGG	780
purl pur2	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCAAAGGGGCTCTTGATCTGGCTGG	780
purl pur2 pur3	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCAAAGGGGCTCTTGATCTGGCTGG	780 780 780
purl pur2 pur3 yel1	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCAAAGGGGCTCTTGATCTGGCTGG	780 780 780 780
purl pur2 pur3 yel1 yel2	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCAAAGGGGCTCTTGATCTGGCTGG	780 780 780 780 780 780
purl pur2 pur3 yel1 yel2 yel3 K122	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCAAAGGGGCTCTTGATCTGGCTGG	780 780 780 780 780 780 780
pur1 pur2 pur3 yel1 yel2 yel3 K122	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCAAAGGGGCTCTTGATCTGGCTGG	780 780 780 780 780 780 780 780
pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCAAAGGGGCTCTTGATCTGGCTGG	780 780 780 780 780 780 780 780 780
pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCAAAGGGGCTCTTGATCTGGCTGG	780 780 780 780 780 780 780 780 780 780
pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 ylLit	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCAAAGGGGCTCTTGATCTGGCTGG	780 780 780 780 780 780 780 780 780 780
purl pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 XlLit XlNach	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCAAAGGGGCTCTTGATCTGGCTGG	780 780 780 780 780 780 780 780 780 780
<pre>pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 XlLit XlNach H3016</pre>	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCAAAGGGGCTCTTGATCTGGCTGG	780 780 780 780 780 780 780 780 780 780
<pre>pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 X1Lit X1Nach H3016 H3173</pre>	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCAAAGGGGCTCTTGATCTGGCTGG	780 780 780 780 780 780 780 780 780 780
<pre>pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 X1Lit X1Nach H3016 H3173 H3179</pre>	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCAAAGGGGCTCTTGATCTGGCTGG	780 780 780 780 780 780 780 780 780 780
<pre>pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 X1Lit X1Nach H3016 H3173 H3179 H3182</pre>	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCAAAGGGGCTCTTGATCTGGCTGG	780 780 780 780 780 780 780 780 780 780
<pre>pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 X1Lit X1Nach H3016 H3173 H3179 H3182 C1</pre>	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCAAAGGGGCTCTTGATCTGGCTGG	780 780 780 780 780 780 780 780 780 780
<pre>pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 X1Lit X1Nach H3016 H3173 H3179 H3182 C1 Meq</pre>	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCAAAGGGGCTCTTGATCTGGCTGG	780 780 780 780 780 780 780 780 780 780
<pre>pur1 pur2 pur3 yve11 yve12 yve13 K122 HF85 HL81 HW79 XlLit XlNach H3016 H3173 H3179 H3182 C1 Meg Habana</pre>	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCCAAAGGGGCTCTTGATCTGGCTGG	780 780 780 780 780 780 780 780 780 780
<pre>pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 X1Lit X1Nach H3016 H3173 H3179 H3182 C1 Meg Habana NZH3</pre>	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCCAAAGGGGCTCTTGATCTGGCTGG	780 780 780 780 780 780 780 780 780 780
<pre>pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 X1Lit X1Nach H3016 H3173 H3179 H3182 C1 Meg Habana NZH3 T327</pre>	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCCAAAGGGGCTCTTGATCTGGCTGG	780 780 780 780 780 780 780 780 780 780
<pre>pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 XlLit XlNach H3016 H3173 H3179 H3182 C1 Meg Habana NZH3 T327 USCA01</pre>	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCCAAAGGGGCTCTTGATCTGGCTGG	780 780 780 780 780 780 780 780 780 780
<pre>pur1 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 XlLit XlNach H3016 H3173 H3179 H3182 C1 Meg Habana NZH3 T327 USCA01 H3210</pre>	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCCAAAGGGGCTCTTGATCTGGCTGG	780 780 780 780 780 780 780 780 780 780
<pre>pur1 pur2 pur3 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 XlLit XlNach H3016 H3173 H3179 H3182 C1 Meg Habana NZH3 T327 USCA01 H3210 CIP108426</pre>	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCCAAAGGGGCTCTTGATCTGGCTGG	780 780 780 780 780 780 780 780 780 780
<pre>pur1 pur2 pur3 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 XlLit XlNach H3016 H3173 H3179 H3182 C1 Meg Habana NZH3 T327 USCA01 H3210 CIP108426 FR32</pre>	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCCAAAGGGGCTCTTGATCTGGCTGG	780 780 780 780 780 780 780 780 780 780
<pre>purl pur2 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 X1Lit X1Nach H3016 H3173 H3179 H3182 C1 Meg Habana NZH3 T327 USCA01 H3210 CIP108426 FR32 H111</pre>	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCCAAAGGGGCTCTTGATCTGGCTGG	780 780 780 780 780 780 780 780 780 780
<pre>purl pur2 pur2 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 X1Lit X1Nach H3016 H3173 H3179 H3182 C1 Meg Habana NZH3 T327 USCA01 H3210 CIP108426 FR32 H111 H267</pre>	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCCAAAGGGGCTCTTGATCTGGCTGG	780 780 780 780 780 780 780 780 780 780
<pre>pur1 pur2 pur3 pur3 yel1 yel2 yel3 Kl22 HF85 HL81 HW79 XlLit XlNach H3016 H3173 H3179 H3182 C1 Meg Habana NZH3 T327 USCA01 H3210 CIP108426 FR32 H111 H267 H295</pre>	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCCAAAGGGGCTCTTGATCTGGCTGG	780 780 780 780 780 780 780 780 780 780
<pre>pur1 pur2 pur3 pur3 yel1 yel2 yel3 Kl22 HF85 HL81 HW79 XlLit XlNach H3016 H3173 H3179 H3182 C1 Meg Habana NZH3 T327 USCA01 H3210 CIP108426 FR32 H111 H267 H295 H3086</pre>	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCAAAGGGGCTCTTGATCTGGCTGG	780 780 780 780 780 780 780 780 780 780
<pre>pur1 pur2 pur3 yel1 yvel2 yvel3 K122 HF85 HL81 HW79 X1Lit X1Nach H3016 H3173 H3179 H3182 C1 Meg Habana NZH3 T327 USCA01 H3210 CIP108426 FR32 H111 H267 H295 H3086 AN6</pre>	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCAAAGGGGCTCTTGATCTGGCTGG	780 780 780 780 780 780 780 780 780 780
<pre>pur1 pur2 pur3 pur3 pur3 yel1 yel2 yel3 Kl22 HF85 HL81 HW79 XlLit XlNach H3016 H3173 H3179 H3182 C1 Meg Habana NZH3 T327 USCA01 H3210 CIP108426 FR32 H111 H267 H295 H3086 AN6 H3107</pre>	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCAAAGGGGCTCTTGATCTGGCTGG	780 780 780 780 780 780 780 780 780 780
<pre>pur1 pur2 pur3 pur3 yel1 yel2 yel3 Kl22 HF85 HL81 HW79 XlLit XlNach H3016 H3173 H3179 H3182 C1 Meg Habana NZH3 T327 USCA01 H3210 CIP108426 FR32 H111 H267 H295 H3086 AN6 H3107 H3240</pre>	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCAAAGGGGCTCTTGATCTGGCTGG	780 780 780 780 780 780 780 780 780 780
purl pur2 pur3 yel1 yel2 yel3 Kl22 HF85 HL81 HW79 XlLit XlNach H3016 H3173 H3179 H3182 C1 Meg Habana NZH3 T327 USCA01 H3210 CIP108426 FR32 H111 H267 H295 H3086 AN6 H3107 H3240 H3014	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCAAAGGGGCTCTTGATCTGGCTGG	780 780 780 780 780 780 780 780 780 780
purl pur2 pur3 yel1 yel2 yel3 Kl22 HF85 HL81 HW79 XlLit XlNach H3016 H3173 H3179 H3182 C1 Meg Habana NZH3 T327 USCA01 H3210 CIP108426 FR32 H111 H267 H295 H3086 AN6 H3107 H3240 H3014 pur1	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCAAAGGGGCTCTTGATCTGGCTGG	780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780
<pre>pur1 pur2 pur3 pur3 yel1 yel2 yel3 Kl22 HF85 HL81 HW79 XlLit XlNach H3016 H3173 H3179 H3182 C1 Meg Habana NZH3 T327 USCA01 H3210 CIP108426 FR32 H111 H267 H295 H3086 AN6 H3107 H3240 H3014 pur1 pur2</pre>	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCAAAGGGGCTCTTGATCTGGCTGG	780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780
<pre>pur1 pur2 pur3 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 X1Lit X1Nach H3016 H3173 H3179 H3182 C1 Meg Habana NZH3 C1 Meg Habana NZH3 T327 USCA01 H3210 CIP108426 FR32 H111 H267 H295 H3086 AN6 H3107 H3240 H3014 pur1 pur2 pur3</pre>	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCAAAGGGGCTCTTGATCTGGCTGG	780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780
<pre>pur1 pur2 pur3 pur3 yel1 yel2 yel3 K122 HF85 HL81 HW79 X1Lit X1Nach H3016 H3173 H3179 H3182 C1 Meg Habana NZH3 T327 USCA01 H3210 CIP108426 FR32 H111 H267 H295 H3086 AN6 H3107 H3240 H3107 H3240 H3014 pur1 pur2 pur3 yel1</pre>	GCAGCCCGTAAAGCACGCGAAATGACACGCCGCAAAGGGGCTCTTGATCTGGCTGG	780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780           780

vel2	$CC_{2}CC_{2}C_{2}C_{2}C_{2}C_{2}C_{2}C_{$	840
yei2		010
yel3	CCAGGCAAACTGGCGGATTGTCAGGAACGTGATCCGGCACTGTCCGAACTCTACTTAGTG	840
K122	CCAGGCAAACTGGCGGATTGTCAGGAACGTGATCCGGCACTGTCCGAACTCTACTTAGTG	840
HF85	CCAGGCAAACTGGCGGATTGTCAGGAACGTGATCCGGCACTGTCCGAACTCTACTTAGTG	840
HL81	CCAGGCAAACTGGCGGATTGTCAGGAACGTGATCCGGCACTGTCCGAACTCTACTTAGTG	840
HW79	CCAGGCAAACTGGCGGATTGTCAGGAACGTGATCCGGCACTGTCCGAACTCTACTTAGTG	840
XlLit	CCAGGCAAACTGGCGGATTGTCAGGAACGTGATCCGGCACTGTCCGAACTCTACTTAGTG	840
XlNach	CCAGGCAAACTGGCGGATTGTCAGGAACGTGATCCGGCACTGTCCGAACTCTACTTAGTG	840
Н3016	CCAGGCAAACTGGCGGATTGTCAGGAACGTGATCCGGCACTGTCCGAACTCTACTTAGTG	840
н3173		840
H2170		010
113179		010
H3102		040
CI		840
Meg		840
Habana	CCCGGCAAACTAGCAGATTGTCAGGAGCGTGATCCGGCACTGTCCGAACTCTACTTAGTG	840
NZH3	CCTGGCAAACTGGCAGATTGCCAGGAGCGTGACCCTGCGCTGTCCGAACTCTACTTAGTG	840
Т327	CCTGGCAAACTGGCAGATTGCCAGGAGCGTGACCCTGCGCTGTCCGAACTCTACTTAGTG	840
USCA01	CCTGGCAAACTGGCAGATTGCCAGGAGCGTGACCCTGCGCTGTCCGAACTCTACTTAGTG	840
H3210	CCCGGCAAACTAGCAGATTGCCAGGAGCGTGATCCGGCACTGTCCGAACTCTACTTAGTG	840
CIP108426	CCCGGCAAACTAGCAGATTGCCAGGAGCGTGATCCGGCACTGTCCGAACTCTACTTAGTG	840
FR32	CCCGGCAAACTAGCAGATTGCCAGGAGCGTGATCCGGCACTGTCCGAACTCTACTTAGTG	840
H111	CCAGGCAAACTGGCGGATTGTCAGGAACGTGATCCGGCACTGTCCGAACTCTACTTAGTG	840
H267	CCAGGCAAACTAGCTGATTGTCAGGAACGTGACCCGGCATTGTCCGAACTCTACTTAGTG	840
H295	CCAGGCAAACTGGCGGATTGTCAGGAACGTGATCCGGCACTGTCCGAACTCTACTTAGTG	840
н3086	ССАСССАААСТАССТСАТТСТСАССААССТСАСССССАТТСТССССААСТСААСТСАСТТАСТС	840
ANG		840
ANO	** ******* ** ** ** ** ***** ** ** ** *	010
112107	CIRCCC 047	
H3107	GAAGGGG 047	
H3240	GAAGGGG 047	
H3014	GAAGGGG 847	
purl	GAAGGGG 847	
pur2	GAAGGGG 847	
pur3	GAAGGGG 847	
yell	GAAGGGG 847	
yel2	GAAGGGG 847	
yel3	GAAGGGG 847	
К122	GAAGGGG 847	
HF85	GAAGGGG 847	
HL81	GAAGGGG 847	
HW79	GAAGGGG 847	
XlLit	GAAGGGG 847	
XlNach	GAAGGGG 847	
н3016	CAACCCC 847	
н3173	GAAGGGG 847	
u2170	CAAGGGG 047	
131/9	GAAGGGG 047	
H3182	GAAGGGG 847	
CI	GAAGGGG 847	
Meg	GAAGGGG 847	
Habana	GAAGGGG 847	
NZH3	GAAGGGG 847	
Т327	GAAGGGG 847	
USCA01	GAAGGGG 847	
H3210	GAAGGGG 847	
CIP108426	GAAGGGG 847	
FR32	GAAGGGG 847	
H111	GAAGGGG 847	
H267	GAAGGGG 847	
Н295	GAAGGGG 847	
Н3086	GAAGGGG 847	
AN6	GAGGGTG 847	
	** ** *	

Figure A. 7: ClustalW Multiple Sequence Alignment of a portion of the gyrB gene from three pur and three yel Photorhabdus temperata isolates from North Bull Island, a selection of P. temperata and P. luminescens species and 1 outgroup (X. nematophila strain AN6). Asterisk (\*) denotes conserved sites

Spaces denote nucleotide polymorphism (See Section 2.3.4 Figure 2.17).

	Number of days to first emergence							
	Nematode infection							
		PU	R IJs			Y	EL IJs	
Bacterial treatment	pur	yel	pur/yel	ctrl	pur	yel	pur/yel	ctrl
	16	16	16	17	17	16	16	16
	17	16	16	17	18	16	16	16
	17	16	17	18	18	16	16	16
	17	16	17	19	19	17	16	16
	18	16	17	19	19	17	18	16
	19	16	18	19	20	17	18	17
	19	17	18	20	20	17	18	17
	19	17	19	20	22	20	18	17
	19	17	19	20	23	21	21	17
	20	21	19	21	23	22	31	17
median	18.5	16.0	17.5	19.0	19.5	17.0	18.0	16.5

**Table A 6:** Time to first emergence in cross-compatibility experiment. Number of days to first emergence is calculated from the day of nematode infection (day 0) (see Section 3.3.1.3).

**Table A 7:** Results of pairwise comparisons of median time to first emergence of (A) PUR1 IJs and (B) YEL3 IJs from insects infected with 100 IJs. Insects had previously been injected with 1 x 10<sup>5</sup> cells of *P. temperata* isolates *pur1, yel3* or a *pur1/yel3* 1:1 mixture. Insects in the no bacteria treatment were injected with an equal volume of <sup>1</sup>/<sub>4</sub> sterile Ringer's solution only. P values in **bold** indicate a significant difference in median time to first emergence (Mann-Whitney U-test, 95% CI Bonferroni-adjusted significance level of  $\alpha = 0.008$ ) (See Section 3.3.1.3).

A	Bacterial treatment	pur1	yel3	pur1/yel3
	yel3	W = 134.5, P = 0.0226		
	pur1/yel3	W = 116.5, P = 0.3874	W = 81.5, P = 0.0679	
	no bacteria	W = 86, P = 0.1480	W = 67.5, <b>P</b> = <b>0.0041</b>	W = 76.5, P = 0.0300

B	Bacterial treatment	rial purl yel3		pur1/yel3
	yel3	W = 132.5, P = 0.0388		
	pur1/yel3	W = 130.0, P = 0.0592	W = 102.5, P = 0.8762	
	no bacteria	W = 152.5, <b>P</b> = <b>0.0003</b>	W = 122.5, P = 0.1625	W = 125.0, P = 0.1176

**Table A 8:** Results of pairwise comparisons of median time to first emergence of IJs from insects following infection with 100 IJs per insect. Insects had previously been injected with 1 x 10<sup>5</sup> cells of *P. temperata* isolates *pur1, yel3*, or a *pur1/yel3* 1:1 mixture. Data was pooled for both nematode types, PUR1 and YEL3, in each bacterial treatment. P values in **bold** indicate a significant difference in median time to first emergence (Mann-Whitney U-test, 95% CI Bonferroni-adjusted significance level of  $\alpha = 0.017$ )(See Section 3.3.1.3)

Bacterial treatment	pur1	yel3	
yel3	W = 517.5, <b>P</b> = <b>0.0031</b>		
pur1/yel3	W = 489.0, P = 0.0308	W = 363.5, P = 0.1965	

**Table A 9:** Results of Two-way ANOVA comparing the number of IJs emerging from insects by day 4 following first emergence as a proportion of the combined emergence by day 4 and at day 19. All insects were infected with 100 IJs per insect. Nematodes used were PUR1 or YEL3 isolates from North Bull Island. Insects had previously been injected with 1 x  $10^5$  cells of *P. temperata* isolates *pur1* or *yel3* (See Section 3.3.1.4).

Factor	Test statistic	Р
Nematode type	$F_{1,12} = 0.53$	0.482
Bacteria type	$F_{1,12} = 1.79$	0.205
Interaction	$F_{1,12} = 0.00$	0.992

**Table A 10:** Results of Two-way ANOVA of total number of IJs emerging from *G. mellonella* larvae over a 26 day period. All insects were infected with 100 IJs per insect. Nematodes used were PUR1 or YEL3 isolates from North Bull Island. Insects had previously been injected with 1 x  $10^5$  cells of *P. temperata* isolates *pur1* or *yel3* (See Section 3.3.1.4).

Factor	Test statistic	Р
Nematode type	$F_{1,16} = 2.39$	0.141
Bacteria type	$F_{1,16} = 0.31$	0.587
Interaction	$F_{1,16} = 0.25$	0.622

**Table A 11:** Two-way comparisons carried out to test whether source cadaver colour tends to predict 1 on 1 bioassay cadaver colour using IJs emerging from *pur1/yel3* 1:1 bacterial treatment. Data is pooled for PUR1 and YEL3 nematodes (See Section 3.3.2).

Source cadaver colour	1 on 1 bioassay cadaver colour		Source cadaver colour	l on l bioassay cadaver colour	
yellow or not	yellow or not	count	purple or not	purple or not	count
Р	р	4	P/PY	ру/у	8
Р	ру/у	6	P/PY	У	8
PY/Y	р	0	Y	ру/у	1
PY/Y	ру/у	9	Y	У	2
Source cadaver colour	1 on 1 bioassay cadaver colour		Source cadaver colour	1 on 1 bioassay cadaver colour	
yellow or not	purple or not	count	purple or not	yellow or not	count
Р	p/py	8	P/PY	р	4
Р	У	2	P/PY	ру/у	12
PY/Y	p/py	1	Y	р	0
PY/Y	у	8	Y	ру/у	3

**Table A 12:** Results of Two-way ANOVA of percentage of insects dead at 44 hours for wild type and hybrid combinations of nematode and bacteria. All insects were infected with 100 IJs per insect using IJs emerging from cadavers in the *pur1* and *yel3* bacterial treatments in the cross-compatibility experiment. There were four replications of each combination with ten insects per replication (See Section 3.3.3).

Factor	Test statistic	Р
Nematode type	$F_{1,12} = 4.27$	0.061
Bacteria type	$F_{1,12} = 0.37$	0.555
Interaction	$F_{1,12} = 0.37$	0.555

**Table A 13:** Results of pairwise comparisons of median number of insects dead at 48 hours post infection with wild type or hybrid nematode – bacteria combinations. (Mann-Whitney U test, 95% Bonferroni-adjusted significance level of  $\alpha = 0.008$ ) (See Section 3.3.3).

Nematode/ bacteria combination	P/p	Y/p	Y/y
Y/p	W = 23.0, P = 0.1635		
Y/y	W = 25.5, P = 0.0372	W = 21.5, P = 0.3778	
P/y	W = 20.0, P = 0.6084	W = 14.0, P = 0.2881	W = 11.0, P = 0.0530

**Table A 14:** Number of IJs with and without detectable levels of bacteria in two experiments. Experiment 1 consisted of six isolates from North Bull Island. Experiment 2 consisted of six isolates North Bull Island and the two hybrid nematode /bacteria combinations. N = 20 for all isolates and hybrids except PUR2 (n = 10 in experiment 1) (See Section 3.3.5).

	Experiment 1		Experiment 2	
Nematode	Bacteria	No bacteria	Bacteria	No bacteria
PUR1	19	1	19	1
PUR2	10	0	18	2
PUR3	19	1	19	1
YEL1	18	2	19	1
YEL2	19	1	20	0
YEL3	19	1	20	0
YEL/pur	-	-	20	0
PUR/yel	-	-	19	1

**Table A 15:** Results of Two-way ANOVA of percentage of insects dead at 56 hours for wild type and hybrid combinations of nematode and bacteria in a one-on-one continuous exposure infection with *G. mellonella* larvae (See Section 4.3.1.1).

Factor	Test statistic	Р
Nematode type	$F_{1,12} = 1.85$	0.198
Bacteria type	$F_{1,12} = 3.87$	0.073
Interaction	$F_{1 12} = 0.02$	0.882

**Table A 16**: Number of IJs emerging per cadaver from *H. abietes* larvae exposed to 50 IJs of each nematode/bacteria combination (See Section 4.3.2.1).

Institution         Exp2         Exp3         Exp4           18218         27983         45933         43700         18667         72333           Pp         47188         22750         98667         22960         59750           65867         50041         14100         79800         56478         30720         48280
Pp 43700 18667 72333 47188 22750 98667 22960 59750 50041 14100 79800 56478 30720 48280
Pp 43700 18007 72333 47188 22750 98667 22960 59750 65867 50041 14100 79800 56478 30720 48280
Pp       47188       22750       98667         22960       59750       65867         50041       14100       79800         56478       30720       48280
22960         59750           65867           50041         14100         79800           56478         30720         48280
65867           50041         14100         79800           56478         30720         48280
50041         14100         79800           56478         30720         48280
56478 30720 48280
37520 48167 17967
49226 25667 79500
28533 68000
19000
108000
105500
Ny 80467
46800
91500
102250
27900
49020
27500
хр 70400
59467
56640

**Table A 17:** Total number of insects dead, total number of adult EPN present and median number of IJs invading per insect at three time points. Data are from two ten-on-one bioassays with *G. mellonella*. In each experiment at each time point up to twelve insects per isolate were randomly chosen to assess the number of adults invading. n = 96 per isolate in experiment 1 and n = 48 per isolate in experiment 2 (See Section 4.3.3.1).

			Number	Number	Total number	
Experiment	Dav	Isolata	of insects	of insects	of adult EPN	Median number (/10) of
1	2	PUR1	18	10	39	3.5
-	-	PUR2	19	9	29	3
		PUR3	18	12	30	2
		YEL1	8	7	22	3
		YEL2	6	6	26	3.5
		YEL3	13	11	36	3
	3	PUR1	50	12	43	3
		PUR2	49	12	29	2
		PUR3	39	12	36	3
		YEL1	53	12	52	4
		YEL2	57	12	44	3.5
		YEL3	55	12	46	3.5
	4	PUR1	19	11	35	3
		PUR2	11	10	29	3
		PUR3	28	12	36	3
		YEL1	19	12	38	3
		YEL2	17	11	39	4
		YEL3	13	11	33	3
2	2	PUR1	5	3	12	3
		PUR2	10	5	20	3
		PUR3	5	3	11	3
		YEL1	0	0	0	0
		YEL2	1	1	3	3
		YEL3	7	2	7	3.5
	3	PUR1	18	9	24	3
		PUR2	20	9	35	4
		PUR3	17	8	21	2.5
		YEL1	21	11	34	3
		YEL2	35	12	39	3
		YEL3	21	10	39	4
	4	PUR1	9	5	15	3
		PUR2	11	6	21	3.5
		PUR3	10	5	12	2
		YEL1	15	7	21	3
		YEL2	6	3	7	2
		YEL3	11	6	21	3



**Figure A. 8:** Mean proportion  $(\pm$  SE) of *G. mellonella* killed by PUR and YEL IJs over eight days. Data shown is from experiment 2 with three isolates of each colour phenotype, n per colour phenotype = 144. Data were arcsine square root transformed for analysis (See Section 4.3.3.1)

**Table A 18:** Median number of IJs invading per insect at three time points. Data are from two experiments with *G. mellonella*. Nematodes were applied at a rate of ten IJs per insect. Up to twelve insects were randomly chosen per isolate in each experiment to assess the number of adults invading. n = 96 per isolate in experiment 1 and n = 48 per isolate in experiment 2 (See Section 4.3.3.2).

			Median number (/10) of IJs invading per
Experiment	Day	Isolate	insect
1	2	PUR1	3.5
		PUR2	3
		PUR3	2
		YEL1	3
		YEL2	3.5
		YEL3	3
	3	PUR1	3
		PUR2	2
		PUR3	3
		YEL1	4
		YEL2	3.5
		YEL3	3.5
	4	PUR1	3
		PUR2	3
		PUR3	3
		YEL1	3
		YEL2	4
		YEL3	3
2	2	PUR1	3
		PUR2	3
		PUR3	3
		YEL1	0
		YEL2	3
		YEL3	3.5
	3	PUR1	3
		PUR2	4
		PUR3	2.5
		YEL1	3
		YEL2	3
		YEL3	4
	4	PUR1	3
		PUR2	3.5
		PUR3	2
		YEL1	3
		YEL2	2
		YEL3	3

**Table A 19:** Results of pairwise comparisons of median number of adult nematodes per cadaver in insects which were dead on days 2, 3 and 4 post infection for each of 3 PUR and 3 YEL isolates. Data are from experiment 1 of a ten-on-one bioassay with *G. mellonella*. P values in bold indicate a significant difference in median number of adults at the 5% level (Mann-Whitney U-test, 95% Bonferroni-adjusted significance level of  $\alpha = 0.003$ ) (See Section 4.3.3.2, Table 4.7).

	PUR Day 2	PUR Day 3	PUR Day 4	YEL Day 2	YEL Day 3
PUR Day 3	W = 1052.5, P = 0.9894				
PUR Day 4	W = 997.0, P = 0.8877	W = 1242.5, P = 0.8282			
YEL Day 2	W = 794.0, P = 0.1914	W = 1001.5, P = 0.1215	W = 884.0, P = 0.2171		
YEL Day 3	W = 839.5, P = 0.0054	W = 1031.5, P = 0.0009	W = 916.5, P = 0.0030	W = 633.5, P = 0.1219	
YEL Day 4	W = 952.5, P = 0.3336	W = 1184.5, P = 0.2418	W = 1055.5, P = 0.3818	W = 738.0, P = 0.6197	W = 1466.5, P = 0.0205

**Table A 20**: Results of pairwise comparisons of median number of adult nematodes per cadaver in insects which were dead on days 2, 3 and 4 post infection for each of 3 PUR and 3 YEL isolates in two ten-on-one bioassays with *G. mellonella*. Data were pooled by colour phenotype for both experiments. P values in bold indicate a significant difference in median number of adults at the 5% level (Mann-Whitney U-test, 95% Bonferroni-adjusted significance level of  $\alpha = 0.003$ ) (See Section 4.3.3.2, Table 4.8).

	PUR Day 2	PUR Day 3	PUR Day 4	YEL Day 2	YEL Day 3
PUR Day 3	W = 2248.0, P = 0.7665				
PUR Day 4	W = 1970.0, P = 0.7539	W = 3471.0, P = 0.9975			
YEL Day 2	W = 1385.5, P = 0.2800	W = 2619.5, P = 0.1062	W = 1755.5, P = 0.1341		
YEL Day 3	W =1998.5, P = 0.0259	W = 3382.0, P = 0.0006	W = 2363.0, P = 0.0016	W = 1176.5, P = 0.2557	
YEL Day 4	W = 1922.5, P = 0.8051	W = 3387.2, P = 0.4746	W = 2359.0, P = 0.5032	W = 1138.5, P = 0.3350	W = 4580.0, P = 0.0131

**Table A 21**: Median time to death, number of cadavers placed on White traps (WT), percentage (%) of cadavers with emergence, and median time to first emergence of infective juveniles for insects infected with one of six isolates of *H. downesi* nematodes from North Bull Island in experiment 2 of a ten-on-one bioassay using *G. mellonella* (See Section 4.3.3.3).

				median time to
		number of	% cadavers	emergence
	median time to	cadavers	with	(days post
Isolate	death (days)	placed on WT	emergence	infection)
PUR1	3	18	100	17
PUR2	3	22	100	17
PUR3	3	20	95	19
YEL1	3	23	100	19
YEL2	3	22	100	17
YEL3	2.5	22	100	17

While a Kruskal-Wallis test indicated there was a highly significant difference in time to first emergence between isolates, (5 df, H = 18.10, P = 0.003), a series of pairwise comparisons failed to show any differences (Mann-Whitney U-test, 95% Bonferroni-adjusted significance level of  $\alpha = 0.003$ ) (See Table A 22).

**Table A 22:** Results of pairwise comparisons of median time to first emergence of infective juveniles for six isolates of *H. downesi* nematodes from North Bull Island in experiment 2 of a ten-on-one bioassay using *G. mellonella*. P values in bold indicate a significant difference in median number of adults at the 5% level (Mann-Whitney U-test, 95% Bonferroni-adjusted significance level of  $\alpha = 0.003$ ) (See Section 4.3.3.3).

	PUR1	PUR2	PUR3	YEL1	YEL2
PUR2	W = 414.5 P = 0.2057				
PUR3	W = 268 P = 0.0233	W = 356.5 P = 0.0051			
YEL1	W = 310.5 P = 0.0724	W = 390 P = 0.0074	W = 404.0 P = 0.9186		
YEL2	W = 395.0 P = 0.4735	W = 470.5 P = 0.5570	W = 631.5 P = 0.0182	W = 631.5 P = 0.0182	
YEL3	W = 405.5 P = 0.3124	W = 493.5 P = 0.9801	W = 641.5 P = 0.0094	W = 641.5 P = 0.0094	W = 514.0 P = 0.6506



**Figure A. 9:** Correlation between time to death and time to first emergence of IJs for (A) PUR isolates and (B) YEL isolates in a ten-on-one bioassay with *G. mellonella* (See Section 4.3.3.3).

**Table A 23:** Total and mean number of IJs emerging over 17 days from *G. mellonella* in experiment 2 of a ten-on-one bioassay. There were two pooled harvests per isolate and three isolates per colour phenotype, with 7 - 12 cadavers per pooled harvest (See Section 4.3.3.4).

Isolate	total harvest	number of cadavers	mean emergence per cadaver
PUR1	892440	7	127491
PUR1	1494670	11	135879
PUR2	1366400	11	124218
PUR2	611800	11	55618
PUR3	1055600	10	105560
PUR3	1515900	9	168433
YEL1	1158720	12	96560
YEL1	1316117	11	119647
YEL2	1362900	11	123900
YEL2	1159667	11	105424
YEL3	1371040	10	137104
YEL3	1744640	12	145387

**Table A 24:** Results of nested ANOVA of mean number of IJs per cadaver in pooled harvests over a 17 day period from *G. mellonella* in a ten-on-one bioassay. There were two pooled harvest per isolate and three isolates per colour phenotype. See Table A.23 for total counts and n per isolate (See Section 4.3.3.4).

Factor	df		Р
Colour phenotype	1	$F_{1,6} = 0.10$	0.925
Isolate	4	$F_{4,6} = 1.207$	0.398



**Figure A. 10:** Mean ( $\pm$  SE) proportion of *H. abietis* larvae killed by PUR and YEL IJs over fourteen days in two experiments. Data shown is mean of three isolates of each colour type in each experiment. (A) n = 16 and (B) n = 24 per colour isolate (See Section 4.3.4.1).

**Table A 25:** Cumulative percentage mortality of *H. abietis* larvae due to EPN infection day 4 - 14 post infection in two experiments with 3 PUR and 3 YEL *H. downesi* isolates from North Bull Island. n = 16 per isolate in experiment 1 and n = 24 per isolate in experiment 2. IJs were applied at a rate of fifty to one. Data from both experiments were combined for each isolate and analysed using a one-way ANOVA. Data were arcsine square root transformed for analysis (See Section 4.3.4.1, Figure 4.5).

Experiment	Isolate					D	ay					
		4	5	6	7	8	9	10	11	12	13	14
1	PUR1	0.31	0.38	0.56	0.69	0.69	0.69	0.69	0.69	0.69	0.94	0.94
	PUR2	0.06	0.38	0.63	0.69	0.75	0.88	0.88	0.88	0.88	0.88	0.88
	PUR3	0.13	0.38	0.38	0.56	0.63	0.69	0.69	0.69	0.69	0.75	0.81
	YEL1	0.19	0.63	0.81	0.88	0.88	0.94	0.94	0.94	0.94	0.94	0.94
	YEL2	0.19	0.31	0.38	0.50	0.63	0.69	0.69	0.69	0.69	0.88	0.88
	YEL3	0.19	0.44	0.56	0.69	0.81	0.81	0.81	0.81	0.81	0.88	0.88
2	PUR1	0.71	0.92	0.92	0.92	0.92	0.92	0.96	0.96	0.96	0.96	0.96
	PUR2	0.21	0.46	0.63	0.75	0.79	0.92	0.92	0.92	0.92	0.92	0.92
	PUR3	0.33	0.67	0.79	0.79	0.79	0.79	0.83	0.83	0.88	0.88	0.88
	YEL1	0.21	0.54	0.71	0.79	0.92	0.92	0.92	0.92	0.92	0.92	0.92
	YEL2	0.13	0.46	0.58	0.75	0.83	0.92	0.92	0.92	0.92	0.92	0.92
	YEL3	0.13	0.50	0.63	0.67	0.67	0.75	0.83	0.83	0.88	0.88	0.88

**Table A 26:** Mean ( $\pm$  SE) or median time to death, number of cadavers placed on White traps (WT), percentage (%) of cadavers with emergence of IJs, mean ( $\pm$  SE) or median time to first emergence of infective juveniles and mean emergence per cadaver for insects infected with one of six isolates of *H. downesi* nematodes from North Bull Island in two fifty-on-one bioassays using *H. abietis* (See Section 4.3.4.2 and Section 4.3.4.3).

					mean/median	
		median	number of		time to	Mean
		time to	cadavers	% cadavers	emergence	emergence
		death	placed on	with	(days post	per
Experiment	Isolate	(days)	WT	emergence	infection)	cadaver†
1	PUR1	6	15	80.0	26 (1.25)	47356
	PUR2	6	14	85.7	25.8 (0.70)	49977
	PUR3	7	13	84.6	25.3 (3.30)	45730
	YEL1	5	14	92.9	25.5 (0.99)	57755
	YEL2	7	14	85.7	25.2 (1.37)	45164
	YEL3	5.5	14	100.0	25.2 (0.95)	29528
2	PUR1	4	23	91.3	18	50305
	PUR2	5.5	22	95.5	23	45167
	PUR3	5	21	100.0	20	64285
	YEL1	5	22	95.5	22	50791
	YEL2	6	22	95.5	21	46545
	YEL3	5	21	95.2	21.5	46751

† A single count was carried out on the pooled harvest for each isolate in each experiment.

			percentage of cadavers with IJ	
			emerg	gence
Experiment			PUR	YEL
1			83.44	92.86
	P value	P = 0.170		
	test statistic	T =-2.10		
2			95.59	95.38
	P value	P = 0.943		
	test statistic	T = 0.08		
Experiment 1 & 2 combined	_		89.51	94.12
	P value	P = 0.237		
	test statistic	T = -1.28		

**Table A 27:** Results of 2-Sample T-test on percentage emergence for insects infected with one of six isolates of *H. downesi* nematodes from North Bull Island in two fifty-on-one bioassays using *H. abietis* (See Section 4.3.4.3).

**Table A 28:** Results of pairwise comparisons of time to first emergence of IJs in experiment 2 of a fifty-on-one bioassay with *H. abietis* larvae. (Mann-Whitney U-test with a 95% Bonferroni confidence interval,  $\alpha = 0.003$ ) (See Section 4.3.4.2, Table 4.10).

	PUR1	PUR2	PUR3	YEL1	YEL2
PUR2	W = 320.0, P = 0.0006				
PUR3	W = 356.0, P = 0.0125	W = 530.0, P = 0.0479			
YEL1	W = 350.0, P = 0.0076	W = 503.5, P = 0.1909	W = 416.5, P = 0.3792		
YEL2	W = 367.0, P = 0.0251	W = 523.0, P = 0.0708	W = 445.0, P = 0.8780	W = 478.5, P = 0.4971	
YEL3	W = 321.5, P = 0.0013	W = 471.5, P 0.4315	W = 397.0, P = 0.2526	W = 424.0, P = 0.6642	W = 402.0, P = 0.3102



**Figure A. 11:** Correlation between time to death and time to first emergence of IJs for (A) PUR isolates and (B) YEL isolates in experiment 1 and (C) PUR isolates and (D) YEL isolates in experiment 2 of a one-on-fifty bioassay with *H. abietis* (See Section 4.3.4.2).

**Table A 29**: Mean ( $\pm$ SE) proportion mortality of kelp fly larvae in a single 48 hour exposure experiment with 3 PUR and 3 YEL *H. downesi* isolates at two concentrations, 1000 or 5000 IJs per 10 larvae. n = 5 for each isolate in each treatment (See Section 4.3.6.1 Figure 4.8A, 4.8B).

	Concentr	Concentration		
Isolate	1000	5000		
PUR1	0.84(0.06)	0.80(0.05)		
PUR2	$0.86_{(0.04)}$	$0.62_{(0.08)}$		
PUR3	$0.90_{(0.04)}$	$0.70_{(0.06)}$		
YEL1	$0.86_{(0.06)}$	$0.68_{(0.05)}$		
YEL2	$0.86_{(0.05)}$	$0.66_{(0.12)}$		
YEL3	$0.90_{(0.03)}$	$0.80_{(0.06)}$		

**Table A 30**: Median percentage cadavers (A) per isolate and (B) per colour phenotype with IJ emergence in tests with kelp fly larvae in a single 48 hour exposure experiment with 3 PUR and 3 YEL *H. downesi* isolates at two concentrations, 1000 or 5000 IJs per 10 larva. n= 5 for each isolate in each treatment (See Section 4.3.6.3).

A	Concentration		
Isolate	1000	5000	
PUR1	100.00	100.00	
PUR2	100.00	100.00	
PUR3	100.00	77.78	
YEL1	100.00	100.00	
YEL2	100.00	100.00	
YEL3	77.78	71.43	
B	Concentration		
Colour phenotype	1000	5000	
PUR	100.00	100.00	
YEL	100.00	90.00	

**Table A 31:** Mean ( $\pm$ SE) emergence per cadaver from Kelp fly larvae infected with 1000 IJs per 10 larvae. Data is from individual harvests with n of 3 - 7 per isolate (See Section 4.3.6.3i).

	Mean (±SE)
Isolate	number of
isolate	IJs per
	cadaver
PUR1	8862 (626)
PUR2	8904 (3032)
PUR3	10384 (2807)
YEL1	10075 (860)
YEL2	3981 (1032)
YEL3	7271 (1655)

	Mean (±SE)
Isolata	number of
Isolate	IJs per
	cadaver
PUR1	4258 (1072)
PUR2	11016 (2106)
PUR3	10301 (2348)
YEL1	11168 (1098)
YEL2	5026 (1231)
YEL3	6135 (1104)

**Table A 32**: Mean (±SE) emergence per cadaver from Kelp fly infected with 5000 IJs per 10 larvae. Data is from individual harvests with n of 4 - 9 per isolate (See Section 4.3.6.3i).

**Table A 33**: Individual (I) or mean (P) emergence per cadaver from Kelp fly larvae in a single 48 hour exposure experiment with 3 PUR and 3 YEL *H. downesi* isolates at two concentrations, 1000 or 5000 IJs per 10 larvae. Emergence from each White trap was counted (I) individually or (P) pooled with up to seven cadavers in pooled counts. There was no difference between individual or mean emergence per cadaver at the isolate or colour phenotype (PUR or YEL) level. (nested GLM with isolate nested within colour and count method (I or P) included as a factor, P > 0.05 for all factors) (See Section 4.3.6.3ii).

Harvest	Isolate	Mean emergence per cadaver	Mean emergence per cadaver
		(1000 IJ concentration)	(5000 IJ concentration)
Ι	PUR1	8862	4258
	PUR2	8904	11016
	PUR3	10384	10301
	YEL1	10075	11168
	YEL2	3981	5026
	YEL3	7271	6135
Р	PUR1	4667	7514
	PUR2	3335	4566
	PUR3	5500	7769
	YEL1	2417	8540
	YEL2	1502	4644
	YEL3	4433	8308
Р	PUR1	5224	10613
	PUR2	4630	10168
	PUR3	5033	4528
	YEL1	3367	7856
	YEL2	3486	6050
	YEL3	2827	6047
Р	PUR1	11939	6976
	PUR2	6333	8094
	PUR3	3146	7904
	YEL1	2665	7060
	YEL2	5313	10941
	YEL3	22647	12767
Р	PUR1		7079
	PUR2		9291
	PUR3		11046
	YEL1		11109
	YEL2		3609
	YEL3		9423

				Day 6			
			Life stages				
			Juve	miles	Ad	ults	Bagged worms <sup>†</sup>
Nematode/bacteria	Moisture	Nematode					
combination	content	abundance	Live	Dead	Live	Dead	
		1	1	0	1	0	0
	0%	2	1	0	1	0	0
		1	1	0	1	0	0
		2	1	0	1	0	0
	1%	3	1	0	1	0	0
		2	1	0	1	0	0
		2	1	0	1	0	0
Рр	2%	3	1	0	1	0	0
		3	1	0	1	0	0
		2	1	0	1	0	0
	3%	2	1	0	1	0	0
		2	1	0	1	0	0
		2	1	0	1	0	0
	8%	3	1	0	1	0	0
		3	1	0	1	0	0
		1	1	0	1	0	0
	0%	2	1	0	1	0	0
		1	1	0	1	0	0
		1	1	0	1	0	0
	1%	2	1	0	1	0	0
		1	1	0	1	0	0
		2	1	0	1	0	0
Ру	2%	1	1	0	1	0	0
		2	1	0	1	0	0
		2	1	0	1	0	0
	3%	2	1	0	1	0	0
		2	1	0	1	0	0
		1	1	0	1	0	0
	8%	1	1	0	1	0	0
		2	1	0	1	0	0

**Table A 34:** Nematode abundance<sup>\*</sup>, survival and life stages<sup>‡</sup> present in *G*. *mellonella* cadavers stored for six days in sand at a range of different moisture levels. n = 3 for each nematode/bacteria combination at each moisture level.

Nematode abundance was assessed semi-quantitatively: 1: < 500 nematodes present; 2: 500 - 1000 nematodes present; 3: 1000 - 10000 nematodes present.</li>

 $\ddagger$  Life stages: 1 = present; 0 = absent for each category.

<sup>†</sup> In a process known as *endotokia matricida*, in each generation some eggs hatch within the maternal body and consume it. These are referred to as bagged worms.
				Day 9			
			Juve	niles	Ad	ults	Bagged worms†
Nematode/bacteria	Moisture	Nematode					
combination	content	abundance	Live	Dead	Live	Dead	
		3	1	0	1	0	0
	0%	3	1	0	1	0	0
		3	1	0	1	0	0
		3	1	0	1	0	0
	1%	3	1	0	1	0	0
		3	1	0	1	0	0
		3	1	0	1	0	0
Рр	2%	3	1	0	1	0	0
		3	1	0	1	0	0
		3	1	0	1	0	0
	3%	3	1	0	1	0	0
		3	1	0	1	0	0
	•	3	1	0	1	0	0
	8%	3	1	0	1	0	0
		3	1	0	1	0	0
	r	2	1	0	1	0	0
	0%	2	1	0	1	0	0
		3	1	0	1	0	0
		3	1	0	1	0	0
	1%	2	1	0	1	0	0
		3	1	0	1	0	0
		3	1	0	1	0	0
Ру	2%	3	1	0	1	0	0
		2	1	0	1	0	0
		3	1	0	1	0	0
	3%	3	1	0	1	0	0
		3	1	0	1	0	0
	•	2	1	0	1	0	0
	8%	3	1	0	1	0	0
		3	1	0	1	0	0

**Table A 35:** Nematode abundance<sup>\*</sup>, survival and life stages<sup>‡</sup> present in *G*. *mellonella* cadavers stored for nine days in sand at a range of different moisture levels. n = 3 for each nematode/bacteria combination at each moisture level.

 Nematode abundance was assessed semi-quantitatively: 1: < 500 nematodes present; 2: 500 - 1000 nematodes present; 3: 1000 - 10000 nematodes present.

 $\ddagger$  Life stages: 1 = present; 0 = absent for each category.

Day 12							
			Life stages				
			Juve	eniles	Ad	ults	Bagged worms*
Nematode/bacteria	Moisture	Nematode					
combination	content	abundance	Live	Dead	Live	Dead	
		3	1	0	1	0	1
	0%	2	1	1	1	1	1
		3	1	0	1	0	1
		3	1	0	1	0	1
	1%	3	1	0	1	0	1
		3	1	0	1	0	1
		3	1	0	1	0	1
Рр	2%	3	1	0	1	0	1
		3	1	0	1	0	1
		3	1	0	1	0	1
	3%	3	1	0	1	0	1
		3	1	0	1	0	1
	*	3	1	0	1	0	1
	8%	3	1	0	1	0	1
		3	1	0	1	0	1
		3	1	0	1	0	1
	0%	3	1	0	1	0	1
		1	1	0	1	0	1
		3	1	0	1	0	1
	1%	3	1	0	1	0	1
		3	1	0	1	0	1
		3	1	0	1	0	1
Ру	2%	3	1	0	1	0	1
		2	1	0	1	0	1
		3	1	0	1	0	1
	3%	3	1	0	1	0	1
		3	1	0	1	0	1
	•	3	1	0	1	0	1
	8%	3	1	0	1	0	1
		3	1	0	1	0	1

**Table A 36:** Nematode abundance<sup>\*</sup>, survival and life stages<sup>‡</sup> present in *G*. *mellonella* cadavers stored for 12 days in sand at a range of different moisture levels. n = 3 for each nematode/bacteria combination at each moisture level.

\* Nematode abundance was assessed semi-quantitatively: 1: < 500 nematodes present; 2: 500 - 1000 nematodes present; 3: 1000 - 10000 nematodes present.

 $\ddagger$  Life stages: 1 = present; 0 = absent for each category.

	Day 15						
			Life stages				
			Juve	eniles	Ad	ults	Bagged worms*
Nematode/bacteria	Moisture	Nematode					
combination	content	abundance	Live	Dead	Live	Dead	
		3	1	1	1	1	1
	0%	3	1	1	1	1	1
		3	1	1	1	1	1
		3	1	0	1	0	1
	1%	3	1	0	1	0	1
		3	1	0	1	0	1
		3	1	0	1	0	1
Рр	2%	3	1	0	1	0	1
		3	1	0	1	0	1
		3	1	0	1	0	1
	3%	3	1	0	1	0	1
		3	1	0	1	0	1
		3	1	0	1	0	1
	8%	3	1	0	1	0	1
		3	1	0	1	0	1
		2	1	0	1	0	1
	0%	1	1	0	1	0	1
		3	1	0	1	0	1
		3	1	0	1	0	1
	1%	3	1	0	1	0	1
		3	1	0	1	0	1
		3	1	0	1	0	1
Ру	2%	2	1	0	1	0	1
		3	1	0	1	0	1
		3	1	0	1	0	1
	3%	3	1	0	1	0	1
		3	1	0	1	0	1
		3	1	0	1	0	1
	8%	3	1	0	1	0	1
		3	1	0	1	0	1

**Table A 37:** Nematode abundance<sup>\*</sup>, survival and life stages<sup>‡</sup> present in *G*. *mellonella* cadavers stored for 15 days in sand at a range of different moisture levels. n = 3 for each nematode/bacteria combination at each moisture level.

\* Nematode abundance was assessed semi-quantitatively: 1: < 500 nematodes present; 2: 500 - 1000 nematodes present; 3: 1000 - 10000 nematodes present.

 $\ddagger$  Life stages: 1 = present; 0 = absent for each category.

	Day 18						
			Juve	eniles	Ad	ults	Bagged worms†
Nematode/bacteria	Moisture	Nematode					
combination	content	abundance	Live	Dead	Live	Dead	
		2	1	0	1	0	1
	0%	3	1	0	1	0	1
		3	1	0	1	0	1
	·	3	1	0	1	0	1
	1%	3	1	0	1	0	1
		3	1	0	1	0	1
		3	1	0	1	0	1
Рр	2%	3	1	0	1	0	1
		3	1	0	1	0	1
		3	1	0	1	0	1
	3%	3	1	0	1	0	1
		3	1	0	1	0	1
	r	3	1	0	1	0	1
	8%	3	1	0	1	0	1
		3	1	0	1	0	1
	r	3	1	0	1	0	1
	0%	3	1	0	1	0	1
		3	1	0	1	0	1
		3	1	0	1	0	1
	1%	3	1	0	1	0	1
		3	1	0	1	0	1
		2	1	0	1	0	1
Ру	2%	3	1	0	1	0	1
		3	1	0	1	0	1
		3	1	0	1	0	1
	3%	2	1	0	1	0	1
		3	1	0	1	0	1
		2	1	0	1	0	1
	8%	2	1	0	1	0	1
		3	1	0	1	0	1

**Table A 38:** Nematode abundance<sup>\*</sup>, survival and life stages<sup>‡</sup> present in *G*. *mellonella* cadavers stored for 18 days in sand at a range of different moisture levels. n = 3 for each nematode/bacteria combination at each moisture level.

\* Nematode abundance was assessed qualitatively: 1 < 500 nematodes; 2 > 1000 nematodes present; 3 > 10,000 nematodes present.

 $\ddagger$  Life stages: 1 = present; 0 = absent for each category.

**Table A 39**: Numbers of IJs emerging from *G. mellonella* cadavers following desiccation for 28 days at 0% RH in three experiments. n = 8 per nematode/bacteria combination in experiment 2, n = 5 per nematode/bacteria combination in experiment 3 and n = 25 per nematode/bacteria combination in experiment 4 (See Section 5.3.4).

Experiment	Nematode	Bacteria	Numbers of IJs emerging <sup>†</sup>
	Р	р	5306
	Р	р	87500
	Р	р	79661
	Р	р	65904
	Р	р	62248
2	Y	р	117491
Ζ	Y	р	99329
	Y	р	87800
	Y	р	77971
	Y	у	24666
	Y	у	1227
	Y	у	18
	Р	р	44920
	Р	р	3099
2	Y	р	1045
3	Y	у	39784
	Y	у	23049
	Р	у	129
	Р	р	76667
	Р	р	61667
	Р	р	15000
	Р	р	10667
4	Y	р	71667
4	Y	р	70800
	Y	р	26667
	Y	р	12333
	Y	р	11667
	Y	у	18667

<sup>†</sup> Overall percentage of cadavers with emergence and total number of IJs emerging by nematode type and bacteria type are shown in Table A 40.

**Table A 40**: Percentage of cadavers with emergence and total numbers of IJs emerging from *G. mellonella* cadavers following desiccation for 28 days at 0% RH in experiments 2, 3 and 4. The data were pooled by nematode type and by bacteria type and were combined for the three experiments with n = 16 per nematode type or bacteria type in experiment 2, n = 10 per nematode type or bacteria type in experiment 3 and n = 50 per nematode type or bacteria type in experiment 4 (See Section 5.3.4)

	Percentage of cadavers with	Total number of IJs
	emergence $(n = 76)$	emerging
Nematode type		
PUR	15.8	512767
YEL	21.1	665513
Bacteria type		
pur	27.6	1089408
yel	7.0	107540

		Nematode/bacteria combination				
Duration (days)	Percentage moisture	Рр	Ру			
	0%	-19.80 (2.505)	-17.92 (1.324)			
	1%	-1.03 (0.453)	-1.69 <sub>(0.147)</sub>			
6	2%	0.90 (0.139)	$0.76_{(0.441)}$			
	3%	0.67 (0.670)	0.63 (0.327)			
	8%	0.45 (0.367)	0.04 (0.228)			
	0%	-23.62 (1.231)	-25.09 (1.892)			
	1%	-1.23 (0.489)	-2.70 (1.909)			
9	2%	3.07 (0.707)	1.48 (0.208)			
	3%	2.06 (0.419)	1.43 (0.382)			
	8%	2.91 (1.035)	3.08 (1.018)			
	0%	-29.65 (5.630)	-32.56 (4.759)			
	1%	-1.45 (0.662)	-1.42 (0.152)			
12	2%	5.90 (4.641)	1.16 (0.228)			
	3%	4.15 (1.299)	3.17 (0.707)			
	8%	6.65 (0.589)	2.75 (0.907)			
	0%	-38.28 (2.789)	-38.74 (4.311)			
	1%	-4.48 (1.333)	-2.31 (1.043)			
15	2%	0.97 (0.161)	3.42 (2.507)			
	3%	1.71 (0.393)	4.28 (1.396)			
	8%	3.95 (0.810)	6.69 (1.213)			
	0%	-32.19 (1.453)	-37.68 (3.379)			
	1%	-4.76 (1.638)	-1.97 <sub>(1.197)</sub>			
18	2%	-11.02 (2.120)	-5.92 (2.967)			
	3%	1.75 (3.074)	1.91 (3.533)			
	8%	-3.50 (4.451)	13.00 (10.028)			

**Table A 41**: Mean ( $\pm$  SE) percentage change in weight of *G. mellonella* cadavers desiccated in sand at varying moisture content for up to 18 days. Data is from experiment 1, chapter 5 (See Section 5.3.5.1).